# Design of Environmental Performance Measurement Systems for Agriculture

## Ha Thanh Pham

A thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy

2016

Accounting Discipline Group University of Technology Sydney

## Certificate of original authorship

I certify that the work in this thesis has not previously been submitted for a degree, nor has it been submitted as part of requirements for a degree except as fully acknowledged within the text.

I also certify that the thesis has been written by me. Any help that I have received in my research work and the preparation of the thesis itself has been acknowledged. In addition, I certify that all information sources and literature used are indicated in the thesis.

Ha Thanh Pham

### Acknowledgements

This thesis would not have been possible without the contribution of a number of people. I would like to thank my supervisors, David Brown, Bruce Sutton and Paul Brown for your mentoring and insight throughout my PhD journey. I would like to thank David for his guidance and feedback, particularly regarding management accounting. I would like to thank Bruce for his support and insightful comments and suggestions regarding water and crop science. I would like to thank Paul for his enthusiasm for my study and his great support, particularly with economic modelling.

I would like to thank the Cotton project team – David Brown, Bruce Sutton, Paul Brown, Paul Thambar, Kai Jin, Nicole Sutton, Dianne Hiles, Anthony Krithinaki and Suzie Nguyen. I also acknowledge the help of Tommaso Armstrong for his assistance in simulation modelling and Ian Ly for his assistance in editing tables and figures of the thesis. I really appreciate the help of Shona Bates in editing the final version of this thesis.

I would like to acknowledge the comments and suggestions from the UTS visitors: Hugh Willmott, Gerhard Speckbacher, Martin Messner, Steven Sutton, Arnold Vicky, Angelo Ditillo, Lars Freemason, Dan Dhaliwal and Jere Francis.

I would like to thank the UTS Management Accounting Research Collaborative (MARC) group, particularly Prabhu Sivabalan and David Bedford, for their encouragement, comments and suggestions. I would like to acknowledge the support as well as comments and suggestions from the UTS Accounting Discipline Group, particularly Martin Bugeja, Steven Taylor, Zoltan Matolcsy, Jonathan Tyler, Bernhard Wieder, Roman Lanis, Robert Czernkowski, Brett Govendir, Anna Loyeung and Helen Spiropoulos.

I would like to acknowledge the institutional support I have received from the Accounting Discipline Group at UTS. I would like to thank Peter Wells for giving me the opportunity to start the PhD and particularly to Martin Bugeja, Jonathan Tyler and Anna Wright for their encouragement and for always ensuring I received adequate support. Special thanks go to the administrative staff, Judith Evans, Katt Robertson, Neil James and Ann-Marie Hopps, for helping me navigate the bureaucracy.

The support and friendship from a number of other people have also contributed to making the completion of this thesis both interesting and enjoyable; in particular, Anna Loyeung, Katt Robertson, Helen Spiropoulos, Amanda White, Rachael Lewis, Ann Usarat, Matthew Grosse, Nelson Ma, Samir Ghannam, James Wakefield, Kai Jin, Brett Govendir, Robert Czernkowski, Ross McClure, Thulaisi Sivapalan, Alexey Feigin, Steven Kean and Jin Sug Yang.

I give special thanks to my wonderful Mum, who has always supported and encouraged me to pursue my academic career. To my Dad and my sister, two incredible scientists who created and nurtured my interest in science and research, and inspired me to complete this work, I am sure that you would both be very proud of me if you knew my achievement.

Finally, and most importantly, I would like to thank my husband and my son, without whom this thesis would not have been completed, nor worth it. Your constant support, love and encouragement over the last 5 years, the sacrifices you made, and your continual understanding of the hard work involved have made this thesis possible. For this reason, this thesis is dedicated to the two of you.

#### Abstract

The research question addressed in this thesis is: how can Environmental Performance Measurement Systems (EPMS) be *designed* and *used* in an agricultural setting to support managers in water and economic sustainability-related decision making and control.

Sustainability and the increasing scarcity of natural resources such as freshwater are of growing social interest. Agriculture has a significant impact on the sustainability of freshwater at both global and local levels. As agriculture is economically and socially significant in meeting human needs for food and clothing, it is surprising that there has been very little management accounting research conducted within an agricultural setting and almost none on its role in environmental sustainability.

Extant EPMS research manifests two underlying theoretical problems, which are also reflected in broader performance measurement systems research. First, the research provides little insight into how to design *valid* environmental performance measures which could provide managers with precise information to enable decision making and control over environmental sustainability. I argue that there are two key reasons for this: that theories from natural science are yet to inform EPMS design; and that while environmental management typically occurs at an operational level, EPMS typically reside at the organisational level. The second theoretical problem is the lack of existing research that considers how environmental performance standards can be developed for use as *targets* to support managers in improving sustainability-related decision making and control.

I address these two problems with a new theoretical construction of a multi-level decomposition EPMS model - which I label, Water and Economic Sustainability Performance Measurement (WESM). The model integrates science into an accounting framework. This design overcomes the two key challenges with EPMS validity. I subsequently examine how the WESM model can be used to support managers in improving sustainability-related decision making and control using a two- phased crop production simulation modelling approach. The simulation results provide significant implications for the cotton industry (and agriculture more broadly) with the potential to save hundreds of gigalitres of water and increase profitability by tens of millions of dollars per crop season for cotton farming in Australia.

The research also makes a theoretical contribution to the accounting literature by developing and applying theory from science to overcome inherent validity and target setting problems in PMS design. In addition, I demonstrate the usefulness of simulation modelling as a research method, which has yet to have a great deal of application in accounting research designs beyond few costing studies.

## Contents

CERTIF	ICATE	OF ORIGINAL AUTHORSHIP	II
ACKNO	OWLED	DGEMENTS	
ABSTR	АСТ		V
CONTE	NTS		VII
LIST OI	F FIGU	JRES	XII
LIST OI	F TABI	LES	XIV
СНАРТ	ER 1.		1
1.1	Rese/	ARCH OBJECTIVE	1
1.2	Мот	IVATIONS FOR THE RESEARCH	1
1	.2.1	Motivation 1 – Management accounting research	1
1	.2.2	Motivation 2 – Agriculture research	11
1.3	Rese	ARCH QUESTION AND APPROACH	13
1	.3.1	The first component of the research question and approach	14
1	.3.2	The second component of the research question and approach	16
1.4	Rese/	ARCH CONTRIBUTIONS	19
1	.4.1	Contribution 1	
1	.4.2	Contribution 2	20
1	.4.3	Contribution 3	21
1	.4.4	Contribution 4	22
СНАРТ	ER 2.	THEORETICAL DISCUSSION OF WATER SUSTAINABILITY	24
2.1	Intro	DDUCTION	24
2.2	Sust	AINABILITY DEFINITION	24
2	.2.1	The Brundtland Commission's definition of sustainable development	24
2	.2.2	The Brundtland definition in practice	26
2.3	WAT	ER SUSTAINABILITY	27
2	.3.1	Global natural water system	27
2	.3.2	Local water systems	31
2	.3.3	Water sustainability principles	
2.4	CONC	CLUSION	

СНАРТ	ER 3.	THEORY OF CROP WATER FLOWS	. 40
3.1	Intro	DDUCTION	.40
3.2	CROP	WATER PROCESS MODEL	.41
3.	2.1	Water balance for a root zone	.42
3.	.2.2	Crop water process model	.43
3.	2.3	Water at the farm scale	. 54
3.3	THEO	RETICAL LINK BETWEEN CROP WATER SUPPLY COMPONENTS AND CROP PRODUCTION	
COM	PONEN	Τ\$	. 59
3.	3.1	Issues of inconsistent definitions and validity of crop water performance	
m	neasur	es in the literature	.60
3.	3.2	Crop water use measurement	. 63
3.	3.3	Crop water use index	. 69
3.4	THEO	RY OF CROP WATER FLOWS	.72
3.5	CONC	CLUSION	.75
СНАРТ	ER 4.	WATER AND ECONOMIC SUSTAINABILITY PERFORMANCE MEASUREMENT	
(WESN	1) MO	DEL	. 77
4.1	INTRO	DDUCTION	.77
4.2	Prof	IT TO WATER COST RATIO AS THE SUMMARY WATER AND ECONOMIC SUSTAINABILITY MEASURE	.78
4.	2.1	Measurement of water and economic sustainability performance for crop	
bi	usines	ses	. 78
4.	.2.2	Profit to water cost ratio as a summary measure for crop water and economic	
SL	ustain	ability	. 80
4.	.2.3	Limitations of profit to water cost ratio as a single measure for evaluating and	
m	nanagi	ing crop water and economic sustainability	.82
4.3	DESIG	sn of Water and Economic Sustainability Measurement Model - A decomposition	
RATIC	) ANAL	YSIS APPROACH	.83
4.	3.1	Overview of the decomposition ratio analysis approach	. 83
4.	3.2	Applying the decomposition analysis approach to EPMS design	.84
4.4	THE A	NALYSIS OF CROP WATER AND ECONOMIC SUSTAINABILITY	.85
4.	4.1	Analysis of the profit to water cost ratio	. 85
4.	4.2	First-level analysis: Decomposition of profit to water cost ratio	. 92
4.	4.3	Second-level analysis: Decomposition of return on water, water use leverage and	
W	eighte	ed average irrigation cost	101

	4.4.4	Third-level analysis: Decomposition of profit margin and economic water use	
	index	110	
	4.4.5	Fourth-level analysis: Decomposition of crop water use index and weighted	
	averag	e cotton price	118
	4.4.6	Fifth-level analysis: Decomposition of total sales revenue, total operating cost	
	and wo	ater input efficiency	121
	4.4.7	Sixth and seventh level analysis	127
4.	5 The I	LINK BETWEEN THE CROP WATER PROCESS MODEL AND THE WESM MODEL	128
4.	6 CON	CLUSION	129
СНА	PTER 5.	RESEARCH METHOD – CROP PRODUCTION SIMULATION MODELLING	131
5.	1 INTR	ODUCTION	131
5.	2 Met	HOD CHOICE	131
	5.2.1	The nature of the phenomenon being studied	131
	5.2.2	Research method alternatives	133
5.	.3 Sele	CTION OF COTTON PRODUCTION SIMULATION MODEL	139
	5.3.1	Overview of cotton production simulation models	139
	5.3.2	Comparison and evaluation of APSIM-OZCOT model and CROPGRO-Cotton-DSSAT	
	model	140	
5.	4 Thei	LINKS BETWEEN DSSAT, THE CROP WATER PROCESS MODEL AND WESM	142
5.	5 CON	CLUSION	147
СНА	PTER 6.	EXPERIMENTAL DESIGN FOR FURROW IRRIGATION SIMULATION	148
6.	1 INTRO	ODUCTION	148
6.	2 Furf	ROW IRRIGATION SYSTEMS: RESEARCH AND PRACTICE	149
	6.2.1	Overview of furrow irrigation systems	149
	6.2.2	The effect of furrow irrigation practices	151
6.	.3 Wat	ER AND CROP PRODUCTION SIMULATION MODELLING DESIGN: TWO-PHASED CROP SIMULATION	
М	ODELLING	Ĵ	151
	6.3.1	Limitations of DSSAT model in simulating furrow irrigated crop production	151
	6.3.2	Simulating a furrow irrigation event	153
	6.3.3	Simulating crop response to furrow irrigation	156
	6.3.4	Data aggregation	160
	6.3.5	Water balance of a cropping system	162
6.	.4 Curf	RENT IRRIGATION PRACTICE	166

6	.5 IM	PROVED IRRIGATION PRACTICE	170
6	.6 Cc	NCLUSIONS	174
СНА	PTER	7. THE ANALYSIS OF WATER AND ECONOMIC SUSTAINABILITY – AN EMPIRICAL	
EXA	MPLE	OF WESM ANALYSIS	175
7.	.1 IN	RODUCTION	
7	.2 Se	ECTION OF THE EMPIRICAL EXAMPLE	
7.	.3 AM	ALYSIS OF WATER AND ECONOMIC SUSTAINABILITY	
	7.3.1	First-level of analysis	
	7.3.2	Second-level of analysis	199
	7.3.3	Third-level of analysis	203
	7.3.4	Fourth-level of analysis	207
	7.3.5	Fifth-level of analysis	212
	7.3.6	Summary of WESM analysis	217
7	.4 Cc	NCLUSION	217
СНА	PTER	3. A STEP TOWARDS SUSTAINABILITY OF CROP BUSINESSES -IMPROVING	
FUR	ROW	RRIGATION MANAGEMENT PRACTICES	219
8	.1 IN	RODUCTION	219
8.	.1 IN <sup>-</sup> .2 Cc	RODUCTION	219
8. 8. PE	.1 IN <sup>-</sup> .2 CC ERFORM	RODUCTION MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION IANCE BETWEEN SCENARIO 1 AND SCENARIO 2	219
8. 8. PE	.1 IN .2 CC ERFORM .3 ST	RODUCTION MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION IANCE BETWEEN SCENARIO 1 AND SCENARIO 2	219 220 225
8 8 PE 8	.1 IN <sup>-</sup> .2 CC ERFORM .3 ST <i>8.3.1</i>	RODUCTION MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION IANCE BETWEEN SCENARIO 1 AND SCENARIO 2 ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2 Description of the WESM input variables	219 220 225 225
8. 8. PE 8.	.1 IN .2 CC ERFORM .3 ST <i>8.3.1</i> <i>8.3.2</i>	RODUCTION MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION IANCE BETWEEN SCENARIO 1 AND SCENARIO 2 ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2 Description of the WESM input variables Statistical test – Level 6 and Level 7 – Low-level environmental variables	219 220 225 225 230
8. 91 8.	.1 IN .2 CC ERFORM .3 ST 8.3.1 8.3.2 8.3.3	RODUCTION   MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION   IANCE BETWEEN SCENARIO 1 AND SCENARIO 2   ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2   Description of the WESM input variables   Statistical test – Level 6 and Level 7 – Low-level environmental variables   Statistical test – Level 6 and Level 7 – Low-level economic variables	219 220 225 225 230 239
8 8 9 8	.1 IN .2 CC ERFORM .3 ST 8.3.1 8.3.2 8.3.3 8.3.4	RODUCTION   MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION   IANCE BETWEEN SCENARIO 1 AND SCENARIO 2   ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2   Description of the WESM input variables   Statistical test – Level 6 and Level 7 – Low-level environmental variables   Statistical test – Level 6 and Level 7 – Low-level economic variables   Statistical test – Level 5 of WESM	219 220 225 225 230 239 246
8. Pf 8.	.1 IN .2 CC ERFORM .3 ST 8.3.1 8.3.2 8.3.3 8.3.4 8.3.5	TRODUCTION   MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION   IANCE BETWEEN SCENARIO 1 AND SCENARIO 2   ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2   Description of the WESM input variables   Statistical test – Level 6 and Level 7 – Low-level environmental variables   Statistical test – Level 6 and Level 7 – Low-level economic variables   Statistical test – Level 5 of WESM   Statistical test – Level 3 and Level 4 of WESM	219 220 225 230 239 246 251
8. 8. PP 8.	.1 IN .2 CC ERFORM .3 ST 8.3.1 8.3.2 8.3.3 8.3.4 8.3.5 8.3.6	TRODUCTION   MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION   IANCE BETWEEN SCENARIO 1 AND SCENARIO 2   ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2   Description of the WESM input variables   Statistical test – Level 6 and Level 7 – Low-level environmental variables   Statistical test – Level 6 and Level 7 – Low-level economic variables   Statistical test – Level 5 of WESM   Statistical test – Level 3 and Level 4 of WESM   Statistical test – Level 1 and Level 2 of WESM	219 220 225 230 239 246 251 258
8. PP 8.	.1 IN .2 CC ERFORM .3 ST 8.3.1 8.3.2 8.3.3 8.3.4 8.3.5 8.3.6 8.3.6	TRODUCTION MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION IANCE BETWEEN SCENARIO 1 AND SCENARIO 2 ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2 Description of the WESM input variables Statistical test – Level 6 and Level 7 – Low-level environmental variables Statistical test – Level 6 and Level 7 – Low-level economic variables Statistical test – Level 6 and Level 7 – Low-level economic variables Statistical test – Level 5 of WESM Statistical test – Level 3 and Level 4 of WESM Statistical test – Level 1 and Level 2 of WESM VIRONMENTAL AND ECONOMIC IMPLICATIONS OF MOVING FROM SCENARIO 1 TO SCENARIO 2	219 220 225 230 239 246 251 258 263
8. PF 8. 8.	.1 IN .2 CC ERFORM .3 ST 8.3.1 8.3.2 8.3.3 8.3.4 8.3.5 8.3.6 8.3.6 8.4.1	TRODUCTION MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION IANCE BETWEEN SCENARIO 1 AND SCENARIO 2 ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2 Description of the WESM input variables Statistical test – Level 6 and Level 7 – Low-level environmental variables Statistical test – Level 6 and Level 7 – Low-level economic variables Statistical test – Level 6 and Level 7 – Low-level economic variables Statistical test – Level 5 of WESM Statistical test – Level 3 and Level 4 of WESM Statistical test – Level 1 and Level 2 of WESM VIRONMENTAL AND ECONOMIC IMPLICATIONS OF MOVING FROM SCENARIO 1 TO SCENARIO 2 Difference in environmental resource use and environmental sustainability	219 220 225 230 239 246 251 258 263
8. PF 8.	.1 IN .2 CC ERFORM .3 ST 8.3.1 8.3.2 8.3.4 8.3.5 8.3.6 .4 EN 8.4.1 perfc	TRODUCTION   MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION   IANCE BETWEEN SCENARIO 1 AND SCENARIO 2   ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2   Description of the WESM input variables   Statistical test – Level 6 and Level 7 – Low-level environmental variables   Statistical test – Level 6 and Level 7 – Low-level economic variables   Statistical test – Level 5 of WESM   Statistical test – Level 3 and Level 4 of WESM   Statistical test – Level 1 and Level 2 of WESM   VIRONMENTAL AND ECONOMIC IMPLICATIONS OF MOVING FROM SCENARIO 1 TO SCENARIO 2   Difference in environmental resource use and environmental sustainability   rmance between Scenario 1 and Scenario 2	219 220 225 230 239 246 251 258 263
8. PF 8.	.1 IN .2 CC ERFORM .3 ST 8.3.1 8.3.2 8.3.3 8.3.4 8.3.5 8.3.6 .4 EN 8.4.1 perfc 8.4.2	RODUCTION MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION IANCE BETWEEN SCENARIO 1 AND SCENARIO 2 ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2 Description of the WESM input variables Statistical test – Level 6 and Level 7 – Low-level environmental variables Statistical test – Level 6 and Level 7 – Low-level economic variables Statistical test – Level 6 and Level 7 – Low-level economic variables Statistical test – Level 5 of WESM Statistical test – Level 3 and Level 4 of WESM Statistical test – Level 1 and Level 2 of WESM Difference in environmental resource use and environmental sustainability rmance between Scenario 1 and Scenario 2 Difference in water cost and economic sustainability performance between	219 220 225 230 239 246 251 258 263 269
8. PF 8.	.1 IN .2 CC ERFORM .3 ST 8.3.1 8.3.2 8.3.3 8.3.4 8.3.5 8.3.6 .4 EN 8.4.1 perfc 8.4.2 Scent	RODUCTION MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION IANCE BETWEEN SCENARIO 1 AND SCENARIO 2 ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2 Description of the WESM input variables Statistical test – Level 6 and Level 7 – Low-level environmental variables Statistical test – Level 6 and Level 7 – Low-level economic variables Statistical test – Level 6 and Level 7 – Low-level economic variables Statistical test – Level 5 of WESM Statistical test – Level 3 and Level 4 of WESM Statistical test – Level 1 and Level 2 of WESM VIRONMENTAL AND ECONOMIC IMPLICATIONS OF MOVING FROM SCENARIO 1 TO SCENARIO 2 Difference in environmental resource use and environmental sustainability rmance between Scenario 1 and Scenario 2 Difference in water cost and economic sustainability performance between ario 1 and Scenario 2 samples	219 220 225 230 239 246 251 263 263 269 269
8. PP 8.	.1 IN .2 CC ERFORM .3 ST 8.3.1 8.3.2 8.3.3 8.3.4 8.3.5 8.3.6 .4 EN 8.4.1 perfc 8.4.2 Scent 8.4.3	RODUCTION MPARISONS OF FURROW IRRIGATION MANAGEMENT PRACTICE AND FURROW IRRIGATION IANCE BETWEEN SCENARIO 1 AND SCENARIO 2 ATISTICAL TEST BETWEEN SCENARIO 1 AND SCENARIO 2 Description of the WESM input variables Statistical test – Level 6 and Level 7 – Low-level environmental variables Statistical test – Level 6 and Level 7 – Low-level environmental variables Statistical test – Level 6 and Level 7 – Low-level economic variables Statistical test – Level 5 of WESM Statistical test – Level 3 and Level 4 of WESM Statistical test – Level 1 and Level 2 of WESM VIRONMENTAL AND ECONOMIC IMPLICATIONS OF MOVING FROM SCENARIO 1 TO SCENARIO 2 Difference in environmental resource use and environmental sustainability rmance between Scenario 1 and Scenario 2 Difference in water cost and economic sustainability performance between arrio 1 and Scenario 2 samples	219 220 225 230 239 246 251 263 263 269 269 276 278

8.6	CONC	CLUSION	282
СНАРТІ	ER 9.	CONCLUSIONS AND IMPLICATIONS	284
9.1	Resea	ARCH SUMMARY	
9.	1.1	Validity issue of EPMS (contribution 1)	
9.	1.2	Target setting issue of EPMS (contribution 2)	
9.	1.3	Agriculture- related management accounting research (contribution 3)	290
9.	1.4	Novel EPMS model for agriculture (contribution 4)	
9.2	IMPLI	CATIONS OF THE RESEARCH	
9.	2.1	Use of new tool	293
9.	2.2	Economic and environmental significance	
9.3	LIMIT	ATIONS, DELINEATIONS AND FUTURE RESEARCH	
9.	3.1	Limitations and future research	
9.	3.2	Delineations and future research	
9.4	CONC	CLUSION	300
REFERE	NCES		303
APPEN	DICES		316
APPENI	DIX A:	SUPPLEMENTARY INFORMATION FOR CHAPTER 4	316
APPENI	DIX B:	SUPPLEMENTARY INFORMATION FOR CHAPTER 5	341
APPENI	DIX C:	SUPPLEMENTARY INFORMATION FOR CHAPTER 6	351
APPENI	DIX D	SUPPLEMENTARY INFORMATION FOR CHAPTER 7	357
APPENI	DIX E:	SUPPLEMENTARY INFORMATION FOR CHAPTER 9	360

## **List of Figures**

FIGURE 2.1 THE HYDROLOGICAL CYCLE SOURCE: ZHANG, WALKER & DAWES 2002	28
FIGURE 2.2 ANNUAL GLOBAL FLUX WATER CYCLE	29
FIGURE 2.3 GLOBAL WATER DISTRIBUTION IN THE HYDROLOGICAL CYCLE	30
FIGURE 2.4 WATER BALANCE AT A WATERSHED (A NATURAL SYSTEM)	32
FIGURE 2.5 WATER BALANCE AT A WATERSHED (WITH HUMAN INTERVENTION)	33
FIGURE 3.1 SCHEMATIC DIAGRAM OF THE WATER BALANCE FOR A ROOT ZONE	42
FIGURE 3.2 CROP WATER PROCESS MODEL	45
FIGURE 3.3 SOIL WATER DIAGRAM	51
FIGURE 3.4 RELATION BETWEEN CROP ECONOMIC YIELD AND WATER SOURCE	62
FIGURE 3.5 THEORETICAL RELATION BETWEEN TRANSPIRATION AND BIOMASS	65
FIGURE 3.6 OPERATIONAL DECISION TREE	71
FIGURE 4.1 LEVELS 1 TO 3 OF THE WESM MODEL	87
FIGURE 4.1 LEVELS 3 TO 5 OF THE WESM MODEL	88
FIGURE 4.1 LEVELS 5 TO 6 OF THE WESM MODEL	89
FIGURE 4.1 LEVELS 6 TO 7 OF THE WESM MODEL (ECONOMIC LOW-LEVEL MEASURES)	90
FIGURE 4.1 LEVELS 6 TO 7 OF THE WESM MODEL (ENVIRONMENTAL LOW-LEVEL MEASURES)	91
FIGURE 4.2 THE LINK BETWEEN THE CROP WATER PROCESS MODEL AND THE WESM MODEL	.129
FIGURE 5.1 EXAMPLE OF SIX TYPICAL COTTON PRODUCTION OPERATIONS IN AN AUSTRALIAN	
COTTON FARMING SYSTEM	. 132
FIGURE 5.2 THE LINKS BETWEEN DSSAT, THE CROP WATER PROCESS MODEL AND WESM	.143
FIGURE 5.3 DETAILED LINKS BETWEEN DSSAT, THE CROP WATER PROCESS MODEL AND WESM	.144
FIGURE 5.4 DESCRIPTION OF INPUT AND OUTPUT DATA OF DSSAT MODEL	.146
FIGURE 6.1 AN EXAMPLE OF FURROW IRRIGATION SYSTEMS IN A COTTON FARM	. 149
FIGURE 6.2 INFILTRATED DEPTH PROFILE UNDER FURROW IRRIGATION	.154
FIGURE 6.3 THE LINK BETWEEN KEY WATER AND CROP PARAMETERS BETWEEN THE TWO	
PHASED-CROP SIMULATION PROCESS AND WESM	.161

FIGURE 6.4 INFILTRATION CURVE_ SCENARIO 1	169
FIGURE 6.5 INFILTRATION CURVE_SCENARIO 2	173
FIGURE 7.1 LEVELS 6 TO 7 OF THE WESM MODEL (ENVIRONMENTAL LOW-LEVEL MEASURES)	186
FIGURE 7.1 LEVELS 6 TO 7 OF THE WESM MODEL (ECONOMIC LOW-LEVEL MEASURES)	192
FIGURE 7.1 LEVELS 1 TO 3 OF THE WESM MODEL	194
FIGURE 7.1 LEVELS 3 TO 5 OF THE WESM MODEL	195
FIGURE 7.1 LEVELS 5 TO 6 OF THE WESM MODEL	196
FIGURE 8.1 INFILTRATION CURVE_SCENARIO 2 VERSUS SCENARIO 1	224
FIGURE 8.2 LEVELS 1 TO 2 OF THE WESM MODEL	264
FIGURE 8.2 LEVELS 3 TO 5 OF THE WESM MODEL	265
FIGURE 8.2 LEVELS 5 TO 6 OF THE WESM MODEL	266
FIGURE 8.2 LEVELS 6 TO 7 OF THE WESM MODEL (ECONOMIC LOW-LEVEL MEASURES)	267
FIGURE 8.2 LEVELS 6 TO 7 OF THE WESM MODEL (ENVIRONMENTAL LOW-LEVEL MEASURES)	268
FIGURE B1 RATIO OF POTENTIAL TRANSPIRATION TO POTENTIAL EVAPOTRANSPIRATION AS A	
FUNCTION OF BOTH SOIL MOISTURE INDEX AND EVAPORATIVE DEMAND (POTENTIAL	
EVAPOTRANSPIRATION)	348
FIGURE B2 SIMULATED RATIO OF POTENTIAL TRANSPIRATION TO POTENTIAL	
EVAPOTRANSPIRATION AS A FUNCTION OF SMI WHEN LEAF AREA INDEX WAS RESTRICTED TO	
1.0-1.5 AND POTENTIAL EVAPOTRANSPIRATION TO 2.5-10 MM/DAY	349
FIGURE B3 RATIO OF POTENTIAL TRANSPIRATION TO POTENTIAL EVAPOTRANSPIRATION	
CALCULATED FROM DSSAT SIMULATION FOR LEAF AREA INDEX OF 1-1.5 AND POTENTIAL	
EVAPOTRANSPIRATION OF 2.5-10 MM/DAY	350
FIGURE C1 AN EXAMPLE OF SUBSURFACE (DRIP) IRRIGATION SYSTEMS IN A COTTON FARM	351
FIGURE C2 AN EXAMPLE OF LATERAL MOVE IRRIGATION SYSTEMS IN A COTTON FARM	353
FIGURE D1 COTTON GROSS MARGIN FUNCTION	358

## List of Tables

Table 4.1 WESM analysis: The first-level analysis	94
Table 4.2 WESM analysis: The second-level analysis	102
Table 4.3 WESM analysis: The third-level analysis	112
Table 6.1 SIRMOD Input: Field and infiltration characteristics	156
Table 6.2 DSSAT set up	158
Table 6.4 WESM Input Variables – Scenario 1(Simulated Data)	168
Table 6.6 WESM Input Variables – Scenario 2 (Simulated Data)	172
Table 7.1 WESM Input Variables – year 2012 (Simulated Data)	177
Table 7.2 WESM Input Variables (Non-Simulated Data)	178
Table 7.3 WESM Analysis (year 2012) – Panel G (Level 6 and Level 7: Low-Level Environmental	
Measures)	180
Table 7.3 WESM Analysis (year 2012) – Panel F (Level 6 and Level 7: Low-Level Economic	
Measures)	187
Table 7.3 WESM Analysis (year 2012) – Panel A (Level 1 of WESM)	198
Table 7.3 WESM Analysis (year 2012) – Panel B (Level 2 of WESM)	201
Table 7.3 WESM Analysis (year 2012) – Panel C (Level 3 of WESM)	205
Table 7.3 WESM Analysis (year 2012) – Panel D (Level 4 of WESM)	210
Table 7.3 WESM Analysis (year 2012) – Panel E (Level 5 of WESM)	214
Table 8.1 Furrow irrigation data: Scenario 1 versus Scenario 2	223
Table 8.2 Statistical Test: Univariate Mean Differences in WESM Simulated Input Data	
between Scenario 1 and Scenario 2	228
Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 –	
Panel G (Level 6 and Level 7: Low-Level Environmental Measures)	233
Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 –	
Panel F (Level 6 and Level 7: Low-Level Economic Measures)	241
Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 –	
Panel E (Level 5 of WESM)	248
Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 –	
Panel D (Level 4 of WESM)	252
Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 –	
Panel C (Level 3 of WESM)	256
Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 –	
Panel B (Level 2 of WESM)	259

Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 –	
Panel A (Level 1 of WESM)	262
Table 8.4 Economic and environmental differences between Scenario 1 and Scenario 2 – Panel	
A (Difference in environmental resource use)	273
Table 8.4 Economic and environmental differences between Scenario 1 and Scenario 2 – Panel	
B (Difference in economic performance)	277
Table 8.4 Economic and environmental differences between Scenario 1 and Scenario 2 – Panel	
C (higher level measures)	279
Table A1 WESM model - Panel D (Level 4 of WESM)	319
Table A1 WESM model - Panel E (Level 5 of WESM)	321
Table A1 WESM model - Panel G - Level 6	324
Table A1 WESM model - Panel G - Level 7	328
Table A2 WESM Input Variables (DSSAT output parameters)	333
Table A3 WESM Input Variables (Non-DSSAT output parameters)	334
Table A4 Summary of High-Level WESM measures	335
Table A5 Summary of Low-Level WESM measures	338
Table C1 Soil properties of the Myall Valve soil (clay vertosol)	354
Table C2 Simulated infiltration – Scenario 1 versus Scenario 2	355
Table D1 The effect of lint yield on gross margin	358
Table E1 Cotton irrigated area	360
Table E2 Environmental and economic gain by moving from Scenario 1 to Scenario 2	361

## **Chapter 1. Introduction**

#### 1.1 Research objective

The objective of this research is to examine how an Environmental Performance Measurement Systems<sup>1</sup> (EPMS) can be designed and used<sup>2</sup> in an agricultural setting to support managers to improve their environmental and economic sustainability-related<sup>3</sup> decision making and control. This thesis examines environmental sustainability performance management of one particular natural resource in the Australian crop agricultural industry – water.<sup>4</sup> Further, while issues of sustainability and their impact on organisations' performance have received growing attention from management accounting (MA) researchers, little is known about how EPMS can be designed and used to support managers to better manage sustainability issues arising from activities at the operational level<sup>5</sup> and to improve the organisation's economic and environmental performance, which is the focus of this thesis.

#### **1.2** Motivations for the research

#### **1.2.1** Motivation 1 – Management accounting research

Since the late twentieth century global environmental issues, such as shortages of natural resources, have drawn increasing attention from scientists, politicians, economists, and the international community (Intergovernmental Panel on Climate Change 2013; The Economics of Ecosystems and Biodiversity 2014). A number of global resource analyses reveal that

<sup>3</sup> At the operating level, this thesis adopts the concept of sustainable development defined by Elkington (1997, p. 70) as development that "*requires dramatic changes in organizations' performance against the triple bottom line, which focus on economic prosperity, environmental quality and social justice*". In particular, the thesis focuses on the environmental and economic aspects of an organization in terms of its sustainability performance. In the theory development contained in Chapter 2, the Brundtland Commission (1987, p. 1) definition is used as the departure point where sustainable development is characterised as "*development that meets the needs of the present without compromising the ability of future generations to meet their own needs*". <sup>4</sup> In this thesis water sustainability is developed in Chapters 2 and 3.

<sup>&</sup>lt;sup>1</sup> A performance measurement system is defined as "*the set of metrics used to quantify both the efficiency and effectiveness of actions*" (Neely, Gregory & Platts 1995, p. 81). More specifically, in this thesis, the term 'environmental performance measurement system' is used to indicate the inclusion of environmental performance indicators into a traditional performance measurement system (which comprises financial and non-financial indicators) (Henri & Journeault 2008).

<sup>&</sup>lt;sup>2</sup> In the context of this study, I am not studying whether farmers/crop managers adopt and use the model. What I am studying is whether it would be useful for farmers/crop managers and how it would be useful for them if they need to make decisions around changing water management practices and how the decisions would impact their crop economic and environmental performance.

<sup>&</sup>lt;sup>5</sup> This thesis adopts the organisation's operational level concept defined by Scherer (2001, p. 5679) as "*the level at which hands-on knowledge and responsibility for implementation are present*". Accordingly, the organisation's operational level discussed in this thesis refers to the level at which organisational activities that relate to the organisational production processes are conducted.

several key natural resources, including freshwater and fossil fuels, will become increasingly scarce over the next few decades (Moore 2013; Raskin et al. 1997; Rijsberman 2006; The Economics of Ecosystems and Biodiversity 2013). At the macro level, the concept of sustainable development was introduced by the Brundtland Commission<sup>6</sup> (1987, p.1) as development that "*meets the needs of the present without compromising the ability of future generations to meet their own needs*", in part to raise public awareness of global environmental issues arising from industrialization and growth. Despite increased awareness of many factors which impede the development of human beings in the long-run, many of the key issues of sustainability remain unsolved.

Scientific evidence demonstrates that global environmental sustainability issues have arisen partly from organisational activities which involve the exploitation and use of natural resources and the discharge of waste into global ecosystems (The Brundtland Commission 1987; The Economics of Ecosystems and Biodiversity 2013). Therefore, while we need to resolve the broad sustainability question at the society level, it is critical to examine environmental sustainability performance at the level *where environmental sustainability issues actually arise – the organisation's operational level*. Measuring environmental performance at the organisational level allows for information to be aggregated at higher levels, enabling the evaluation of sustainability issues at the industry, regional, national and global level (Chenoweth, Hadjikakou & Zoumides 2013; Henri & Journeault 2008).

Environmental performance measurement systems (EPMS) research emerged more than a decade ago, often examining environmental performance in the context of more traditional organisational performance and integrating concepts from environmental management (Burritt & Schaltegger 2001; Burritt, Hahn & Schaltegger 2002; Burritt & Saka 2006; Epstein & Roy 2001; Figge & Hahn 2013; Henri & Journeault 2008, 2010; Schaltegger & Synnestvedt 2002; Wagner & Schaltegger 2004; Wei, Roger & Gary 2011)<sup>7</sup>. While EPMS research has received growing attention from management accounting researchers, much of it

<sup>&</sup>lt;sup>6</sup> The Brundtland Commission is also formally known as the World Commission on Environment and Development (WCED).

<sup>&</sup>lt;sup>7</sup> EPMS literature is a subset of the broader sustainability accounting literature (Bartolomeo et al. 2000; Bebbington & Larrinaga 2014; Bebbington & Thomson 2013; Burritt & Schaltegger 2010; Gray 1992, 2010; Herbohn 2005; Hopwood 2009; Lamberton 2005; Milne 1996) and environmental management control systems (Brown & Sundin 2016; Henri & Journeault 2010; Pondeville, Swaen & De Rongé 2013). The thesis focuses on a particular issue that sits in the EPMS literature - validity measurement of EPMS - which is underpinning the entire infrastructure of environmental sustainability performance and management.

has focused on the examination of the role and use of EPMS in driving environmental and economic performance within organisations (Henri & Journeault 2008, 2010; Pondeville, Swaen & De Rongé 2013) or the relations between environmental disclosure, environmental performance, and economic performance (Al-Tuwaijri, Christensen & Hughes Ii 2004; Clarkson et al. 2008; Patten 2002)<sup>8</sup>.

Though this research contributes to MA literature in creating an understanding of how existing EPMS may have some influence on organisations' environmental and economic performance, little is known about how the examined EPMS are (or could be) *designed* in a way that more precisely capture the concept of economic and environmental sustainability at the organisational level. Furthermore, there are problems that emerge due to this lack of focus on EPMS design in studies which investigate EPMS. If the EPMS studied do not provide measures that capture and quantify the performance of an underlying phenomenon, activity or system, then studies evaluating how the EPMS are used to support decision making and control seem unlikely to provide meaningful outcomes/implications.

The roots of EPMS lie in performance measurement systems (PMS) - a subset of management control systems (MCS), which are traditionally defined as the systems that provide financial and non-financial information that is intended to be useful to managers for decision making and control to ensure the attainment of organisational objectives (Franco-Santos et al. 2007; Ittner & Larcker 2001; Malmi & Brown 2008; Otley 1999). The theoretical roots of how PMS design enable control are in cybernetic<sup>9</sup> control theory<sup>10</sup> (Fisher 1995; Flamholtz, Das & Tsui 1985; Green & Welsh 1988) with PMS operating as a cybernetic control system when it possesses five characteristics (Malmi & Brown 2008). The first characteristic is the ability to provide measures that more precisely capture the performance of an underlying activity or system. Second is the ability to create standards of performance, which could be a target to guide managers on what to achieve or aim for. Third

<sup>&</sup>lt;sup>8</sup> I recognise that there is a broader debate in the accounting literature about business cases for sustainability (Brix-Asala, Hahn & Seuring 2016; Dias, Vianna & Felby 2016; Haslam et al. 2014; Jones & Corral de Zubielqui 2016; Kang, Ryu & Kim 2010; Salzmann, Ionescu-somers & Steger 2005; Schaltegger & Burritt 2010; Weber 2008). However, it is beyond the scope of the thesis

<sup>&</sup>lt;sup>9</sup> Cybernetic is defined as a process in which a feedback loop is represented by using standards of performance, measuring system performance, comparing that performance to standards, feeding back information about unwanted variances in the system, and modifying the system's components (Hofstede 1978).

<sup>&</sup>lt;sup>10</sup> Green & Welsh 1988 define control as "a cybernetic, regulatory process that directs or contains an interactive activity to some standard or purpose" (p. 289).

is a feedback process which involves quantifying the actual performance of the activity or the system and comparing it with the targets (the expected performance). Fourth is the ability to perform variance analysis, which involves examining and identifying the key factors of difference between actual and expected performance (if any) that are examined in the feedback process. Fifth is the ability to modify the underlying activities or systems to enable improvement in their performance in the next activity cycle. Among the five characteristics of the cybernetic control systems, the first and second relate to the design of the systems whereas the later three relate to the use of the systems as a cybernetic control tool for decision making and control.

PMS literature has recognised the crucial role of the design phase among the three phases of PMS development (system design, system implementation and system use) laying out the theoretical foundation for the studies of the later phases of the PMS cycle (Artz, Homburg & Rajab 2012; Bourne et al. 2000; Ferreira & Otley 2009; Neely et al. 2000; Neely et al. 1997). However, from the research to date, in the relatively few studies on the design of EPMS (Figge & Hahn 2013), two broad problems were identified: first, the validity of EPMS design and second, target setting in EPMS design.

#### **1.2.1.1 Validity issue of EPMS design**

The first problem in this line of research relates to the validity of EPMS in relation to designing performance measures. Validity is defined as *"the extent to which a metric succeeds in capturing what it is supposed to capture"* (Ittner & Larcker 2003, p. 92). Accordingly, there are two questions that need to be considered when designing performance measures; what is supposed to be captured, and how can a measure(s) be designed to successfully capture this.

Extant PMS research has long addressed the validity issue of traditional performance measures (Bhargava, Dubelaar & Scott 1998; Dubelaar, Bhargava & Ferrarin 2002; Flamholtz, Das & Tsui 1985; Ittner & Larcker 2003; Nørreklit, Nørreklit & Israelsen 2006; Otley 1999). However, in the context of EPMS, the validity issue of environmental performance measures appears even more of a challenge to researchers with the need to understand natural science (particularly environmental science) and the integration between environmental performance measures and traditional financial performance measures. Since designing measures that capture the performance of an underlying activity or system is the first and essential step of PMS design, which is then used as a cybernetic control for better decision making and control, invalid environmental sustainability measures will prevent organisations from improving their environmental and economic performance. Extant EPMS research reveals that validity, one of the most important properties of sustainability measures, is generally missing in many environmental sustainability reporting and management practices (Burritt & Schaltegger 2010; Epstein 1996; Gray 2010). More specifically, Gray (2010) argues that most environmental sustainability reporting shows little implications of organisations' actual sustainability actions and emphasizes that "Only on those rare occasions when the indicators capture the key issues – and the issues are equally key to all parties – can the range of indicators necessarily be said to provide a full narrative" (Gray 2010, p.51).

The gap in our knowledge about validity of performance measures in extant EPMS research is due to three main reasons. First, there is still little work on the integration of PMS and environmental science (Bebbington & Thomson 2013). This means that the EPMS researchers do not always have a good understanding of the environmental science behind the phenomenon they study. In recognition of this kind of problem, Bebbington and Thomson (2013) call for more engagement and/or stronger interactions between the natural environment and sustainability oriented management accounting studies. In his study of the development of a framework for performance management, Otley (1999, p. 381) articulates a widely accepted concept which is relevant for EPMS, that *"attempting to design control systems without having a detailed knowledge of how business works is likely to prove a recipe for disaster*". In respect to EPMS, this means attempts to design applicable environmental issues being examined is unlikely to lead to valid measures, and hence improved organisational performance and/or meaningful practical implications associated with the use of the designed EPMS.<sup>11</sup> Two key components of the question of whether an

<sup>&</sup>lt;sup>11</sup> For example, in their more recent study of improving the efficient use of environmental resources, Figge and Hahn (2013) provide an empirical analysis of the carbon-efficiency of international car manufacturers, which is not consistent with their theoretical framework of corporate eco-efficiency. The problem lies in the confusion between natural resources efficiency and  $CO_2$  efficiency, which capture two different aspects of environmental sustainability. While the former relates to the degree of natural resources used for production processes, the later relates to the degree of pollution due to the discharge of wastes (gaseous, liquid or solid wastes) from the processes. This illustrates that while the authors have made a good attempt to integrate natural science into their study, there is still lack of good scientific understanding of the environmental science behind the phenomenon they study.

environmental sustainability measure(s) is valid are: (i) what is supposed to be captured in respect to environmental sustainability, and (ii) which measure(s) can capture this.

From the MA research to date there is still little understanding of what environmental sustainability means and how to account for environmental sustainability and its impact at the organisational level (Bebbington & Thomson 2013; Grav 2010; Grav 1992)<sup>12</sup>. For example, extant EPMS research mainly considers eco-efficiency or environmental resource use efficiency as a sustainability measure for businesses (Figge & Hahn 2013). While these measures reflect economic return on the use of an environmental resource, the broader concept of sustainable development, which includes consideration of environmental impacts (among other things) of businesses on the environment, is largely ignored. As discussed previously, scientific evidence also shows that many sustainability issues (including unsustainable water use) have arisen due to organisational activities (The Brundtland Commission 1987; The Economics of Ecosystems and Biodiversity 2013). This is manifest in two ways. First, organisations may operate in an inefficient manner (i.e., not using natural resource efficiently and effectively) resulting in inefficient use of Earth's limited natural resource capacity and increased costs. Second, even under conditions of efficiency, continuing withdrawals of natural resources from the environment will lead to the reduction of non-renewable resources or cause imbalance of the ecosystems. Consequently, one way to solve the sustainability issues at the global level is to manage the sustainability issues at the level they happen -the organisational level – by improving natural resource use efficiently and minimising interference in the natural environment<sup>13</sup>. Therefore, in order to deal with the validity issue of EPMS, it is of crucial important to incorporate environmental science into the first step of PMS design.

The second reason in relation to the validity of measures used in EPMS is that EPMS researchers typically do not look at the detail in the operational level of organisations when

<sup>&</sup>lt;sup>12</sup> This is a long-standing problem in environmental management accounting, and its most evolved form – sustainability oriented management accounting. More than two decades ago, Gray (1992) raised a concern of failures of conventional accounting with regard to its ability to deal with accounting for environmental issues and/or answering the questions of how and why we need to account for sustainability. Nearly twenty years later, Gray (2010) raises another concern in relation to accounting for sustainability, which is whether accounting for sustainability is actually accounting for sustainability and how we would know.

<sup>&</sup>lt;sup>13</sup> For example, in the context of crop water sustainability, for a crop business to be sustainable, it is not just about reducing the amount of water used to produce a crop but also about reducing organizational impacts on the natural environment. The thesis focuses on how to reduce as much water abstracted from the environment (which is irrigation) as possible, and from that I come up with my definition of water sustainability and crop water sustainability measurement which is discussed in more details in Chapters 2 and 3.

studying EPMS design. Due to focusing primarily at the organisation level, researchers know little about the design of EPMS at the level where environmental issues need to be managed. Scientific evidence shows that some of the key global environmental sustainability issues, such as the depletion of natural resources and the degradation of ecosystem services, have arisen directly from organisational production processes and activities, through the consumption of physical resources and the discharge of gaseous, liquid and solid waste into the natural environment (Intergovernmental Panel on Climate Change 2011; The Economics of Ecosystems and Biodiversity 2010). Therefore, in order to gain insights into environmental sustainability issues within organisational research, the focus must be on the level where organisational production processes and activities are carried out and, thus, environmental sustainability issues actually occur – the operational level of an organisation. By doing this, the question of what needs to be captured in respect to an organisation's environmental sustainability could be resolved.

However, one of the most common approaches that MA researchers adopt in EPMS studies is focusing on the corporate level or business level of an organisation and analysing financial and environmental data stemming from annual and environmental reports published by the companies under analysis (Figge & Hahn 2013; Pondeville, Swaen & De Rongé 2013). While EPMS studies at the higher organisational levels provide aggregate information about environmental performance of an organisation, the aggregate, high level information may not capture the underlying phenomenon being examined. In contrast, the examination of EPMS at the operational level will provide disaggregated and detailed information in relation to environmental sustainability that allows us to better understand the actual drivers of environmental performance of the organisation.

The third reason is that there is still a lack of a theoretical connection between summary performance measures, which sit at the organisational level, and environmental-related activities happening at the organisation's operational level – the level at which environmental issues occur and need to be managed. On the one hand, valid summary performance measures representing the overall (environmental and economic) sustainability performance of an organisation are an important measure that allow high-level managers to have more informed sustainability-related decision making and control. On the other hand, environmental-related activities and their performance can be managed and improved by operational level managers. A higher-level manager cannot assess the overall sustainability-related

performance of their organisation unless they see clearly the impact of management practices and/or management decisions at the operational level. Therefore, it is necessary to establish stronger and theoretical connections between these two levels to enable MA researchers to gain better and valid understanding of the routines, actions and outcomes of sustainabilityrelated management and accounting activities within organisations.

In summary, in order to deal with the validity issue of EPMS design it is important for PMS researchers to integrate environmental science into PMS theory, examine environmental issues at the organisational operational level, and build a theoretical connection between the summary measures sitting at the organisational level and environmental-related activities capturing environmental issues at the organisation's operational level. This will enable MA researchers to gain more insights into what is supposed to be captured and which measures can successfully capture this in relation to environmental sustainability performance of an organisation.

#### **1.2.1.2 Target setting issue of EPMS design**

The second problem in this line of research relates to target setting of EPMS design. Target setting refers to a process of developing targets (e.g. baselines, standards or goals) for performance measures to assist managers in identifying what to aim for and directing organisational change (Dekker, Groot & Schoute 2012; Ittner & Larcker 2001). In respect to EPMS research, target setting relates to a process of developing sustainability targets to support managers in improving organisational sustainably performance and/or moving towards sustainability.

Extant MA research has addressed the important role of target setting in managing and directing organisational performance (Dekker, Groot & Schoute 2012; Ittner & Larcker 2001). Furthermore, as discussed above, in EPMS design, setting of performance targets is the second step in designing a cybernetic control tool (Green & Welsh 1988). There is a considerable literature in accounting on target setting that spans topics such as, managerial and supervisory influences on targets (Bol et al. 2010), participation in the target setting process (Anderson, Dekker & Sedatole 2010), the relation between incentives, ratcheting and target setting (Aranda, Arellano & Davila 2014; Indjejikian, Matějka & Schloetzer 2014) performance level and effort (Fisher et al. 2015), and target difficulty, flexibility and performance (Arnold & Artz 2015). However, little attention has been paid to the methods of

determining performance targets with it being identified as being a topic that has been underresearched (Bol et al. 2010; Dekker, Groot & Schoute 2012; Ittner & Larcker 2001).

A number of PMS researchers have raised concerns about a lack of appropriate methods for target setting in the MA literature (Dekker, Groot & Schoute 2012; Ittner & Larcker 2001). The target setting research to date has mainly focused on two main methods to establish goals for future performance measures: (i) historical results based target setting and (ii) benchmarking based target setting (Ahmed & Rafiq 1998; Dekker, Groot & Schoute 2012; Elnathan, Lin & Young 1996; Figge & Hahn 2013; Ittner & Larcker 2001; Josée & Louis 2004). There are three limitations to the two methods. First, historical results and benchmarking methods do not always provide theoretically justified or reliable targets that organisations can aim for in the long-term. This is largely because they are either based on a firm's historical or past performance (under the historical results based target setting method) or rely on the performance of others within the industry to establish goals for future performance; none of which are a theoretically or scientifically justified maximum level of performance that a firm should aim for. Second, in relation to the benchmarking based target setting method, most companies do not have performance data from outside entities, such as their competitors (Dekker, Groot & Schoute 2012). Third, data that are publicly available<sup>14</sup>, might not be reliable, comprehensive or valid (Dekker, Groot & Schoute 2012; Ittner & Larcker 2001).

One way to overcome the limitations of the two methods mentioned above when studying organisational performance and target setting is applying analytical and simulation<sup>15</sup> modelling methods to determine optimal performance (Balakrishnan & Sivaramakrishnan 2002). For example, a small body of relevant costing, product pricing and capacity planning decisions research applies analytical and simulation modelling methods to approach questions that relate to optimization of decisions and performance (Balachandran, Balakrishnan & Sivaramakrishnan 1997; Balakrishnan & Sivaramakrishnan 2001, 2002; Christensen & Demski 1995; Dhavale 2005; Hansen & Banker 2002; Kaplan & Thompson 1971; Labro &

<sup>&</sup>lt;sup>14</sup> Including surveys conducted by either third parties, such as consulting firms, or researchers.

<sup>&</sup>lt;sup>15</sup> Balakrishnan and Sivaramakrishnan (2002, p. 24) define that "To simulate is to attempt to duplicate the features, appearance, and characteristics of a real system. Scientists often use physical models to investigate phenomenon (e.g., airplane wind tunnel simulations). In a similar manner, accounting researchers use computational experiments (with mathematical models) to estimate the effects of various alternatives. The idea behind simulation is to model a situation mathematically, use numerical methods to study its properties, and to draw conclusions and make decisions based on the results of the simulation".

Vanhoucke 2007; Noreen & Burgstahler 1997). Compared to the analytical method, the simulation method is an appropriate method to investigate the efficiency of various decision rules (i.e. cost-based decision rules), and characterise the performance of various planning rules (i.e. capacity planning rules or analyse trade-offs between performance measures) when dealing with a complex environment, as the complexity of the phenomenon or problem makes it difficult to mathematically compare performance (Balachandran, Balakrishnan & Sivaramakrishnan 1997). However, the analytical and simulation modelling methods have still not been widely used in PMS research; perhaps due to a paucity of appropriate/relevant simulation models and technical expertise.

In the development of EPMS, issues related to appropriate target values for performance measures become an even greater challenge as decision-makers have to deal with a more complex environment where economic and environmental performance aspects are simultaneously taken into consideration.<sup>16</sup> The limitations of historical results-based and benchmarking-based target setting methods discussed above make them problematic choices in setting targets for improving organisational sustainability performance. More specifically, in respect to environmental sustainability performance, the two methods mentioned above are limited in their ability to provide insights into which current management areas/practices need to be changed and to which extent to minimise the environmental issues occurring at the organisation's operational level in order to improve the overall organisational sustainability performance. For example, in one of the most recent studies on energy sustainability performance using benchmarking based target setting method, using the average performance of the car industry sector as the benchmark, Figge & Hahn (2013) admit that this approach does not provide insights into the level of sustainability of the sector or help understand how firms can move from the status quo to a more environmentally sustainable performance. In addition, while one way to improve organisations' performance is to estimate and evaluate the effects of various alternatives, the effect of a particular decision choice on performance of a business cannot be studied if it has not yet been used in practice, especially when a longterm effect which links to the sustainability concept is considered.

<sup>&</sup>lt;sup>16</sup> For example, in their study of energy efficiency, Virtanen, Tuomaala and Pentti (2013, p. 401) raise a concern that setting targets for energy efficiency is problematic due to the complexities involved in the measurement and management of the efficient use of energy.

In summary, while setting target is the second and necessary step in designing EPMS to be used as cybernetic control, there is still little research on how to develop appropriate targets for sustainability performance measures that can help organisations improve their sustainability performance. There are two main reasons for this. First, valid sustainability measures need to be designed. Second, most studies on environmental- related target setting in extant EPMS literature focus on historical results-based and benchmarking-based target setting methods which as discussed do not always provide theoretically justified and reliable targets that organisations can aim for in the long-term.

#### **1.2.2** Motivation 2 – Agriculture research

The agricultural sector is a potentially fruitful setting to investigate EPMS and water management in particular for three main reasons. First, from an economic perspective, the agricultural sectors play an important role in the global economy. Agriculture is the third largest contributor to the global GDP (after services and industry) (FAO 2013) and the valueadded of world's agricultural production is estimated at US\$ 2 trillion as of 2013 (FAO 2013; UNCTAD 2014). In 2010-11, the Australian agricultural sector, with approximately 134,000 farm businesses (99% of which are family owned and operated), contributed around AU\$48.7 billion to the Australian economy (at farm-gate) (National Farmers' Federation 2012). Moreover, when post-farm gate value<sup>17</sup> is added, the Australian agricultural sector contributes around AU\$155 billion, or 12%, each year to the GDP (National Farmers' Federation 2012). In addition, the Australian agricultural sector provides about 300,000 jobs at the farm gate, and 1.6 million jobs across the agricultural supply chain, to the Australian economy. In the US, the agricultural sector accounts for approximately one-fifth of national economic activity (US Department of Homeland Security 2013). It is estimated that the US agricultural sector consists of 2.2 million farms, and impacts more than 400,000 food manufacturing, processing, and storage facilities, and 900,000 restaurants (US Department of Homeland Security 2013).

Second, from a social point of view, agriculture is of crucial importance to the existence and development of human beings as it provides food and fibre to meet two of the most essential needs of human beings (Henzell 2007).<sup>18</sup> Jack (2005, p. 60) emphasizes that "*this is one industry that has every single person as a stakeholder, relying on it for food and clothing*".

<sup>&</sup>lt;sup>17</sup> Value of all economic activities supporting farm production processes of food and fibre from the farm-gate.

<sup>&</sup>lt;sup>18</sup> Agriculture also provides critical inputs for bio-fuel, which is also of importance.

One-third of the world's population still obtains its life-essential income from agriculture (FAO 2013). Due to population growth and rising per capita income, it is projected that the world annual agricultural production needs to further increase by approximately 60-70 percent in order to satisfy the future demand (Alexandratos & Bruinsma 2012; Conforti 2011). Australian farmers produce food for an estimated 60 million people. Of its total agricultural production, 40% is for Australia's daily food supply and 60% is exported (National Farmers' Federation 2012). The growth in world population since the 1950s requires substantial increases in global food and fibre production (Rijsberman 2006; van Ittersum et al. 2012). This means sustaining crop agriculture is critical to maintaining human society and the global economy.

Third, agronomic research shows that water is one of the key limiting factors to crop yield and quality (Cai et al. 2011; Hochman, Holzworth & Hunt 2009; Passioura 2006). While water plays an important role in improving crop yield and crop quality for sustaining crop businesses' profitability (Bange et al. 2010; Long et al. 2010), it is predicted that a water shortage will become a key constraint for agriculture in coming decades (Hamdy, Ragab & Scarascia-Mugnozza 2003; Jackson et al. 2011). Only 2.5 percent of the planet's water is freshwater, however only 0.76% can be considered as usable fresh water (Ibaraki, 2010). Increased human consumption has contributed to the severe stress and deterioration of the limited freshwater resources worldwide(WWAP 2009, 2012, 2014). Agriculture is the biggest contributor towards freshwater use - currently irrigated agriculture accounts for about 70 percent of freshwater withdrawals globally (FAO 2013). Furthermore, among the three main water consuming sectors (domestic, industrial and agricultural sectors), irrigated agriculture is the largest consumer of water (Jackson, Khan & Hafeez 2010; Rijsberman 2006; van Ittersum et al. 2012). In Australia, it is reported that agriculture consumes a large proportion of water resources (i.e. the agriculture industry accounted for 52% of national water use in 2009-10) (Pink 2012). Therefore, improvement of water management in agriculture is vital for global economic, social and environmental sustainability.

Despite the important role of water sustainability management in agriculture, the issues around how to design appropriate EPMS to support crop growers in improving their sustainable use of water are unresolved. One of the main reasons for this is a lack of theoretical framework of crop water sustainability performance at the crop business level. Extant research mainly focuses on water performance measures for improving the efficiency in water use<sup>19</sup> (Hochman, Holzworth & Hunt 2009; van Ittersum et al. 2012), without incorporating economic performance of a crop business; or considering the trade-off between water-related environmental and economic performance to provide a more complete picture of water sustainability. Moreover, literature which focuses on efficiency of water used by plants is plagued by inconsistent terminology in such areas as, the relationship between crop yield and water use, different terms for the water component and crop output, and differing concepts of water efficiency (Blum 2009; Cossani, Slafer & Savin 2012; Hochman, Holzworth & Hunt 2009; Igbadun et al. 2006; Liu-Kang & Hsiao 2004; Moore, Robertson & Routley 2011; Passioura 2006; Pereira, Cordery & Iacovides 2012; Tennakoon & Milroy 2003; van Halsema & Vincent 2012). The above challenges with validity of measures creates limitations on the capacity of researchers and growers to develop 'appropriate' targets to support crop growers in improving their crop water sustainability performance.

Though the development of crop simulation models since the 1980s provides opportunities for the simulation-based target setting approach to be adopted in crop science research (Hearn et al. 1981; Keating et al. 2003), this body of research mainly focuses on crop yield as the main output. This means that a number of determinants of environmental sustainability generally, and water sustainability performance particularly, are still missing in crop simulation models (Cheeroo-Nayamuth et al. 2000; Grassini, Hall & Mercau 2009; van Ittersum et al. 2012). Therefore, without a crop water sustainability framework coupled with crop water performance measurement, the question of how crop businesses can move towards water sustainability will remain unanswered.

#### **1.3** Research question and approach

The above arguments outline three unresolved issues from extant research on EPMS design for crop agriculture. First is the lack of validity of EPMS to provide managers more precise environmental performance information about what environmental sustainability issues need to be captured and measured. Second is the lack of environmental performance standards that can be used as targets to direct managers as to what they need to do for better decision making and control. Third is the lack of a theoretical framework of crop water measurement

<sup>&</sup>lt;sup>19</sup> Crop yield per unit of crop water use.

systems that can support crop managers and inform better water sustainability-related decision making and control<sup>20</sup>.

Therefore, the core research question addressed in this thesis is:

How can Environmental Performance Measurement Systems (EPMS) be designed and used in an agricultural setting to support managers in water and economic sustainability-related decision making and control?

This question focuses on the concept of economic and environmental sustainability at the level of organisation, and more specifically, economic and water sustainability of crop production. In this context, there are two components to the research question. The first is how EPMS can be designed; the second is how the EPMS can be used to better support sustainability-related decision making and control.

#### **1.3.1** The first component of the research question and approach

Chapters 2, 3 and 4 address the first component of the research question – relating to how EPMS can be designed. In order to overcome the first issue of EPMS design, which is the validity issue of EPMS, in Chapter 2, a set of water sustainability principles, based on water science and contemporary sustainable thinking, are developed to provide a theoretical foundation of what water sustainability means generally. Chapter 3 then provides a theoretical foundation for conceptualising water sustainability at the business level, in the context of crop agriculture, and developing some key environmental sustainability measures that capture the underlying crop water sustainability issues arising from irrigation management practices and activities. This is done through a number of steps. First, a crop water process model, based on crop science, is developed to lay out the inflow and outflow components of a crop production process occurring at the operational level of the crop business. Second, based on the water sustainability principles developed in Chapter 2 and the constructed crop water process model, four key water sustainability measures (including

<sup>&</sup>lt;sup>20</sup> I recognise that there is accounting literature which deals with water sustainability and water accounting at an industry and higher levels (Chenoweth, Hadjikakou & Zoumides 2013; Gleick 2003; Ibaraki 2010; Rijsberman 2006; The Economics of Ecosystems and Biodiversity 2013) and political issues related water accounting (Bell & Quiggin 2008; Chalmers, Godfrey & Lynch 2012; Moore 2013; Ogden 1997; Tingey-Holyoak 2014). However, it is beyond the scope of the thesis. My thesis focuses on a technical issue of sustainability measurement which is dealt with at the organisational level.

water input efficiency, water resource efficiency, environmental index, and crop water use index), which capture sustainability-related activities/management practices occurring at the crop production level (the level at which water sustainability issues occur and need to be managed) are developed and theoretical links between crop water and crop production is established. Furthermore, a discussion of the theoretical value of some of the developed sustainability measures is included in this chapter to lay out theoretical target development for the second step of EPMS design.

In Chapter 4 I develop a Water and Economic Sustainability Measurement (WESM) model a form of EPMS specifically for the context of water and economic sustainability performance measurement for crop agriculture. More specifically, I apply the decomposition ratio analysis approach – a way to present performance measures in a transparent, structured and hierarchical way<sup>21</sup> – to design a hierarchical decomposition model that has measures at the organisational level connecting down to the operational level. The design of the WESM comprises two main steps. First, by incorporating the economic aspect into the discussion of water sustainability developed in the previous two chapters, a theoretical summary measure of water and economic sustainability – the profit to water cost ratio – is developed. This measure is informative about both economic and water sustainability performance in one equation. Together with the four key water sustainability measures (water input efficiency, water resource efficiency, environmental index and crop water use index), I argue how this summary measure can be used to support managers in better sustainability-related decision making and control. Second, the profit to water cost ratio that sits in the highest level (e.g. the first level) of the WESM model can be decomposed into its key components (sitting in the second and third levels of the WESM model), and further decomposed into some of the key water sustainability measures developed in Chapter 3 (sitting in the fourth and fifth levels of WESM model) which link to water and crop production components of the crop water process model (through various water and economic measures sitting in the sixth and seventh levels of WESM model).

This application of the decomposition analysis approach allows the summary measure of water and economic sustainability and its drivers to be tied in a logical and structured way to

<sup>&</sup>lt;sup>21</sup> One of the key features of the decomposition ratio analysis approach is that the aggregate measures can be decomposed into more detailed and specific descriptors of an organization's performance, and hence, changes in overall performance can be traced in terms of changes in specific aspects of the organization's performance (Penman 2003).

form the architecture of the WESM.<sup>22</sup> This also enables establishing of a theoretical and logical connection between the summary sustainability performance at the business level, and water sustainability-related activities/management practices occurring at the crop businesses' operational level. This allows us to better understand how the profit to water cost ratio will change as a result of a particular management decision at the operational level, and how the change translates into economic and environmental sustainability value/gain created for the crop business. The WESM provides a platform for the empirical work in later chapters of the thesis.

#### **1.3.2** The second component of the research question and approach

The second component of the research question addressed in this thesis is how the designed EPMS can be used in an agricultural setting to support managers in water and economic sustainability-related decision making and control. This focusses on the target setting feature of the WESM, followed by the feedback process, variance analysis and system modification features of the WESM. This is addressed in Chapters 5, 6, 7 and 8.

First, Chapter 5 lays out the method applied in this thesis. The focus of the empirical work is on cotton production in Australia. This is for four key reasons. First, cotton is a crop of economic significance. Second, cotton provides fibre to meet one of the most essential needs of human beings - clothing. Third, cotton yield and quality is sensitive to water availability. Finally, cotton is a comparatively high user of water.

In relation to the research method, crop production simulation modelling is argued as the most appropriate approach, among three common methods that can be applied to collect crop data and information (field experiments, publicly available data and crop simulation modelling). This is due to the complexity of the phenomenon being studied and the lack of reliable, valid and specific crop data from public sources. Most importantly, crop simulation modelling allows model users to simulate a cropping system in response to various climate and soil factors, examine spatial and temporal variation, and evaluate management intervention through a wide range of management practices, which is not feasible (if not impossible) with field experiments or collection of historical and/or publicly available data.

<sup>&</sup>lt;sup>22</sup> While this approach is similar to Penman styled ratio analysis (Nissim & Penman 2001; Penman 2003) or DuPont styled ratio analysis (Soliman 2008) in the financial statement analysis and security valuation literature, the set of variables in Chapter 4 extend beyond the information available in financial statements.

The Decision Support System for Agrotechnology Transfer (DSSAT) model with the CROPGRO-Cotton module<sup>23</sup> is applied for cotton production simulation modelling. The DSSAT system is selected as the most appropriate crop simulation model for this study; it is well-recognised and widely-used by researchers worldwide and proven to generate valid and reliable crop information which can facilitate the evaluation of water and economic sustainability performance of a crop business from various crop management decision choices. The DSSAT system provides a mechanism to simulate the real-world cropping system represented by the crop water process model. Accordingly, input data to DSSAT includes information about crop genotype (*G*), environmental conditions (*E*) and a set of management rules for crop production processes (*M*), whilst output data from DSSAT comprise crop water and production components of the crop water process model. Output data generated by the DSSAT model and accounting data are fed into the WESM model to allow quantification of key water and economic sustainability measures presented in the WESM model, as well as the effect of specific water management decisions/activities on the overall water and economic sustainability performance of a crop business.

Second, building on Chapter 5, Chapter 6 constructs a two phased-crop simulation modelling design which enables capture of a more comprehensive picture of a real-world cropping system. To do this, I first select furrow irrigation systems as an irrigation method applied in examining the use of the designed WESM in this thesis since furrow irrigation is currently the most common irrigation system applied in cotton production. Second, I address some of key limitations of the DSSAT model in simulating cotton production systems with furrow irrigation management practices. This leads to the need to incorporate a furrow irrigation simulation model, SIRMOD, into the DSSAT model to build a two phased-crop simulation modelling design with integration of two simulation modelling processes – furrow irrigation simulation modelling using SIRMOD and crop production simulation modelling using DSSAT. This approach enables generation of more accurate, valid, useful and higher quality information under various scenarios of *G* x *E* x *M* conditions, and over a long period of time, to better support sustainability-related decision making and control.

Given the development of a new EPMS (WESM) in the first part of the thesis which includes a range of new water and economic performance measures, the question of what the targets of

<sup>&</sup>lt;sup>23</sup> CROPGRO-Cotton is the cotton specific module in DSSAT. In the thesis I will refer to this as being the DSSAT model.

these performance measures look like is difficult to answer as no data currently exists for this. In Chapter 7, the WESM model will be run the first time to enable me to create initial targets for the model. This relates to the second stage of the cybernetic control. More specifically, based on the designed two phased-crop simulation modelling method, a cropping system using current water management practice is simulated using data reflective of current practice in order to quantify performance. For each of the key water and economic sustainability measures presented in the WESM model, their simulated results reported in Chapter 7 represent the range of values that can be obtained under current operating conditions which provides some initial evidence on performance and guidance for target setting for the new WESM. This will then support the decisions related to how current water management practices can be changed to improve water and economic sustainability performance of crop businesses, which relates to the stage 5 of cybernetic control.

Chapter 8 then creates a scenario (Scenario 1) with 86 years of weather data (from 1924 to 2014 excluding four drought years in which no crops were grown) using the same cropping system as outlined in Chapter 7. Then for purposes of comparison, sets up a more advanced (improved) water management practice (Scenario 2) using the same cropping system. The simulated results over 86 years of weather data under Scenario 2 are compared against that of Scenario 1 and variance analysis is performed to examine and identify the driving factors of difference in the summary water and economic sustainability performance between Scenario 1 and Scenario 2. This represents the feedback process and variance analysis process which are the third and fourth features of a cybernetic control system. The average value over 86 years of weather data obtained from Scenario 2 can be set as the second stage of target setting for the new EPMS.

Chapter 8 also provides a test of the usefulness of the WESM model in supporting crop managers in making better sustainability- related decisions in order to improve their businesses' environmental and economic sustainability performance. More specifically, using the simulation method designed in this thesis, I provide evidence that there is a statistically, economically and environmentally significant difference when moving from the current water management practice to the improved water management practice.

Finally, Chapter 9 provides the summary, discussion, limitations, delineations and conclusions of this study and suggestions for future research.

#### **1.4 Research contributions**

Based on our knowledge gaps addressed in the motivation section, the thesis is expected to make three broad contributions to MA research and a contribution to crop agricultural research and practice.

#### 1.4.1 Contribution 1

This thesis provides MA research with a new theoretical construction of EPMS design -WESM model - by building off PMS and cybernetic control theory along with agricultural science.

The WESM model is constructed based on science and accounting theories to deal with theoretical issues in EPMS design - validity issues - to improve quality of sustainability information to support better sustainability related decision making. It this respect, it provides a theoretical contribution to sustainability management accounting research. More specifically, the construction of WESM addresses the validity issues of EPMS design with regard to crop water sustainability measurement in three ways.

First, using an interdisciplinary approach and integrating crop science into PMS and accounting theories, this research provides insights into the science behind crop water sustainability to better understand the underlying water sustainability issues of the phenomenon (e.g. cropping systems) being studied and capture more precisely both environmental aspect and economic aspect of crop water sustainability. Furthermore, the thesis makes a theoretical and methodological contribution to MA research by demonstrating how to operationalise an interdisciplinary approach in sustainability-related MA research. This also responds to a call for broader cross-disciplinary study by Bebbington & Thomson (2013) who address a number of limitations that have arisen when MA research is treated as a stand-alone discipline in studying managerial approaches to sustainability.

Second, this research examines water sustainability issues at the operational level of a crop business (e.g. the crop production level). By undertaking this approach, I am able to identify the links between operational level decisions (i.e., irrigation method choices) and the overall economic and environmental sustainability performance of a crop business. The significance of this is that overall organisational decisions with regards to economic and environmental sustainability that are made at the business or strategic level are often distant from where the sustainability issues that actually arise and key operational or management decisions that are made and where environmental impacts occur. The research demonstrates the value of studying environmental issues at the level where environmental issues actually arise. The findings support the contention of Ittner and Larker (2001) that it would be fruitful to study MA at different organisational levels where the value drivers are more closely tied to the management action.

Finally, this research establishes theoretical links between sustainability measures at the organisational level and the operational level of crop businesses. By applying a decomposition ratio analysis approach, which is commonly used for financial statement and profitability analysis in financial accounting research, I articulate theoretical and logical links between the summary sustainability measure and operational level measures and tie those measures in a logical and structured way. This allows me to identify drivers of the summary sustainability measure (profit to water cost ratio) to gain better understand how profit to water cost ratio will change as a result of a particular management decision at the operational level, and how the change translates into economic and environmental sustainability value created for the crop business.

#### 1.4.2 Contribution 2

The second contribution relates to the second problem of EPMS design identified in this thesis - developing appropriate methods for designing sustainability targets to steer managerial decisions towards sustainability. In this thesis, water sustainability targets will be developed based on a crop model simulation. By doing this, the thesis makes a contribution to sustainability- related management accounting research by providing a new method to set valid targets, the second feature of a cybernetic system, to support better sustainability-related decision making and control.

In MA research to date, most simulation studies have focused on cost accounting systems generally and activity-based costing particularly. It is anticipated that this study will be the first simulation study on sustainability performance measures and target setting. The application of the simulation method to answer the question of how organisations' performance can be improved will provide opportunities to expand our research in performance measurement systems. Extant management accounting research discusses the

value of using a range of research methods to address the same research question (Atkinson et al. 1997; Ittner & Larcker 2001). It is proposed that opportunities to expand our understanding of the phenomenon being studied are created when "*researchers use the synergy that exists among research methods and across disciplines to study complementary issues*" (Atkinson et al. 1997, p. 81).

Furthermore, by designing a two phased-crop simulation modelling process, integrating two simulation modelling phases (furrow irrigation simulation modelling using SIRMOD and crop production simulation modelling using DSSAT), a more comprehensive picture of a real-world cropping system can be captured and hence the quality of simulated information can be improved to support better sustainability-related decision making and control. This is because by coupling SIRMOD and DSSAT together, it allows DSSAT to get pass a single point description of furrow irrigation and capture the key characteristics of furrow irrigation systems which is the non-uniformity of water distribution along the furrows. In this respect, I enhance the realism of cotton production simulation and improve the quality of crop information input to WESM.

Moreover, it is believed that the empirical results reported in this thesis are the first to be based on two-phase simulation modelling for studying how economic and environmental sustainability performance of a crop business can be improved by moving from the status quo. The results include how the summary water and economic sustainability measure changes as a result of the change in water management decisions and how the change can be translated into better environmental sustainability performance and, at the same time, increase economic value created for the crop business. In this respect, the designed WESM is a useful tool for quantifying, assessing and forecasting the impact of water management decisions at the operational and organisational level of crop businesses using simulation modelling.

#### 1.4.3 Contribution 3

The third contribution relates to the setting selected in the thesis – the agricultural sector. Much of the research in environmental management accounting to date focuses on the manufacturing sector (Figge & Hahn 2013; Henri & Journeault 2008, 2010; Pondeville, Swaen & De Rongé 2013). In contrast, the agricultural accounting topic is still underresearched (Jack 2005), despite the vital role of agricultural sectors to global economic, social
and environmental sustainability. Jack (2005, p. 60) argues that "agricultural academics tend not to be interested in accounting and accounting researchers tend to stay clear of agriculture and related industries". Argilés and Slof (2001) also raise concerns about the sparseness of accounting research in the agricultural setting despite the relative importance of agriculture in the economy of many countries. This study is one of the first to address in detail the design and use of PMS and more specifically EPMS in an agricultural setting. I provide science based theoretical development in relation to how EPMS can be designed and used in agriculture and empirical evidence on how water and economic sustainability performance can be improved. Insight is provided into how management accounting has a role to play in understanding how this important sector may move towards water sustainability. In doing so, a new avenue of investigation is opened up for studying environmental performance management in agriculture.

## 1.4.4 Contribution 4

The contribution to crop agriculture research broadly relates to the theoretical framework of crop water sustainability performance measurement systems at the crop business level. More specifically, I contribute to this agricultural research in two key areas. Both of these areas also have the potential to contribute to practice.

First, I contribute towards overcoming the ambiguity, inconsistency and validity issues associated with water use efficiency measurement in the extant literature (Cammarano et al. 2012; Igbadun et al. 2006; Pereira, Cordery & Iacovides 2012). I do this by developing (based on water and crop science) a set of key water sustainability measures that more precisely reflect water sustainability-related activities and management practices. These exist where water sustainability issues arise - at the crop production level. As part of this I also establish a clear theoretical connection between water coming into a cropping system and crop production coming out the cropping system. In addition, through building the WESM model I link the organisational summary measure (profit to water cost ratio) to its drivers across all levels of the organisation and in particular down to the operational level. To the author's knowledge, this study is the first study to provide a theoretical development of a comprehensive conceptual framework for crop water and economic sustainability performance, which is of interest to both agricultural researchers and practitioners.

Second, using the WESM model and drawing upon current water and crop simulation models, I develop a novel two-phased crop simulation modelling process. This contributes to the crop simulation modelling literature with a new approach by establishing an end-to-end connection between irrigation strategies and their economic and environmental outcomes, improving the quality of simulated information. This approach also contributes to literature and practice in the development of more valid and precise standards and targets for crop businesses' economic and environmental performance measures. This will enable researchers and crop businesses to evaluate actual performance against targets and identify areas for improvement in water and economic sustainability performance – including how to drive better economic and environmental performance simultaneously. Furthermore, this has the potential to provide crop growers greater product value post-farm gate by creating more transparent on-farm sustainability information to the supply chain downstream. As the demand for sustainability related information signals increases for product branding this will become an increasingly important issue.

Moreover, the developed model can provide crop growers with simulation ability for more accessible scenario analysis for crop production. This can translate to the regional level as the model has the capacity to establish how profitable a crop is in comparison to other crops in relation to water resources used. This has the potential to support growers and policy makers in relevant decisions related to water allocation under water constraints.

# Chapter 2. Theoretical Discussion of Water Sustainability

# 2.1 Introduction

The first component of the research question addressed in this thesis is how EPMS can be *designed* in an agricultural setting to support managers in water and economic sustainability-related decision making and control. To answer this, the first issue considered is the development of sustainability measures that more precisely capture performance, both in terms of economics and water sustainability, of crop businesses. In order to do this, it is of crucial importance to have a clear understanding of what water sustainability means generally and how the concept of water sustainability at different levels (globally and locally) can be translated and applied to the individual crop business.

This chapter develops a theoretical discussion of water sustainability at the global and local levels, based on water science and contemporary sustainable thinking and discourse, which can be used as a foundation for conceptualising crop water sustainability at the business level in Chapter 3 and designing valid EPMS in Chapter 4. Section 2.2 reviews the Brundtland Commission's definition of sustainability; this is followed by the review of global hydrological cycle and water mass balance concepts in Section 2.3. Based on this, some of the key issues related to water sustainability at the global and local levels are addressed and a number of key water sustainability principles are developed to provide a theoretical foundation for discussion of water sustainability for crop production in the next chapter of the thesis.

# 2.2 Sustainability definition

#### 2.2.1 The Brundtland Commission's definition of sustainable development

In an attempt to raise public awareness of global environmental issues arising from industrialization and growth, the Brundtland Commission<sup>24</sup> developed and published a broad concept of sustainable development, defined as "*development that meets the needs of the present without compromising the ability of future generations to meet their own needs*" (The Brundtland Commission 1987, p. 1). This definition of sustainable development has become

<sup>24</sup> The Brundtland Commission is also formally known as the World Commission on Environment and Development (WCED).

one of the most commonly used definitions for sustainability research in both natural and social science (Ibaraki 2010).

The concept of human needs is at the heart of the Brundtland definition. The definition has as its emphasis that the main objective of sustainable development is ensuring adequate environmental capacity to satisfy human needs and aspirations (such as food, clothing, shelter and jobs), at present and in the future (The Brundtland Commission, 1987). In addition, the idea of social equity is inherit in the definition so as to ensure equitable opportunities for all people, both in developed and developing countries, to satisfy their basic needs and aspirations for a better life. There are three key aspects embedded in this concept:

- (i) Human beings: The definition is anthropocentric, it sees human beings are the central species and more intrinsically valuable than animals and all the other non-human things on the planet (Brennan & Lo, 2010). This provides the underlying reason why humanity dominates and is the centre for the need to develop.
- (ii) Capacity of the environment: This refers both to the availability of physical resources as well as the assimilative capacity of the environment, defined as the extent to which the natural environment is capable of receiving, degrading and converting waste matter to useful substances for ecosystems (Pearce, 1976), to allow human survival and development.
- (iii) Time: This reflects on the extent to which environmental capacity still meets human needs. In addition, time can also be considered in terms of expected improvements in technology over time.

A number of key sustainability principles can be drawn from the first aspect of the Brundtland definition and the related discussion in the Brundtland Report (The Brundtland Commission, 1987). First, sustainability is about the relation between human needs and environmental capacity. Human development involves exploitation of natural resources (i.e. fossil fuels, minerals and freshwater) and discharge of waste into the environment. Human intervention in natural systems (either via withdrawing natural resources from or discharging waste into the environment) reduces the availability of natural resources and reduces the assimilative capacity of the natural environment, which in turn may constrain the ability of human beings to continue to develop (Greene 2010; Ibaraki 2010; Pearce 1976; The Brundtland Commission 1987; The Economics of Ecosystems and Biodiversity 2010, 2013). Therefore, the discussion of sustainability needs to be placed in the context of anthropogenic impact (human impact) on natural systems of the planet.

Second, due to human intervention in these natural systems, the capacity of the environment to absorb waste (solid, liquid or gas) will change over time. Scientific evidence shows that this capacity is currently a decreasing trend (Ibaraki 2010; Intergovernmental Panel on Climate Change 2011; Patzek 2004; Patzek & Pimentel 2005; Pearce 1976; The Brundtland Commission 1987; The Economics of Ecosystems and Biodiversity 2010, 2013). This trend is predicted to reach a critical level (i.e. at which natural resources become alarmingly scarce and waste residuals greatly impact on human health and hinder economic development) if no actions are taken in the near future (Pearce 1976). While several key natural resources (such as freshwater, fossil fuels, minerals) are still available at the present, some of them will not be available in the next several decades (Patzek 2004). The fact that Earth is still capable of receiving waste does not guarantee that the volume of waste residuals will not exceed the assimilative capacity of the natural environment in the near future (Patzek 2004; Pearce 1976). Therefore, a time dimension must be included in the discussion of sustainability.

Third, while its focus is to meet the present needs, the Brundtland definition also suggests that current generations owe a duty to future generations and other living beings to conserve natural capital (The Brundtland Commission, 1987). Even though human beings might be able to meet their needs at the present, if current generations continue to exploit and consume natural resources and emit waste to the environment at the current rate without implementing appropriate plans to preserve environmental capacity, future generations may not be able to meet their own needs due to a shortage of key natural resources and low quality of ecosystem services (Massoud et al. 2010; Patzek 2004; Pearce 1976; Schäfer & Beder 2006; The Brundtland Commission 1987; The Economics of Ecosystems and Biodiversity 2013).

#### 2.2.2 The Brundtland definition in practice

Although the Brundtland definition is widely accepted, in part due to its appeal to principles of equity and self-preservation, it has two main of limitations – especially when applied to the sustainability of individual natural resources. First, as its focus is at the macro level (such as economy or society), the definition is too abstract and complex to provide an exact meaning

of sustainability at an organisational level (Azapagic & Perdan 2000; Farrell & Hart 1998). Evidence shows that some of the key global environmental issues have arisen directly from organisational activities through their consumption of physical resources and discharge of waste into the environment (The Economics of Ecosystems and Biodiversity 2013). This means it is the organisation's operational level where human activities are actually carried in order to meet the needs of the present generation (Patzek & Pimentel 2005). Therefore, discussion of sustainability of natural resources needs have a focus at the organisation's operational level to capture the issues of sustainability actually occurring at that level.

Second, it is a challenge to persuade individual organisations to act in the "common interest" as addressed in the Brundtland definition. In addition, it is difficult to see how intergenerational decisions can be made efficiently by individuals. Furthermore, environmental and ecological interactions do not respect the boundaries of individual ownership and political jurisdiction. From a classical economics point of view, the goal of business is to maximise shareholders' value with the key objective being to increase productivity. However, in the contemporary business environment, organisations in many industries, such as agriculture, manufacturing or petroleum industries, have faced challenges of operating in a more efficient and more environmentally friendly manner in order to move towards sustainable development (Barney 1991; Hart 1995; Joshi, Krishnan & Lave 2001; Rainey 2010; Williams & Martin 2011; Wisner, Epstein & Bagozzi 2006). Therefore, the link between economic performance and environmental performance of a business needs to be demonstrated explicitly to encourage individual businesses to pursue an evolutionary trajectory towards sustainability.

In summary, while the Brundtland definition provides useful guidance in how to think about sustainability at a macro level, it's translation into a more meaningful, clear, rigorous and relevant definition at the organisational level still remains a challenge.

# 2.3 Water sustainability

#### 2.3.1 Global natural water system

As discussed in Chapter 1, this thesis focuses on water sustainability for crop production. In order to gain an understanding of what water sustainability means and identify potential approaches to the sustainable use and management of water, both globally and locally, it is

necessary to have a fundamental understanding of hydrology (Barth et al. 2010; De Wever 2010; Graedel & van der Voet 2010; Ibaraki 2010; Kanae 2010; Knepper & Ternes 2010; Lindner et al. 2010). This section reviews the water cycle and water distribution to provide background on global water flows and water stocks. This is used as a foundation for discussing issues related to water sustainability when considering human impact on the natural water systems.

# 2.3.1.1 Global water cycle

Water is considered one of the most important natural resources essential for human life (Ibaraki 2010). In addition, water resources are fundamental elements for economic development (including transport) and the maintenance of ecological systems (Ibaraki 2010).Of the three main uses of water, including domestic, industry and agriculture, agriculture is currently the highest consumer of freshwater (i.e. for irrigation) (Kanae 2010).

Water can exist in three phases: gas (i.e. water vapour or moisture that exists in the atmosphere), liquid (water drops in a form of rain, surface water, groundwater or sea water) and solid (i.e. snow, ice), and can change from one phase to another in the hydrological cycle<sup>25</sup> (Ibaraki 2010). Figure 2.1 illustrates the hydrological cycle.



Figure 2.1 The hydrological cycle Source: Zhang, Walker & Dawes 2002

<sup>&</sup>lt;sup>25</sup> Water circulation that occurs naturally near the Earth's surface.

The hydrological cycle refers to water circulation that occurs naturally near the Earth's surface (Ibaraki 2010; Zhang, Walker & Dawes 2002). It is an endless recirculatory system that links water in the atmosphere, oceans and on the continents though several natural processes, including evaporation and transpiration (liquid water vaporized from the surface water or from plant stomata) (Avissar et al. 1985), precipitation and water run-off (Barth et al. 2010; De Wever 2010). The four key processes of the global hydrological cycle can be quantified as presented in Figure 2.2.



## Figure 2.2 Annual global flux water cycle Source: Barth et al. (2010)

Annual global evaporation (from both the ocean and the continents) equals the amount of annual global precipitation (to the ocean and the continents) (see Figure 2.2). This is the natural water mass balance at the global scale; in the natural hydrological cycle (the global-scale, closed system). Water does not disappear but moves from one place to another around the Earth surface in the hydrological cycle (Ibaraki 2010).

In addition, as the global hydrological cycle is a closed system, water does not disappear but changes from one phase to another by melting (solid to liquid), evaporating (liquid to gas), sublimating (solid to gas), condensing (gas to liquid) or freezing (liquid to solid) and moves from one place to another around the Earth's surface (Ibaraki 2010). Notably, water is unevenly distributed on the Earth, determined by a number of geographical factors such as

gravity, energy (i.e. solar or wind energy), climatic conditions, and other forces (De Wever 2010).

# 2.3.1.2 Global water distribution

Understanding the distribution of freshwater and saltwater in the hydrological cycle helps us to quantify the availability of freshwater resources and examine water sustainability (Ibaraki 2010). Figure 2.3 shows global water distribution in the hydrologic cycle.



Figure 2.3 Global water distribution in the hydrological cycle Source: (Ibaraki, 2010)

Global water distribution in the hydrological cycle reveals that, while only freshwater resources can be used for domestic, industrial and agricultural purposes, freshwater comprises only 2.5% of global water (see Figure 2.3). Furthermore, while glaciers and permanent snow cover, stored in polar and mountainous regions (i.e. Antarctica and Greenland), are the main components of freshwater (making up approximately 70% of global freshwater), groundwater (making up 30% of global freshwater) is the largest source of freshwater that is readily accessible. This means only 0.76% of global water resources, in the form of groundwater and surface water, can be considered as usable freshwater.

While it is evident that the amount of freshwater to meet all human water needs is limited (De Wever 2010; Ramankutty 2010), Barth et al. (2010) predict that demand for freshwater will increase with future population growth and associated agricultural and industrial development, in addition to the higher living standards (and consumption of water) in some parts of the world. Therefore, one of the main foci for water sustainability examination is

groundwater, including its characteristics and its availability (in both temporal and spatial dimensions), and how it is being used and managed at the local level.

#### 2.3.2 Local water systems

In order to evaluate the availability and sustainability of freshwater resources at a regional or local level, the application of the principle of conservation of mass to flows of water in a natural system (e.g. watershed, water basin) and change of its water storage volume is needed.

## **2.3.2.1** The principle of conservation of mass of water

The principle of conservation of mass of water is that the water mass that enters a system (within the *control volume*) must leave the system. Accordingly, the annual rate of water inflow (*I*) to the control volume must equal the annual rate of water outflow (*O*) from the system (Ibaraki 2010). Water balance in a natural system (i.e. watershed) can be expressed as follows<sup>26</sup>:

$$I = 0$$
 [2.1]

Or

$$I - O = 0 \tag{2.2}$$

Furthermore, by assuming that the density of water, therefore it's volume, is constant, under average conditions, water stored in a natural system does not change its volume significantly. Accordingly, dV/dt equals zero.

$$\frac{dV}{dt} = I - 0 = 0 \tag{2.3}$$

where V is the volume of water stored within the control volume and t represents the time dimension; dV/dt represents the time rate of change of mass inside of the control volume.

Moreover, the two water flow components, *I* and *O*, can be further decomposed into their components, as expressed in Equation 2.4.

$$\frac{dV}{dt} = p - (r_s + r_g + et) = 0$$
[2.4]

 $<sup>^{26}</sup>$  The equations (2.1) to (2.6) are derived from Ibaraki (2010).

where *p* denotes the average annual rate of precipitation  $(L^{3}T^{-1})$ , the only component of volumetric inflow; *r<sub>s</sub>*, *r<sub>g</sub>* and *et* denote the average annual surface flow water rate, the average annual groundwater discharge (or groundwater run-off) rate, and the average annual evapotranspiration rate  $(L^{3}T^{-1})$ , respectively. They are the three natural components of volumetric outflow.

Equation 2.4 illustrates that in a natural system, the water inflow component (precipitation) is equal to water outflow components (surface water, groundwater discharge and evapotranspiration); and water stored (water in rivers, lakes, aquifers and the soil) in a natural system (within the control volume) does not change significantly. Figure 2.4 represents the water inflow and water outflow components of a natural system (e.g. watershed).



Figure 2.4 Water balance at a watershed (a natural system)

More importantly, Equations 2.3 and 2.4 include water flows, a storage term (V) and time scale (t), which are key elements for discussion in relation to water sustainability in the next section and in particular the impact humans have when withdrawing V from the system.

## 2.3.2.2 Human impact on local water systems

Freshwater from the environment (including groundwater and surface water) is withdrawn by human beings to meet their water needs for agriculture (irrigation), industrial and household purposes (Barth et al. 2010). Human impact on freshwater availability and sustainability needs to be included in the water balance equation (Ibaraki 2010). Accordingly, Equation 2.4 can be modified to reflect water balance at a watershed – that is the intervention from human activities:

$$p + i_r + i_a = r_s + r_q + et + r_w$$
[2.5]

where  $i_r$  and  $i_a$  denote recycled sewage water and agriculture return to the system ( $L^3 T^{-1}$ ), respectively, and  $r_w$  denotes the rate of water withdrawal from surface water and groundwater ( $L^3 T^{-1}$ ).

Figure 2.5 visualises the water inflow (I) and water outflow (O) components of a watershed with human intervention.



Figure 2.5 Water balance at a watershed (with human intervention)

Human-induced flow components ( $i_r$ ,  $i_a$  and  $r_w$ ) play an important role in the change in the volume of water stored within a local water system, which is implied as zero in the natural system (see Equation 2.6<sup>27</sup>).

$$\frac{dV}{dt} = (p + i_r + i_a) - (r_s + r_g + et + r_w)$$
[2.6]

In some parts of the world, freshwater (e.g. surface water and groundwater) is withdrawn significantly by humans for irrigation. Over time, this leads to a considerable negative change

 $<sup>^{27}</sup>$  Assuming parameters on the right hand side of Equation 2.6 are measured at the same time scale as t.

of water stock in these local catchments ( $dV/dt \ll 0$ ), resulting in water shortage and competition for water among users.

In addition, water can be withdrawn from one system but be discharged into another. Consequently, significant artificial water recharge leads to unexpected positive change of water storage within the local water systems (dV/dt >> 0), resulting in groundwater salinization (as groundwater levels rise) and flooding problems in these areas.

# 2.3.2.3 Issues with sustainability of water

As discussed above, at the global level, water is considered sustainable in the context of the hydrological cycle as a whole. For example, unlike other commodities such as oil and coal that are defined as non-renewable resources<sup>28</sup>, water exists in the endless recirculatory hydrological process. In addition, water that is used for irrigation, industrial and household purposes will return to the hydrological cycle and eventually will be available for reuse (Ibaraki 2010). However, when considering human impact on local water systems in respect of time, space and usability<sup>29</sup> of water, and in particular freshwater, there are a number of issues related to its sustainability.

First, human intervention causes imbalance in the water system. For example, water used for crop irrigation may lead to greater evapotranspiration and runoff (therefore affecting *t*) than in the natural system. As a result, annual global flux water cycle and global water distribution changes over time with a decrease in the freshwater reservoir volume (e.g. water stocks in river and lakes over time may be depleted) (Ibaraki 2010).

Second, the stocks (V) of freshwater resources upon which humanity can withdraw are limited. From the discussion in the previous section, only available freshwater resources (mainly surface water and groundwater), can be used by humans to meet their water demands (for example, for irrigation, industrial and household purposes). It is projected that water abstraction in 2050 will be from 50,000 to 80,000 km<sup>3</sup> per year due to growing world

<sup>&</sup>lt;sup>28</sup> In the context that the rate of use is much greater than the rate of generation.

<sup>&</sup>lt;sup>29</sup> Usability in a sense of both quantity (available and ready to be used) and quality (can be used for particular purposes).

population. Moreover, water abstracted<sup>30</sup> is not necessarily water actually being used to meet human needs. While a small proportion of water is consumed for irrigation, industrial and household purposes, overdrawing of water occurs due to inefficient use of water and/or poor water management (Barth et al. 2010).

Third, water abstraction leads to intense water competition among regions and among water users within a region. Distribution of available freshwater varies to a more or less significant degree in both spatial and temporal dimensions. Water can be withdrawn in one place but discharged in another place, ranging from several to some thousand kilometres in distance. Yet, when water is abstracted from a groundwater system (i.e. from a well), water at the discharge/pumping area will be recharged from surface water bodies (i.e. aquifer storage). For example, the timescale for water being recharged in the source water body can vary tremendously from less than a day to more than a million years, depending on the velocity of water which is driven by the hydrological properties of the aquifer and the distance between the discharge and recharge points (Ibaraki 2010).

Fourth, there is a time issue in respect to dV/dt where V is the volume of a necessary resource or waste remediation process. So far, we have, as a society, been thinking as though V was infinite, but it was not and now V has been more or less seriously reduced. The dV/dt is probably accelerating in a negative way. Therefore, the first element of time is when dVdecreases towards zero. The second element of time is that we need to reverse dV/dt, that is, reduce its negative growth and eventually, not only get dV/dt to equal zero, but go beyond it to dV/dt greater than zero as water stocks are rebuilt. This will occur either through human catastrophe or the development of new technology, and probably social conventions.

Finally, water abstraction leads to water quality issues. A proportion of withdrawn water that is consumed for human activities gradually returns to the environment but generally with lower quality (Ibaraki 2010). This not only degrades systems but also decreases future freshwater availability as a considerable amount of water discharged is not ready to be reused or is costly to treat and recycle (given the current technology).

<sup>&</sup>lt;sup>30</sup> Water withdrawals are defined as "the amount of water taken out of rivers, streams or groundwater aquifers to satisfy human needs for water" (Rijsberman, 2006, p8).

Ibaraki's (2010) work, in conjunction with several water sustainability studies mentioned above, provides a number of key concepts for water sustainability. First, it is argued that water is a resource only if it is available: (i) at the place where it is required, (ii) at the time when it is needed, (iii) at the quality required for its particular use, (iv) in sufficient quantity, and (v) at a low enough cost.<sup>31</sup> Therefore, the discussion of water sustainability is not only about available water at the global scale but mainly about whether it can be ready and suitable for use at a local level. In addition, unlike other natural resources, water sustainability is more about maintaining the balance in the flow of water through the cycle than it is about water stocks as it is, ultimately, naturally recirculated.

#### 2.3.3 Water sustainability principles

Based on the Brundtland definition of sustainable development (The Brundtland Commission 1987), the concept of the hydrological cycle and water mass balance, and the discussion of some of the key theoretical issues with water sustainability in the previous sections, this section aims to propose a number of water sustainability principles to provide a foundation for discussion of water sustainability for crop production in the next sections.

The heart of water sustainability principles is the consideration of the use of available freshwater resources by the current generation to meet their own needs, which will cause negative impacts on the global hydrological system and availability of freshwater resources for future generations (The Economics of Ecosystems and Biodiversity 2013).

## **2.3.3.1 Principle 1**

Discussion of water sustainability needs to be placed in the context of human intervention in the natural environment through abstraction (e.g. water withdrawals from the natural environment) as well as discharges impacting on natural systems (e.g. impacting on the quality of water in the natural environment or adding to the system). This causes imbalance in the natural hydrological system and shifts the equilibrium point of the water cycle to another state (in which available freshwater at a particular place reduces over time).

<sup>&</sup>lt;sup>31</sup> For example, though ocean water is the major water resource and abundant, it is currently costly to convert salted water into freshwater using desalination techniques.

## 2.3.3.2 Principle 2

Available freshwater resources (groundwater and surface water) are the water resources that can be consumed by human beings to meet their own needs. In order to conserve water stock in the long-term, minimizing water abstractions from these natural resources is necessary. Therefore, efficient management of groundwater and surface water abstraction needs to be considered as a step towards water sustainability.

## 2.3.3.3 **Principle 3**

In the previous section, it is argued that water can be considered as a usable and useful resource only if it is available (i) where it is required, (ii) when it is needed, (iii) with a quality for its particular uses, (iv) in sufficient amounts to meet the particular needs, and (v) at a reasonable cost<sup>32</sup> (Ibaraki 2010; Kanae 2010). Therefore, the discussion of water sustainability is not really about available water at the global scale but mainly about whether it is available and suitable for use at a local level.

## 2.3.3.4 Principle 4

Water stored within a local water system is the resource from which water is withdrawn. However, change in volume of water storage over time, dV/dt, does depend on water flows of the system. Therefore, discussion of water sustainability needs to take into consideration of both water stock and flows of water.

## 2.3.3.5 **Principle 5**

Water plays a critical role in production processes of agriculture and several important industries (e.g. food and manufacturing) (Barth et al. 2010; Rijsberman 2006). This means inadequate water supply will impact business production outcomes, and hence, its profitability (Cammarano et al. 2012; Jackson et al. 2011; Long et al. 2010). However, it is evident that we face the risk of a shortage of freshwater resources (Jackson, Khan & Hafeez 2010; Raskin et al. 1997; Rijsberman 2006). This leads to increasing competition among water users, increased water costs, and more stringent environmental regulation, which makes it more difficult for businesses with high water needs to maintain their economic growth in the long-term (Cammarano et al. 2012; Jackson et al. 2011; Khan et al. 2006). Therefore, the

<sup>&</sup>lt;sup>32</sup> For example, though ocean water is the major water resource and abundant, it is currently costly to convert salted water into freshwater using desalination techniques.

link between business profitability and water use needs to be taken into consideration to capture both economic sustainability (e.g. as a going concern) and the sustainable use of water (the environmental sustainability of the enterprise with respect solely to its water use) of the business.

# 2.4 Conclusion

This chapter provides a theoretical discussion of water sustainability at the global scale which lays out the foundation for conceptualising water sustainability at the crop business level discussed in Chapter 3. This includes development of a set of water sustainability principles, based on hydrology and contemporary sustainable thinking, to provide a clearer and better understanding of what defines water sustainability generally. In summary, the heart of the set of water sustainability principles is the consideration of human intervention in the natural environment (e.g. water abstraction from the natural environment) as well as human impact on natural systems. In addition, it is argued that water sustainability is not really about available water at the global scale but mainly about whether it is available and suitable for use at a local level. Therefore, discussion of water sustainability needs to take into consideration of both water stock and flows of water, and efficient management of groundwater and surface water abstractions needs to be considered to minimise disturbance on the natural environment as a step towards sustaining water systems. Furthermore, for crop businesses, in order to maintain going concern, they need to consider both economic and environmental sustainability. In this respect, the link between business profitability and water use needs to be taken into consideration to capture both economic sustainability and water sustainability of the business. These water sustainability principles will be applied to a specific local focus which is cotton businesses in Australia.

Furthermore, this chapter theoretically addresses key parameters that define an overall definition of what constitutes water sustainability, which provides overall targets for Chapter 3. One of the key parameters that defines water sustainability which has not yet been considered in extant water sustainability literature is that if an unsustainable human activity results in V decreasing to a point where other legitimate uses (i.e. ecosystem stability, domestic use) become insufficient then the unsustainable activity should be modified so that its contribution to decreased dV/dt is reduced. This definition also allows for unsustainable drinking water abstraction to be modified by return of wastewater. At the crop business level,

such general definition of water sustainability can be applied to address the extent to which crop production withdraws water from the environment and how runoff from irrigation can be managed and recycled to reduce the water impost of crop cultivation on the environment.

# Chapter 3. Theory of Crop Water Flows

# 3.1 Introduction

The previous chapter discussed issues related to water sustainability at the global and regional levels. The environmental impact on water in a natural ecosystem was described in terms of its contribution to the negative dV/dt characteristic of such systems. Sustainability requires that dV/dt approaches zero, and in the case of severely disrupted systems, it should become positive.

Water use in crop production will now be assessed in this context, with the goal of identifying the reasons for the tension generally assumed to lie between environmental and economic sustainability. Broadly they can be sketched as follows. Economic sustainability requires profitability and, in a water-limited cropping environment, that will mean abstraction of water for irrigation. However, this leads to a negative dV/dt, and, as we have just discussed, environmental sustainability requires a positive dV/dt.

While the issue of improving water sustainability in agriculture, a setting that has the largest human imposts on *V*, has drawn increasing attention (Moore, Robertson & Routley 2011; Pereira, Cordery & Iacovides 2012; Tennakoon & Milroy 2003), most research has focused on crop water use efficiency.<sup>33</sup> However, when a holistic view is taken, from the point of water abstraction through to making a profit, a more nuanced perspective is necessary.

Section 3.2 lays the foundation of such a view, describing, from a theoretical basis, water flows through a cropping system.<sup>34</sup> In aggregate, this description encompasses what I term the *crop water process model* which provides a theoretical link between water flow components and crop yield at an operational level. The model also allows identification of a number of key sustainability performance measures in Section 3.3 which will form part of the EPMS designed in Chapter 4. Finally, the conclusion contains a summary of the above and lays out how this provides the theoretical basis for the design of a decomposition ratio model in Chapter 4.

<sup>&</sup>lt;sup>33</sup> Crop water use efficiency is defined as the ratio of crop yield to crop water use (Tennakoon & Milroy 2003).

<sup>&</sup>lt;sup>34</sup> A cropping system is defined as the volume of land, soil and atmosphere where a crop is grown.

## **3.2** Crop water process model

Water is considered at the heart of crop agriculture; the primary process of crop production is photosynthesis through which transpiration (the evaporation of water from plants) and atmospheric carbon dioxide are exchanged and the latter is used in the synthesis of organic carbon compounds (i.e. sugar) to feed the plants using energy obtained from sunlight (Pereira, Green & Villa Nova 2006). Some of the carbon is used to provide metabolic energy to the plant with the remainder used for growing structures (stems, leaves, roots and so on) and reproductive structures (fruit, seeds) which are generally the products valued by mankind.

Insufficient water supply during crop growth significantly affects crop yield and quality (Bange & Constable 2006; Cammarano et al. 2012; Jackson et al. 2011; Long et al. 2010). There is some scientific evidence that shows crop agriculture is facing a challenge of rainfall variability (in terms of both rainfall amount and frequency), which partly arises from changes in the climate (Jackson et al. 2011; The Economics of Ecosystems and Biodiversity 2013) and partly from growing competition from the environment and urban growth (Gleick 1998; Ibaraki 2010). Rainfall variability affects crop yield and quality of rainfed crops<sup>35</sup>, and increases reliance on environmental abstraction of water<sup>36</sup> for irrigated crops (Grassini et al. 2011; Jackson et al. 2011; Jackson, Khan & Hafeez 2010; Mateos et al.; Mehta et al. 2013; Oweis, Farahani & Hachum 2011; Pereira, Green & Villa Nova 2006; Smith, Raine & Minkevich 2005; Tennakoon & Milroy 2003; Yeates, Constable & McCumstie 2010). For example, it is estimated that around 85% of Australian cotton crops are irrigated (Cammarano et al. 2012). Similarly, in California, agriculture is heavily dependent on irrigation water which is withdrawn from surface and groundwater and accounts for 80% of total freshwater withdrawals in the state (Mehta et al. 2013). Therefore, irrigation water is considered as one of the key water inputs for a crop and may need to be added in a cropping system.

This thesis considers two sustainability aspects of crop businesses – water and economic sustainability, and how to incorporate them into a single assessment of crop businesses' sustainability performance. To do this, first it is necessary to gain an understanding of how water flows across a cropping system and how water relates to crop production output.

<sup>&</sup>lt;sup>35</sup> Rainfed (or dryland) crops are crops that rely only on rainfall not irrigation for water.

<sup>&</sup>lt;sup>36</sup> Environmental abstraction of water refers to irrigation water, which is water withdrawn from natural water systems, such as groundwater or rivers (Ibaraki 2010).

## **3.2.1** Water balance for a root zone

This section provides a background of water balance for a root zone under natural conditions. Figure 3.1 is a schematic diagram of the water balance for a root zone. The root zone is a crucial focus as this is the soil volume, explored by roots, which holds the water resource on which the plant draws, by root uptake. The plant depends explicitly on this and, with rare exceptions, requires that the soil water resource is reasonably frequently replenished.



Figure 3.1 Schematic diagram of the water balance for a root zone Source: Zhang et al. 2002 Precipitation or rainfall is the key water inflow component to the root zone<sup>37</sup>; whereas soil evaporation, transpiration<sup>38</sup>, deep drainage and run-off are the four key water outflow components (see Figure 3.1). The difference between water inflow to and water outflow from the root zone is change in soil water. Soil water storage is defined as the volume of water that is held in the soil within the plant's root zone (Zhang, Walker & Dawes 2002). Accordingly, water balance for a root zone can be expressed as follows:

Change in soil water = Rainfall - (Transpiration + Soil evaporation + Deep drainage +Runoff) [3.1]

Of the four water outflow components, transpiration refers to the amount of water lost from plants through stomata, whereas soil evaporation refers to water evaporated from wet soil surfaces. Water lost by transpiration is exchanged for carbon assimilation, the primary process of plant growth (Passioura 2006; Pereira, Green & Villa Nova 2006). In addition, deep drainage refers to water that drains beyond the plant root zone. Deep drainage occurs when water inputs exceed the soil water storage capacity of a plant. In contrast, run-off refers to excess water from rain that flows over the land – also called as surface runoff. This water loss occurs when water is applied to the soil surface faster than it can infiltrate the soil (Zhang, Walker & Dawes 2002).

Not all water held in the soil is equally available to support plant growth. Excess water is held only temporarily and some water, at the drier end of the range, cannot be extracted by actively growing plants. This issue is elaborated in Section 3.2.2.4.

#### 3.2.2 Crop water process model

In this section, a water process model for an irrigated crop is constructed, based on water balance for a root zone described above, to lay out three main water components of a cropping system; crop water supply (the flow of water supply to the cropping systems), transpiration (the flow of water contributing to productive output), and crop water loss (the flow of water representing non-productive water).

<sup>&</sup>lt;sup>37</sup> For irrigated crop, irrigation is an additional water input to supplement precipitation when the latter is insufficient to adequately meet crop needs. <sup>38</sup> The amount of water lost from plants through stomata.

# **3.2.2.1** The cropped field

An irrigation<sup>39</sup> farm generally has two scales at which water can be considered. The field scale encompasses the operations and flows in a defined sub-area of the farm, growing a particular crop. There may be many fields, not all operating in synchrony. The farm, in contrast, has a unitary aspect, with generally one central water storage and one, or a few, water sources replenishing the storage. Irrigation water for each field is drawn from the storage. There are exceptions to this, but it is a generally useful construct to identify key water processes. I begin at the field scale. Figure 3.2 demonstrates a water process model for a cropped field.

<sup>&</sup>lt;sup>39</sup> Irrigation is referred as the supply of additional water to a crop so that any inadequacies in supply of water from rain are alleviated.



Figure 3.2 Crop water process model

Based on the crop water process model, discussions of each type of water resource, including how it is sourced, will be laid out to provide insights into its environmental sustainability which will be discussed in Section 3.3. Furthermore, the link between water for irrigation applied to a cropped field and water storage at the farm scale will be discussed in Section 3.2.3.

#### 3.2.2.2 **Crop water supply**

Water supply to the crop includes all types of water resources flowing into the cropping system. As described in Eq.  $(3.2)^{40}$ , crop water supply for irrigated crops typically comprises rainfall, irrigation applied to the cropping system, and run-on from adjacent areas. The formula for crop water supply of a cropped field, *i*, in a particular crop season, *t*, is therefore:

Crop water supply<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Rainfall<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Irrigation applied<sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
+ Run - on<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [3.2]

In this thesis, crop water flow components are measured in mega litres<sup>41</sup> per hectare (ML/ha).

#### Rainfall

Rainfall comprises two components; rainfall directly on the field and rainfall on catchments, from which runoff and seepage contribute to the inflow of water systems, such as reservoirs, dams and rivers. In this thesis, the term "rainfall" is defined as comprising the rainfall component that falls directly on the cropped field while the rainfall component on catchments is defined within the irrigation component of water supply to a cropping system.

Rainfall frequency and amounts are a characteristic of a given location, with long-term climatic trends and familiar short-term variation around those trends (Langousis & Kaleris 2012). Historical data of the climate factors of a particular geographical area (climate station)

 $<sup>^{40}</sup>$  The equations (3.2) to (3.11) are derived from the crop water process model constructed in Section 3.2.2 (Figure 3.2). <sup>41</sup> 1 mega litre is 1,000, 000 litres. Therefore 1 mm of applied (or lost) water equals 100 ML/ha.

can be obtained from climatic databases, for example SILO climate data<sup>42</sup>, for Australian locations.

The major problem of rainfall availability is that the long-term availability of water in any location may be affected by changes in climate (Jackson et al. 2011). Therefore, relying on rainfall as the main water source for crop production may lead to low production yield and low quality in unfavourable climatic seasons (Hochman et al. 2013; Jackson et al. 2011). In dryland cropping, irrigation is not available and the crop must be managed so that its transpirational demands do not outstrip water supply from rainfall at the site (Hochman et al. 2013; van Ittersum et al. 2012).

Rainfall falling on the crops<sup>43</sup> is considered as a free water resource<sup>44</sup>(Cammarano et al. 2012). This means crop businesses will not incur direct water costs for crop production if solely rainfed. However, crop businesses have to manage water risk and consider the tradeoff between water cost savings (when irrigation is available) and crop profitability (Cammarano et al. 2012; Jackson et al. 2011).

#### *Irrigation*

Irrigation is the supply to the field of supplementary water sufficient to alleviate any water deficits resulting from inadequate rainfall. In this respect, irrigation is carried out to potentially eliminate the risk of not completely satisfying transpirational demands (Cammarano et al. 2012; Gaydon, Meinke & Rodriguez 2012; Grassini et al. 2011). The intent is to distribute water evenly to the soil so that the entire root zone can be uniformly rewet. Successful irrigation therefore comprises two key aspects; distribution and scheduling.

#### Distribution a.

A mechanism is used to transport water from one entry point to all points of the field. In the case of pressurised systems such as sprinklers, their design should achieve the necessary uniformity. Surface flow systems such as flood or furrow irrigation use the soil surface itself as the transport system with water flowing down a slight gradient engineered into the surface.

<sup>&</sup>lt;sup>42</sup> SILO is an enhanced climate database hosted by The Science Delivery Division of the Department of Science, Information Technology, Innovation and the Arts (DSITA), providing Australian climate data from 1889 to the <sup>43</sup> There are legislative restrictions on retention of runoff from non-cropped areas.

<sup>&</sup>lt;sup>44</sup> In terms of direct water cost.

In all systems, water distribution is followed by infiltration, the movement of liquid water into the soil, generally in a downward direction. Sources of inefficiency, particularly for furrow irrigation, will be discussed later in Chapter 6.

#### b. Scheduling

For most irrigation systems, water is applied in a batch-wise way, refilling the storage capacity of the root zone in a short period. The plants draw water from soil water reservoirs over the next period, which may be in the order of days or weeks. The key management decisions are when to replenish the crop soil reservoirs and how much water to apply. Inadequate rewetting and excessive rewetting both can harm crop growth and the efficiency with which water is used. Relevant details will also be discussed in Chapter 6.

#### Run-on

Run-on refers to the amount of water run-on to a farming system from up-slope land. The amount of run-on depends on soil type, slope and condition, up-slope of the crop site (Zhang, Walker & Dawes 2002). Typically, water run-on is a small component<sup>45</sup> and is ignored in this thesis.

#### 3.2.2.3 Crop water loss

Water loss from the system occurs as either water vapour lost in evaporation (soil evaporation and transpiration) or as liquid water loss. These two components are classified as evaporation and water output. The separation is made because while a liquid water flow can be a mechanism for transporting soil, salts, pesticides and nutrients out of the cropping system into the environment, water vapour cannot, of itself, be polluting. The crop water process model demonstrates a link between liquid water flows (e.g. deep drainage and run-off) and quality of water output from the cropping system (see equation 3.3).

The formula for crop water loss can be expressed as follow:

Crop water 
$$loss_{it} \left(\frac{ML}{ha}\right) = Water output_{it} \left(\frac{ML}{ha}\right) + Evaporation_{it} \left(\frac{ML}{ha}\right)$$
 [3.3]

<sup>&</sup>lt;sup>45</sup> We ignore replenishment of the soil water reserves by flooding across river floodplains.

#### **Evaporation**

Two categories of evaporative water loss from the cropped field system are defined as evaporation from soil surfaces (soil evaporation) and from plants as (transpiration).<sup>46</sup> As discussed earlier, during carbon assimilation, transpired water is exchanged for carbon dioxide which is converted into plant biomass comprising grain, fibre or other harvestable product (Passioura 2006). While transpiration is considered as a productive water component, soil evaporation is considered as a non-productive component. In other words, evaporation can be partitioned into productive (transpiration) and non-productive (soil evaporation) components.

Traditionally, soil evaporation and transpiration are combined as **evapotranspiration**, which is calculated with the Penman–Monteith equation (Pereira, Green & Villa Nova 2006). In this thesis, soil evaporation and transpiration are presented separately in order to distinguish between productive water (transpiration) to produce an economic product (e.g. cotton lint) and non-productive water (water lost from the soil due to evaporation).

Evaporation<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
 = Transpiration<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Soil evaporation<sub>it</sub>  $\left(\frac{ML}{ha}\right)$ 
[3.4]

As discussed above, wet soil surfaces, for example, after rain and following surface or sprinkler irrigation, initially evaporate water at a rate similar to an open water surface (Khan & Hanjra 2008; Salvador et al. 2011). However, once surface drying occurs, the evaporation rate decays at the square root of time (Sommer et al. 2012). Therefore, frequent rewetting of the surface potentially exposes the field to significant evaporative losses. Such non-productive water loss needs to be taken into consideration in the discussion of water sustainability performance of a crop business.

#### Water Output

In the crop water process model (see equation 3.5), two categories of liquid water discharged from the system are presented; deep drainage and run-off.

<sup>&</sup>lt;sup>46</sup> At the field level, we do not consider evaporation from free water surfaces (e.g. distribution channels, on-farm storage). In addition, because free water surfaces exist in the field for relatively short periods in the crop cycle, a simplifying assumption was made that free water surface evaporation is equal to zero at the field scale for this thesis.

Water output<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
 = Deep drainage <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Runoff <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [3.5]

#### a. Deep drainage

Deep drainage from the field refers to water loss that occurs when either rainfall or irrigation water inputs to the root zone exceed the water deficit in the soil<sup>47</sup> (Moore, Robertson & Routley 2011; Smith, Raine & Minkevich 2005). As a result, excess water drains from the soil store to deep soil strata where the roots cannot reach it (Zhang, Walker & Dawes 2002). Loss through deep drainage may be substantial for surface irrigated crops, which potentially leads to significant environmental harm and an annual loss of 2.5 ML/ha of water that could be beneficially used for growing additional crop (Smith, Raine & Minkevich 2005).

#### b. Runoff

When water is applied to the soil surface at a rate greater than the infiltration rate, the excess water moves across the field surface as gravity-driven flow and runs off the field, hence its name. Runoff can have negative impact, both because it can cause soil and nutrient lost by erosion and also because it represents a non-productive loss of water from the crop. Surface flow systems depend on this same process for water distribution and hence are particularly susceptible to runoff losses.

At the field scale, runoff is recycled. This is discussed later in Section 3.2.3

#### 3.2.2.3 Change in soil water

Soil water (*SW*) is defined as the amount of water held in the soil at a particular point in time. Figure 3.3 shows a soil water diagram with various soil water-related parameters, including:

- (i) **Saturation** (*SAT*): amount of soil water at saturation, that is, all soil pores are filled with water.
- (ii) Drained upper limit (DUL): amount of soil water at drained upper limit or field capacity. DUL is also known as field capacity. Larger pores have drained under gravity and smaller pores retain water.

<sup>&</sup>lt;sup>47</sup> Soil water deficit will be discussed in detail in the next section.

- (iii) Lower Limit at 15 bar (*LL15*): amount of soil water corresponding to a soil water potential at 15 bar. LL15 roughly represents the minimum water content to which plants can dry the soil.
- (iv) **Plant available water** (*PAW*): the difference between drained upper limit and lower limit at 15 bar.
- (v) Extractable soil water (*ESW*): the difference between soil water and lower limit at 15 bar.
- (vi) Soil water deficit (*SWD*): the difference between drained upper limit and soil water.



Figure 3.3 Soil water diagram

Figure 3.3 lays out some of the key soil water-related parameters to provide a foundation for setting up irrigation schedules to support plant growth free of water stress.

As defined above, plant available water (*PAW*) represents potential plant available water - the volume of water held in the soil depth, explored by the crop roots, between the maximum water content after free drainage (field capacity) and the minimum water content at which plants cannot continue to support growth (permanent wilting point). Furthermore, a fraction

of *PAW*, extractable soil water  $(ESW)^{48}$  represents the soil water storage or actual plant available water which can be withdrawn by the crop roots to support plant growth. The sum of *ESW* and soil water deficit (*SWD*) is *PAW*.

Irrigation and rainfall replenish soil water to field capacity (e.g. replenish *ESW*), while the growing crop continuously depletes this store (e.g. reduces *ESW*). Good management ensures the store is neither overfilled to the point where soil is waterlogged nor depleted to the point where plant growth is affected (water stress). Both cases will reduce the ability of the plant to achieve its potential yield. The ratio of *ESW* to *PAW*, which is called fraction of *PAW*, is typically used as a trigger for irrigation, i.e. to determine when irrigation should take place to support plant growth.

## Soil water storage under water deficit conditions

The soil provides a practical interfacial zone between water supply and root water uptake. If one could supply water continuously at low rates to match instantaneous root water uptake, directly to the roots, then soil water content would have little influence on crop cultivation. However, in the case of dryland cultivation, rainfall is intermittent and most forms of irrigation tend to supply water at rates much faster than plant uptake. Therefore, soil water storage provides the buffer between water supply and plant uptake.

#### **Drivers of PAW**

The key parameters controlling potential PAW are: soil type, as texture has a great influence on the values for field capacity and permanent wilting point; and management, to the extent that compaction is avoided as it both limits water storage and root exploration. Soil health is shown as having an influence, both positive in terms of the beneficial effects of soil microflora on soil structure, and negative in terms of the effects of soil-borne pathogens that may compromise plant water uptake capacity (Bone et al. 2010).

#### Change in soil water

The volume of soil water within the root zone may change over the life of the crop, depending on where one draws the time boundaries delimiting the crop (Zhang, Walker & Dawes 2002). This change may represent either an additional source of water for the crop or

<sup>&</sup>lt;sup>48</sup> Which is also called **Readily available water** (RAW).

a net drain on water supply, depending on the direction of the net change over the season. In this thesis, the change in soil water is defined as follows:

Change in soil water<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
 = Soil water<sub>it,at harvest date</sub>  $\left(\frac{ML}{ha}\right)$   
- Soil water<sub>it,at sowng date</sub>  $\left(\frac{ML}{ha}\right)$  [3.6]

## **3.2.2.4 Crop water balance**

Based on the water mass balance principle, changes in soil water are equal to the difference between total crop water supply and total crop water loss. Accordingly, the water balance of a cropping system can be expressed as:

Change in soil water<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Crop water supply <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  - Crop water loss<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [3.7]

Equation 3.7 can be rearranged as follows:

Crop water supply 
$$_{it}\left(\frac{ML}{ha}\right) = \text{Crop water } \log_{it}\left(\frac{ML}{ha}\right) + \text{Change in soil water}_{it}\left(\frac{ML}{ha}\right)$$
[3.8]

In this thesis, based on the crop water process model (see Fig. 3.2) and the discussion above, transpiration, the productive water loss component, is separated from non-productive crop water loss components. Furthermore, change in soil water is considered as a loss because, generally, the soil is wetter at the end of the season. Therefore, crop water supply equals sum of transpiration and crop water loss, which comprises soil evaporation, deep drainage, run-off and change in soil water. Equation 3.9 below provides a foundation for further discussion of crop water use efficiency in the next section.

Crop water supply<sub>it</sub> 
$$\left(\frac{ML}{ha}\right) = \text{Transpiration}_{it} \left(\frac{ML}{ha}\right) + \text{Crop water loss}_{it} \left(\frac{ML}{ha}\right)$$
 [3.9]

Whereas crop water supply and crop water loss are defined in Equations 3.10 and 3.11.

Crop water supply<sub>it</sub> 
$$\left(\frac{ML}{ha}\right) = \text{Rainfall}_{\text{it}} \left(\frac{ML}{ha}\right) + \text{Irrigation applied}_{\text{it}} \left(\frac{ML}{ha}\right)$$
 [3.10]

Crop water 
$$loss_{it} \left(\frac{ML}{ha}\right)$$
  
= Soil evaporation  $_{it} \left(\frac{ML}{ha}\right)$   
+ Deep drainage  $_{it} \left(\frac{ML}{ha}\right)$  + Runoff  $_{it} \left(\frac{ML}{ha}\right)$   
+ Change in soil water  $_{it} \left(\frac{ML}{ha}\right)$   
(3.11)

#### 3.2.3 Water at the farm scale

The key characteristic of water flows at the farm scale, particularly on irrigated cotton farms, is a central storage which provides a single locus of water flow. Water for irrigation is drawn from this and it is replenished by recycled water and water abstracted from the environment.

## 3.2.3.1 Recycled runoff

Heavy rainfall events and surface irrigation systems may generate runoff (Jun et al. 2011; Zhang, Walker & Dawes 2002), as illustrated in Fig. 3.2. The regulatory treatment of runoff aims to encourage crop growers to mitigate runoff to the environment for two reasons (Jun et al. 2011). First, runoff can be treated as a resource (after recycling). Minimizing runoff to the environment means both minimising water loss from a farming system and minimising expenditure (abstraction costs). Second, recycling of run-off helps to prevent runoff carrying pollutants off-farm.

Total water runoff is a sum of runoff from irrigation and runoff from rainfall:

Total water runoff<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Runoff from rainfall<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Runoff from irrigation<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [3.12]

In Australia, legislation requires that water runoff from a cropping system is recycled proportionally back to the cropping system<sup>49</sup>. Ratios of recycled runoff vary between 75%–85% depending on water management practices applied by crop businesses.

With a given ratio of recycled runoff, the amount of water runoff that is recycled back to water storage systems of the crop business can be calculated as follows:

Recycled runoff<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Total water runoff<sub>it</sub>  $\left(\frac{ML}{ha}\right) \times$  Ratio of recycled runoff<sub>it</sub> (%) [3.13]

The difference between total water runoff and the amount of recycled runoff determines the amount of water loss from runoff:

Loss from water runoff<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Total water runoff<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  – Recycled runoff<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [3.14]

## 3.2.3.2 Abstracted water

Water abstracted from the environment includes surface water (e.g. water pumped from a river) or groundwater (e.g. water pumped from a bore) (Ibaraki 2010).

Since the beginning of the twenty-first century, where increasing demand for irrigation water worldwide has led to potential negative impacts on the natural environment, management of water supplies has drawn growing attention from local and state governments (Gaydon,

<sup>&</sup>lt;sup>49</sup> http://www.recycledwater.com.au/

Meinke & Rodriguez 2012; Khan & Hanjra 2008; Mehta et al. 2013; Moreno-Pérez & Roldán-Cañas 2012; Rijsberman 2006). In California (US), increasing water competition among agriculture, industry and urban areas has led to water over-allocation in most watersheds in the state (Mehta et al. 2013). As a result, a number of water institutions have been developed to adapt to new water scarcity conditions and manage water competition among water users, particularly at the river basin level (or catchment level), in an attempt to conserve ecosystems and biodiversity and to sustain water resources while maintaining sustainable agriculture (Cai et al. 2011; Khan et al. 2006; Mehta et al. 2013; Naramngam & Tong 2013; Reidsma et al. 2012; Rifkin 2005; Rijsberman 2006).

In Australia, for example, in order to obtain irrigation water, farmers are required to have a licence to abstract surface water or groundwater (Department of Environment and Primary Industries 2014). In areas where water sharing plans are implemented, licensed surface water users can apply for a surface water licence from their local rivers (Australasian Legal Information Institute 2014). In contrast, in areas where water sharing plans have not commenced, farmers need to apply for licences to construct a bore to take water from groundwater sources (Department of Natural Resources and Mines 2014).

#### Water availability

Availability of freshwater from surface and groundwater sources is driven by environmental factors such as climate and geography (Jackson et al. 2011; Zhang, Walker & Dawes 2002). For example, in drought years, fresh water availability is very low (Jackson et al. 2011; Oweis, Farahani & Hachum 2011). Furthermore, the amount of water allocation for irrigation (if water is available) is not only dependent on physical water availability in those water sources, but also water competition and the water allocation policy of local governments or states (Department of the Environment 2014). For example, decisions on how water resources of the Murray River System are shared are based on competition among human needs, especially during periods of extremely low water availability in the Murray River (Department of the Environment 2014). The NSW system allows water entitlement holders to carry over unused water from one season to the next (with a given percentage of upper limit) so that they can manage water scarcity risk better (Australasian Legal Information Institute 2014).

Given that recycled runoff contributes to the total amount of water available to supply the cropping system (crop water supply), the amount of water abstracted from the environment to supplement rainfall to ensure sufficient water to completely satisfy transpirational demand is reduced by the amount of recycled runoff<sup>50</sup>:

Irrigation abstracted <sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Irrigation applied<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  – Recycled runoff<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [3.15]

When replacing the irrigation applied component in the Equation 3.10 by irrigation abstracted and recycled runoff components from the Equation 3.15, total crop water supply can be expressed as the sum of three water components:

Crop water supply<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Rainfall<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Recycled runoff<sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
+ Irrigation abstracted<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [3.16]

As the amount of water abstracted from the environment for irrigation has a significant impact on dV/dt, the key parameter that defines water sustainability, reducing the amount of water abstracted by improving water management practices with regard to recycling of runoff (i.e. improving ratios of recycled runoff) is a step towards water sustainability.

#### Types of environmental water

Water can be accessed through entitlement under a water licence or purchased another licence holder who sells their unused allocation in a water trading market (Department of Environment and Primary Industries 2014; Department of Natural Resources and Mines 2014). In this thesis, such water is classified in two groups: licensed water<sup>51</sup>, which is defined

<sup>&</sup>lt;sup>50</sup> Assuming that there is no water loss from the water storage systems.

<sup>&</sup>lt;sup>51</sup> In order to gain access to and use licensed irrigation water, the crop business is required to have a water licence issued by statutory water rights administered by State and Territory governments: http://www.nationalwatermarket.gov.au/about/rights.html.
as water allocation where delivery entitlements are owned by irrigators  $5^{2}$  (i.e. crop businesses), and traded water, which is defined as water bought from other businesses via water trading markets (for example, Murray Irrigation's Exchange) (Department of the Environment 2014), a market mechanism that enables the buying and selling water access licenses or annual allocation water.<sup>53</sup>

Accordingly, total water abstracted from the environment is a sum of licensed and traded water:

Irrigation abstracted<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Licensed irrigation<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Traded irrigation<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [3.17]

Climate and geographic factors (seasonal rainfall amount, farm location), farm characteristics (i.e. soil type, slope of land, farm size), and operational management decisions (specifically irrigation-related management decisions, such as irrigation methods, ratio of water recycled, irrigated crop area, when water sustainability is considered), will determine the total amount of irrigation water that needs to be abstracted, as well as the proportion of licensed and traded water that need to be accessed for crop production in a given year.

### Environmental water cost

Since the beginning of the twenty-first century, there has been increasing competition for water abstracted from the natural environment. This has led to demand, particularly from industry (including agriculture), exceeding water availability, leading to increased water costs, more stringent environmental regulation, making it more difficult for crop businesses to be sure of their economic sustainability and growth in the long-term (Blum 2009; Jackson, Khan & Hafeez 2010; Mehta et al. 2013; Naramngam & Tong 2013; Passioura 2006; Rijsberman 2006; van Ittersum et al. 2012). Therefore, the problem of water availability for sustaining and developing a business does not only lie in calculating the optimal amount of

<sup>&</sup>lt;sup>52</sup> For example, for areas in NSW, water licences are generally issued by the NSW Government: http://www.water.nsw.gov.au/water-licensing/about-licences. <sup>53</sup> See an example of general water dealings including the trading of water access licenses under the *Water* 

Management Act 2000 (NSW) in : http://www.water.nsw.gov.au/water-licensing/dealings-and-trade

water for irrigation to ensure optimal growth, but also in sourcing that water, either through existing allocations (licensed) or water available on the water market (traded), and the cost associated with each type of water.

The price paid for water (i.e. \$/ML) varies between licensed water and traded water, across years and locations.<sup>54</sup> In Chapter 4 the water cost will be discussed in detail, taking into consideration the ratio of licensed and traded water.

## 3.2.3.3 Simplifying assumptions

A complete treatment of farm scale water flows would include both several asynchronously irrigated fields and natural factors influencing the stored volume of water, such as precipitation falling on the surface of the dam, evaporation from the water surface, and deep drainage through the floor of the storage. In this thesis, the simplified farm is assumed to have, effectively, only one field.

As the intent is to compare the effects of different management practices on the field in relation to water sustainability, the influence of precipitation, water surface evaporation and deep drainage on the stored volume is assumed to be constant across cases and is therefore not included.

# **3.3** Theoretical link between crop water supply components and crop production components

In this section, I provide a theoretical discussion on the link between crop water supply components and crop production components. Earlier, transpiration was defined as the water lost from the crop during photosynthesis. In a real sense, the plant is obliged to exchange water for carbon, which ultimately is partially used to generate economic products. This section examines the parameters that have been used, or should be used, to describe this crucial exchange.

<sup>&</sup>lt;sup>54</sup> See an example of *State Water Pricing Application: 2014-15 – 2016-17* issued by the Australian Government in <u>https://www.accc.gov.au/regulated-infrastructure/water/state-waters-regulated-charges-2014-17-review/final-decision</u>

## 3.3.1 Issues of inconsistent definitions and validity of crop water performance measures in the literature

This section provides a brief review on how the concept of crop water efficiency is defined in the crop literature. Based on this, issues of inconsistent definitions and validity of water efficiency measures are discussed.

### **3.3.1.1** Inconsistent terminology and definitions of water efficiency measures

The literature surrounding efficiency of water used by plants is bedevilled by inconsistent terminology. For instance, water use efficiency (WUE) has been used, as noted by (Sinclair, Tanner & Bennett 1984, p. 36) for scales "*ranging from gas exchange by individual leaves over a few minutes to grain yield response to irrigation treatments through an entire season*". Usually, the context makes clear what scale is being considered and hence what variables are important, but ambiguity remains. It is also noted that the same parameter might be used for quite different purposes. One common meaning of transpiration ratio is that equivalent to small-scale WUE, i.e. the amount of dry matter produced by a crop per unit of mass of water used (Forbes & Watson 1992). However, transpiration ratio has also been used to describe the ratio of transpiration to evapotranspiration (Kato, Kimura & Kamichika 2004).

Moreover, while another body of the crop literature has used crop water productivity (CWP) terminology to express the relationship between the amount of crop produced and volume of water used (Igbadun et al. 2006; Pereira, Cordery & Iacovides 2012), the WUE terminology has also been used to express the concept of CWP (Blum 2009; French & Schultz 1984). For example, Blum (2009) defines WUE as a ratio of biomass to water-use and transpiration efficiency (TE) as WUE at the leaf level, which is "*the amount of water transpired per given unit of CO*<sub>2</sub> *fixation*" (p.120). In addition, several definitions and terms are used to express the concept of CWP and WUE. For example, water use (technical) efficiency is defined as the ratio of biomass to water consumed, while water use (economic) efficiency and water use (hydraulic) efficiency are defined as the value of crop product(s) per unit of water volume consumed and the volume of water actually used to the volume of water supply to irrigated crops, respectively (Igbadun et al. 2006).

Another term that has also been used to express the relationship between crop production and the volume of water input is the water use index (WUI). More particularly, three different terms are used to express this relationship, including agronomic water use index (AWUI),

defined as the ratio of crop yield to the volume of water input; crop water use index (CWUI), defined as the ratio of crop yield to the volume of water used by the crop (evapotranspiration) in production; and economic water use index (EWUI), defined as gross revenue per water input (Igbadun et al. 2006).

In brief, the water use efficiency concept has been used in crop science research to describe how efficient crop water use is converted into crop output, or how water adds value to a crop business (Blum 2009; Cossani, Slafer & Savin 2012; Hochman, Holzworth & Hunt 2009; Liu-Kang & Hsiao 2004; Moore, Robertson & Routley 2011; Passioura 2006; Pereira, Cordery & Iacovides 2012; Tennakoon & Milroy 2003; van Halsema & Vincent 2012). However, there are three main issues in the literature surrounding crop water efficiency measurement. First, different terminologies are used to express the concept of "water use efficiency" or the relationship between crop yield and crop water use, including WUE, CWP and CWUI. Second, different water terms are used to define the denominator (e.g. water component) of water efficiency measures; for example, water used by the crop in production (evapotranspiration), water transpired (transpiration), water use, water input, water consumed, irrigation water, and water supply. In addition, the numerator of water efficiency measures (e.g. crop output) can be either crop yield or revenue of the crop. This means the water use concept is expressed with different meanings (and in relation to various scales), making the concept of water use efficiency ambiguous and inconsistent. Third (and in contrast to the first issue), the same WUE measure is used to express two different concepts of water efficiency; the first is the extent to which crop yield is produced per unit of water use, while the second is the percentage of a water component (e.g. transpiration) to another aggregate water component (e.g. evapotranspiration).

### **3.3.1.2** The validity issue of irrigation water efficiency measures

Improving water-efficient irrigation has become an important issue in irrigation research and practice (Pereira, Cordery & Iacovides 2012). New indicators have been developed with an attempt to capture water-efficient irrigation performance. For example, Pereira et al. (2012) propose two different indicators to express water use efficiency (or water productivity) of irrigated crops, *WP* and *WP*<sub>*Irrig*</sub>, both of which are defined as the ratio of the actual crop yield achieved ( $Y_a$ ) and the water used. However, the denominator of *WP* refers to the total water

use (*TWU*), including rainfall and irrigation, while the denominator of  $WP_{Irrig}$  only refers to the irrigation water use (*IWU*) (see Equations 3.18 and 3.19).

$$WP_{it}\left(\frac{kg}{ML}\right) = Y_{a,it}\left(\frac{kg}{ha}\right) \div TWU_{it}\left(\frac{ML}{ha}\right)$$
 [3.18]

$$WP_{Irrig,it}\left(\frac{kg}{ML}\right) = Y_{a,it}\left(\frac{kg}{ha}\right) \div IWU_{it}\left(\frac{ML}{ha}\right)$$
[3.19]

Pereira et al. (2012) argue that the same amount of crop yield (e.g.  $Y_a$ ) are driven not only by the amount of irrigation water applied to the crop, but also by the amount of rainfall water provided to the crop. However, I argue that there is a validity issue associated with the proposed  $WP_{Irrig}$  indicator in this study.

As discussed in Section 3.2.2.2, rainfall and irrigation are the two main water sources that are used for irrigated crops. Irrigation is applied to supplement rainfall when rainfall is not sufficient to meet transpirational demands of the crop (Cammarano et al. 2012; Gaydon, Meinke & Rodriguez 2012; Grassini et al. 2011). The relation between economic yield of a crop and these two water sources is illustrated in Figure 3.4.



Water Source (ML/ha)

Figure 3.4 Relation between crop economic yield and water source

As rainfall can support some crop growth (see line 1 in Fig. 3.3), irrigation activities contribute an incremental amount of economic yield that can be generated by the application of irrigation (see line 2 in Fig. 3.3), which captures the accurate irrigation water use efficiency. Therefore, while the irrigation *WUE* measure (e.g.  $WP_{Irrig}$  indicator in Pereira et al.'s 2012 study) is proposed as a measure of how efficient irrigation water is used to generate economic yield, an irrigation water use efficiency measure that is expressed as the ratio between total economic yield obtained by the crop business and irrigation amount suffers a validity issue since the economic value of irrigation use is over-represented. This is because the measure does not exclude the proportion of economic yield generated by rainfall (see line 4 in Fig. 3.3). This raises an issue with the validity of irrigation water use efficiency measures, which relates to the extent to which the measure can capture the efficient use of irrigation with respect to crop production output.

If irrigation depends on water abstracted from the environment, and sustainability directs that this should be reduced towards zero, perhaps through better use of rain, then *WPI*<sub>rrig</sub>, as defined, will tend to infinity as improvements occur, while *WPI*<sub>rrig</sub> (actual) will tend to zero. Targets at infinity are practically useless for managers.

Furthermore, in their study of economic evaluation of cotton lint yield in different cotton areas in Australia, Cammarano et al. (2012) provide the comparisons of lint yield (kg/ha) response to irrigation amount (mm) among three different cotton locations in which rainfall amount (and other environmental factors) are different. This leads to invalid comparison and hence invalid evaluation of economic outcomes of cotton product with respect to irrigation application across regions, since the contribution of rainfall to the economic outcomes of the crop is not separated from that of irrigation.

#### 3.3.2 Crop water use measurement

In this section, theoretical discussion as to how to overcome the ambiguity and inconsistency in water use efficiency definitions in the extant literature is developed, as well as how to overcome the validity issue of water use efficiency measurement.

First, in order to avoid confusion on water use efficiency definitions, the convention proposed by Purcell & Currey (2003) has been adopted. In this the ratio of two similar qualities is designated as an efficiency measure, and the ratio of dissimilar quantities as an index. Second, the generally accepted use of irrigation is to ensure crop productivity is not limited by water deficits that might otherwise occur during the crop life cycle. In this respect, both rainfall and irrigation contribute to the total economic outcomes of the crop. This means when total economic yield/value is considered, total crop water supplied to the crop (e.g. sum of rainfall and irrigation applied) should be incorporated in one assessment. By doing this, the validity issue of irrigation water use efficiency is overcome. Furthermore, when irrigation is applied to supplement rainfall when this resource is not sufficient to meet the transpirational demands of the crop, the abstraction of water from the environment for irrigation constitutes an impost on the environment, questioning the sustainability of irrigated production.

The first sentence of the paragraph above provides a central position from which the sustainability of irrigated cotton production can be objectively and quantitatively evaluated, by considering processes both up and downstream from this position. Before doing this, however, the central position needs to be somewhat more development. In the following subsections, theoretical links among water coming into the crop system, water actually used by the crop to generate biomass and cash (generated from economic yield) coming out from the crop system, lay out the foundation for discussion of the sustainability of irrigated cotton production.

To begin, I assume that, as sufficient water will be supplied to avoid water-related limitations to crop growth, other limitations that can be reduced by management, for example, by providing sufficient fertilisers and adequate pest and disease control, will also be avoided. The crop should therefore grow as well as the prevailing weather conditions allow.

## **3.3.2.1** Transpiration index

Crop research reports that the evaporative loss of water from within plant leaves, transpiration, is a necessary corollary of the primary process of photosynthetic carbon dioxide assimilation and subsequent biomass accumulation (Blum 2009; Passioura 2006). The ratio of transpiration to biomass is termed the transpiration index. Over a cropping season, we will use transpiration index (TI) as a ratio between cumulative transpiration and cumulative biomass. The relation between transpiration and biomass accumulation can be demonstrated in Equation 3.20.

Transpiration index<sub>it</sub> 
$$\left(\frac{\text{kg}}{\text{ML}}\right)$$
 = Biomass yield<sub>it</sub>  $\left(\frac{\text{kg}}{\text{ha}}\right)$   
 $\div$  Transpiration<sub>it</sub>  $\left(\frac{\text{ML}}{\text{ha}}\right)$ 
[3.20]

The transpiration index is affected by ambient  $CO_2$  concentration, which we will ignore as our focus is on any given single crop for which this will be constant; humidity and leaf temperature will be, across a season, characteristic of the site and climate (Sinclair, Tanner & Bennett 1984). It will not be affected by crop nutrition and water supply, nor by insect and disease burden (Liu-Kang & Hsiao 2004). Therefore, in this thesis, the transpiration index is treated as a constant.

We can therefore visualise the transpiration index for a crop in a given year and location as a linear relation between transpiration and biomass to a point defined by ( $T_{max}$ ,  $Y_{B,max}$ ) which represents the maximum biomass for that crop and the transpiration needed to achieve it (see Equation 3.21. This point also represents for that year, the best use of the environment, the major uncontrolled variable in farming, by the cotton crop for the least water use.



Figure 3.5 Theoretical relation between transpiration and biomass

Transpiration index<sub>it</sub> 
$$\left(\frac{\text{kg}}{\text{ML}}\right)$$
 = Maximum Biomass yield<sub>it</sub>  $\left(\frac{\text{kg}}{\text{ha}}\right)$   
 $\div$  Maximum Transpiration<sub>it</sub>  $\left(\frac{\text{ML}}{\text{ha}}\right)$ 
[3.21]

A key assumption is that sufficient water can be provided such that weather is the only limitation to growth.  $T_{max}$  represents the minimal amount of water that the crop will need to achieve maximum growth (sourced from rainfall and/or irrigation). The transpiration index can be decomposed into its two variables, water on the upstream side of our logical chain and biomass yield on the downstream side, which will be discussed in Sections 3.3.2.2 and 3.3.2.3, respectively.

## **3.3.2.2** Crop water supply components

Two components of crop water supply are considered in this subsection: the proportion of the crop water supply that is used for transpiration and the adequacy of this to support the desired growth of the cotton crop. To describe this, we introduce two terms – water input efficiency and water resource efficiency.

### Water input efficiency

Water input efficiency is defined as the ratio between transpiration and crop water supply (Purcell & Currey 2003). This measure describes how effectively water at the field boundary is distributed to all crop roots, used only for uptake by the plants and transpiration, and is expressed as follow:

Water input efficiency<sub>it</sub> (%) = Transpiration<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
÷ Crop water supply<sub>it</sub>  $\left(\frac{ML}{ha}\right)$ 
[3.22]

Water input efficiency has a theoretical maximum of 100% (or 1.0) when crop water supply equals transpiration. Based on Section 3.2, and more particularly Equation 3.9, it is argued that there may be four reasons for water input efficiency to be less than 1.0. First, some of the rainfall may runoff and be partially lost before it can be recycled. Second, heavy or protracted rain may wet the profile so that drainage below the root zone occurs. Third, after rain and

surface irrigation, the soil surface is wet leading to surface evaporation. Finally, poor irrigation management, or choice of irrigation system, may lead to runoff and deep drainage losses.

While values of water input efficiency less than 1.0 might indicate inefficiencies, we need to isolate those that may not be susceptible to improved management. For instance, deep drainage losses from rainfall may be reduced by irrigation management that maintains soil wet enough for crop growth, but sufficiently below the drained upper limit to allow storage of water infiltrating during rain. However, depending on the soil and climate, deep drainage may not be completely eliminated. Depending on the mechanisms involved in the irrigation system, it may not be possible to prevent some deep drainage or runoff, no matter how proficiently the system is managed. The only recourse then is to change the system.

Water input efficiency can be further decomposed, particularly if calculations show it to be lower than expected, to determine where unexpected drainage, runoff and evaporative losses are diverting water from root uptake and hence transpiration. This is a well-known analysis in agriculture and will not be developed further here.

### Water resource efficiency

A farm, by definition, has exclusive use of the climate incident on it. We may imagine that this exclusively places some responsibility on the farm manager to make the most efficient use possible of these environmental resources. The result ( $T_{max}$ ,  $Y_{B,max}$ ) represents the best growth possible in the environment with just adequate water, and hence best use of the environmental resources (see Figure 3.5). Poorer growth will have a result (T,  $Y_B$ ), smaller than ( $T_{max}$ ,  $Y_{B,max}$ ), representing a less efficient use of the environment.

The efficiency with which environmental resources are used, with water supply as the controlling factor, we term as water resource efficiency, which is quantified as:

Water resource efficiency<sub>it</sub>(%) = Biomass yield<sub>it</sub> 
$$\left(\frac{kg}{ha}\right)$$
  
 $\div$  Maximum Biomass yield<sub>it</sub>  $\left(\frac{kg}{ha}\right)$ 
[3.23]

and, because the transpiration index is a constant for that crop and season,

Water resource efficiency<sub>it</sub>(%) = Transpiration<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
 $\div$  Maximum transpiration<sub>it</sub>  $\left(\frac{ML}{ha}\right)$ 
[3.24]

Water resource efficiency has a theoretical maximum of 100% (or 1.0). The ratio between transpiration and maximum transpiration describes how adequately the water taken up by the crop allows it to grow to its environmentally-limited potential, for which transpiration is equal to maximum transpiration and hence the ratio is equal to 1.0.

There may be circumstances when a farm manager intentionally aims for water resource efficiency less than 1.0. In rainfed cropping, for instance, the total crop water supply may be less than maximum transpiration for the site. A crop managed for maximum transpiration and hence maximum biomass yield will experience severe water deficits before maturity, and biomass yield may tend towards zero. Different management, for example, sowing the crop at a lower plant density, will intentionally reduce transpiration so that it more closely matches the water supply and hence biomass yield becomes higher than zero.

For irrigated cotton, the general assumption is that water, while having a finite cost, can be managed so that the operator's target is for ( $T_{max}$ ,  $Y_{B,max}$ ). Given this, a necessary crop water supply volume can be calculated.

#### 3.3.2.3 Harvest index

Turning to yield, the downstream side of our logical chain, the frequent circumstance in agriculture is that only part of the total biomass provides an economic product to be harvested. For cotton, only the lint and seed provide economic returns. Of the two economic products produced by cotton crops, lint has a higher economic value and hence improving lint yield is a subject of interest to crop scientists. The relationship between lint yield and biomass is described by the harvest index (Blum 2009; Passioura 1996):

Harvest index<sub>it</sub> = Lint yield<sub>it</sub> 
$$\left(\frac{kg}{ha}\right)$$
 ÷ Biomass yield<sub>it</sub>  $\left(\frac{kg}{ha}\right)$  [3.25]

68

In Equation 3.25, the harvest index represents the fraction of primary economic yield of total biomass. It can be expressed as units of kg economic product per kg biomass or as a percentage. To be consistent with our convention described earlier, this should be designated as efficiency as both lint yield and biomass have the same units. However, the term harvest index is so well entrenched in agricultural literature and is so lacking in ambiguity that our use of it is non-controversial.

By definition, the harvest index describes the efficiency with which the plant converts assimilated carbon into economic product. This conversion will be affected by crop management decisions a,s well as choice of genotype and direct environmental effects on reproductive development (Blum 2009; Stewart et al. 2010). An estimate of the harvest index can be gained from breeding trials and simulations (Stewart et al. 2010). As it encompasses the effects of a broad range of management decisions, it is a measure of plant physiological efficiency that should be of significance to managers. In this thesis, it is assumed that harvest index is a constant for a cotton crop in any year.

It is necessary to note that water resource efficiency and water input efficiency must be considered independently in evaluating whether crop water supply is equal to maximum transpiration, and more importantly whether a crop business is sustainable in respect to economic and water sustainability.

#### 3.3.3 Crop water use index

In considering the overall management of the cotton crop, a manager will have an interest in the summary position of how much lint is generated for the water supplied. This summary is captured as the crop water use index:

Crop water use 
$$index_{it}\left(\frac{kg}{ML}\right)$$
  
= Lint yield<sub>it</sub>  $\left(\frac{kg}{ha}\right)$  ÷ Crop water  $supply_{it}\left(\frac{ML}{ha}\right)$   
[3.26]

Given the preceding sections, this can be decomposed to:

Crop water use 
$$index_{it} \left(\frac{kg}{ML}\right)$$
  
= Biomass yield<sub>it</sub>  $\left(\frac{kg}{ha}\right)$  x Havest index<sub>it</sub> ÷ Crop water supply<sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
= Transpiration<sub>it</sub>  $\left(\frac{kg}{ML}\right)$  x Transpiration index<sub>it</sub> x Havest index<sub>it</sub>  
÷ Crop water supply<sub>it</sub>  $\left(\frac{ML}{ha}\right)$ 

= Water input efficiency<sub>it</sub>(%) x Transpiration index <sub>it</sub> x Havest index<sub>it</sub>

Crop water use index can be related to a range of management decision domains as shown in Figure 3.6.

The decision tree allows farmers to go back and examine their performance indicators and make relevant operational decisions to improve their sustainability index. For example, when a farmer looks at the crop water use index, if the index in respect to water input efficiency turns out to be poor, irrigation management needs to be examined. On the other hand, if the harvest index is poor (i.e. very low compared to the theoretical value), it suggests that things such as pest control, disease control need to be better managed. Alternatively, if the transpiration index is poor, this means the farmer needs to improve nitrogen management practices.



High = Theoretical targets

#### Figure 3.6 Operational decision tree

Figure 3.6 illustrates that by incorporating the transpiration index, harvest index and water input efficiency into one equation, the crop water use index provides a broad guide to managers wishing to improve crop system performance. First, low water input efficiency would suggest excessive water losses between supply point and crop root uptake, with remedies likely to be achieved in choice and engineering of the irrigation system used, in scheduling irrigation, and giving attention to water recycling systems. Second, a low transpiration index may be due to poor crop growth resulting in inadequate light interception (poor establishment, poor nutrition, unexpected restrictions in water supply to the roots, or waterlogging). Additionally, fungal root pathogens, to which cotton is susceptible, may reduce the ability of the crop to take up water. Third, the harvest index is subject to a variety of agronomic influences which can lead to its reduction. For example, insect or disease attack on foliage or developing bolls may reduce lint yield relative to biomass. Other aggregated indices of water-related crop performance might be able to indicate the presence of management problems without the ability to specifically identify possible remedies.

While the crop water use index summarises how effectively the crop exchanges water for its yield, of itself it provides inadequate information about how well that water use captures the environmental potential of the site and year. This is described, as earlier, in terms of water

resource efficiency. This is closely related to another term use in the agricultural literature, yield gap, which is defined as the difference between maximum lint yield (potential yield) and actual lint yield (van Ittersum et al. 2012).

Yield 
$$gap_{it}\left(\frac{kg}{ha}\right) = Maximum lint yield_{it}\left(\frac{kg}{ha}\right) - Actual lint yield_{it}\left(\frac{kg}{ha}\right)$$
 [3.28]

Given both the transpiration index and the harvest index as constants for that crop:

Yield gap:
$$\left(\frac{kg}{ha}\right)$$
=[Maximum biomass yield\_{it}  $\left(\frac{kg}{ha}\right)$ ]-Actual biomass yield\_{it}  $\left(\frac{kg}{ha}\right)$ ] x Havest index<sub>it</sub>=[Maximum transpiration\_{it}  $\left(\frac{ML}{ha}\right)$ -Actual transpiration\_{it}  $\left(\frac{kg}{ha}\right)$ ] x Transpiration index<sub>it</sub> x Havest index<sub>it</sub>=Maximum transpiration\_{it}  $\left(\frac{ML}{ha}\right)$  x [1-Water resource efficiency<sub>it</sub>(%)] x Transpiration index<sub>it</sub> x Havest index<sub>it</sub>

## 3.4 Theory of crop water flows

From earlier sections, I can summarise a number of key parameters of the constructed crop water process model that define water sustainability, and some of the key water sustainability measures at a field level.

**a. Water abstracted from the environment:** defined as dV/dt, which relates to water sustainability from farm.

Irrigation abstracted it 
$$\left(\frac{ML}{ha}\right) = \text{Irrigation}_{\text{it}}\left(\frac{ML}{ha}\right) - \text{Recycled runoff}_{\text{it}}\left(\frac{ML}{ha}\right)$$
[3.30]

## b. Crop water supply: defined as water input efficiency of a cropping system.

Crop water supply<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Rainfall <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Irrigation <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  ( )  
= Rainfall <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Recycled runoff <sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
+ Irrigation abstracted <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [3.31]  
= Transpiration <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Soil evaporation <sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
+ Deep drainage <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  Runoff <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Change in soil water<sub>it</sub>  $\left(\frac{ML}{ha}\right)$ 

Water input efficiency<sub>it</sub> (%) = Transpiration<sub>it</sub> 
$$\left(\frac{ML}{ha}\right) \div$$
  
Crop water supply<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [3.32]

c. Theoretical link between crop water supply components and crop production components: defined as water resource efficiency and crop water use index.

Biomass yield<sub>it</sub> 
$$\left(\frac{\text{kg}}{\text{ha}}\right)$$
 = Transpiration<sub>it</sub>  $\left(\frac{\text{ML}}{\text{ha}}\right)$  x Transpiration index<sub>it</sub>  $\left(\frac{\text{kg}}{\text{ML}}\right)$  [3.33]

Lint yield<sub>it</sub> 
$$\left(\frac{\text{kg}}{\text{ha}}\right)$$
 = Biomass yield<sub>it</sub>  $\left(\frac{\text{kg}}{\text{ha}}\right)$  x Harvest index<sub>it</sub>  
= Transpiration<sub>it</sub>  $\left(\frac{\text{ML}}{\text{ha}}\right)$  x Transpiration index<sub>it</sub>  $\left(\frac{\text{kg}}{\text{ML}}\right)$  x Harvest index<sub>it</sub> [3.34]

Water resource efficiency<sub>it</sub> (%) = Transpiration<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  $\div$  Maximum transpiration<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  Crop water use index<sub>it</sub>  $\left(\frac{kg}{ML}\right)$  [3.35]

Crop water use 
$$\operatorname{index}_{it}\left(\frac{\mathrm{kg}}{\mathrm{ML}}\right)$$
  
=  $\operatorname{Lint}\operatorname{yield}_{it}\left(\frac{\mathrm{kg}}{\mathrm{ha}}\right) \div \operatorname{Crop}\operatorname{water}\operatorname{supply}_{it}\left(\frac{\mathrm{ML}}{\mathrm{ha}}\right)$   
=  $\operatorname{Water}\operatorname{input}\operatorname{efficiency}_{it}\left(\frac{\mathrm{ML}}{\mathrm{ha}}\right) \times \operatorname{Transpiration}\operatorname{index}_{it}\left(\frac{\mathrm{kg}}{\mathrm{ML}}\right) \times \operatorname{Harvest}\operatorname{in}$ 
[3.36]

Taken together with Figure 3.6, this provides a complete summary of the key aspects of water supply, use and loss in a cotton crop and the resulting yield.

The beginning point is driven by environmental sustainability, which is improved as dV/dt for the farm tends towards zero. The end point is the production of cotton lint which has an economic value and hence is a key input to economic sustainability. However, the farm business, to be economically sustainable, must make a profit. The additional considerations necessary for that decision are laid out in the next chapter.

The summary provides some useful management targets, shown below and described earlier.

 $dV/dt \longrightarrow 0$ 

Water input efficiency  $\longrightarrow 1.0$ 

Water resource efficiency  $\longrightarrow$  1.0

Target values for the transpiration index and harvest index are those best reported on simulated results.

A manager, planning future operations, might use the crop water process model (Figure 3.2). In the case of a predetermined allowance for dV/dt (known as the allocation of water), decisions can be made about the extent of the crop that can be adequately supported by the likely available water. Alternatively, if a desired yield and profit is required, then the necessary dV/dt can be calculated. The extent of traded purchases of water might need to be calculated to ensure this yield.

As Figure 3.6 illustrates, different management decisions will affect different components in the crop water process model. A theoretically ideal rainfed or micro-irrigated crop may have water input efficiency equal to 1.0, i.e. achieves water sustainability, but limitations on crop water supply may lead to less than maximum transpiration, and hence produce less than maximum biomass, meaning that the crop business has not capitalised on its environment. It is also possible to conceive of an irrigation system to which a theoretically adequate crop water supply is made available. However, if water input efficiency is lower than expected, transpiration is still less than maximum transpiration. In both cases, the crop is considered be sub-optimal with regards to economic sustainability in the first instance, or both economic sustainability and environmental sustainability in the second instance.

## 3.5 Conclusion

This Chapter provides theoretical discussion of water sustainability in the context of cotton production and development of some of the key measures of crop water and economic sustainability performance. More specifically, three main constructs are developed. First, a crop water process model is constructed based on the theoretical discussion of water sustainability in Chapter 2 and crop science theory. Second, the theoretical discussion of water sustainability of irrigated cotton production is developed. This provides a fundamental foundation for understanding the sustainability issues behind crop water use and crop production, and the theoretical link between components of crop water sustainability measurement are discussed, leading to the necessity of development of a set of water sustainability performance measures to better capture the concept of crop water sustainability for crop production . These measures will form part of the water and economic sustainability performance model developed in Chapter 4.

The crop water process model and crop water use index explain how crop water flows across a cropping system, how an end-to-end connection between resource utilisation and farm profitability and sustainability is established, and define a linear relationship between water resources coming in (to crop production) and cash coming out (from crop production).

One of the key contributions of the research is an attempt to define water input efficiency, water resource efficiency, and a crop water use index. Furthermore, the discussion of theoretical value of these three measures, including clearly pointing to maximising profit and minimizing footprint, imply that the developed crop water use index does not only cover the efficiency but also covers the scale of sustainability. Moreover, the work shows that we now have a theoretically defined sustainability index that can enable farmers to identify areas for improving sustainability performance and make optimal operational decisions. For example, the operational decision tree enables farmers to identify areas for improvements and make appropriate/optimal operational decisions to achieve better economic and environmental sustainability performance.

## Chapter 4. Water and Economic Sustainability Performance Measurement (WESM) Model

## 4.1 Introduction

In the previous chapter, a number of key water sustainability measures at a field level are developed. These measures closely link to sustainability-related activities and management practices occurring at the crop production level, the level at which water sustainability issues occur and need to be managed. The measures establish a connection between water coming into a cropping system and crop production coming out the cropping system. However, a more holistic approach is necessary to develop a measure(s) that enables capture of both the economic and water sustainability performance of a crop business. Such a measure(s) can be used to support crop managers in making sustainability-related decisions at the organisational level.

As laid out in Chapter 1, one of the challenges in designing and using an EPMS is that the overall organisational decisions with regards to economic and environmental sustainability are made at the strategic level; this may, in organisational terms, be distant from where the key management decisions are made at an operational level. Ideally, the EPMS should span across all levels of the organisation, providing guidance at an operational level as well as informing and being informed by the strategic level.

In Section 4.2, the profit to water cost ratio is developed as a reasonable summary measure of water and economic sustainability performance of a crop business – this incorporates information about both economic sustainability and environmental sustainability. However, the measure itself suffers from a lack of transparency as to how sustainability activities and decisions at the operational level of the crop business impact the business' overall sustainability performance.

Section 4.3 reviews a decomposition ratio analysis approach and provides the rationale behind the application of this approach in this chapter to establish a theoretical connection between the summary level and operational level of a crop business. Taking this approach, in Section 4.4, I decompose the profit to water cost ratio over seven levels and articulate the logical and theoretical links between the summary water and economic sustainability performance measure at the business level and its drivers at the operational level. The building block of the profit to water cost ratio is called the Water and Economic Sustainability Performance Measurement (WESM) model. This approach enables me to provide explanation of how and why the developed WESM model - a new form of EPMS – can be used to support crop managers in managing, evaluating and improving crop water and economic sustainability performance.

Furthermore, the link between the crop water process model developed in Chapter 3 and the designed WESM model is described in Section 4.5. This lays out how the WESM model is integrated with the crop water process model and will be used in later chapters of this thesis to provide empirical evidence on how the designed EPMS (WESM) can be used to assist managers with better sustainability-related decision making and control. The designed WESM will provide the basis for empirical work (crop simulation modelling) in the later chapters of the thesis.

# 4.2 Profit to water cost ratio as the summary water and economic sustainability measure

## 4.2.1 Measurement of water and economic sustainability performance for crop businesses

In Chapter 2, I argue that to provide a valid assessment to encourage a business to move towards sustainability, business profitability and environmental resource use need to be considered in one equation to capture both economic and environmental sustainability of the business. More specifically, in the agricultural context, water shortage and increasing water cost are predicted to become key constraints for agriculture, making managing risks associated with water shortage and high water cost crucial for crop businesses to be sustainable (Hamdy, Ragab & Scarascia-Mugnozza 2003; Jackson et al. 2011; Khan et al. 2006). This means managing risks associated with water shortage risks associated with water cost is of crucial importance for crop businesses to move towards sustainability.

As outlined in Chapter 3, crop water supply to a farming system generally comes from two main sources; rainfall and irrigation. From the environmental sustainability perspective, rain falling directly on cropping systems is not considered to be disturbing the natural systems.<sup>55</sup>

<sup>&</sup>lt;sup>55</sup> However, crop businesses do need to consider the uncontrollable factors associated with rainfall supply.

In contrast, irrigation water is considered to be *water abstracted from the environment* and needs to be withdrawn from rivers or groundwater sources for irrigation. Sustainability Principle 2 in Section 2.3.3 emphasises that managing and minimising groundwater and surface water abstraction needs to be considered as a step towards water sustainability.

From an economic perspective, excluding water capture and storage costs, rainfall is a "free" water resource, which is free of cost while irrigation water abstracted from the environment and distributed mechanically is costly. Given the assumption that rainfall is a free water resource, total water cost represents irrigation cost. Hence, low water cost indicates less irrigation water in both volume and dollar terms. In this respect, crop management practices resulting in low water cost signal that the crop business moves a step towards both economic sustainability and environmental sustainability. However, another aspect needs to be considered, which is the economic return associated with those crop management practices. A business cannot be considered as a sustainable business if it improves environmental sustainability but incurs an economic loss.

The discussion above leads to the necessity of developing a summary measure that can capture both the concept of crop water sustainability for crop production (i.e. water cost) and its relation to economic sustainability (i.e. profitability) of the crop business.

Extant crop water literature proposes a number of crop water performance measures in an attempt to establish a link between economic return and water resourcing, including the economic water use index for irrigation and the economic water use index for crop water supply (Cammarano et al. 2012; Igbadun et al. 2006; Pereira, Cordery & Iacovides 2012). However, these measures are either invalid as they do not capture the proportion of rainfall that also contributes to crop production or are insufficient since the amount of economic return (rather than total sales revenue) and total water cost (rather than volume of crop water supply) are not presented explicitly in these measures.

Furthermore, while the developed crop water use index in Chapter 3 can incorporate lint yield and crop water supply in one equation, and hence can be used as a valid water and economic sustainability measure to support crop managers in evaluating and managing their sustainable manage practices at the operational level, it still does not demonstrate the link between economic and water sustainability performance of a crop business explicitly.

## 4.2.2 Profit to water cost ratio as a summary measure for crop water and economic sustainability

Based on the discussion in Section 4.2.1, a summary measure of water and economic sustainability performance must be a comprehensive measure. This should include an economic sustainability dimension, such as operating profit, and an environmental dimension, such as water cost.

Therefore, the profit to water cost ratio cost ratio, which is defined as the amount of operating profit generated by each dollar spent on water resource (see Equation 4.1 below) is an appropriate summary sustainability measure for three main reasons.

$$\begin{aligned} Profit to water cost ratio_{it} \left( \frac{\$ operating profit}{\$ water cost} \right) \\ &= Operating profit_{it} \left( \frac{\$}{ha} \right) \div Total water cost_{it} \left( \frac{\$}{ha} \right) \end{aligned}$$

$$[4.1]$$

First, as expressed in Equation 4.1, operating profit is a measure of economic return of a crop business and irrigation water cost is an indicator of the water impost of cotton cultivation on the environment. Accordingly, the profit to water cost ratio is proposed to emphasise the link between economic sustainability (captured by profit, a summary of the economic status of the business as a going concern) and environmental sustainability (captured by water cost, reflecting both the value and volume of water abstracted from the environment) of a crop business.

It is a ratio indicating the overall sustainability performance of a crop business and can be used as part of the performance measurement system to support crop managers in evaluating, managing and improving their economic and environmental sustainability performance. A business that is increasingly economically and environmental sustainable would be increasing profit and decreasing water consumption. Therefore, when a crop business can maintain its economic sustainability performance while improving its environmental sustainability performance, the business increases its business sustainability.

Second, building upon economic theory, when there is a resource constraint, a business is better off maximising economic return per unit of constraint (Barney 1991; Hart 1995). In the context of agriculture where water is evidenced as a resource constraint (Rijsberman 2006), profit to water cost ratio is a useful measure to optimise water resource use. Therefore, when water is the constraint, improvement of economic return per unit of water resource, taking climate factors into consideration, is a step towards sustainability.

Third, there are two implications of water cost within irrigation. As profit to water cost ratio uses water cost rather than water volume, it better reflects when a business intentionally uses higher security and therefore higher cost water which, by definition, has a potentially greater environmental impact when abstracted. In this respect, profit to water cost ratio demonstrates that from an economic and environmental perspective, a crop business needs to take into consideration not only the amount of water resource used (i.e. the volume of crop water supply) but also the cost of the environmental resource (i.e. cost of irrigation water).

One could argue that the ratio of operating profit to irrigation amount can be proposed as a summary sustainability measure for crop businesses. However, this ratio does not convey a similar story to the profit to water cost ratio. In fact, it provides invalid information as the ratio does not capture the proportion of operating profit generated by rainfall (as discussed in Chapter 3). On the other hand, one could argue that the ratio of operating profit to total crop water supply can be proposed as a summary sustainability measure for crop businesses since it captures both rainfall and irrigation water sources. However, as the ratio does not distinguish between 'free' water sources and those that cost money to withdraw from the environment, it does not capture the concept of water sustainability explicitly. In this respect, it is argued that the proposed profit to water cost ratio is a more comprehensive summary sustainability measure.

81

## 4.2.3 Limitations of profit to water cost ratio as a single measure for evaluating and managing crop water and economic sustainability

The profit to water cost ratio (I argue) is a good summary sustainability measure that can be used as an informative high-level measure at the business level for supporting crop managers in assessing the overall water and economic sustainability performance of the crop business. However, there are two limitations to this measure when used as a single measure for evaluating, improving and forecasting economic and water sustainability performance of crop businesses.

First, while such an aggregate measure can provide an overall figure about economic return per dollar of water cost spent for crop production, and the extent to which crop cultivation imposes on the environment, it provides limited theoretical explanation about what actually drives the business' water and economic sustainability performance. As a ratio, the profit to water cost ratio changes because of changes in one or both terms involved. However, by only looking at the profit to water cost ratio and its change, the crop manager does not know explicitly whether the change is driven by the change in economic return on crop water supply or the change in water cost ratio is useful for managers to gain better underlying drivers of the profit to water cost ratio and, hence, how to manage the overall sustainability performance of the crop business. Therefore, it is necessary to decompose the profit to water cost ratio into its components and translate changes of these components to inform important intermediary outcomes or implications for management decisions.

Second, as outlined in Chapter 1, sustainability issues need to be managed at the operational level of a business as it is the level at which business activities actually occur and sustainability issues arise. In this respect, a connection between the developed summary measure and measures that closely link to the operational level is necessary.

Therefore, without knowing the cause and effect of the overall sustainability performance, particularly the driving factors that directly relate to businesses' activities at the operational level, it seems to be impossible for managers to identify areas for improving crop businesses' sustainability performance to move towards sustainability. In other words, in order to support managers in evaluating, managing and improving their businesses' sustainability

performance, it is of crucial importance to identify and establish the theoretical links between factors at the operational level that affect the overall sustainability performance of the business.

One way to articulate the logical relations between the high level measures (i.e. the summary measure) and the low level measures (i.e. drivers of the summary measure and/or measures that are closely linked to activities at the operational level) is to use a decomposition ratio analysis approach.

## 4.3 Design of Water and Economic Sustainability Measurement Model -A decomposition ratio analysis approach

## 4.3.1 Overview of the decomposition ratio analysis approach

Financial ratio analysis (e.g. decomposition ratio analysis) is an accounting technique for financial statement analysis, based on a structural approach to allow the ratio scheme to be presented in a clear, simple, structured and hierarchical way (Nissim & Penman 2001; Soliman 2008). One of the distinct features of the ratio analysis approach is that a single financial ratio presented as a high level performance measure for analysis (for example, return on common equity, a summary profitability ratio in the financial analysis) can be decomposed into more detailed and specific descriptors of a firm's performance. Ratios in lower levels of the ratio scheme are identified as finer information about those higher up. Therefore, the approach not only helps to determine relevant ratios but also provides a structured and organized way for the analysis of the summary ratio (Nissim & Penman 2001; Soliman 2008).

Decomposition ratio analysis also allows for a transparent and accurate analysis of a firm's operations (Nissim & Penman 2001; Soliman 2008). For example, in the analysis of profitability, the ratio analysis approach allows separation of the different components of operating profitability and the profitability identified with the financing activities. It also allows decomposition of return on net operating assets into two components; profit margin and asset turnover, which are accounting signals that measure different aspects of a firm's operation (Penman 2003; Soliman 2008). Moreover, as the ratio scheme can be decomposed into more detailed and specific drivers of a firm's performance, changes in overall

performance can be traced in a more detailed manner (Nissim & Penman 2001; Soliman 2008).

Financial ratio analysis is applied for financial statement analysis in financial accounting (Nissim & Penman 2001; Soliman 2008) for comparison and valuation purposes. For example, Nissim and Penman (2001) apply the structural approach of ratio analysis technique to financial statement analysis for equity valuation. Soliman (2008) also reveals that the DuPont analysis technique is used by market participants for forecasting future earnings as its components contain an incremental and viable form of information about the operating characteristics of a firm and provide exploratory power about changes in future profitability. In addition, Penman (2003) suggests that by clearly articulating the various links between operational level activities and firm value, the relative materiality of different actions and/or strategies can be evaluated by market participants and researchers who are interested in evaluating various accounting and other decisions.

## 4.3.2 Applying the decomposition analysis approach to EPMS design

While the decomposition ratio analysis approach outlined in the previous section provides a number of advantages in relation to establishment of a clear and structured way to link a set of measures, this approach has not been considered thoughtfully with several issues of PMS design, including the validity issue and the useability issue of EPMS. For example, while Figge and Hahn (2013) attempt to apply the DuPont analysis technique as an integrated tool to support decision-making for more efficient use of environmental resources, there are a number of issues with their study. First, Figge and Hahn (2013) only analyse the first level breakdown of the eco-efficiency measure, which means more detailed and specific drivers of a firm's eco-efficiency performance are not determined. Second, in their empirical work, they provide an analysis of the carbon-efficiency of international car manufacturers. This is not consistent with their theoretical work in respect to the creation of 'Sustainable Value' - as natural resources efficiency and CO<sub>2</sub> efficiency convey two different aspects of environmental sustainability. While the former relates to the degree of natural resources used for organisational production processes; the later relates to the degree of pollution to the natural environment due to the discharge of wastes (gaseous, liquid or solid wastes) from the organisational production processes. This illustrates the important role of determining the

actual drivers of the summary/aggregate measure (in this case the eco-efficiency measure) to successfully capture environmental sustainability performance of an organisation.

Furthermore, the usability of a performance measurement system refers to the balance between the meaningfulness and the usefulness of the PMS. These two important characteristics of PMS seem to pull in opposite directions (Floridi et al. 2011; Veleva 1994). On the one hand, a meaningful set of measures need to provide valid and adequate information about specific performance aspects of an organisation that they are intended to represent (Bhargava, Dubelaar & Ramaswami 1994; Ittner & Larcker 2003). This means, in most case, the set of measures need to be relatively complex to be able to fully reflect what it is supposed to capture (Floridi et al. 2011; Veleva 1994). On the other hand, the same set of measures needs to be simple and clear enough to provide useful information to enable system users to understand and use the information to make better management decisions and control (Veleva 1994). In the context of EPMS, in addition to increasing the use of multiple financial and non-financial performance measures (Ittner & Larcker 2001; Kaplan & Norton 1996; Malmi & Brown 2008), the inclusion of environmental performance measures increases the complexity of the system (Floridi et al. 2011; Veleva 1994). The question of how to design different measures to capture different performance aspects of an organisation, which are expressed in different dimensions, and present them in a way to provide both overall and detailed assessment still remains a challenge (Boiral & Sala ; de Oliveira, Serra & Salgado ; Ittner & Larcker 2001; Jabbour 2010; Miles, Munilla & McClurg 1999).

One way to deal with the usability of PMS is to apply the decomposition ratio analysis approach to enable a range of ratios to be tied together in a hierarchal, logical and structured way in which the high level measures contain information about the lower level measures. By doing this, a building block of a set of linked measures is established for better information provision to support better decision making and control.

## 4.4 The analysis of crop water and economic sustainability

### 4.4.1 Analysis of the profit to water cost ratio

In this thesis, the ratio analysis approach is applied for analysis of crop water and economic sustainability for three main reasons. First, it allows more thorough analysis of water and economic sustainability performance of a crop business through decomposition of the

proposed summary water and economic sustainability measure – the profit to water cost ratio – into its levels in order to gain insights into the causes and effects of the change in the overall sustainability performance.

Second, it enables establishment of the theoretical links between the summary sustainability measure and information down the hierarchy towards the operational level where water related environmental issues actually occur. By doing this, the key drivers of the overall water and economic sustainability performance of crop businesses are identified and (un)sustainability crop management practices are identified for improvement of profit to water cost ratio – a step towards water sustainability.

Third, the ratio analysis approach allows us to distinguish between different aspects/decisions of the crop business's operation; for example, decisions related to water use, irrigation systems or crop production. This supports managers in evaluating and/or comparing the outcomes of various management practices at the operational level for better sustainability decision making and control.

The analysis of the drivers of the profit to water cost ratio in the following section is called *water and economic sustainability analysis*. While I apply the decomposition ratio analysis approach that is traditionally applied to analysis of profitability, several modifications are proposed to enable capture of various crop water and economic sustainability aspects of a business.

In the following sections, the profit to water cost ratio is broken down into its drivers over seven levels of analysis. These seven levels are depicted in the WESM model diagram (see Figure 4.1). Each level contains information about the lower level. Theoretical justifications of the decomposition are provided with the analysis of these seven levels, including explanations of how high level measures from Level 1 to Level 5 are defined and decomposed into its drivers, as well as how low level measures in Level 6 and Level 7 are calculated and linked to input data of WESM (i.e. key parameters of the crop water process model, accounting data, irrigation data, crop data which can be generated via either available information sources in the real-world or simulation modelling).



Figure 4.1 Levels 1 to 3 of the WESM model



Figure 4.1 Levels 3 to 5 of the WESM model



Figure 4.1 Levels 5 to 6 of the WESM model



Figure 4.1 Levels 6 to 7 of the WESM model (economic low-level measures)



Figure 4.1 Levels 6 to 7 of the WESM model (environmental low-level measures)

### 4.4.2 First-level analysis: Decomposition of profit to water cost ratio

In the first level breakdown, to gain more insights into whether the change in the profit to water cost ratio is driven by its numerator (operating profit) or its denominator (total water cost), the summary measure is decomposed into its three components named as return on water, water use leverage, and weighted average irrigation cost. The logical relation between profit to water cost ratio and its three drivers is expressed as follows:

Profit to water cost ratio<sub>it</sub> 
$$\left(\frac{\$ \text{ operating profit}}{\$ \text{ water cost}}\right)$$
  
= Return on water<sub>it</sub>  $\left(\frac{\$}{\text{ML}}\right)$   
 $\div \left[\text{Water use leverage}_{\text{it}} (\%)\right]$ 
  
× Weighted average irrigation cost<sub>it</sub>  $\left(\frac{\$}{\text{ML}}\right)$   
[4.2]

Where: return on water, water use leverage, and weighted average irrigation cost are defined in Equation 4.3, Equation 4.4 and Equation 4.5, respectively.

Return on water<sub>it</sub> 
$$\left(\frac{\$}{ML}\right)$$
  
= Opening profit<sub>it</sub>  $\left(\frac{\$}{ha}\right)$  ÷ Crop water supply<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [4.3]

Water use  $leverage_{it}$  (%)

$$= \left[ \text{Irrigation abstracted}_{it} \left( \frac{ML}{ha} \right) \right]$$

$$\div \text{Crop water supply}_{it} \left( \frac{ML}{ha} \right) \times 100\%$$

$$(4.4)$$

Weighted average irrigation cost <sub>it</sub> 
$$\left(\frac{\$}{ML}\right)$$
  
= Total water cost<sub>it</sub>  $\left(\frac{\$}{ha}\right)$   
 $\div$  Irrigation abstracted <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [4.5]

Table 4.1 summarises the formulas related to the first-level of analysis.

Comparing Equation 4.1, which presents the formula for calculating the profit to water cost ratio, to Equation 4.2, which shows its break-down into three drivers, taking into consideration of Equations 4.3, 4.4 and 4.5, it is proved that they are equal.

Profit to water cost ratio<sub>it</sub> 
$$\left(\frac{\$ operating profit}{\$ water cost}\right)$$
  
= Operating profit<sub>it</sub>  $\left(\frac{\$}{ha}\right) \div Total water costit  $\left(\frac{\$}{ha}\right)$   
= Return on water<sub>it</sub>  $\left(\frac{\$}{ML}\right)$   
 $\div \left[Water use leverage_{it} (\%)$   
 $\times Weighted average irrigation cost_{it} \left(\frac{\$}{ML}\right)\right]$   
=  $\left[Opening profit_{it} \left(\frac{\$}{ha}\right) \div Crop water supply_{it} \left(\frac{ML}{ha}\right)\right]$   
 $\div \left[\left[Irrigation abstracted_{it} \left(\frac{ML}{ha}\right)\right]$   
 $\div Crop water supply_{it} \left(\frac{ML}{ha}\right)\right]$   
 $\times \left[Total water cost_{it} \left(\frac{\$}{ha}\right) \div Irrigation abstracted_{it} \left(\frac{ML}{ha}\right)\right]\right]$   
 $= Operating profit_{it} \left(\frac{\$}{ha}\right) \div Total water cost_{it} \left(\frac{\$}{ha}\right)$$
Table 4.1 WESM analysis: The first-level analysis

Variable	Level	Formula
<b>Profit to water cost ratio</b> <sub>it</sub> $\left(\frac{\text{$ operating profit}}{\text{$ water cost}}\right)$	Level 1	= Opening $\operatorname{profit}_{it}\left(\frac{\$}{ha}\right)$ ÷ Total water $\operatorname{cost}_{it}\left(\frac{\$}{ha}\right)$
		$= \text{Return on water}_{it}\left(\frac{\$}{\text{ML}}\right)$
		÷ [Water use leverage <sub>it</sub> (%)
		× Weighted average irrigation cost it $\left(\frac{\$}{ML}\right)$
Return on water <sub>it</sub> $\left(\frac{\$}{ML}\right)$	Level 2	= Opening profit <sub>it</sub> $\left(\frac{\$}{ha}\right)$ ÷ Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$
Water use leverage <sub>it</sub> (%)	Level 2	$= \left[ \text{ Irrigation abstracted}_{\text{it}} \left( \frac{\text{ML}}{\text{ha}} \right) \div \text{ Crop water supply}_{\text{it}} \left( \frac{\text{ML}}{\text{ha}} \right) \right] \times 100\%$
Weighted average irrigation cost it $\left(\frac{\$}{ML}\right)$	Level 2	= Total water $\operatorname{cost}_{it}\left(\frac{\$}{ha}\right)$ ÷ Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$

The first level breakdown of the profit to water cost ratio enables us to distinguish between two sustainability aspects: economic sustainability (captured by return on water and weighted average irrigation cost) and environmental sustainability (captured by water use leverage)<sup>56</sup>. More specifically, it provides us a better understanding of how change in economic return on crop water supply, and/or change in proportion of crop water sources (i.e. rainfall versus irrigation abstracted), and/or or change in proportion of irrigation allocation (i.e. licensed irrigation versus traded irrigation)<sup>57</sup>, affect the change of the overall water and economic sustainability performance of a crop business.

Furthermore, the decomposition of the profit to water cost ratio into its three components demonstrates that by relying solely on the summary measure for evaluating sustainability performance of a crop business, a crop manger may end up with suboptimal/inaccurate decisions. The following example is used to support this argument.

To control for climate factors, two farms in the same location are considered in this example. Farm A gains an economic return (in terms of operating profit) of \$100 per ML of Crop water supply by sourcing 60% of crop water from irrigation water. Given the cost of irrigation water being \$60/ML (NSW Department of Primary Industries 2015), using Equation 4.2, profit to water cost ratio of Farm A equals to \$2.77 of operating profit per \$ of water cost (= 100/ML÷ (60% x \$60/ML)).

In contrast, Farm B gains an economic return (in terms of operating profit) of \$85 per ML of Crop water supply by only sourcing 50% of crop water supply from irrigation water. Given the same irrigation cost of \$60/ML, profit to water cost ratio of Farm B equals to \$2.66 of operating profit per \$ of water cost (=  $$80/ML \div (50\% \times $60/ML)$ ). The difference in profit to water cost ratio between Farm A and Farm B is \$0.11 of operating profit per \$ of water cost (= \$2.77 - \$2.66). Assuming that both Farm A and Farm B spent \$573 (NSW Department of Primary Industries 2015), or irrigation cost per ha and the size of two farms are 400 irrigated

<sup>&</sup>lt;sup>56</sup> The return on water measure is a type of eco-efficiency measures while the water use leverage measure is a measure to capture crop businesses' impacts on the natural water systems. Therefore, in the WESM model, the eco-efficiency measure forms part of the model but the measure itself is not sufficient to capture crop water sustainability.

<sup>&</sup>lt;sup>57</sup> As discussed in Chapter 3, irrigation water can be obtained from two sources: licensed irrigation – irrigation water allocation with delivery entitlements owned by the crop business, and traded irrigation – irrigation water bought from other licensed irrigators via water trading market.

ha, Farm A gains an additional \$25,212 (=\$0.11 x 573 x 400) of operating profit for a crop season. From an economic perspective, it can be concluded that Farm A is more sustainable than Farm B. However, from an environmental perspective, Farm A relies on water withdrawn from the environment more than Farm B, which is argued to be a less sustainable practice. Consequently, when a crop business improves its profit to water cost ratio by sourcing a higher proportion of irrigation in total Crop water supply, it is a less sustainable business.

This example illustrates that achievement of higher profit to water cost ratio does not always mean that the crop business is moving towards sustainability. While a profit to water cost ratio is argued as a valuable summary indicator, a crop manager who uses the ratio must be prepared to decompose it into its drivers to gain more insight into the cause and effect of its change. Furthermore, for more valid evaluation of sustainability of the crop business, the manager needs to consider the trade-off between economic sustainability and water sustainability if the two sustainability aspects move in two different directions.

#### 4.4.2.1 The effect of return on water

In the first-level breakdown, return on water is isolated as an important driver of the profit to water cost ratio. Return on water is defined as a measure of economic return on the water resource.

In this thesis, operating profit represents crop businesses' economic return. Businesses' activities comprise of operating activities and financial activities (Nissim & Penman 2001; Penman 2003). Financial activities are irrelevant in water sustainability analysis which focuses on operating decisions, such as water source decisions and water management decisions in relation to crop production. In addition, volume of Crop water supply, comprising rainfall water and irrigation water, represents water resources that are supplied to a crop business for crop production. Accordingly, return on water is expressed as the ratio of operating profit generated by total amount (volume)<sup>58</sup> of Crop water supply (see Equation 4.3).

<sup>&</sup>lt;sup>58</sup> Amount of water refers to volume (ML) of water. Whereas, unit cost of water refers to cost of water (\$/ML)

Mathematically, as illustrated in Equation 4.2, assuming the other drivers hold constant, improving return on water will result in increased profit to water cost ratio, and hence increased overall water and economic sustainability performance of the crop business. In addition, from an economic point of view, since water is considered as an environmental resource constraint (as discussed previously), the return on water measure reflects the degree to which operating profit is generated by one unit of this resource constraint. This measure provides useful information for economic sustainability-related decision making by a crop business, as maximising profitability generated per unit of resource constraint is a way for the business to be sustainable. In this respect, improvement of return on water is argued as a step towards to sustainability.

However, other factors also need to be considered in the analysis of water and economic sustainability of the crop business. Since the denominator of the return on water ratio captures total Crop water supply, the ratio reveals the amount of operating profit generated by all sources of water supplied to the crop business. This means it does not provide information on how water is sourced (e.g. rainfall versus irrigation) to capture a distinction between a *water coming naturally element* (rainfall) and a *water abstracted from the environment element* (irrigation), as well as a distinction between two components of irrigation water (licensed irrigation water and traded irrigation water). In addition, the measure does not reveal the cost structure of irrigation water (e.g. licensed irrigation water cost versus traded irrigation water cost). These two pieces of information are captured by the other two drivers of the profit to water cost ratio.

Furthermore, while the return on water describes the relation between profitability and environmental resource (e.g. water resource in this thesis) that a business uses for its production, it does not inform explicitly the extent to which economic return can be created from each dollar of sales, or the ability of water resource used in crop production to generate sales (e.g. sales from lint product and sales from cotton seed product). The former relates to operating activities of a crop business whilst the later relates to water management practices at the operational level of the crop business. In the second-level of analysis that is discussed in a following section (Section 4.4.3), the return on water is decomposed into its drivers to gain insights into how operating activities and operational management practices affect the return on water of the crop business.

#### 4.4.2.2 The effect of weighted average irrigation cost

The second component of the profit to water cost ratio is weighted average irrigation cost. This is a measure that informs the water rate (i.e. \$ of water cost per 1 ML of water) that a crop business has to pay on average for all its water sources to supply to its cropping system for crop production. Weighted average water cost is expressed as the ratio of total water cost incurred by a crop business to the total amount of Crop water supply to the crop business (see Equation 4.7).

Weighted average irigation cost <sub>it</sub> 
$$\left(\frac{\$}{ML}\right)$$
  
= Total water cost<sub>it</sub>  $\left(\frac{\$}{ha}\right)$  ÷ Crop water supply <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [4.7]

As discussed in Section 4.2.1, it is assumed in this thesis that rainfall is a free water resource, with no costs associated with capturing, recycling or storing rainfall to supply the crop business. Taking this assumption, per unit water cost measure becomes weighted average irrigation cost, demonstrating average water cost for the crop business for every unit (i.e. ML) of irrigation water withdrawn. This measure is expressed in Equation 4.5. Accordingly, the second component of the profit to water cost ratio considered in this thesis is a special form of the weighted average water cost measure, which is weighted average irrigation cost.<sup>59</sup>

Mathematically, as illustrated in Equation 4.2, given the other drivers are held constant, reducing weighted average irrigation cost will result in an increase in the profit to water cost ratio. In addition, from an economic point of view, as total water cost forms a (negative) part of operating profit created by a crop business, reducing the weighted average irrigation cost will lead to increased operating profit – this helps to reinforce the value of profit to water cost ratio. In this respect, reducing the weighted average irrigation cost is a step towards crop water and economic sustainability.

<sup>&</sup>lt;sup>59</sup> As a special form of the weighted average water cost measure when rainfall cost is assumed zero, Equation 4.7, compared with Equation 4.5, shows that the rainfall components in both a dollar term and quantity term (i.e. ML of water) are excluded from the numerator and denominator of the measure, respectively.

As outlined in Chapter 3, irrigation water can be obtained by crop businesses from two main sources – licensed irrigation water and traded irrigation water. While the weighted average irrigation cost measure provides useful information about the unit irrigation water cost (i.e. \$/ML) that a crop business has to pay on average for its two irrigation water sources, it is still unclear what actually drives weighted average irrigation cost. In the next section (Section 4.4.3), the weighted average irrigation cost is further decomposed into the two individual irrigation components to allow us to gain more insight into how irrigation water is sourced and specifically how the relative weights (in volume) of each component of irrigation water and their associated unit costs (i.e. \$/ML) affect the weighted average water cost, as well as the total water cost of the crop. This provides the foundation for further analysis on the magnitude effects of licensed and traded water volume and cost on weighted average irrigation cost, and the trade-off between traded irrigation and licensed irrigated sourcing decisions, when evaluating the overall water and economic sustainability performance of a crop business.

Furthermore, since the weighted average irrigation cost measure (in assumption) only captures the volume and cost of the irrigation water component, while excluding the rainfall component of Crop water supply, it is necessary to include another driver of profit to water cost ratio that measures the degree to which Crop water supply to a cropping system is sourced by irrigation and rainfall components.

#### 4.4.2.3 The effect of water use leverage

The third driver of the profit to water cost ratio is water use leverage. This ratio is defined as the degree to which Crop water supply to a cropping system is sourced by water abstracted from the environment (i.e. irrigation abstracted). The higher the water use leverage, the more dependent a crop business is on abstracted irrigation sources to produce a crop.

Water use leverage is expressed as the ratio of irrigation abstracted over Crop water supply (see Equation 4.4). Water use leverage builds on an analogy with the financial leverage described in the financial accounting literature (Nissim & Penman 2001; Penman 2003; Soliman 2008). While the financial leverage indicates the use of borrowed capital to lever up the return on equity of the business, the water use leverage reveals the use of abstracted water to improve crop yield, and thus economic return, for the crop business.

Mathematically, Equation 4.2 shows that, where other drivers are held constant, reducing water use leverage will lead to increased profit to water cost ratio, and hence increased overall water and economic sustainability performance of the crop business. This is because water cost (the denominator of the profit to water cost ratio) is directly related to amount of abstracted irrigation (the numerator of the water use leverage). In addition, from an environmental sustainability point of view, the more irrigation water withdrawn from the environment, either via surface water (e.g. water withdrawn from a river) or groundwater (e.g. water pumped from a bore), the bigger the impact of the crop business on water systems is. In this respect, reducing water abstraction for irrigation, and hence reducing the water use leverage, is a step towards crop water sustainability.

However, from an economic point of view, irrigation is carried out to potentially eliminate the risk of the crop suffering water deficits when rainfall receipts are insufficient to completely satisfy transpirational demands. This means irrigation contributes proportionally to the economic return of the crop business.

The discussion above highlights that the water use leverage measure reflects the risk of rainfall resource shortage and the risk of economic return. As the effect of irrigation use on economic sustainability and environmental sustainability of a crop business may go in opposite directions, it is of crucial importance to consider trade-offs between these two sustainability aspects and the potential risks associated with them when evaluating the overall sustainability performance of the crop business.

Furthermore, one way to reduce the proportion of abstracted irrigation in total Crop water supply is to improve water utilisation efficiency by improving water management practices at the operational level. For a crop business that uses furrow irrigation systems, better management of water runoff from rainfall and furrow irrigation systems results in more water runoff being recycled to the water system of the farm which can be recirculated for crop production. In the second-level of analysis that is discussed in the next section (Section 4.4.3), the water use leverage is decomposed into its drivers to provide a better understanding of how water management practices at the operational level can help to improve the water use leverage so that the overall sustainability performance of the crop business can be improved.

## 4.4.3 Second-level analysis: Decomposition of return on water, water use leverage and weighted average irrigation cost

The first-level of analysis uncovers the three drivers of the profit to water cost ratio, but each of these drivers refers to different aspects of sustainability and relates to different crop management practices/decisions. In the second-level analysis, the three drivers of the profit to water cost ratio are further decomposed into their drivers. The analysis is summarised in Table 4.2.

#### 4.4.3.1 Drivers of return on water

Return on water can be decomposed into its two components: profit margin and economic water use index as expressed in Equation 4.8.

Return on water<sub>it</sub> 
$$\left(\frac{\$}{ML}\right)$$
  
= Profit margin<sub>it</sub>(%) × Economic water use index<sub>it</sub>  $\left(\frac{\$}{ML}\right)$  [4.8]

Where profit margin and economic water use index are defined in Equation 4.9 and Equation 4.10 respectively.

$$Profit \ margin_{it}(\%) = \left[ Opearing \ profit_{it}\left(\frac{\$}{ha}\right) \div Total \ sales \ revenue_{it}\left(\frac{\$}{ha}\right) \right] \times 100\%$$

$$[4.9]$$

Economic water use 
$$index_{it}\left(\frac{\$}{ML}\right)$$
  
= Total sales  $revenue_{it}\left(\frac{\$}{ha}\right) \div Crop water  $supply_{it}\left(\frac{ML}{ha}\right)$  [4.10]$ 

### Table 4.2 WESM analysis: The second-level analysis

Variable	Level	Formula			
Panel A: Analysis of return on water					
<b>Return on water</b> <sub>it</sub> $\left(\frac{\$}{ML}\right)$	Level 2	= Operating $\operatorname{profit}_{it}\left(\frac{\$}{ha}\right)$ ÷ Crop water $\operatorname{supply}_{it}\left(\frac{ML}{ha}\right)$			
		= Profit margin <sub>it</sub> (%) × Economic water use index <sub>it</sub> $\left(\frac{\$}{ML}\right)$			
Profit margin <sub>it</sub> (%)	Level 3	$= \left[ \text{Operating profit}_{\text{it}} \left( \frac{\$}{\text{ha}} \right) \div \text{Total sales revenue}_{\text{it}} \left( \frac{\$}{\text{ha}} \right) \right]$			
		$\times$ 100%			
Economic water use index <sub>it</sub> $\left(\frac{\$}{ML}\right)$	Level 3	= Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$ ÷ Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$			
Panel B: Analysis of water use leverage					
Water use leverage (%)	Level 2	$= \left[ \text{ Irrigation abstracted}_{it} \left( \frac{ML}{ha} \right) \right]$			
		÷ Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$ × 100%			
		= <b>100</b> %			
		– [Rainfall to crop water supply ratio (%)			
		+ Recycled runoff to crop water supply ratio (%)]			
Rainfall to crop water supply ratio (%)	Level 6	= Rainfall it $\left(\frac{ML}{ha}\right)$ ÷ Crop water supply it $\left(\frac{ML}{ha}\right)$			

Recycled runoff to crop water supply ratio (%)	Level 6	= Recycled runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$ ÷ Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$			
Panel C: Analysis of weighted average irrigation cost					
Weighted average irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ML}}\right)$	Level 2	$= \text{Total water } \operatorname{cost}_{it} \left( \frac{\$}{ha} \right) \div \text{Irrigation } \operatorname{abstracted}_{it} \left( \frac{ML}{ha} \right)$			
		$= \left  \text{Licensed irrigation to Irrigation abstracted ratio}_{it} (\%) \right $			
		× Unit licensed irrigation $cost_{it} \left(\frac{\$}{ML}\right)$			
		+ Traded irrigation to Irrigation abstracted ratio <sub>it</sub> (%)			
		× Unit traded irrigation $cost_{it} \left(\frac{\$}{ML}\right)$			
Licensed irrigation to Irrigation abstracted ratio <sub>it</sub>	Input	$= \left[ \text{Licensed irrigation}_{it} \left( \frac{\text{ML}}{\text{ha}} \right) \div \text{Irrigation abstracted}_{it} \left( \frac{\text{ML}}{\text{ha}} \right) \right]$			
		imes 100%			
Traded irrigation to Irrigation abstracted ratio <sub>it</sub> (%)	Input	$= \left[ \text{Traded irrigation}_{\text{it}} \left( \frac{\text{ML}}{\text{ha}} \right) \div \text{Irrigation abstracted}_{\text{it}} \left( \frac{\text{ML}}{\text{ha}} \right) \right]$			
		$\times 100\%$			
Unit licensed irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ML}}\right)$	Input	Accounting Data			
Unit traded irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ML}}\right)$	Input	Accounting Data			

Comparing Equation 4.3, which shows the formula for calculating the return on water measure, to Equation 4.8, which demonstrates the break-down of return on water into its two drivers, taking into account of Equations 4.9 and 4.10, it is proved that they are equal.

$$\begin{aligned} \text{Return on water}_{it}\left(\frac{\$}{ML}\right) \\ &= Operating \ profit_{it}\left(\frac{\$}{ha}\right) \div \ Crop \ water \ supply_{it}\left(\frac{ML}{ha}\right) \\ &= Profit \ margin_{it}(\%) \times \ Economic \ water \ use \ index_{it}\left(\frac{\$}{ML}\right) \\ &= \left[Operating \ profit_{it}\left(\frac{\$}{ha}\right) \div \ Total \ sales \ revenue_{it}\left(\frac{\$}{ha}\right)\right] \\ &\times \left[\text{Total sales } revenue_{it}\left(\frac{\$}{ha}\right) \div \ Crop \ water \ supply_{it}\left(\frac{ML}{ha}\right)\right] \\ &= Operating \ profit_{it}\left(\frac{\$}{ha}\right) \div \ Crop \ water \ supply_{it}\left(\frac{ML}{ha}\right) \end{aligned}$$

The decomposition of return on water into profit margin and economic water use index enables us to distinguish between operating activities, which drive profitability of the crop business, and resource use - related decisions, which drive the economic water use index of the crop business.

#### The effect of profit margin

As demonstrated in Equation 4.9, the profit margin is the ratio of operating profit to total sales revenue. Profit margin is well-defined and widely used in the financial accounting literature to measure the net profit as a percentage of sales revenue (Nissim & Penman 2001; Soliman 2008).

While the profit margin reveals the profitability of each dollar of sales, this measure per se only refers to the economic aspect of a business and not its environmental sustainability which in this case relates to how efficient water resource is used in the crop production process in generating sales revenue for the crop business. The later aspect is captured by the second driver of the return on water. Furthermore, from Equation 4.8, it is seen that where the economic water use index is unchanged, an increase in profit margin will lead to an increase in return on water. In this respect, it is necessary to identify the driving factors of profit margin and understand how changes in the profit margin's drivers affect the return on water and ultimately the profit to water cost ratio. In the next section, the numerator (operating profit) and denominator (sales revenue) of the profit margin ratio will be decomposed into their components to allow the establishment of the information hierarchy from the business level down to the lower level of a crop business.

#### The effect of economic water use index

As described in Equation 4.10, economic water use index is defined as the value of crop products (total sales revenue) produced per unit of water volume (i.e. ML) supplied to the cropping system. Similar to the asset turnover ratio, presenting the sales revenue generated for each dollar of net operating assets, described in the financial ratio analysis literature to measure asset utilization and efficiency<sup>60</sup> (Figge & Hahn 2013; Soliman 2008), the economic water use index is referred to as a measure of efficient use of an environmental resource (e.g. water resource in this thesis).

One of the key contributions of this research is that the economic water use index allows an end-to-end connection between water resource utilisation and crop profitability. As the numerator of the economic water use index captures total sales revenue (including sales revenue of all crop products) and its denominator includes all water sources input to the cropping system, the index explicitly demonstrates a relationship between water resource coming in (to the crop production) and cash coming out (from the crop production). As such, the economic water use index is a useful measure for quantifying the use of water resource by a crop business to create its value.

Equation 4.8 also indicates that improvement in the economic water use index will improve the return on water measure. In order to gain insights into the impact water management decisions have, including irrigation decisions and water storage and recycling decisions, on

<sup>&</sup>lt;sup>60</sup> For example, the efficient use of property, plant and equipment, efficient inventory processes, and working capital efficiency (Penman 2003).

the economic water use index, this measure will be further broken down into its drivers in the next section (Section 4.4.4).

#### Analysis of the drivers of return on water

In the second-level breakdown of the WESM model I establish the relationship between the profitability of each dollar of sales (profit margin) and the use of water to the generation of sales (economic water use index). The decomposition of the return on water measure reveals that the economic return of a crop business in respect to water resource comes from two sources: profit margin and economic water use index. This means that for a crop business, as a higher proportion of each dollar of sales ends up in operating profit and/or the more sales are generated from water resource supplied to the crop business, the higher the economic return on water resource.

A crop business can produce a given level of return on water (for example, \$120/ML or \$120 of operating profit generated for every 1 ML of water supplied) in two ways. First, the business can have a relatively high profit margin (for example, 30% or \$0.3 of operating profit is generated for every \$1 of sales revenue) but low economic water use index (for example, \$400/ML or \$400 of sales revenue is generated for every 1 ML of water supplied). Second, the business can have a relatively high economic water use index (for example, \$600/ML or \$600 of sales revenue is generated for every 1 ML of water supplied). Second, the business can have a relatively high economic water use index (for example, \$600/ML or \$600 of sales revenue is generated for every 1 ML of water supplied) but low profit margin (for example, 20% or \$0.2 of operating profit is generated for every \$1 of sales revenue).

While the first driver of return on water is predominantly a profitability measure, the second component is an efficiency measure. On the one hand, profitability measurement is related to operating cost, including water cost.<sup>61</sup> On the other hand, the economic water use index is driven by water management decisions/practices which inform how efficiently a crop business uses water resources to generate crop product(s). External factors (e.g. climate factors, regional characteristics, water policy), which impact availability of water resource (both rainfall and irrigation) and water cost, are considered contributing factors to the two components of the return on water.

<sup>&</sup>lt;sup>61</sup> As the focus of this thesis is water and economic sustainability, in the lower levels of analysis, water cost will be isolated to allow examination of the impact of cost of water resources on businesses' profitability.

#### 4.4.3.2 Drivers of weighted average irrigation cost

In the second-level of analysis, weighted average irrigation cost is decomposed into unit costs of individual irrigation components and the relative weights (in volume) of each type of irrigation water of the total irrigation water withdrawn by a crop business.

Weighted average irrigation cost is calculated in very much the same way as weighted average cost of capital in financial accounting research. For example, given that debt and equity are two main financing (capital) sources of a firm, carrying two different types of cost of capital, namely cost of debt and cost of equity respectively, the weighted average cost of capital is calculated (see Equation 4.8) taking into account the relative weights of each component of the capital structure of the firm.

Weighted average cost of capital (%) = [Equity (\$) ÷ Total capital (\$)] × Cost of Equity (%) [4.12] + [Debt (\$) ÷ Total capital (\$)] × Cost of Debt (%)

In the context of irrigation sources in agriculture, as discussed in Chapter 3, crop businesses can obtain irrigation water from two main sources – licensed irrigation water and traded irrigation water. Part of the decision to allow water trading was a hope that this would lead to water being used by the most efficient or profitable farmers, who could afford the trades thus returning maximum dollar benefit to society for use of a public good.

Furthermore, as water prices (i.e. \$/ML) vary between licensed water and traded water, across years and locations<sup>62</sup>, average irrigation costs are calculated based on the relative weights (in volume) of each component of total irrigation water withdrawn as follows:

<sup>&</sup>lt;sup>62</sup> See an example of State Water Pricing Application: 2014-15 – 2016-17 issued by the Australia Government in <u>https://www.accc.gov.au/regulated-infrastructure/water/state-waters-regulated-charges-2014-17-review/final-decision</u>

Weighted average irrigation  $cost_{it} \left(\frac{\$}{ML}\right)$ 

$$= \left[ \text{Licensed irrigation to Irrigation abstracted ratio}_{it} (\%) \right]$$

$$\times \text{Unit licensed irrigation cost}_{it} \left(\frac{\$}{ML}\right) \right]$$

$$+ \left[ \text{Traded irrigation to Irrigation abstracted ratio}_{it} (\%) \right]$$

$$\times \text{Unit traded irrigation cost}_{it} \left(\frac{\$}{ML}\right) \right]$$

In Equation 4.13, unit licensed irrigation cost (\$/ML) and unit traded irrigation cost (\$/ML) represent water prices (\$/ML) of licensed water and traded water respectively. In addition, licensed irrigation to Irrigation abstracted ratio (%), and traded irrigation to Irrigation abstracted ratio (%), which capture the relative weights (in volume) of each type of irrigation water of the total irrigation water withdrawn by a crop business respectively, are expressed below:

Licensed irrigation to Irrigation abstracted ratio $_{it}$  (%)

$$= \left[ \text{Licensed irrigation}_{it} \left( \frac{ML}{ha} \right) \right]$$

$$\div \text{Irrigation abstracted}_{it} \left( \frac{ML}{ha} \right) \times 100\%$$

$$(4.14)$$

Traded irrigation to Irrigation abstracted ratio $_{it}$  (%)

$$= \left[ Traded \ irrigation_{it} \left( \frac{ML}{ha} \right) \right]$$

$$\div \ Irrigation \ abstracted \ _{it} \left( \frac{M}{ha} \right) \times 100\%$$

$$(4.15)$$

As indicated in the WESM model diagram (see Figure 4.1) and Table 4.2, these four components of the weighted average irrigation cost are input into the WESM model to enable quantification of this second-level measure.

#### 4.4.3.3 Drivers of water use leverage

As expressed in Equation 4.4, water use leverage is the ratio of irrigation water abstracted to total crop water supply. In the second-level of analysis, the water use leverage measure is decomposed into various water sources supplied to the cropping system of a crop business. This allows us to establish the link between the measure of water sustainability at the high level and components at the operational level of the crop business which are driven by water management decisions.

The crop water process model constructed in Chapter 3 shows that two main sources of crop water supply to a cropping system are rainfall and irrigation, and irrigation is needed to supplement rainfall where there is insufficient rainfall supplied to the cropping system (see Equation 3.10). From a management perspective, the rainfall component is considered as an uncontrollable factor while the irrigation component is determined and controlled by water management by irrigation decisions, including irrigation method, irrigation scheduling decisions, and runoff recycling practice.

Furthermore, as discussed in Section 3.2.3, recycled runoff can be treated as a source of water supply, and hence reduces the water impost of cotton cultivation on the environment (see Equation 3.15). Therefore, crop water supply comprises three components, rainfall, recycled runoff, and irrigation water abstracted (see Equation 3.16).

When water recycling is taken into account, the water use leverage measure expressed in Equation 4.4 can be rearranged as follow:

*Water use leverage*<sub>*it*</sub> (%)

- = 100%
- [Rainfall to crop water supply ratio (%)
   [4.16]
- + Recycled runoff to crop water supply ratio (%)]

Equation 4.16 demonstrates more recycled runoff means less water needs to be abstracted from the environment. As such, the water use leverage measure of the crop business can be improved for better water sustainability performance.

As discussed above, the level of irrigation applied to a cropping system, and the level of recycled runoff are driven by a range of water management decisions that are made at the operational level of a crop business; this includes irrigation system decisions, irrigation scheduling decisions, and water recycling decisions. The impact of such decisions with regard to water management is captured by the drivers of the water use leverage that are laid out in Level 7 of the WESM model, including irrigation applied per event, runoff from irrigation per event, number of irrigation events, ratio of recycled runoff. By establishing the link between the water use leverage and its drivers, the impact of water management decisions at the operational level of a crop business on its water sustainability performance can be quantified and managed.

## 4.4.4 Third-level analysis: Decomposition of profit margin and economic water use index

The first-level breakdown of the WESM model shows that profit to water cost ratio is driven by economic return of a crop business in respect to water resource (return on water), the degree to which a crop business is dependent on irrigation sources to produce crop products (water use leverage), and how irrigation is sourced (weighted average irrigation cost). In the second-level breakdown, the drivers of weighted average irrigation cost and water use leverage are identified. Since these drivers directly link to water management decisions at the crop production level, including how crop water supply is sourced and how irrigation is sourced, the relationship between operational decisions with regard to water management and these two drivers of profit to water ratio is established to enable quantification and management of the impact of such operational decisions on the overall sustainability performance of a crop business.

In addition, also in the second-level analysis of WESM model, it is revealed that return on water is driven by the profitability of a crop business (profit margin) and how efficient water resource is used to generate sales revenue for the crop business (economic water use index). While measures, such as profit to water cost ratio, water use leverage, weighted average irrigation cost, return on water, profit margin, economic water use index, are argued in this thesis as higher level sustainability measures because they do not directly link to activities of the crop production (or capture the science behind crop production output at the operational level), such high level measures provide limited explanation about the drivers of water and

economic sustainability performance. Therefore, in the third-level analysis, the two drivers of return on water are further decomposed into their components that are directly related to the operational level of a crop business where sustainability issues are argued to occur.

#### 4.4.4.1 Drivers of profit margin

Mathematically, ratios change because of changes in one or both terms involved (e.g. the numerator and denominator). In order to gain insights to what actually drives the profit margin ratio, it is required to decompose each of its terms into their components. As defined in Equation 4.9, operating profit is the numerator and total sales revenue is the denominator of the profit margin. Accordingly, these two components of the profit margin ratio are arranged in the next level down of the WESM model (i.e. Level 3) (see Figure 4.1) and also presented in Table 4.3.

## Table 4.3 WESM analysis: The third-level analysis

Variable	Level	Formula		
Panel A: Analysis of profit margin				
Profit margin <sub>it</sub> (%)	Level 3	$= \left[ \text{Operating profit}_{it} \left( \frac{\$}{ha} \right) \div \text{Total sales revenue}_{it} \left( \frac{\$}{ha} \right) \right] \times 100\%$		
Operating $\operatorname{profit}_{it}\left(\frac{\$}{ha}\right)$	Level 4	= Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$ - Total operating cost <sub>it</sub> $\left(\frac{\$}{ha}\right)$		
Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$	Level 5	= Sales revenue <sub>it</sub> (lint) $\left(\frac{\$}{ha}\right)$ + Sales revenue <sub>it</sub> (cotton seed) $\left(\frac{\$}{ha}\right)$		
Panel B: Analysis of economic water use index				
Economic water use $index_{it}\left(\frac{\$}{ML}\right)$	Level 3	= Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$ ÷ Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$		
		= Crop water use index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{ML}}\right) \div \text{Lint percentage}_{it}(\%)$		
		× Weighted average cotton price <sub><i>it</i></sub> $\left(\frac{\$}{\text{kg of total product}}\right)$		
Crop water use $index_{it} \left( \frac{kg \text{ of } lint}{ML} \right)$	Level 4	$= \text{Lint yield}_{\text{it}} \left(\frac{\text{kg of lint}}{\text{ha}}\right) \div \text{Crop water supply}_{\text{it}} \left(\frac{\text{ML}}{\text{ha}}\right)$		
Weighted average cotton price <sub><i>it</i></sub> $\left(\frac{\$}{\text{kg of total yield}}\right)$	Level 4	= Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right) \div$ Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{ha}\right)$		
Lint percentage <sub>it</sub> (%)	WESM Input	$= \left[ \text{Lint yield}_{\text{it}} \left( \frac{\text{kg of lint}}{\text{ha}} \right) \div \text{Total yield}_{\text{it}} \left( \frac{\text{kg of total product}}{\text{ha}} \right) \right] \times 100\%$		

From an accounting perspective, operating profit is calculated by deducting the amount of operating cost<sup>63</sup> from sales revenue.

Operating profit<sub>it</sub> 
$$\left(\frac{\$}{ha}\right)$$
  
= Total sales revenue<sub>it</sub>  $\left(\frac{\$}{ha}\right)$  – Total operating cost<sub>it</sub>  $\left(\frac{\$}{ha}\right)$  [4.17]

Equations 4.9 and 4.17 reveal that total sales revenue and total operating cost are the two drivers of operating profit and profit margin measures.

Operating cost consists of various costs associated with activities undertaken for crop production, including cultivation cost, planting cost, irrigation cost, fertiliser cost, and so on (see Appendix A1). One way to improve the profitability of a crop business is to cut costs incurred by the business. More specifically, in respect to water resources, improvement of water management would result in less irrigation water required for crop production and hence reduced irrigation costs. In the fifth-level of analysis (see Appendix A2: Table A1, Panel E), total operating costs will be decomposed into their components in order to isolate irrigation costs from other operating cost. This enables crop managers to quantify the impact of water management decisions on the profitability of the crop business.

Furthermore, as demonstrated in Equation 4.9 and Equation 4.17, where the total operating cost is held constant, the more the total sales revenue generated by a crop business, the higher the profitability of the crop business. From a crop production perspective, total sales revenue generated by a crop business comprises of sales revenue generated from the primary product (i.e. lint product in cotton production) and sales revenue generated from the by-product (i.e. cotton seed product in cotton production). For each of the sales revenue components, its value is driven by two factors: a crop factor (production output/yield), and an economic factor (i.e. selling price). In the fifth-level of analysis (see Appendix A2: Table A1, Panel E),), total sales revenue will be further decomposed into its components for examining its driving factors and identifying areas for improvement.

 $<sup>^{63}</sup>$  As mentioned above, this thesis only focuses on operating activities, not financing activities, when examining water and economic sustainability since it is argued that sustainability issues arise as a result of unsustainable activities undertaken in the process of producing crop products. Accordingly, only operating cost – cost incurred to undertake operating activities at the operational level of a crop business – is considered in the analysis.

#### 4.4.4.2 Drivers of economic water use index

In the third-level of analysis, the economic water use index is decomposed into its three drivers: crop water use index, weighted average cotton price, and lint percentage, as described below:

Economic water use 
$$index_{it}\left(\frac{\$}{ML}\right)$$
  
= Crop water use  $index_{it}\left(\frac{kg \ of \ lint}{ML}\right) \div Lint \ percentage_{it}(\%)$ 
(4.18)  
× Weighted average cotton  $price_{it}\left(\frac{\$}{kg \ of \ total \ product}\right)$ 

Where crop water use index is defined in Chapter 3 (Section 3.3.3, Equation 3.26), while lint percentage and weighted average cotton price are defined below:

$$Lint \ percentage_{it} (\%) = \\ = \left[Lint \ yield_{it} \left(\frac{kg \ of \ lint}{ha}\right) \\ \div \ Total \ yield_{it} \left(\frac{kg \ of \ total \ product}{ha}\right)\right] \times 100\%$$

$$(4.19)$$

Weighted average cotton 
$$price_{it}\left(\frac{\$}{kg \text{ of total yield}}\right) =$$
  

$$= Total \text{ sales } revenue_{it}\left(\frac{\$}{ha}\right)$$

$$\div Total \text{ yield}_{it}\left(\frac{kg \text{ of total product}}{ha}\right)$$
[4.20]

In Equation 4.19 and Equation 4.20, total yield refers to sum of the output of all crop products. In the context of cotton production, total yield comprises lint yield and cotton seed yield.

$$Total yield_{it} \left(\frac{kg \ of \ total \ product}{ha}\right) = Lint \ yield_{it} \left(\frac{kg \ of \ lint}{ha}\right) + Seed \ yield_{it} \left(\frac{kg \ of \ seed}{ha}\right)$$

$$[4.21]$$

Since the crop water use index measure only defines the relationship between the output of the primary product of a cotton crop (e.g. lint yield) and the amount of water supplied for crop production, lint percentage is incorporated into crop water use index in order to capture the total economic yield of the crop.



In addition, by incorporating the weighted average cotton price into the Equation 4.22, the relationship between total sales revenue generated by a crop business and the water resource provided to the crop business for crop production processes is established.

Crop water use 
$$index_{it}\left(\frac{kg \ of \ lint}{ML}\right) \div Lint \ percentage_{it}(\%)$$

$$\times Weighted \ average \ cotton \ price_{it}\left(\frac{\$}{kg \ of \ total \ product}\right)$$

$$= \left[Total \ yield_{it}\left(\frac{kg \ of \ total \ product}{ha}\right)\right]$$

$$\div \ Crop \ water \ supply_{it}\left(\frac{ML}{ha}\right)\right]$$

$$\times \left[Total \ sales \ revenue_{it}\left(\frac{\$}{ha}\right)$$

$$\div \ Total \ yield_{it}\left(\frac{kg \ of \ total \ product}{ha}\right)\right]$$

$$= Total \ sales \ revenue_{it}\left(\frac{\$}{ha}\right) \div \ Crop \ water \ supply_{it}\left(\frac{ML}{ha}\right)$$

Comparing Equation 4.10, which defines the economic water use index measure, to Equation 4.23, where the economic water use index measure is decomposed into its three components, it is proved that they are equal.

The decomposition of the economic water use index measure into its three drivers is useful to crop managers in terms of looking at potential increases in economic return of the crop business on water resource through improvements in those three drivers.

#### The effect of crop water use index

The crop water use index is defined as the amount of primary crop product produced per unit of water volume supplied to the cropping system. In the third-level breakdown, the crop water use index is isolated as an important driver of the economic water use index measure, focusing on the primary (high-value) product of cotton production. If crop economic factors (e.g. lint price and cotton seed price) and lint percentage in cotton products remain constant, improving the crop water use index is the key factor to drive increases in the economic water use index.

As discussed in detail in Section 3.3.3, the crop water use index provides a means of evaluating the ability of a crop business to generate primary crop product from each unit of

water volume supplied to the cropping system. In addition, it can be a useful measure for crop managers to look at potential increases in crop yield, and hence economic return, per unit of the resource constraint (water resource). However, by merely looking at the value of the crop water use index, it is not known what (management) factors actually drive this measure. Therefore, in the fourth-level of analysis, the crop water use index will be further decomposed into its components to allow us to identify (management) areas for improvements of crop businesses' performance.

#### The effect of lint percentage

The lint percentage component, the second driver of the economic water use index, is a measurement of the weight ratio of lint to total lint and cotton seed. This measure is mainly determined by the type of cotton cultivars selected in a particular field (Verhalen, Greenhagen & Thacker 2003). For varieties that are currently supplied and used for cotton production in Australia, lint percentage varies between 39% and 44%.<sup>64</sup>

As expressed in Equation 4.23, incorporation of lint percentage into crop water use index enables us to establish the link between total yield of a crop, which contributes to the overall economic outcome of the crop business, and the Crop water supply to the cropping system. As illustrated in the WESM model diagram, the lint percentage variable is an input variable, forming a part of the economic water use index.

### The effect of weighted average cotton price

The third component of the economic water use index is weighted average cotton price. This measure represents the average selling price of cotton products (i.e. \$/kg) that a crop business offers its customers. Since a cotton business produces two economic products (lint product and cotton seed product), the constructed weighted average cotton price provides a simple way to incorporate both products into one dimension for capturing the total income a crop business generates from cotton production.

<sup>&</sup>lt;sup>64</sup> http://www.csd.net.au/varietyguide.

In very much the same way as the weighted average irrigation cost is calculated, weighted average cotton price of a crop business is calculated by dividing total sales revenue generated for a period by total yield produced in that period (see Equation 4.20).

Weighted average cotton price captures the economic dimension of the economic water use index. If crop management factors (e.g. captured by the crop water use index) and lint percentage in cotton products remain constant, higher weighted average cotton price will result in a higher economic water use index. Since weighted average cotton price consists of two crop production components, lint and cotton seed, each of which carries a different economic value, it is essential to further decompose this measure into its components to gain more insights into what drives the total economic value of a crop business. Accordingly, in the next section, weighted average cotton price will be further broken down into its components.

## 4.4.5 Fourth-level analysis: Decomposition of crop water use index and weighted average cotton price

The breakdown of the economic water use index measure enables crop managers to distinguish between the managerial effect, which is captured by the crop water use index, and economic effect, which is captured by weighted average cotton price on this measure. In the fourth-level analysis, the crop water use index and the weighted average cotton price are further decomposed into their drivers in order to enable us to gain insights into what affects change in these two measures. The analysis is summarised in Appendix A2 – Table A1 – Panel D.

#### 4.4.5.1 Decomposition of crop water use index

The decomposition of the crop water use index into its three drivers (water input efficiency, transpiration index and harvest index) was conducted in Chapter 3 (see Equation 3.27). As discussed in Section 3.3.3, the three drivers of the crop water use index is useful to crop managers for identifying management areas for continuous improvement in order to move towards sustainability.

#### Improving water input efficiency is a step towards crop water sustainability

In respect to the water aspect of crop production, the theoretical link between lint yield and transpiration described in Equation 4.37 demonstrates that plant transpiration is the only water outflow component that contributes to crop production, and hence, adds value to a crop

business. However, the crop water process model illustrated in Chapter 3 (see Figure 3.2) reveals that a portion of Crop water supply to the cropping system does not contribute to crop production and is considered as crop water loss (non-productive water component).

Crop water loss negatively affects economic and water sustainability performance of a crop business in a number of ways. First, as irrigation is one of the two main water sources supplied to a cropping system, as indicated in the crop water process model (see Figure 3.2), water lost from the cropping system (e.g. deep drainage, water run-off and soil evaporation) means more irrigation water is required for crop production (given the same amount of rainfall). Increasing irrigation abstraction from surface or groundwater within a water system potentially causes imbalance in the water system (Ibaraki 2010). This is one of the key issues related to water sustainability addressed in Chapter 2.

Second, since irrigation is costly, increasing irrigation leads to increasing cost of water incurred by the crop business (Cammarano et al. 2012). Furthermore, increasing the amount of water abstracted for irrigation leads to increasing water competition among water users – agriculture, industry, environment and urban areas (Mehta et al. 2013; Rijsberman 2006). This will impact the ability of crop businesses to sustain their economic growth in the long-run. Third, the amount of water lost could be beneficially used to grow more crops. This not only helps to improve the profitability of the crop business but also allows increased supply of cotton products to society. Finally, increasing water loss means more water being discharged to the natural environment. The quality of water output from cropping systems has become a great concern in respect to water sustainability (Chapagain et al. 2006; Ibaraki 2010). It is evident that water loss from deep drainage causes significant environmental harm (Smith, Raine & Minkevich 2005).

Therefore, a crop business to move towards economic and water sustainability requires improving the efficient use of crop water input to a cropping system to capture more of the water supply for use in transpiration. However, while the water input efficiency provides useful information on how well water resource is managed, how efficient crop water resource is used, and how changes in water input efficiency affect changes in crop water use index, it is not yet clear, among various water components defined in the crop water process model, which are the key driving factors of water input efficiency with regard to water management generally and irrigation decisions particularly. In the next section, the water input efficiency will be further decomposed into its water components.

#### 4.4.5.2 Decomposition of the weighted average cotton price

Similar to how the weighted average irrigation cost measure is decomposed into its components (see Equation 4.13), the weighted average cotton price is broken down into two components of crop production, taking into consideration their relative weights to the total production, and their associated selling prices.

Weighted average cotton 
$$price_{it}\left(\frac{\$}{kg \text{ of total yield}}\right)$$
  

$$= \left[Lint \ percentage_{it} (\%) \times Lint \ price_{it} \left(\frac{\$}{kg}\right)\right]$$

$$+ \left[Seed \ percentage_{it} (\%) \times Seed \ price_{it} \left(\frac{\$}{kg}\right)\right]$$
[4.24]

Where seed percentage is the weight ratio of cotton seed to total lint and cotton seed and defined as below.

Seed percentage<sub>it</sub> (%) =  
= 
$$\left[Seed \ yield_{it} \left(\frac{kg \ of \ seed}{ha}\right)$$
  
 $\div \ Total \ yield_{it} \left(\frac{kg \ of \ total \ product}{ha}\right)\right] \times 100\%$ 
[4.25]

Since total yield consists of lint yield and seed yield, the sum of lint percentage<sup>65</sup> and seed percentage equal to 100%.

Seed 
$$percentage_{it}$$
 (%) = 100% – Lint  $percentage_{it}$  (%) [4.26]

The decomposition of the weighted average cotton price above is based on the three components that affect this measure are lint percentage, lint price and cotton price. When the

<sup>&</sup>lt;sup>65</sup> See Equation 4.29 for definition of lint percentage.

effect of lint percentage, which is mainly driven by types of crop variety selected for growing cotton crops, is isolated, the weighted average cotton price is driven by two economic factors: lint price and cotton seed price. As indicated in the WESM model diagram (see Figure 4.1), lint price and cotton seed price are input variables to the WESM model to enable quantification of the weighted average cotton price measure.

Furthermore, cotton lint is a significantly higher value product than cotton seed. <sup>66</sup>Consequently lint price is considered as the main economic factor that drives the average price of cotton products. Compared to cotton seed, a smaller change in lint price can result in greater change in the weighted average cotton price, leading to greater change in the economic value of a crop business. The selling price of cotton lint is typically driven by both external factors, such as market competition, demand for crop products at a particular time of the year, futures contracts, as well as internal factors, such as fibre quality which is driven by irrigation management and other crop management.

The decomposition of the weighted average cotton price allows us to identify the key drivers of the average selling price of cotton products. In addition, another benefit of decomposing the weighted average cotton price is that break-even analysis can be conducted to determine the break-even point in lint yield (e.g. what level of lint yield produced does the crop business start to make profit) and/or the break-even point in lint price (e.g. at which lint price does the crop business start to make profit). This analysis can be a useful tool to support crop managers in planning crop production and managing risks associated with the business' profitability.

## 4.4.6 Fifth-level analysis: Decomposition of total sales revenue, total operating cost and water input efficiency

#### 4.4.6.1 Drivers of total sales revenue

In the fifth-level of analysis, the two sources of crop sales revenue will be further decomposed into their crop and economic factors.

<sup>&</sup>lt;sup>66</sup> For example, from a sample budget for furrow irrigated cotton for Central and Northern NSW area (2014 – 2015) (NSW Department of Primary Industries 2015), the selling prices of lint product and cotton seed product are \$2.09/kg of lint and \$0.32/kg of cotton seed, respectively.

Total sales revenue 
$$_{it}\left(\frac{\$}{ha}\right)$$
  
= Sales revenue  $(lint)_{it}\left(\frac{\$}{ha}\right)$   
+ Sales revenue  $(cotton seed)_{it}\left(\frac{\$}{ha}\right)$  [4.27]

Where sales revenue from lint product and cotton seed product are determined based on their associated production output and selling price.

Sales revenue 
$$(lint)_{it}\left(\frac{\$}{ha}\right)$$
  
= Lint yield<sub>it</sub>  $\left(\frac{kg \ of \ lint}{ha}\right) \times Lint \ price_{it}\left(\frac{\$}{ha}\right)$  [4.28]

Seed revenue (cotton seed)<sub>it</sub> 
$$\left(\frac{\$}{ha}\right)$$
  
= Seed yield<sub>it</sub>  $\left(\frac{kg \ of \ seed}{ha}\right) \times Seed \ price_{it} \left(\frac{\$}{ha}\right)$  [4.29]

Given lint percentage and total yield are WESM model input variables, lint yield and seed yield can be calculated as below.

$$Lint \ yield_{it} \left(\frac{kg \ of \ lint}{ha}\right)$$

$$= \left[Total \ yield_{it} \left(\frac{kg \ of \ total \ product}{ha}\right) \times Lint \ percentage_{it} (\%)\right] \div 100\%$$
[4.30]

Seed yield<sub>it</sub> 
$$\left(\frac{kg \text{ of seed}}{ha}\right)$$
  
=  $\left[Total yield_{it} \left(\frac{kg \text{ of total product}}{ha}\right) \times (1 - Lint percentage_{it} (\%))\right] \div 100\%$  [4.31]

Therefore, the economic value (e.g. total sales revenue) of a crop business is driven by crop factors (total yield and lint percentage) and economic factors (lint price and cotton seed price).

$$Total \ sales \ revenue_{it}\left(\frac{\$}{ha}\right)$$

$$= \ Lint \ price_{it}\left(\frac{\$}{ha}\right)$$

$$\times \left[Total \ yield_{it}\left(\frac{kg \ of \ total \ product}{ha}\right) \right]$$

$$\times \ Lint \ percentage_{it} \ (\%) \ \div \ 100\% + Seed \ price_{it}\left(\frac{\$}{ha}\right)$$

$$\times \left[Total \ yield_{it}\left(\frac{kg \ of \ total \ product}{ha}\right) \right]$$

$$\times (1 - \ Lint \ percentage_{it} \ (\%)) \ \div \ 100\%$$

The decomposition of total sales revenue into its components enables us to isolate economic factors and crop factors when analysing the effect of crop management practices on the economic value of a crop business. Equation 4.32 reveals that if other components are constant, total yield is the key driver of total sales revenue. Furthermore, crop research discusses that water is one of the key driving factors for achieving potential yield (Hochman et al. 2013; van Ittersum et al. 2012). Therefore, increased water availability and improved water management practices need to be considered by crop managers for potential increase in crop yield.

#### 4.4.6.2 Drivers of total operating cost

When crop managers look for potential increases in the profitability of the crop business through operational cost savings, it is of crucial importance for them to understand what factors drive the total operating cost and how to manage them. In this section, total operating cost is first decomposed into its two cost components: variable operating cost and fixed operating cost.

Total operating 
$$cost_{it}\left(\frac{\$}{ha}\right)$$
  
= Variable operating  $cost\left(\frac{\$}{ha}\right)$ 
  
+ Fixed operating  $cost_{it}\left(\frac{\$}{ha}\right)$ 
[4.33]

In the WESM analysis, the unit analysis chosen is a unit of crop area (e.g. a hectare) and the cost object considered is lint production. In this respect, fixed operating cost is considered as operating cost that do not change in relation to the level of lint production. In contrast, variable operating cost is defined as cost that changes in proportion to the level of lint production.

Furthermore, with regard to water management, it is essential for crop managers to isolate irrigation cost from other variable operating costs in order to gain a better understanding of the effect of water management, including irrigation management, on the total operating cost and hence profitability of the crop business. Accordingly, three cost items that form total operating costs are defined in Equation 4.34.

$$Total operating cost_{it} \left(\frac{\$}{ha}\right)$$

$$= Total water cost_{it} \left(\frac{\$}{ha}\right) + Other variable cost_{it} \left(\frac{\$}{ha}\right) \qquad [4.34]$$

$$+ Fixed operating cost_{it} \left(\frac{\$}{ha}\right)$$

By definition, fixed operating costs remain unchanged (within a relevant range of lint production). This component is treated as an input variable to the WESM model. In addition, apart from irrigation cost, other variable cost is assumed as a function of lint yield with unit other variable cost as its coefficient.

$$Other \ variable \ cost_{it} \left(\frac{\$}{ha}\right)$$

$$= \ Unit \ other \ variable \ cost_{it} \left(\frac{\$}{kg \ of \ lint}\right)$$

$$\times Lint \ yield_{it} \left(\frac{kg}{ha}\right)$$

$$[4.35]$$

The most important driver of total operating cost, with regard to the water management aspect, is decomposed into its two components: licensed irrigation cost and traded irrigation cost.

$$Total water cost_{it} \left(\frac{\$}{ha}\right) =$$

$$= Licensed irrigation cost_{it} \left(\frac{\$}{ha}\right)$$

$$+ Traded irrigation cost_{it} \left(\frac{\$}{ha}\right)$$

$$[4.36]$$

Where licensed irrigation cost and traded irrigation cost are driven by their irrigation volume and unit irrigation cost.

Licensed irrigation  $cost_{it}\left(\frac{\$}{ha}\right)$ = Licensed irrigation  $volume_{it}\left(\frac{ML}{ha}\right)$  × Unit licensed irrigation  $cost_{it}\left(\frac{\$}{ML}\right)$ [4.37] Traded irrigation  $cost_{it}\left(\frac{\$}{ha}\right)$ 

$$= Traded irrigation volume_{it} \left(\frac{ML}{ha}\right)$$

$$\times Unit traded irrigation cost_{it} \left(\frac{\$}{ML}\right)$$
[4.38]

#### 4.4.6.3 Drivers of water input efficiency

Water input efficiency is defined in Chapter 3 as a measure that captures the degree to which crop water supply is actually used for carbon assimilation to generate biomass. As demonstrated in Equation 3.22, transpiration is a numerator and crop water supply is a denominator of the ratio.

However, by only looking at transpiration, which is the amount of water required by plants to produce biomass, and total crop water supply, it does not provide information on how well crop managers manage and use crop water. It is the crop water loss components or non-productive water components that make the water input efficiency poor. Therefore, in the fifth-level of analysis, the water input efficiency measure is further broken down into productive and non-productive water components.

$$Water input efficiency_{it}(\%) = \left[ Crop water supply_{it} \left( \frac{ML}{ha} \right) - Crop water loss_{it} \left( \frac{ML}{ha} \right) - recycled runoff \right] \div Crop water supply_{it} \left( \frac{ML}{ha} \right)$$

$$[4.39]$$

Equation 4.39 reveals that maximising water input efficiency will ensure that the maximum possible proportion of available fresh water supplied is used for transpiration to yield biomass. Ideally, when no water loss<sup>67</sup> or use other than transpiration occurs, actual crop water use is equal to total crop water supply. In this case, water input efficiency achieves its theoretical maximum value of 100%.

<sup>&</sup>lt;sup>67</sup> Therefore, there is also no water runoff that needs to be recycled.

In reality, water input efficiency is less than 100% due to evaporation from soil and/or evaporation from free water surfaces (such as dams and channels), deep drainage, loss from run-off, and change in soil water. Equation 3.11 describes four non-productive water components of water loss.

Water supplied that is lost from the farming system by these processes does not contribute to yield and can therefore be considered a loss from the system. Where that water is purchased and abstracted from the environment, this represents a financial cost of some significance. Therefore, for a crop business to be environmentally (in terms of water) and economically sustainable, it is essential to improve water management practices to enable water input efficiency value to move closer to its theoretical value.

While soil evaporation is partly driven by non-controlling factors (e.g. climate factors, soil characteristics of the region) and loss from water runoff can be managed through better water recycling management practices, the deep drainage component, which contributes significantly to total water lost from crop production under furrow irrigation systems, can be managed/controlled via better irrigation scheduling management.

### 4.4.7 Sixth and seventh level analysis

The sixth and seventh level look at the effect of economic factors and factors at the crop production level described in the crop water process model in Chapter 3 on higher level measures of the WESM model diagram. This includes:

- a. The effect of lint price, cotton seed price and harvest index on weighted average cotton price
- b. The effect of plant transpiration and transpiration index on biomass yield
- c. The effect of individual operating cost items other than water cost on other operating costs
- d. The effect of weighted average irrigation cost and irrigation amount on total water cost
- e. The effect of crop water supply and crop water loss on water input efficiency
- f. The effect of different types of water sources, including rainfall, irrigation (licensed) and irrigation (traded) on crop water supply.

In the WESM model diagram and Appendix 4.2 (Level 6 and Level 7 of the WESM model), it is shown that most of the measures laid out in Level 6 and 7 of the analysis link to the components (crop water flow components and crop yield) described in the crop process model and other economic factors. As discussed in Chapter 3, these components are driven by crop production activities at the operational level of a crop business and relate to on-farm management decisions.

Furthermore, Appendix A2 (Tables A2 and A3) present a list of WESM input variables – accounting data and crop water process model parameters, respectively. In addition, Appendix A2 (Tables A4 and A5) summarise WESM high-level measures (from Level 1 to Level 5) and WESM low-level measures (Level 6 and Level 7), respectively.

# 4.5 The link between the crop water process model and the WESM model

This section describes the link between the two models, the crop water process model and the WESM model, incorporating on-farm accounting data. This provides an end-to-end view that allows crop managers and researchers to identify and calculate key water and economic sustainability indicators, which when linked together can contribute to more sustainability-related decision making.

The starting point is the crop water process model which demonstrates the phenomenon being studied in this thesis – a cotton cropping system. As described in Chapter 3, the crop water process model presents the supply of water to the cotton crop from a range of water sources and its partitioning into productive (transpiration) and non-productive water losses. The essential feature of crop production is the transaction enacted by the plant where transpiration is exchanged for carbon assimilation, the primary process which leads to the production of biomass consisting of economic yield – lint and cottonseed – and other dry matter components (Pereira, Green & Villa Nova 2006). In addition, there is a range of operational decisions, as well as the variability of the weather experienced by the crop, that drive the outcome of the crop production (Pereira, Green & Villa Nova 2006). Key components of the crop water process model, including water sources (rainfall and irrigation), transpiration (T), non-productive water loss (evaporation, deep drainage and

runoff), and biomass yield (lint yield and cotton seed yield)<sup>68</sup>, are input to the ratio decomposition model (at the lowest level, i.e. level seven). This provides a logical link between aggregate water and economic sustainability indicators and crop production parameters which capture the phenomenon behind crop production and issues of sustainability of water at the operational level of the crop business. In addition, accounting data, including the selling price of lint and cotton seed, unit irrigation cost and other operating costs, are required to allow the calculation of key water sustainability performance indicators presented in this chapter. Figure 4.2 presents the link between the two models.





The WESM model designed in this chapter, which is integrated with the crop water process model developed in Chapter 3 and builds upon the decomposition ratio analysis approach, provides a new theoretical construction of EPMS design. By doing that I answer the first component of the research question considered in this thesis.

### 4.6 Conclusion

In this Chapter, I apply the decomposition ratio analysis technique to the development of the water and economic sustainability performance measurement (WESM model) - a form of

<sup>&</sup>lt;sup>68</sup> While fibre quality plays an important role in determining economic value of a cotton crop business (Long et al. 2010), the simulation model I use (which I will outline in the methods chapters) DSSAT has not yet simulated fibre quality. Therefore, fibre quality is not included in the studied models.
EPMS that can be used in the context of water and economic sustainability for cotton production in crop agriculture.

The starting point is the summary water sustainability measure, profit to water cost ratio, decomposed into seven levels of analysis. The lowest level directly links to the crop water process model and is driven by on-farm crop management decisions. The development of the water sustainability performance model is motivated by two reasons.

First, extant literature suggests that the issue of sustainability is so complex and abstract that sustainability cannot be understood at the superior level (organisational or industrial level) and/or be captured (explicated) in a single definition. Therefore, there is a need of a framework or model for sustainability measurement which captures the link between sustainability- related activities at the operational level and the sustainability- related performance at the organisational level. Accordingly, the WESM model is designed to enable the theoretical link between the crop output and crop water components (low level measures) that were outlined in Chapter 3 and the aggregate measures (high level measures) to be established to provide useful information as to how decisions- related to water sustainability issues at the crop production level affect the overall economic and water sustainability performance of a crop business.

Second, there is a need to establish the link between some of the high level measures to capture both economic return on crop water supply and irrigation cost and irrigation leverage. This will help to overcome the issue of validity of irrigation measures in extant crop research and provide useful information on which to base irrigation- related decisions. This will allow the logical link between the high level and low level measures to be established, building upon existing knowledge of crop water measurement.

The WESM model, which lays out the key water and economic sustainability measures and describes the theoretical and logical link between the measures, will provide the basis for empirical work (crop simulation modelling) in the later chapters of the thesis. It will also enable demonstration of how to provide more useful information to support agricultural managers' decision making in relation to crop water and economic sustainability.

## Chapter 5. Research Method – Crop Production Simulation Modelling

## 5.1 Introduction

This purpose of this chapter is to explain and justify the research method used to address the study's research question. Section 5.2 starts with an explanation as to why a crop production simulation method is chosen. This explanation includes a discussion of the underlying phenomenon and how alternative research methods make answering the research question posed in relation to this phenomenon difficult. Section 5.3 provides an overview of common crop production simulation models and the rationale behind the selection of the particular simulation model, the Decision Support System for Agrotechnology Transfer (DSSAT), used in this thesis. This is then expanded in Section 5.4 where an explanation of how the selected crop DSSAT is linked to the crop water process model, accounting information, and the water and economic sustainability measurement model (WESM) developed in Chapter 4. Section 5.5 summarises the chapter.

## 5.2 Method choice

The core research question addressed in this thesis is: *how can Environmental Performance Measurement Systems (EPMS) be designed and used in an agricultural setting to support managers in water and economic sustainability-related decision making and control?* While the first part of this question is addressed in the design of the WESM model, how this is model is used and how decisions around sustainability are made also need to be investigated using an appropriate method. The choice of method to provide answers to this question is going to be largely influenced by the nature of the phenomenon being studied.

#### 5.2.1 The nature of the phenomenon being studied

The research setting being examined in the thesis is crop agriculture. There are a range of factors in crop production that make addressing the above research question a real challenge. Crop production is defined as a process of growing cultivated plants for food, fibre and feed (Henzell 2007). There are three essential physiological processes involved in crop production; seed germination, seedling establishment and reproductive growth of the plant (Stewart et al. 2010). One of the key features of crop production is that crop output, such as

crop yield and crop quality<sup>69</sup>, is driven by crop genotypes<sup>70</sup> (*G*), environmental conditions (*E*), and a wide range of management practices (*M*), and the interactions between these (Bange & Constable 2006; Hochman, Holzworth & Hunt 2009; Jones et al. 2003; Keating et al. 2003; Milroy, Bange & Hearn 2004; van Ittersum et al. 2012).

When focusing on environmental conditions that impact cotton production, the key factors are climate, water availability<sup>71</sup>, and soil conditions (Jones et al. 2003; Stewart et al. 2010; van Ittersum et al. 2012). These key factors are time and location-specific as they are highly variable across regions and years (Bange & Constable 2006; van Ittersum et al. 2012).

Crop production typically consists of six main production operations, including (i) soil preparation (tillage), (ii) planting, (iii) nutrient and pest management, (iv) irrigation, (v) harvest, and (vi) post-harvest (Chen & Baillie 2009; Khabbaz 2010). Figure 5.1 illustrates the main production processes in an Australian cotton farming system<sup>72</sup>. Each production process involves various management practices and decisions, and the choices between these practices will drive economic and environmental outcomes (Keating et al.(2003)<sup>73</sup>. More specifically, management decisions and practices involved in the irrigation operation are considered as key to determining economic yield and crop quality, and water use efficiency of a crop (Hochman, Holzworth & Hunt 2009; Keating et al. 2003; Stewart et al. 2010; van Ittersum et al. 2012).



# Figure 5.1 Example of six typical cotton production operations in an Australian cotton farming system

<sup>&</sup>lt;sup>69</sup> For a fibre crop, such as cotton, crop quality is characterised by fibre length, strength, micronaire, colour, purity, uniformity, moisture content and chemical content (Abbott et al. 2010; Long et al. 2010). For grain crop, crop quality is characterized by protein content, grain size (Roer et al. 2012; Zhao et al. 2010).

 $<sup>^{70}</sup>$  Genotype is the combination of genes that particular living things carry.

<sup>&</sup>lt;sup>71</sup> Including rainfall amount and irrigation accessibility

<sup>&</sup>lt;sup>72</sup> See Appendix B1 for detailed description of the six cotton production operations.

<sup>&</sup>lt;sup>73</sup> For example, total sales revenue earned from a crop, total water use for a crop, and crop yield produced per unit of water (Cammarano et al. 2012; Naramngam & Tong 2013).

The relationships between  $G \ge E \ge M$  interactions and crop output are non-linear relationships. There are two reasons for this. First, many of the physiological processes are non-linear, in the sense that they often exhibit a threshold response where plant performance is unaffected over some range of stimulus and then declines as the range of stimulus extends (Avissar et al. 1985; Forbes & Watson 1992). Crop response to soil water deficit is an example of this. Second, plants exhibit a high degree of homeostasis even in an environment rich in both abiotic and biotic stimuli. Part of the reason for this is the high degree of interdependency among growth processes. As a result, crop production is, by nature, a non-linear and complex process for which simple analytical descriptions are only approximations (Avissar et al. 1985; Forbes & Watson 1992).

As discussed in Chapters 3 and 4, information required from a cropping system for quantifying the summary crop water and economic sustainability measures includes crop water supplied (rainfall and irrigation), productive water (plant transpiration), non-productive water components (such as deep drainage, water runoff, soil evaporation), and economic yield of the crop (for example lint yield and cotton seed yield from cotton production). Due to the nature of crop production described above, the value of these input and output data of a crop production process are variable across regions and years, and dependent on  $G \times E \times M$  interactions. Such variability has important consequences. I argue that it represents the risk to which farmers are exposed, particularly for variation in *E*. Agriculture is much more exposed to the effects of environmental variation than other industries. In agriculture, *E* is also one of the major driving variables and farmers have essentially no control over it. Any valid research method that addresses the above question must accommodate this variation.

#### 5.2.2 Research method alternatives

Given the above context, there are three methods that can be used; field experiments, publicly available data, and crop simulation modelling (Keating et al. 2003; van Ittersum et al. 2012). The advantages and disadvantages of these methods are now discussed, followed by a discussion about why crop simulation modelling method is the most appropriate method to be applied in this thesis.

### 5.2.2.1 Field experiments

Field experimentation is generally defined as a scientific method applied by researchers to experimentally examine an intervention in the real world as opposed to in the laboratory. In the context of crop production, a field experiment involves the process of replication of crop production operations including soil preparation, planting, fertiliser treatments, irrigation applications and crop harvesting, conducted in a cropping system (Zhang, Walker & Dawes 2002). In addition, in a field experiment, one or more variables or practices is varied in a systematic way while all other factors are held constant. During a field experiment, a range of crop data are collected, including collection of information on plant growth, soils, water and nitrogen (Cassman et al. 2003; van Ittersum et al. 2012; Zhang, Walker & Dawes 2002). The field experiment is of crucial importance in crop exploratory studies (for example, exploring the effect of a new crop variety on crop performance) as well as crop simulation model validation studies (Jones et al. 2003; Keating et al. 2003; van Ittersum et al. 2012).

However, the crop literature has identified a number of limitations of the field experimental method, especially for topics related to the choices of management practices to obtain optimal yield (Hochman, Holzworth & Hunt 2009; van Ittersum et al. 2012) and the effect of various management choices on crop yield, and the economic and environmental outcomes of the crop business (Cammarano et al. 2012; Naramngam & Tong 2013), making it difficult (and sometimes impossible) to generate required crop data in an accurate and valid manner. First, a field experiment is an expensive, time-consuming and labour intensive technique for examining crop yield under a range of choices of management practices (Braunack, Bange & Johnston 2012; Cammarano et al. 2012). Second, field experiments are conducted at particular points in time and space (Jones et al. 2003; van Ittersum et al. 2012). This means crop data and information derived from field experiments is site and season specific. As a result, examinations of crop performance (both economic and environmental performance) are limited by the small number of field trials in terms of soil type, time and space. Third, this method requires well-managed field studies and replications over several years to obtain a robust estimate of average yield (Cassman et al. 2003). Fourth, it is difficult to know for certain if all biotic and abiotic stresses (such as water deficit, nitrogen deficit) are controlled when determining optimal yield in a specific location (van Ittersum et al. 2012). Finally, it is difficult (if not impossible) to quantify one of the key crop water parameters - plant transpiration – using a field experiment approach (Avissar et al. 1985).

## 5.2.2.2 Historical/ public data collection

Another approach to collecting crop information is from publicly available information. Some economic and environmental information including crop yield, irrigation applications and irrigation cost, crop revenue and operating costs, can be collected from public data sources (e.g. from consulting firms<sup>74</sup> or the crop literature). They include industry reports (Cotton Australia 2008; Samson & Cotton CRC 2009; The Cotton Industry Development and Delivery Team 2012), research papers (Braunack, Bange & Johnston 2012; Cammarano et al. 2012), case studies and conference proceedings (Baillie 2009; Baillie & Chen 2008), and farm data collected from surveys.

While crop data collected from public sources are less expensive, less time-consuming, and less labour intensive compared to field experiment methods, the public data collection approach also has several limitations. First, the data collected are historical data, which means information needed for crop production planning and forecasting is not available. Second, information needed for evaluating the effect of management decision choices on crop economic and environmental performance of a crop business cannot be obtained if such management practices have not been used. For example, studies considering how a crop business can move towards more sustainable decision making from the current state cannot be examined if the proposed more sustainable practices have not been conducted in practice. Third, similar to the field experiment method, information related to plant transpiration, one of the key crop parameters that is used as an input for calculating the crop water use index, is not available from public data sources.

Fourth, while some crop data can be derived from publicly available information, most of them represent averages across regions or for the whole industry (Boyce Chartered Accountants 2012), making it difficult (if not impossible) to examine the relationship between two key crop variables, total water supplied and crop yield. This is because crop production is site-specific, contingent on various environmental factors and  $G \ge X$  M interactions, resulting in different irrigation applied to and different economic yield produced from different cropping systems. In this respect, averages are not suitable.

<sup>&</sup>lt;sup>74</sup> For example, Australian cotton comparative analysis crop reports provided by Boyce Chartered Accountants (Boyce Chartered Accountants 2012).

Finally, most of the publicly available crop data provide aggregate information – at the organisational level. This limits examination of the effect of different operational management decisions on crop performance.

#### 5.2.2.3 Crop simulation modelling

The third approach that can be considered to collect crop information is crop simulation modelling. Crop simulation models are generally defined as "*mathematical representations of our current understanding of biophysical crop processes (phenology, carbon assimilation, assimilate partitioning) and of crop responses to environmental factors*" (van Ittersum *et al.* 2012, p.4).

For cotton production studies, crop simulation models (also called cropping system simulation models) have been used since the 1970s with an attempt to simulate fundamental cropping system processes including soil water redistribution, plant transpiration and soil evaporation, nutrient dynamics, crop growth and development (Thorp et al. 2014). To do so, developers of crop simulation models have synthesised the knowledge gained from crop science, laboratory, field, and controlled-environment experiements accumulated over several decades in order to produce computer algorithms that simulate the real world (Thorp et al. 2014). Crop simulation models also need to be validated by comparing simulated results against measured experimental data with various treatments to ensure they can represent the real world (Jones et al. 2003; Keating et al. 2003; Thorp et al. 2014; Wang et al. 2002). Once model developers are confident that the models can provide scientifically defensible results, particularly sound scientific calculations of key parameters of cropping system processes, and simulate the real world adequately, the models can be applied to integrate knowledge about soil, climate, crops and management at any point in time and space (Jones et al. 2003).

The need to develop process-based simulation models for cotton production stems from limited information available from traditional agronomic experiments and public sources of information about agricultural decision making and control (Jones et al. 2003; Thorp et al. 2014). Many of the issues that are facing the cotton industry and cotton crop businesses include issues related to the use of natural resources (such as land, water, energy), environmental and economic risks, profitabilility, and sustainability of practices, can all be better understood and managed by applying process-based cropping system simulation

models. They are unlikely to be addressed using field experiments or public data colection methods (Thorp et al. 2014). Furthermore, information needs for environmental and economic sustainability- related decision making and control within the cotton industry are increasingly urgent due to increased pressures on land, water, energy and other natural resources. This impacts on both quantity and costs, making crop simulation models important and powerful tools for manging risks and guiding cotton management and research to make better decisions and move towards sustainability (Cammarano et al. 2012; Jackson et al. 2011; Jones et al. 2003; Thorp et al. 2014).

The simulation modelling method is not only being applied in natural science research but also in accounting research. For example, a body of costing systems, product pricing and capacity planning decisions research applies analytical and simulation<sup>75</sup> modelling methods to address questions that relate to efficiency of various decision rules (i.e. cost-based decision rules), performance of various planning rules (i.e. capacity planning rules), and optimisation decisions (Balachandran, Balakrishnan & Sivaramakrishnan 1997; Balakrishnan & Sivaramakrishnan 2001, 2002; Christensen & Demski 1995; Dhavale 2005; Hansen & Banker 2002; Kaplan & Thompson 1971; Labro & Vanhoucke 2007; Noreen & Burgstahler 1997). Furthermore, compared to the analytical method, the simulation method is an appropriate method when dealing with a complex environment as the complexity of the phenomenon or problem makes it difficult to mathematically compare the performance of the rules (Balachandran, Balakrishnan & Sivaramakrishnan 1997).

## 5.2.2.4 Crop simulation method as the most appropriate method

Based on the discussion in Section 5.2.1 regarding the nature of the phenomenon being studied, I argue that crop simulation modelling is the most appropriate method to be applied in this thesis.

Compared to field experiments method, computer simulations can perform thousands of scenarios in a short period (i.e. several hours), making crop simulation modelling a less expensive, less-time consuming and less labour-intensive technique to examine the effect of

<sup>&</sup>lt;sup>75</sup> Balakrishnan and Sivaramakrishnan (2002, p. 24) define that "To simulate is to attempt to duplicate the features, appearance, and characteristics of a real system. Scientists often use physical models to investigate phenomenon (e.g. airplane wind tunnel simulations). In a similar manner, accounting researchers use computational experiments (with mathematical models) to estimate the effects of various alternatives. The idea behind simulation is to model a situation mathematically, use numerical methods to study its properties, and to draw conclusions and make decisions based on the results of the simulation".

various management decision choices on economic and environmental performance of a crop business (Cammarano et al. 2012; Jones et al. 2003; Keating et al. 2003; Thorp et al. 2014; van Ittersum et al. 2012).

In order to study long-term performance of a crop business<sup>76</sup>, or predict the impact of climate change and local weather conditions on crop performance which can be used to support crop managers in making a wide range of decisions (i.e. asset maintenance, risk management, cost-benefit analysis), it is neccesary to collect crop data and information over long periods of weather data, the primary uncontrolled driver of crop output. As discussed earlier, crop production is subject to uncontrollable variation in input variables, particularly the weather. This means the data required by a manager is stochastic, not only in terms of mean performance under any given set of management conditions, but also the in terms of the variability of the result. Therefore, generating these data from public data is simply not possible. Field experiments are also inadaquate since a sufficient number of experiments must be run to test an adequate range of weather data. Crop simulation modelling allows the researchers do analysis using many yeas of weather data. Preliminary simulations in the experimental section of this thesis demonstrate that a 90-year range of weather data was just sufficient to capture rare events on both tails of the weather distribution. In this respect, estimating the possible outcomes over a hundred years of crop production without simulation modelling may prove exhausting and time consuming (if not impossible). Therefore, valid simulation models provide the most accesible, if not the only, way of providing stochastic data for managers.

Apart from some rare examples, e.g. the Rothamsted l Rotation, long-terrm field experiments are not feasible(Zhang, Walker & Dawes 2002). Apart from cost, the climate itself is now changing and genetic change will also occur in respose to biotic pressure. For example, the evolution of a new strain of disease part way through such a series of field experiments will probably mean that the original genoypes cannot be used for the duration of the study. Using crop model simulation, biotic and abiotic stresses can be captured to allow increased precision in quantification of crop yield and resource use efficiency (Cheeroo-Nayamuth et al. 2000; Grassini, Hall & Mercau 2009; Hochman, Holzworth & Hunt 2009; Lisson et al. 2005; van Ittersum et al. 2012).

<sup>&</sup>lt;sup>76</sup> By definition, sustainability relates to the concept of long-term performance

Finally, crop simulation modelling is also considered extremely helpful when field experimental data are not available (Keating et al. 2003). For example, plant transpiration, one of the key parameters for water and economic sustainability analysis, can be estimated accurately via well-calibrated and validated simulation models(Hearn 1994; Wells & Hearn 1992).

In summary, crop simulation modelling is the most appropriate method for this research as it overcomes the serious limitations of other methods<sup>77</sup>. However, one of the key challenges of crop production simulation modelling research is selecting good (valid and reliable) models that enable model users to determine meaningful and reliable estimates of several crop parameters and predict the behaviours of the crop system under given conditions for better decision making and control.

## 5.3 Selection of cotton production simulation model

## 5.3.1 Overview of cotton production simulation models

The development and application of cotton production simulation models initially began in the US in the 1960s (Thorp et al. 2014). Building upon a range of crop functions, associated fundamental equations were developed by scientists in an attempt to capture different aspects of cotton growth and development. The GOSSYM simulation model was first developed and used across the US cotton belt in the 1980s to provide guidance about on-farm cotton management (Thorp et al. 2014). Since then, several cotton production simulation models have been developed, continuously refined and applied to major cotton production regions globally, including GOSSYM, Cotton2K, COTCO2, OZCOT and CSM-CROPGRO-Cotton, respectively (Thorp et al. 2014).

The common characteristics of these five cotton simulation models are they are *mechanistic* (describing plant growth and development processes with a significant level of scientific understanding), *dynamic* (including a time dimension in their equations when quantifying crop variables and parameters, such as plant transpiration and plant growth), and *deterministic* (rather than stochastic with probability distribution based-calculations for crop

<sup>&</sup>lt;sup>77</sup> I recognise that the simulation models are built upon field experiments data and public data. However, the primary method that I use to test the two scenarios in thesis is simulation since it would take a very long time to test my two scenarios using field experiments or case studies.

variables) (Thorp et al. 2014). As process-based models, these models share the same goal of estimating crop output (i.e. crop yield) by simulating plant growth and developmental processes over time (daily or hourly) in response to weather conditions, soil water and other soil characteristics, and crop management actions (Jones et al. 2003; Keating et al. 2003; Thorp et al. 2014). However, simulation approaches for these processes and their associated factors, including phenology, potential carbon assimilation, respiration, partitioning, canopy size, yield components, plant stress factors, and atmospheric and soil processes, vary widely among these models (Jones et al. 2003; Keating et al. 2003; Thorp et al. 2014).

In relation to management impacts on cotton production, the five models mentioned above share several common simulated management practices, including sowing date, cultivar selection, row spacing, plant density, irrigation, fertiliser (except COTCO2), defoliation (except COTCO2), and insect damage. Other management considerations, such as skip rows, crop residue, tillage, growth regulators, disease impact, climate change, cropping sequences, and geospatial analysis, vary among these models (Thorp et al. 2014).

Of the five existing crop production simulation models described above, APSIM-OZCOT model (the OZCOT module embedded in the APSIM system) and CROPGRO-Cotton – DSSAT model (the CROPGRO-Cotton module embedded in the DSSAT system) are widely used in Australia (Keating et al. 2003) and internationally (Jones et al. 2003), respectively. In Section 5.3.2 below, an analysis and evaluation of the two models are performed to provide a theoretical justification as to which cotton production simulation model is the most appropriate model to be used in this thesis.

## 5.3.2 Comparison and evaluation of APSIM-OZCOT model and CROPGRO-Cotton-DSSAT model

## 5.3.2.1 APSIM-OZCOT model

The Agricultural Production Systems Simulator (APSIM) cropping system modelling software was developed by the Agricultural Production Systems Research Unit in Australia to simulate biophysical processes of a crop incorporating the dynamics of plant-soilmanagement interactions (Keating et al. 2003; Wang et al. 2002). The development of APSIM in the 1990s arose from a need for crop modelling tools that "*provided accurate predictions of crop production in relation to climate, genotype, soil and management factors,*  whilst addressing long-term resource management issues in farming systems" (Keating et al. 2003, p.268).

APSIM is described as a modular modelling framework, providing a range of crop modules for the majority of the grain and fibre crops grown in temperate and tropical areas (Keating et al. 2003). In the context of cotton production simulation, the original cotton plant model, which is named OZCOT, was developed by CSIRO Plant Industry (Wells & Hearn 1992) and embedded into the APSIM framework as a particular module for cotton crop simulation in an Australian setting (Keating et al. 2003). The integration of the plant aspect of the OZCOT model in the generic soil model in APSIM (named SOILWAT) makes it available for cotton crop simulation (Keating et al. 2003; Wang et al. 2002). In this respect, the term APSIM-OZCOT model is used in this thesis referring to the APSIM framework with the OZCOT model for cotton crop simulation.

In preliminary use of the APSIM-OZCOT model, estimation of transpiration was closely observed given its importance in the crop water process model described in Chapter 3. Serious underestimates of transpiration were found (see detailed description in Appendix B2), for which there was no theoretical basis, so the decision was made not to use the APSIM-OZCOT model in this thesis.

## 5.3.2.2 CROPGRO-Cotton-DSSAT model

The Decision Support System for Agrotechnology Transfer (DSSAT), a software application program comprising crop simulation models for simulating more than 42 crops<sup>78</sup>, has been well-recognised and widely-used by researchers, crop managers, growers, consultants and policy makers in over 100 countries (Hoogenboom et al. 2015; Jones et al. 2003; Thorp et al. 2014) . DSSAT was originally developed from the International Benchmark Sites Network for Agrotechnological Transfer (IBSNAT) Project funded by the US Agency for International Development in the period of 1982-1993 (Thorp et al. 2014). The DSSAT system has been continuously developed and redesigned through interdisciplinary research collaboration among scientists worldwide to allow simulation of a wide range of crops and various applications to meet increased information needs for agricultural decision making (Hoogenboom et al. 2015; Jones et al. 2003; Thorp et al. 2014).

<sup>&</sup>lt;sup>78</sup> As of Version 4.6 released on September 1, 2015

One of the key features of DSSAT is the design of the cropping system model (DSSAT-CSM) with a modular structure (similar to APSIM). The CSM model allows dynamic simulation of soil, plant growth and development, and yield for individual crops, as well as continuous simulation of crop rotations using different crop modules (Jones et al. 2003; Thorp et al. 2014). Also, similar to the APSIM framework, the CROPGRO-Cotton-DSSAT model has a generic structure and uses a daily time step to simulate key underpinning physiological processes in response to climate, soil conditions, and crop management actions (Thorp et al. 2014).

This model was subjected to the same evaluation as that of the APSIM-OZCOT model. The results of the evaluation are set out in Appendix B3. As the results show that there were no estimates of transpiration in conflict with the underlying science, so the conclusion was made that DSSAT provided a more reliable estimate of this key parameter. Consequently this model was selected for this thesis.

## 5.4 The links between DSSAT, the crop water process model and WESM

In previous chapters (Chapters 3 and 4), the designed WESM integrated with the crop water process model were developed. The discussion of why and how the selected cotton production simulation model – DSSAT model – can be used to simulate the real world (i.e. the crop water process model) to generate information needed for WESM analysis is also provided in the previous sections in this chapter. This section integrates these three models and discusses how they link together.

The scope of the devised WESM covers events from the abstraction of water from the environment to the declaration of a profit from cultivating cotton, with many of the processes between these steps being mediated by biological processes in a managed ecosystem. No single extant simulation model has this scope. Consequently, the WESM model requires that several discrete modules be linked and relevant data passed between them to enable the necessary quantitative links to be established. Figure 5.2 illustrates the links between DSSAT, the crop water process model and WESM.

The DSSAT model plays a role as a cropping system simulator to replicate a real-world cropping system in order to facilitate evaluation of environmental and economic sustainability performance from different crop management practices. As discussed above, crop simulation modelling is a method that mathematically captures biological and physiological processes of crop growth and crop responses to environmental factors and management decisions (Keating et al. 2003; Wang et al. 2002). Therefore, the DSSAT model is applied in this thesis as a mechanism to simulate crop growth and production in response to this range of environmental factors and management decisions. This provides an estimate of crop output (e.g. the key components of the crop water process model) to feed into the WESM model to enable calculations of the aggregate water and economic sustainability performance indicators. The integration of the DSSAT model into the crop water process model and the WESM model enables the process of creating targets, which is the second feature of a cybernetic control.





Greater detail of the links between DSSAT, the crop water process model and WESM are shown in Figure 5.3. A key feature not readily illustrated in each of these figures is the exposure of agricultural process to the weather. Temperature, for example, not only directly influences water flow processes but also influences plant growth which also control key steps in the flow. Fluctuations in temperature occur in short timeframes and repeat sequences are rare. The same considerations apply to other weather parameters. Such variation in key driving variables underpins the high levels of risk and uncertainty associated with agriculture.



#### Figure 5.3 Detailed links between DSSAT, the crop water process model and WESM

DSSAT integrates this variable input data into the parameters that determine, for each season, the volume of water abstracted from the environment and the mass of saleable product generated. Consequently, it occupies a key role in the WESM, for which no real substitutes exist (Figure 5.4).

Figure 5.4 describes the key input data required for and output data resulting from cotton simulation modelling with the DSSAT model. These key crop parameters, which are driven by various management practices (given a specific crop variety, at a particular location and in a particular season), will be passed to the WESM model for environmental and economic sustainability analysis.

Another key feature not evident in Figures 5.2, 5.3 and 5.4 is the alignment of scales between models. The final summation is performed on unit area (hectare) and for a single season, symbolised by year of harvest. Input data are not necessarily provided at these scales (e.g. rainfall in mm, radiation in rad/m<sup>2</sup>, temperature in average per day) and much of the scale alignment on which the WESM depends, is performed within DSSAT.



Figure 5.4 Description of input and output data of DSSAT model

## 5.5 Conclusion

This chapter outlines the research method applied in this thesis. The crop production simulation modelling is argued to be the most appropriate method to address the research question considered in this thesis. More specifically, the DSSAT model is selected as the most appropriate crop simulation model since it is well-recognised and widely-used by researchers worldwide. The model is proven to generate valid and reliable crop information which can facilitate the evaluation of water and economic sustainability performance of a crop business from various crop management decision choices.

The integration of the DSSAT model into the crop water process model and WESM model, incorporated with accounting data, enables the process of creating targets, which is the second feature of a cybernetic control. This will benefit cotton growers in a number of ways; to provide growers the ability to access simulations of different scenarios of cotton production to analyse and improve on-farm decision making and control for environmental and economic crop management, asset maintenance, and cost benefit analysis.

Building on Chapter 5, in Chapter 6, a furrow irrigation system is selected as an irrigation method applied in this thesis to examine the second component of the research question, how the designed EPMS can be *used* in an agricultural setting to support managers in water and economic sustainability-related decision making and control. To do this, the crop production simulation model (DSSAT) will be incorporated into a furrow irrigation simulation model. This will enable capture of a more comprehensive picture of a real-world cropping system, and hence, generation of more accurate and higher quality information under various scenarios of  $G \ge X \ge M$  conditions and over a long period of time to support belter sustainability-related decision making and control.

## **Chapter 6. Experimental Design for Furrow Irrigation Simulation**

## 6.1 Introduction

The second part of the thesis focuses on the second component of the research question, which is how the designed WESM can be *used* in an agricultural setting to support managers in water and economic sustainability-related decision making and control.

Section 6.2 starts with an explanation of furrow irrigation systems and practices. Furrow irrigation systems are currently the most common irrigation systems used in cotton production making it an appropriate empirical example. In Section 6.3 I address some limitations of the DSSAT model in simulating cotton production systems with furrow irrigation management practices. This identifies the need for incorporating a furrow irrigation simulation model – SIRMOD – into the DSSAT model. Consequently, I build a two-phased crop simulation model which enables us to capture of a more comprehensive picture of a real-world cropping system, and provides more accurate and valid crop information.

Sections 6.4 and 6.5 describes the details and rationale for an experimental design of the two cropping system scenarios using a current furrow irrigation practice (Scenario 1) and an improved furrow irrigation practice (Scenario 2), respectively. Results of furrow irrigation and crop simulation modelling are also presented in these sections.

The final section (Section 6.6) summarises the chapter, including how the designed twophased simulation modelling allows the generation of more accurate and reliable crop information. This enables analysis and evaluation of crop sustainability performance from the analysis which will be outlined in Chapters 7 and 8. Overall this provides an empirical example demonstrating how the WESM designed in this thesis can be used to support crop managers in more sustainable decision making and control.

## 6.2 Furrow irrigation systems: research and practice

## 6.2.1 Overview of furrow irrigation systems

It is estimated that around 85% of Australian cotton crops are irrigated (Cammarano et al. 2012; Jackson, Khan & Hafeez 2010). There are three main types of irrigation systems that are being used in cotton cultivation, furrow irrigation, subsurface (drip) irrigation systems, and lateral move (centre pivot) irrigation systems (Chen & Baillie 2009; Jackson, Khan & Hafeez 2010; Khabbaz 2010). Of those three irrigation systems, furrow irrigation systems are currently the most common method for irrigated cotton production (Smith, Raine & Minkevich 2005; Walker 2003). Figure 6.1 provides an example of furrow irrigation systems applied in an Australian cotton farm.



Figure 6.1 An example of furrow irrigation systems in a cotton farm (Source: Khabbaz 2010)

As illustrated in Figure 6.1, a furrow irrigation system works by simply supplying water from a head ditch, which is filled with water from a bore, river or storage with a pump, into the furrows where water is distributed over the field by gravity flow (Walker 2003). More specifically, furrow irrigation systems applied in Australian cotton regions, which are mostly characterised by cracking clay soils, are typically designed with relatively long furrows, low flow rates, and long irrigation application time (or cut-off time) (Smith, Raine & Minkevich 2005). As a gravity-fed irrigation system, one of the key advantages of the furrow irrigation system such as

subsurface irrigation systems<sup>79</sup> or lateral move irrigation systems<sup>80</sup> which require increased head pressure to operate (Khabbaz 2010). Furthermore, capital investment in furrow irrigation systems is currently much lower than the other two irrigation systems.

However, compared to modern irrigation systems like subsurface irrigation and lateral move irrigation systems, the traditional furrow irrigation systems are considered the least efficient with respect to water use (Jackson, Khan & Hafeez 2010). While it was assumed for a long time within the Australian cotton industry that furrow irrigation systems on clay soils only resulted in a small amount of deep drainage, and hence these systems were relatively water use efficient, recent research has proven that these assumptions are invalid (Jackson, Khan & Hafeez 2010; Smith, Raine & Minkevich 2005). One of the major problems of furrow irrigation systems that have been addressed in recent research is that there is substantial water loss from furrow irrigation to deep drainage (i.e. in average 42.5 mm per irrigation event or 2.5 ML per ha), leading to low irrigation application efficiency<sup>81</sup> (as low as 50%) (Smith, Raine & Minkevich 2005). This results in two problems. First, the environmental issues of deep drainage under irrigated crops leading to serious soil degradation and salinity. Second, the economic costs arising from the loss of an economic resource (water). The costs are two fold, (i) the direct cost of water, and (ii) lost opportunity costs of producing more crops (where the amount of water available to the farmer is capped or subject to a step-cost function).

<sup>&</sup>lt;sup>79</sup> See Appendix C.1 for an example of subsurface (drip) irrigation systems.

<sup>&</sup>lt;sup>80</sup> See Appendix C.2 for an example of lateral move and centre pivot irrigation systems.

<sup>&</sup>lt;sup>81</sup> Irrigation application efficiency is defined as "the depth or volume of water added to the root zone store expressed as a ratio of the depth or volume of water applied to the field" (Smith, Raine & Minkevich 2015, p.118).

#### 6.2.2 The effect of furrow irrigation practices

One way to improve water use efficiency in irrigated cotton production is shifting from furrow irrigation systems to more efficient irrigation systems, such as subsurface irrigation or lateral move irrigation systems which are proven to offer significant water savings from irrigation of up to 2.2ML/ha (Pratt Water 2004). Despite significant reduction in water application of modern irrigation systems, they have not yet been used widely in practice but may be feasible if the cost of water becomes higher than the cost of the investment. Another way to reduce the amount of deep drainage and improve water use efficiency in furrow irrigation systems is refining the design of furrow irrigation systems to improve their efficiency and/or changing furrow irrigation practices of cotton businesses.

Smith, Raine & Minkevich (2005) studied the effect of 79 furrow irrigation events on a range of soils under a range of irrigation management conditions (such as flow rate, time to cut-off, furrow length, soil moisture deficit) on crop irrigation performance. The study applied the surface irrigation model SIRMOD for each irigation event. It was empirically shown that given soil conditions (type, hydraulic properties/texture/clay content, moisture deficit) and furrow characteristics (furrow length and slope), flow rates and irrigation time are the two key driving factors that influence the maginitude of deep drainage and irrigation application efficiency (Smith, Raine & Minkevich 2005). While the Smith, Raine & Minkevich (2005) study recognised the role of irrigation management in controlling the depth of deep drainage, and demonstrates the possibility of significant improvements in irrigation performance by the application of more advanced irrigation management practices (through the optimisation of the flow rate and irrigation time for a particular soil type and furrow design), it does not provide an end to end connection between irrigation management practices and water resource utilisation and crop profitability.

## 6.3 Water and crop production simulation modelling design: Twophased crop simulation modelling

#### 6.3.1 Limitations of DSSAT model in simulating furrow irrigated crop production

As discussed in Chapter 5, the DSSAT model can be used to simulate biophysical process of a cotton crop under the dynamics of plant-soil-management. However, there are two limitations of DSSAT in simulating furrow irrigated crop production which prevent the model from generating accurate crop water and yield components for a field.

The first limitation is that the DSSAT model treats the entire furrow as a single point. This means field characteristics, including field length, which are evident as driving factors of irrigation performance, are not captured in the crop simulation. More importantly, while water distributed through the surface of the land (infiltration) varies along the furrow, leading to the potential for variance in yield and fibre quality among different points of the furrow, this is not captured in DSSAT and a single yield value is generated.

The second limitation is that, because DSSAT does not itself calculate infiltration in a rigorous way, the effect of furrow irrigation on rewetting the soil profile and generating runoff is not reliable<sup>82</sup>. For these reasons, when using DSSAT as a stand-alone model to simulate the effect of irrigation strategies and/or management practices on irrigation performance, that DSSAT isn't sensitive enough to recognise differences in some irrigation strategies. Accordingly, DSSAT users will face the validity issue with crop data generation (inaccurate irrigation water outflow components and hence crop yield).

However, a solution is available. DSSAT is a modular system and many of the current modules may have begun as stand-alone models which were gradually incorporated into the DSSAT code. In this thesis, I propose to simulate the furrow irrigation event separately and manually transferring the necessary data to DSSAT as input data to its routines – thereby creating an additional module. This is essentially what existing modules do internally within DSSAT.

There are three stages to this. First, I simulate the irrigation event. Second, I simulate the crop response to the event. Finally, I aggregate results and account, at the field scale, for all water inputs and losses, and crop outputs.

<sup>&</sup>lt;sup>82</sup> More specifically, the DSSAT model only simulates the movement of irrigation water applied to the field under the soil surface, from the first layer of the soil profile to the root zone area, not along the furrow. This means the DSSAT model can only capture the vertical dimension of water movement but not the horizontal dimension. Therefore, given irrigation water inflow, DSSAT cannot simulate how much irrigation water will go through the surface of the soil and how much irrigation water will go out of the system as runoff. Typically, DSSAT users will make an assumption of how much infiltration water and how much runoff there is given the amount of irrigation applied to the field. Furthermore, two different irrigation strategies, having different irrigation input conditions (for example, inflow rate and cut-off time), can lead to the same irrigation in the model. While it is evident that changing inflow rates and cut-off-times affect the magnitude and variation of infiltration along the furrow, the amount of runoff from irrigation, as well as deep drainage (Smith 2005), the DSSAT model itself does not capture this and will generate the same yield figure given a particular irrigation application. Consequently, DSSAT does not recognize the role of irrigation management/strategies in determining or controlling the irrigation performance of a crop.

#### 6.3.2 Simulating a furrow irrigation event

#### 6.3.2.1 Characteristics of furrow irrigation

In furrow irrigation, water is applied to the soil in the furrow at the upslope end. The water begins to infiltrate into the soil. Assuming that the application (or inflow) rate is greater than the infiltration rate of the initial soil surface, there will be excess water which will run, under gravity, along the furrow to initiate infiltration at the adjacent downslope soil surface. This process effectively repeats until the aggregate infiltration rate of the wetted soil equals the application rate. The advancing water front may not have reached very far down the furrow at this point.

Infiltration at any point along the wetted surface is initially rapid for the dry soil but the rate of infiltration declines with time as the soil wets, so that the furrow system, as described above, is never in equilibrium. As infiltration in the first wetted soil slows, a water excess is created in the furrow which causes the water front to advance further down the furrow. Eventually, the advancing water front reaches the furrow end and runoff begins.

For each irrigation event, the relationship between water inflow (irrigation applied) and water outflow (infiltration and runoff from irrigation) can be expressed as follow.

Irrigation applied per event it 
$$\left(\frac{ML \text{ per event}}{ha}\right)$$
  
= Irrigation infiltrated per event it  $\left(\frac{ML \text{ per event}}{ha}\right)$   
+ Runoff from irrigation per event it  $\left(\frac{ML \text{ per event}}{ha}\right)$ 
[6.1]

Furrow irrigation therefore has a number of characteristics which may lead to suboptimal performance. First, wetting is sequential along the furrow with the rate of advance depending, inter alia, on the application rate. Second, soil at the upslope end of the furrow is able to infiltrate for a longer time than soil at the downslope end. Even though the infiltration rate declines with time, the general result is that more water infiltrates into the upslope soil than the downslope soil, resulting in the variable infiltration with furrow distance shown in Figure 6.2.



Figure 6.2 Infiltrated depth profile under furrow irrigation (Source: Walker 2003)

Irrigation management with furrow irrigation is therefore something of a compromise. If water is allowed to run in the furrow long enough (by adjusting irrigation time or cut-off time) that soil at the downslope end is adequately rewet, then excess water will have been applied at the upslope end. This will result in deep drainage from that part of the field. Alternatively, inadequate water supplied to the downslope crop area may result in crop water deficit, and hence poor crop growth there.

## 6.3.2.2 Furrow irrigation simulation modelling

A well-recognised model for simulating furrow irrigation is SIRMOD, which has been used to simulate the hydraulics of surface<sup>83</sup> irrigation systems at the field level (Gillies 2008; Gillies & Smith 2005; Smith, Raine & Minkevich 2005; Walker 2003). One of the key features of SIRMOD is its simulation of the irrigation event along the furrow, estimating depth of water infiltrated at each point of the furrow and aggregate runoff from the furrow end (see Figure 6.2).

Three key inputs of SIRMOD are field topography and geometry (field length and slope), inflow controls (inflow, time of cut-off) and infiltration characteristics (infiltration

<sup>&</sup>lt;sup>83</sup> The term 'surface irrigation' represents a broad category of irrigation methods, including basin irrigation, border irrigation and furrow irrigation, in which water spreads over the field due to gravity and the slope of the field. As explained earlier, I focus on the furrow irrigation method, which is the most common surface method used in practice.

coefficients which are driven by soil characteristics)(Walker 2003). Based on these input parameters, SIRMOD simulates the hydraulics of the furrow irrigation system and reports water outflow components (infiltration, runoff) and irrigation performance (distribution uniformity, tailwater fraction) (Walker 2003).

SIRMOD can be applied in crop water management research and practice for identifying optimal irrigation strategies that would lead to improvements in irrigation efficiency and reductions in deep drainage losses (Smith, Raine & Minkevich 2005; Walker 2003). The applications include studies of the effect of changes of characteristics of the furrow system (furrow characteristics, soil properties) and the effect of changes of input conditions of an irrigation event (inflow controls and cut-off time) on irrigation performance of the system. Extant research shows evidence that changing these factors can lead to improvements in furrow irrigation performance (Gillies & Smith 2005; Smith, Raine & Minkevich 2005; Walker 2003). While factors that are related to soil type and soil properties, and physical characteristics of the furrow (length, slope, cross-section shape), are not easy to change in the short-term, factors that are related to irrigation management at the operational level, including inflow and irrigation time (or time to cut-off), can be changed in the short-term by crop managers to increase irrigation efficiency and reduce water loss due to deep drainage.

#### 6.3.2.3 Design for simulation of a furrow irrigation event

As discussed above, three key inputs of SIRMOD are field topography and geometry, inflow controls, and infiltration characteristics. In this thesis, only one cropped field with one soil type is examined. Accordingly, input parameters associated with field topography, and geometry and infiltration characteristics, remain constant and are presented in Table 6.1 below. In Sections 6.4 and 6.5, two cropping system scenarios using two different sets of inflow controls, representing two different furrow irrigation management practices/strategies, will be discussed.

SIRMOD was run for a single irrigation event, assuming that all events would be the same, on the same soil type with the same inflow. Data<sup>84</sup> transferred to DSSAT and WESM were as follows. First, for each 10 metre length of furrow, average infiltration was calculated and passed to DSSAT as irrigation water infiltrated at that point. Second, runoff data was passed

<sup>&</sup>lt;sup>84</sup> Water and crop parameters in SIRMOD and DSSAT are presented on a per ha basis.

to the WESM to be included with runoff from rainfall simulated by DSSAT as an input for calculating volume of recycled runoff.

Field Topography and geometry <sup>85</sup>		Infiltration Characteristics (clay vertosol) <sup>86</sup>	
Field length, m	710.00	a	0.238000
Field width, m	800.00	K, m <sup>3</sup> /m/mn <sup>a</sup>	0.018180
Furrow spacing, m	2.00	Fo, m <sup>3</sup> /m/mn	0.000120
Slope of the field	0.08	C, m <sup>3</sup> /m	0.000000

Table 6.1 SIRMOD Input: Field and infiltration characteristics

### 6.3.3 Simulating crop response to furrow irrigation

## 6.3.3.1 Design for simulation of crop response to furrow irrigation

As discussed in Section 6.3.1, SIRMOD can simulate the movement of irrigation water applied to the field (inflow) in both vertical and horizontal dimensions simultaneously. However, for the vertical movement of water, SIRMOD can only simulate the total amount of irrigation water going down through the first layer of the soil (infiltration) but not how much water will go through each layer of the soil profile, how much water will stay in the root zone, or how much water will go beyond the root zone (deep drainage). In contrast, the DSSAT model can simulate, given the total amount of infiltration, how water is distributed through several layers of the soil profile and how water is used by the root zone for carbon assimilation.

Therefore, I integrate the SIRMOD irrigation modelling software with the DSSAT crop production modelling software. This provides a more comprehensive picture of the hydraulics of furrow irrigation systems and overcome a validity issue of the DSSAT model by generating more accurate crop output data (crop yield and crop water components).

Furthermore, as crop water supplied by irrigation, the amount of water infiltrating the soil, varies with distance along the furrow, it follows that crop response should also vary.

<sup>&</sup>lt;sup>85</sup> J. Purcell, Aqua-Tec Consulting (<u>http://www.aquatechconsulting.com.au/jim-purcell.htm</u>), personal communication, 2015.

<sup>&</sup>lt;sup>86</sup> Details of this soil type and its properties will be discussed in Section 6.3.3.2.

Accordingly, the furrow was considered to consist of 71 x 10 metre length or 72 furrow points. For each of these, irrigation infiltration for each furrow point was calculated by SIRMOD. Rainfall was assumed to be constant across each length.

DSSAT was then run, in parallel, 72 times to generate a total yield (lint yield and cotton seed yield), transpiration, soil evaporation and deep drainage profile along the furrow, with 72 data points for each parameter, for each year of weather records. This was considered to be a more valid and accurate simulation of crop response than simply using, in a single annual simulation, the average infiltration amount, as it could respond to nonlinearities in crop response to crop water supply. A simple average would not do this.

In addition, in reality, these furrow points should share the same planting date, irrigation dates and harvest date. In order to capture this realism, there are two steps were considered. First, a particular furrow point<sup>87</sup> was selected to allow DSSAT to simulate planting dates, irrigation dates and harvest dates over 86 years of weather records, based on the management rules described in Table 6.2. Second, DSSAT was run, in parallel, 72 times using the fixed planting dates, irrigation dates and harvest dates determined in step 1.

## 6.3.3.2 DSSAT Setup

This section provides a brief description of DSSAT setup with respect to choices of cotton location, soil type, cultivar and crop management practices (i.e. scheduling planting, irrigation, nitrogen application and harvest). Details of DSSAT setup is shown on Table 6.2.

Cotton simulation were performed using CROPGRO-Coon model in DSSAT 4.6 (Hoogenboom et al. 2015).

<sup>&</sup>lt;sup>87</sup> In this thesis, I selected furrow point 24, which is 240 metres from the top end of the furrow, 1/3 of the furrow length, for doing step 1. Future research could select other furrow points, for example the furrow points in the middle or at 2/3 of the furrow length in the process of finding planting, irrigation and harvest dates and examine a better strategy to gain improved output of the crop as a whole.

## Table 6.2 DSSAT set up

Soil	Soil type	Vertosol
	Surface texture	Clay
	Depth, cm	180
Cultivar	Сгор	Cotton
	Cultivar type	Deltapine 555 B
Planting rule		
Sowing window	Start date	1 <sup>st</sup> October
	End date	30 <sup>th</sup> November
Soil temperature for planting	Maximum soil temperature, <sup>0</sup> C	40
	Minimum soil temperature, <sup>0</sup> C	10
Soil water for planting	Soil depth, cm	20
	Lower soil water percentage, %	40
	Upper soil water percentage, %	100
Planting management	Planting method	Dry seed
	Planting distribution	Rows
	Planting population at seeding	12 plants/m <sup>2</sup>
	Row spacing, cm	100
	Plant depth, cm	5
Fertiliser management	Fertiliser material	Urea
	Fertiliser application method	Bended beneath surface
	Soil depth, cm	60
First fertiliser application	Date	On planting date
	Amount, kg N/ha	250
Second fertiliser application	Date	60 days after planting date
	Amount, kg N/ha	50 kg
Irrigation management	Irrigation option	Fixed amount applied
	Management depth	40 cm
	Threshold, % of max available	60
	Method	Furrow
	End of application	GS009 (50% of bolls have
		attained 0.5 final size)
Harvest management	Harvest option	At maturity

Daily weather data used as an input to the model were recorded at the Myall Vale Weather Station, New South Wales, Australia (30<sup>0</sup>200'S, 149<sup>0</sup>600'E) by the Australian Bureau of Meteorology (www.longpaddock.qld.gov.au/silo/) for the period 1924–2014<sup>88</sup>. Weather data comprised rainfall (mm day<sup>-1</sup>), maximum and minimum air temperature (°C), and solar radiation (MJm<sup>-2</sup>day<sup>-1</sup>).

Clay Vertosol<sup>89</sup> is selected. Details of soil properties are provided in Appendix C – Table C1.

Deltapine 555 B cultivar was selected. This is a US cultivar available in DSSAT, which has similar characteristics of Bollgard II cultivars, the most common cultivar used in the Australian cotton industry (more than 90%) (Braunack, Bange & Johnston 2012).

Cotton was planted 0.05 m deep in 1 m rows with a plant density of 12 plants (Braunack, Bange & Johnston 2012).

Nitrogen was applied two times during a crop season; 250 kg N/ha on the planting date and an additional 50kgN/ha 60 days after planting. The way the nitrogen management was set up in this thesis also ensures that cotton was not nitrogen stressed at any point during the growing season (D. Johnston, CSIRO Agriculture<sup>90</sup>, personal communication, 2015).

The furrow irrigation scheduling was constructed based on the irrigation rule of refilling the soil profile when 40% of the plant-available water (PAW) was depleted. The way the furrow irrigation was set up in the scenarios modelled ensures that cotton was not water stressed at any point during the growing season<sup>91</sup>.

When the threshold of 60% was reached, 72 different irrigation depths (infiltration amount, mm) generated by SIRMOD were applied for each furrow point for each crop season; this

<sup>&</sup>lt;sup>88</sup> 90 cotton crop-year observations were obtained, from1924-1925 cotton crop year to 2013 - 2014 cotton crop year.

<sup>&</sup>lt;sup>89</sup> The soil information was obtained from <u>https://researchdata.ands.org.au/south-eastern-australia-vertosol-partical-size-dirtibution/155154.</u>

 <sup>&</sup>lt;sup>90</sup> CSIRO Agriculture Flagship, Locked Bag 59, Narrabri NSW 2390 michael.bange@csiro.au,
 <u>david.b.johnston@csiro.au</u>, <u>michael.braunack@csiro.au</u>
 <sup>91</sup> In Australia, a cotton growing season is typically from September to June, depending on cotton

<sup>&</sup>lt;sup>91</sup> In Australia, a cotton growing season is typically from September to June, depending on cotton locations(Braunack, Bange & Johnston 2012).

provided 6,192 simulations run under DSSAT ( $86^{92}$  crop years x 72 irrigation furrow points). In the next section (Section 6.3.4), I will discuss how simulated data from DSSAT are aggregated and passed to the WESM model.

The DSSAT model used simulated phenology to estimate harvest date. For each crop season, daily values of each of key crop parameters (i.e. rainfall, transpiration, soil evaporation, deep drainage, runoff from rainfall, total yield) were cumulated from the planting date to the harvest date.

## 6.3.4 Data aggregation

The data aggregation and passing of parameter values to the WESM model followed the paths described in Figure 6.3. A statistical software package (STATA)<sup>93</sup> was used to aggregate and pass data from SIRMOD and DSSAT to WESM. Values averaged over 72 stations were passed to WESM and included economic yield, soil evaporation, transpiration and deep drainage. Irrigation was calculated for WESM as average infiltration plus runoff from irrigation, estimated by SIRMOD. Runoff from rainfall, which is calculated within DSSAT, does not consider slope length and is considered the same across the furrow. Therefore, runoff from rainfall was taken into WESM from the last station in the furrow.

Preliminary simulations showed that the irrigation events did not induce deep drainage themselves. Rain, followed closely by an irrigation event might, however, exhibit more deep drainage as a result of an initially wet soil profile.

<sup>&</sup>lt;sup>92</sup> DSSAT results show that of 90 crop-year runs there are four years (1928, 1940, 1945 and 2002) which had no crop production due to severe drought. <sup>93</sup> StataCorp. 2015. *Stata Statistical Software: Release 14*. College Station, TX: StataCorp LP.



Figure 6.3 The link between key water and crop parameters between the two phased-crop simulation process and WESM

#### 6.3.5 Water balance of a cropping system

As discussed in Section 6.2.3, SIRMOD can simulate the movement of irrigation water applied to the field (inflow) in both the vertical dimension and horizontal dimension simultaneously. However, for the vertical movement of water, SIRMOD can only simulate the total amount of irrigation water going down through the first layer of the soil (infiltration) but not how much water will go through each layer of the soil profile, how much water will stay in the root zone, and how much water will go beyond the root zone (deep drainage). In contrast, the DSSAT model can simulate, given the total amount of infiltration, how water is distributed through several layers of the soil profile and how water is used by the root zone for carbon assimilation. Therefore, in this thesis, I will integrate the SIRMOD irrigation modelling software with the DSSAT crop production modelling software to provide a more comprehensive picture of the hydraulics of furrow irrigation systems. This overcomes the validity issue of the DSSAT model by generating more valid/accurate crop output data (crop yield and crop water components). By doing this, I will provide more valid, useful and reliable information that can support Australian crop managers in better water and economic sustainability decision making.

#### 6.3.5.1 Water balance under the SIRMOD system

The model starts by considering the field and the water crossing that boundary for an irrigation event. When we consider water crossing that boundary for an irrigation event at the field level, for each irrigation event, SIRMOD provides three simulated parameters; irrigation applied to the upslope end of the furrow (water inflow, in ML), irrigation infiltrated (infiltration, in mm) at any point along the furrow, and runoff from irrigation at the downslope end of the furrow (in ML/ha). Based on the mass balance, water inflow to the system should be equal to water outflow of the system.<sup>94</sup>

Irrigation applied per event<sub>it</sub> 
$$\left(\frac{ML \text{ per event}}{ha}\right)$$
  
= Irrigation infiltrated per event<sub>it</sub>  $\left(\frac{ML \text{ per event}}{ha}\right)$  [6.2]  
+ Runoff from irrigation per event<sub>it</sub>  $\left(\frac{ML \text{ per event}}{ha}\right)$ 

 $<sup>^{94}</sup>$  Given the length and the width of the furrow, volume of irrigation applied and runoff from irrigation (ML) can be converted into depth of water (mm). Then all parameters are converted into ML/ha (1 ML/ha = 100 metre).

The inflow and outflow parameters in Equation 6.1 can be aggregated for a crop season. Given a number of irrigation events:

Irrigation applied <sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Irrigation applied per event <sub>it</sub>  $\left(\frac{ML \text{ per event}}{ha}\right)$   
× Number of irrigation events <sub>it</sub>  
= Irrigation infiltrated <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Runoff from irrigation <sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
(6.2)

Whereas total infiltration (irrigation infiltrated) and total runoff from irrigation for a crop season can be calculated by:

Irrigation inflitrated it 
$$\left(\frac{ML}{ha}\right)$$
  
= Irrigation infiltrated per event it  $\left(\frac{ML \text{ per event}}{ha}\right)$  [6.3]  
× Number of irrigation events it

Runoff from irigation it 
$$\left(\frac{ML}{ha}\right)$$
  
= Runoff from irrigation per event it  $\left(\frac{ML \text{ per event}}{ha}\right)$   
× Number of irrigation events it [6.4]

## 6.3.5.2 Water balance under the DSSAT system

For DSSAT, I am replacing the DSSAT infiltration calculations with SIRMOD because, as discussed above, that gives use better realism, particularly down the furrow. For each of 72 furrow stations, infiltration at that point is input from SIRMOD and partitioned into soil evaporation (ML/ha), deep drainage (ML/ha) and transpiration (ML/ha). Infiltration is therefore simply an internal parameter that has bearing outside this data exchange.

Given rainfall supplied to the cropping system, I use the DSSAT runoff calculation to calculate runoff from rainfall, as it is the only one on offer.

Therefore, for all points in the field, rainfall is input by DSSAT and infiltration is input from SIRMOD, and the result in transpiration, soil evaporation, deep drainage, runoff from rainfall<sup>95</sup> and change in soil water is simulated.

Rainfall <sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
 + Irrigation infiltrated <sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
= Transpiration<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Soil evaporation <sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
+ Deep drainage <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Runoff from rainfall <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [6.5]  
+ Change in soil water<sub>it</sub>  $\left(\frac{ML}{ha}\right)$ 

## 6.3.5.3 Water balance for the field

In order to examine the water balance for the field for each year of weather data, there are two steps involved. First, for each year, average transpiration, soil evaporation, deep drainage, runoff from rainfall and change in soil water over the 72 furrow stations are calculated. Second, the water balance under the DSSAT model needs to be integrated into the water inflow (irrigation applied) and the water outflow (runoff from irrigation) components of SIRMOD to establish the overall water balance for the field.

This can be done by simply adding the runoff from irrigation component to both the left and right hand-sides of Equation 6.5, such that:

 $<sup>^{95}</sup>$  Because the runoff calculation does not involve slope length, we use just the runoff from the last station in the furrow.

Rainfall <sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
 + Irirgation infiltrated  $\left(\frac{ML}{ha}\right)$  + + Runoff from irrigation <sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
= Transpiration<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Soil evaporation <sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
+ Deep drainage <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Runoff from rainfall <sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
+ + Runoff from irrigation <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Change in soil water<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [6.6]

Given irrigation applied is the sum of irrigation infiltrated and runoff from irrigation, and total water runoff is a sum of runoff from rainfall and runoff from irrigation:

Rainfall<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
 + Iringation applied  $\left(\frac{ML}{ha}\right)$   
= Transpiration<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Soil evaporation<sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
+ Deep drainage<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Total water runoff<sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
+ Change in soil water<sub>it</sub>  $\left(\frac{ML}{ha}\right)$ 
[6.7]

If we move the boundary out to the farm, as discussed in Chapter 3, a portion of total water runoff needs to be recycled to supplement rainfall and reduce the water impost of cotton cultivation on the environment.

Irrigation applied it 
$$\left(\frac{ML}{ha}\right)$$
  
= Irrigation abstracted it  $\left(\frac{ML}{ha}\right)$  + Recycled runoff it  $\left(\frac{ML}{ha}\right)$  [6.8]
Total water runoff <sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Recycled runoff <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Loss from water runoff <sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [6.9]

Therefore, crop water supply and crop water loss are defined in Equations 6.10 and 6.11

Crop water supply<sub>it</sub> 
$$\left(\frac{ML}{ha}\right)$$
  
= Rainfall<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  + Irrigation abstracted<sub>it</sub>  $\left(\frac{ML}{ha}\right)$   
+ Recycled runoff<sub>it</sub>  $\left(\frac{ML}{ha}\right)$  [6.10]

Crop water 
$$loss_{it} \left(\frac{ML}{ha}\right)$$
  
= Soil evaporation  $_{it} \left(\frac{ML}{ha}\right)$  + Deep drainage  $_{it} \left(\frac{ML}{ha}\right)$   
+ Loss from water runoff  $_{it} \left(\frac{ML}{ha}\right)$  + Change in soil water  $_{it} \left(\frac{ML}{ha}\right)$  [6.11]

## 6.4 Current irrigation practice

This simulation, as described earlier, requires two phases. The key input data for each phase are described below.

For current practice, a water inflow rate of 2.8 L/s was used for each irrigation event. Water was cut-off to the furrow at 745 minutes (see Table 6.3 – Panel A). The data were initially recorded in commercial cotton fields and kindly provided to me for this study (J. Purcell, Aqua-Tec Consulting<sup>96</sup>, personal communication, 2015).

<sup>&</sup>lt;sup>96</sup> http://www.aquatechconsulting.com.au/jim-purcell.htm

#### Table 6.3 Furrow irrigation data – Scenario 1

Panel A: SIRMOD Input – Scenario 1	
Inflow controls	
SIRMOD Input Variables	Value
Furrow Inflow, l/s	2.80
Time of Cut-off, min	745.00

#### Panel B: Simulated results (SIRMOD Output) – Scenario 1

Irrigation performance		
SIRMOD Out Variables	Value	
Advance time, min	572.10	
Distribution Uniformity, %	85.45	
Tailwater Fraction	10.42	
Average irrigation amount per event, ML/ha	0.79	
Runoff from irrigation per event, ML/ha	0.09	

## Panel C: Furrow irrigation input to WESM – Scenario 1

WESM Input Variables	Value	Information source	Note
Irrigation per event $\left(\frac{ML \text{ per event}}{ha}\right)$	0.79	SIRMOD Output	Scenario 1 – Current irrigation management practice
Runoff from irrigation per event $\left(\frac{ML \text{ per event}}{ha}\right)$	0.09	SIRMOD Output	Scenario 1 – Current irrigation management practice
Ratio of recycled runoff	75%	Assumption	Scenario 1 – Current irrigation management practice

Infiltration as a function of furrow distance for current practice is illustrated in Figure 6.4. This represents a distribution uniformity of 85%. The average infiltration amount was 0.79 ML/ha, with runoff to the tailwater drain of 0.09 ML/ha (Table 6.3 – Panel B). From Figure 6.4, an infiltration or irrigation supplied for each of the 71 stations was calculated as input to DSSAT (Table 6.3 – Panel C). Details of infiltration amounts of each of the 71 stations under Scenario 1 are presented in Appendix C -Table C2.

Using DSSAT for simulating cotton production over 86 years, a range of crop production and water components are generated (Table 6.4). WESM analysis for the current irrigation practice will be outlined in Chapter 7.

Variable	Information	Value
	Source	(Mean-S1)
Panel A: Furrow irrigation simulated data		
Irrigation required per event it $\left(\frac{ML \text{ per event}}{ha}\right)$	SIRMOD	0.79 ML/ha
Runoff from irrigation per event it $\left(\frac{ML \text{ per event}}{ha}\right)$	SIRMOD	0.09 ML/ha
Panel B: Crop production simulated data		
Biomass yield <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ha}}\right)$	DSSAT	11,948 kg/ha
Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$	DSSAT	4,507 kg/ha
Panel C: Crop water simulated data		
Rainfall it $\left(\frac{ML}{ha}\right)$	DSSAT	4.57ML/ha
Transpiration <sub>it</sub> $\left(\frac{ML \text{ of transpiration}}{ha}\right)$	DSSAT	4.61 ML/ha
Soil evaporation it $\left(\frac{ML}{ha}\right)$	DSSAT	6.24 ML/ha
Deep drainage <sub>it</sub> $\left(\frac{ML}{ha}\right)$	DSSAT	1.01 ML/ha
Runoff from rainfall it $\left(\frac{ML}{ha}\right)$	DSSAT	0.52 ML/ha
Change in soil water <sub>it</sub> $\left(\frac{ML}{ha}\right)$	DSSAT	-0.06 ML/ha
Number of irrigation events it $\left(\frac{\text{events}}{\text{crop concor}}\right)$	DSSAT	9.8 irrigation
(crop season/		events per
		season

Table 6.4	WESM	Input	Variables –	Scenario	1(Simulated	Data)
					(	



Figure 6.4 Infiltration curve\_ Scenario 1

## 6.5 Improved irrigation practice

The SIRMOD input data provided to me for this study (J. Purcell, Aqua-Tec Consulting, personal communication, 2015) also included an altered inflow management which resulted in a more uniform water distribution. The resulting infiltration along the furrow is shown in Figure 6.5. Details of infiltration amounts of each of the 71 stations under Scenario 2 are presented in Appendix C -Table C2.

The altered management parameters were an inflow of 6.0 L/s and a cut-of time of 310 minutes (Table 6.5 - Panel A). This gave, as output, the average infiltration amount of 0.55 ML/ha, with runoff to the tailwater drain of 0.24 ML/ha (Table 6.5 - Panel C).

The DSSAT simulation and data aggregation was as described earlier (Section 6.3) except that altered infiltration values for each station were calculated from Fig. 6.5. A range of crop production and water components are generated (Table 6.6). In Chapter 8, simulated results generated by DSSAT for Scenario 2 are presented to show input data for WESM analysis and comparison between the two scenarios.

# Table 6.5 Furrow irrigation data – Scenario 2

Panel A: SIRMOD Input – Scenario 2	
Inflow controls	S2
Furrow Inflow, l/s	6.00
Time of Cutoff, min	310.00

Panel B: Simulated results (SIRMOD Output) – Scenario 2				
Key SIRMOD Output Variables	S2			
Irrigation performance				
Advance time, min	225.50			
Distribution Uniformity, %	93.23			
Tailwater Fraction	30.72			
Volume Balance				
Inflow, m <sup>3</sup>	111.60			
Infiltration, m <sup>3</sup>	77.30			
Outflow, m <sup>3</sup>	34.20			

Panel C: Furrow irrigation input to WESM – Scenario 2		
WESM Input Variables	S2	Information source
Irrigation required per event $\left(\frac{ML \text{ per event}}{ha}\right)$	0.55	Calculated based on SIRMOD Output
Runoff from irrigation per event $\left(\frac{ML \text{ per event}}{ha}\right)$	0.24	Calculated based on SIRMOD Output
Ratio of recycled runoff	85%	Assumption
Deep drainage per event $\left(\frac{ML \text{ per event}}{ha}\right)$	<mark>0.004</mark>	DSSAT Simulated Output (Average over 86 years)

Variable	Information	Value
	Source	(Mean-S2)
Panel A: Furrow irrigation simulated data		
Irrigation required per event it $\left(\frac{ML \text{ per event}}{ha}\right)$	SIRMOD	0.55 ML/ha
Runoff from irrigation per event it $\left(\frac{ML \text{ per event}}{ha}\right)$	SIRMOD	0.24 ML/ha
Panel B: Crop production simulated data		
Biomass yield <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ha}}\right)$	DSSAT	12,270 kg/ha
Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$	DSSAT	4,514 kg/ha
Panel C: Crop water simulated data		
Rainfall it $\left(\frac{ML}{ha}\right)$	DSSAT	4.57 ML/ha
Transpiration <sub>it</sub> $\left(\frac{ML \text{ of transpiration}}{ha}\right)$	DSSAT	4.69 ML/ha
Soil evaporation it $\left(\frac{ML}{ha}\right)$	DSSAT	6.10 ML/ha
Deep drainage <sub>it</sub> $\left(\frac{ML}{ha}\right)$	DSSAT	0.04 ML/ha
Runoff from rainfall it $\left(\frac{ML}{ha}\right)$	DSSAT	0.53 ML/ha
Change in soil water <sub>it</sub> $\left(\frac{ML}{ha}\right)$	DSSAT	-1.19 ML/ha
Number of irrigation events it ()	DSSAT	10.2 irrigation
"\crop season/		events per
		season

## Table 6.6 WESM Input Variables – Scenario 2 (Simulated Data)



Figure 6.5 Infiltration curve\_Scenario 2

## 6.6 Conclusions

The example used in this chapter to illustrate the utility of the WESM developed and described in Fig. 5.2 is that of a conventionally (furrow) irrigated cotton crop, because this is the most common form of irrigation used in the industry. The chapter describes furrow irrigation practices, sets up DSSAT to simulate this practice accurately, and considers its application to current furrow irrigation practice and to improved irrigation management.

The two-phased crop simulation modelling approach is designed with an attempt to provide a more comprehensive picture of the hydraulics of furrow irrigation systems and to overcome the validity issue of the DSSAT model. By doing this, I will provide more valid, useful and reliable information which enables the second component of the research question posed in this thesis to be addressed.

The data generated from SIRMOD and DSSAT for current and improved practices will be input to the WESM model for analysis in Chapters7 and 8, respectively.

# Chapter 7. The Analysis of Water and Economic Sustainability – An empirical example of WESM Analysis

#### 7.1 Introduction

This chapter does two things. First, it outlines an empirical example of the how the developed WESM model works, demonstrating the theoretical development of the model as contained in earlier chapters of the thesis. The data includes how information is presented at seven levels of the WESM model and how they are linked together in a clear, logical and structured way. The output from this model, as explained earlier, has the potential to enable crop managers to make more informed sustainability- related decision making and control.

Second, this chapter contains an initial set of outputs of performance measures so as to inform target construction, further contributing to the design of the EPMS. As argued in Chapter 1, the construction of targets is a key cybernetic control feature; however, in the initiation of the system it is impossible to specify targets in many instances as the model is designed from the ground up. While some targets are informed by scientific theory (such as water input efficiency, water resource efficiency) or estimated by simulation (such as transpiration), and some data inputs are supplied from market sources (such as lint price, and the cost of irrigation water), many of the measures are completely new theoretical developments; as a result, the construction of targets without initialising the model is impossible. This chapter therefore contains a set of output data that creates an initial set of targets to be used for future target setting and demonstrates the use of the model for cybernetic control.

In order to illustrate how the WESM model works, a sample year (Year 2012) is selected as the sample. Section 7.2 contains crop water and production data that result from furrow irrigation simulation modelling (using SIRMOD software) and crop production simulation modelling (using DSSAT software) for the year 2012. In addition, economic data that is obtained from a sample budget for furrow irrigated cotton in 2014–2015 provided by the NSW Department of Primary Industries are used for this example. These data will be fed into the WESM model for quantifying all the measures presented in the WESM model. Furthermore, a summary of calculated low-level measures (Level 7 and Level 6) are also

presented to support the analysis of the five main levels of the WESM model (Level 1 to Level 5) in Section 7.3.

In Section 7.3, the analysis of the summary water and economic sustainability – the profit to water cost ratio – over the five levels (Level 1 to Level 5) is presented. Using the empirical example selected, this section illustrates how changes in a higher level measure are driven by changes in the lower level.

Section 7.4 concludes the chapter and contains the implications of WESM model analysis for better management decisions in relation to water and economic sustainability performance.

## 7.2 Selection of the empirical example

The current furrow irrigation practice that is described in Chapter 6 (see Table 6.3) is applied in this chapter for generating irrigation and crop data for the WESM model analysis. Based on WESM input data (mean value) described in Table 6.4, all WESM measures across seven levels of WESM are calculated. The mean of the summary water and economic sustainability measure – profit to water cost ratio – is *2.62*.

In order to illustrate how the WESM model works, a sample year is selected to create the sample. The year 2012 is selected for two reasons. First, its profit to water cost ratio value (2.57) is close to the profit to water cost ratio of the Scenario 1 sample. Second, the economic data is obtained from a sample budget for furrow irrigated cotton in 2014 -2015(NSW Department of Primary Industries 2015), therefore a sample year that is as close as possible to 2014–2015 is preferable. Table 7.1 reports the SIRMOD and DSSAT output simulated for the sample year 2012.

Table 7.2 summarises the economic data and irrigation management data, accompanied by simulated water and crop data to input to the WESM model, to enable quantification of WESM measures across seven levels. For the sample year 2012, Table 7.3 – Panel F and Panel G, accompanied by figure 7.1, present the calculations of low-level (levels 6 and 7) economic measures and low-level (levels 6 and 7) environmental measures, respectively. The data outlined in Tables 7.1, 7.2 and Panel F and Panel G of Table 7.3 are used for quantifying all high-level WESM measures that are analysed in Section 7.3.

Table 7.1	WESM I	Input Vari	ables – year	2012 (	Simulated	Data)
				-		,

Variable	Information Source	Value
Panel A: Furrow irrigation simulated data		
Irrigation infiltrated per event it $\left(\frac{ML \text{ per event}}{ha}\right)$	SIRMOD	0.79 ML/ha
Runoff from irrigation per event it $\left(\frac{ML \text{ per event}}{ha}\right)$	SIRMOD	0.09 ML/ha
Panel B: Crop production simulated data		
Biomass yield <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ha}}\right)$	DSSAT	12,201 kg/ha
Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$	DSSAT	4,734 kg/ha
Panel C: Crop water simulated data		
Rainfall <sub>it</sub> $\left(\frac{ML}{ha}\right)$	DSSAT	4.17 ML/ha
Transpiration <sub>it</sub> $\left(\frac{ML \text{ of transpiration}}{ha}\right)$	DSSAT	4.82 ML/ha
Soil evaporation it $\left(\frac{ML}{ha}\right)$	DSSAT	5.83 ML/ha
Deep drainage <sub>it</sub> $\left(\frac{ML}{ha}\right)$	DSSAT	1.87 ML/ha
Runoff from rainfall it $\left(\frac{ML}{ha}\right)$	DSSAT	0.38 ML/ha
Change in soil water <sub>it</sub> $\left(\frac{ML}{ha}\right)$	DSSAT	-0.02 ML/ha
Number of irrigation events $_{it}\left(\frac{events}{crop season}\right)$	DSSAT	11 irrigation events per season

Panel A: Accounting data			
WESM input variables	Value	Information source	Note
Lint price. $\left(\frac{\$}{2}\right)$	\$2.09/kg of lint	NSW Department of Primary Industries	Equivalent to \$474/bale
kg/		(2015)	(1  bale = 227  kg)
			This is the 5-year average cotton lint price
			2010–2015
Seed price. $\left(\frac{\$}{1}\right)$	\$0.32/kg of seed	NSW Department of Primary Industries	Equivalent to \$320/t
seeu price <sub>it</sub> (kg)		(2015)	
Fixed operating cost. $\left(\frac{\$}{-}\right)$	\$2,812/ha	NSW Department of Primary Industries	Only considers fixed operating cost on a per
ha)	(ha)	(2015)	ha basis. Overhead costs and permanent
			labour are not considered in calculation of
			operating profit per ha.
			Fixed operating cost is determined using the
			regression technique97 based on the sample
			budget for furrow irrigated cotton in 2014-
			2015 for Central and Northern NSW area.
Unit other variable $\operatorname{cost}_{it}\left(\frac{\$}{\operatorname{kg of lint}}\right)$	\$0.12/kg of lint	NSW Department of Primary Industries	Equivalent to \$27.33 /bale of lint. Exclusive

<sup>&</sup>lt;sup>97</sup> Details of the calculations are presented in the Appendix D1

		(2015)	irrigation cost.
			Unit other variable cost is determined using
			the regression technique98 based on the
			sample budget for furrow irrigated cotton in
			2014–2015 for Central and Northern NSW
			area.
Unit licensed irrigation cost. $\begin{pmatrix} \$ \\ - \end{pmatrix}$	\$60.30/ML	NSW Department of Primary Industries	Including licence fees of \$32.24 /ML and
$\frac{1}{ML}$		(2015)	pumping cost of \$28.06/ML
Unit traded irrigation cost. $\begin{pmatrix} \$ \\ \hline \end{pmatrix}$	Not applicable		Assuming that traded irrigation to irrigation
one traded in igation cost it (ML)			abstracted ratio is equal to zero
Panel B: Other input data			
WESM input variables	Value	Information source	Note
Licensed irrigation to	1000/		
(%)	100%	Assumption	Water management practice: Irrigation
Irrigation infiltrated ratio <sub>it</sub> (%)	100%	Assumption	Water management practice: Irrigation allocation assumption
Irrigation infiltrated ratio <sub>it</sub> (%) Traded irrigation to (%)	0%	Assumption Assumption	Water management practice: Irrigationallocation assumptionWater management practice: Irrigation
Irrigation infiltrated ratio <sub>it</sub> (%) Traded irrigation to Irrigation infiltrated ratio <sub>it</sub> (%)	0%	Assumption Assumption	Water management practice: Irrigationallocation assumptionWater management practice: Irrigationallocation assumption
Irrigation infiltrated ratio <sub>it</sub> <sup>(%)</sup> Traded irrigation to Irrigation infiltrated ratio <sub>it</sub> <sup>(%)</sup> Lint percentage <sub>it</sub> (%)	100% 0% 40%	Assumption Assumption Cotton Seed Distributors (2013)	<ul> <li>Water management practice: Irrigation         <ul> <li>allocation assumption</li> </ul> </li> <li>Water management practice: Irrigation         <ul> <li>allocation assumption</li> </ul> </li> <li>Crop variety assumption: SICOT 43BRF</li> </ul>
Irrigation infiltrated ratio <sub>it</sub> <sup>(%)</sup> Traded irrigation to Irrigation infiltrated ratio <sub>it</sub> <sup>(%)</sup> Lint percentage <sub>it</sub> (%)	100% 0% 40%	Assumption Assumption Cotton Seed Distributors (2013)	Water management practice: Irrigationallocation assumptionWater management practice: Irrigationallocation assumptionCrop variety assumption: SICOT 43BRF

<sup>&</sup>lt;sup>98</sup> Details of the calculations are presented in the Appendix D1.

## Table 7.3 WESM Analysis (year 2012) – Panel G (Level 6 and Level 7: Low-Level Environmental Measures)

Panel G1– Calculation of Irrigation required			
Variable	WESM Level	Formula	
Irrigation infiltrated it $\left(\frac{ML}{ha}\right)$	Level 7	= Irrigation required per event it $\left(\frac{ML \text{ per event}}{ha}\right)$ × Number of irrigation events <sub>it</sub> = 0.79 * 11 = <b>8.69 ML</b> / ha	
Irrigation infiltrated per event it $\left(\frac{ML \text{ per event}}{ha}\right)$	WESM Input	SIRMOD Output (Simulation) 0.79 ML/ ha	
Number of irrigation events	WESM Input	DSSAT Output (Simulation)	
		11 irrigation events	
Panel G2– Calculation of Runoff from irrigation		·	
Runoff from irrigation it $\left(\frac{ML}{ha}\right)$	Level 7	= Runoff from irrigation per event it $\left(\frac{ML \text{ per event}}{ha}\right)$	
		× Number of irrigation events	
		= 0.09 * 11 = <b>0.99 ML</b> / ha	
Runoff from irrigation per event $\frac{(ML \text{ per event})}{(ML \text{ per event})}$	WESM Input	SIRMOD Output (Simulation)	
ha /		0.09 ML/ ha	
Number of irrigation events	WESM Input	DSSAT Output (Simulation)	
		11 irrigation events	

Panel G3– Calculation of Irrigation applied		
Irrigation applied it $\left(\frac{ML}{ha}\right)$	Level 7	= Irrigation infiltrated it $\left(\frac{ML}{ha}\right)$
		+ Runoff from irrigation it $\left(\frac{ML}{ha}\right)$
		= 8.69 + 0.99 = <b>9.68 ML/ ha</b>
Irrigation infiltrated : $\left(\frac{ML}{m}\right)$	Level 7	Calculated in Panel G1
ha /		= 8.69 ML/ ha
Runoff from irrigation : $\binom{ML}{m}$	Level 7	Calculated in Panel G2
ha/		= 0.99  ML/ ha
Panel G4– Calculation of Total water runoff		
Total water runoff $_{it}\left(\frac{ML}{ha}\right)$	Level 7	= Runoff from rainfall it $\left(\frac{ML}{ha}\right)$
		+ Runoff from irrigation it $\left(\frac{ML}{ha}\right)$
		= 0.38 + 0.99 = 1.37 ML/ ha
Runoff from rainfall it $\left(\frac{ML}{ha}\right)$	WESM Input	DSSAT Output (Simulation)
		0.38 ML/ ha
Runoff from irrigation it $\left(\frac{ML}{L_{res}}\right)$	Level 7	Calculated in Panel G3
(na)		= 0.99  ML/ ha

Panel G5– Calculation of Recycled runoff		
Recycled runoff it $\left(\frac{ML}{ha}\right)$	Level 7	= Total water runoff $_{it}\left(\frac{ML}{ha}\right)$
		× Ratio of recycled runoff
		= 1.37 x 75% = <b>1.02 ML/ ha</b>
Total water runoff $\frac{ML}{M}$	Level 7	Calculated in Panel G4
ha/		= 1.37 ML/ ha
Ratio of recycled runoff <sub>it</sub> (%)	WESM Input	Water management practice (assumption – Current
		Farming Practice)
		= 75%
Panel G6– Calculation of Irrigation abstracted		
Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$	Level 7	= Irrigation applied it $\left(\frac{ML}{ha}\right)$
		$-$ Recycled runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$
		= 9.68 - 1.02 = <b>8.66 ML</b> / ha
(ML)	Level 7	Calculated in Panel G3
Irrigation applied it $\left(\frac{1}{ha}\right)$		= 9.68  ML/ ha
Recycled runoff $\left(\frac{ML}{m}\right)$	Level 7	Calculated in Panel G5
ha/		= 1.02  ML/ ha

Panel G7– Calculation of Loss from water runoff			
Loss from water runoff it $\left(\frac{ML}{ha}\right)$	Level 6	= Total water runoff $_{it}\left(\frac{ML}{ha}\right)$	
		- Recycled runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$	
		= 1.37 - 1.02 = <b>0.35 ML/ ha</b>	
Total water runoff $\left(\frac{ML}{m}\right)$	Level 7	Calculated in Panel G4	
ha)		= 1.37 ML/ ha	
Recycled rupoff $\left(\frac{ML}{ML}\right)$	Level 7	Calculated in Panel G5	
ha/		= 1.02 ML/ ha	
Panel G8– Calculation of Crop water loss			
$Crop water loss\left(\frac{ML}{ha}\right)$	Level 6	= Soil evaporation it $\left(\frac{ML}{ha}\right)$ + Deep drainage <sub>it</sub> $\left(\frac{ML}{ha}\right)$	
		+ Loss from water runoff it $\left(\frac{ML}{ha}\right)$	
		+ Change in soil water <sub>it</sub> $\left(\frac{ML}{ha}\right)$	
		= 5.83 + 1.87 + 0.35 - 0.02 = 8.03 ML/ ha	
Soil evaporation $(\frac{ML}{m})$	WESM Input	DSSAT Output (Simulation)	
ha/		5.83 ML/ ha	
Deep drainage: $\left(\frac{ML}{M}\right)$	WESM Input	DSSAT Output (Simulation)	
ha/		1.87 ML/ ha	

(ML)	Larval 7	Calculated in Danal C7
Loss from water runoff it $\left(\frac{ML}{h_{r}}\right)$	Levei /	Calculated in Panel G/
(ha)		= 0.35  ML/ ha
Change in soil water: $\binom{ML}{m}$	WESM Input	DSSAT Output (Simulation)
ha/		= - 0.02 ML/ha
Panel G9– Calculation of Crop water supplied	•	
Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 6	$= \text{Rainfall}_{\text{it}} \left(\frac{\text{ML}}{\text{ha}}\right) + \text{Irrigation abstracted}_{\text{it}} \left(\frac{\text{ML}}{\text{ha}}\right)$
		+ Recycled runoff
		= 4.17 ML/ha + 8.66 ML/ha + 1.02 ML/ha
		= 13.85 ML/ha
$\operatorname{Rainfall}_{\leftarrow} \left( \frac{\operatorname{ML}}{} \right)$	WESM Input	DSSAT Output (Simulation)
hall ha		= 4.17 ML/ha
Irrigation abstracted $\left(\frac{ML}{m}\right)$	Level 7	Calculated in Panel G6
ha/		= 8.66 ML/ha
Recycled runoff $\left(\frac{ML}{m}\right)$	Level 7	Calculated in Panel G5
hetyeleu ruhon it ha /		= 1.02 ML/ha
Panel G10– Calculation of Rainfall to crop water supplied ratio (%)		
Rainfall to crop water supplied ratio (%)	Level 6	$= \text{Rainfall}_{\text{it}} \left(\frac{\text{ML}}{\text{ha}}\right) \div \text{Crop water supplied}_{\text{it}} \left(\frac{\text{ML}}{\text{ha}}\right)$
		= 4.17 ML/ha ÷ 13.85 ML/ha = <b>30.1%</b>
Rainfall <sub>it</sub> $\left(\frac{ML}{ha}\right)$	WESM Input	DSSAT Output (Simulation) = 4.17 ML/ha

(ron water supplied, $\left(\frac{ML}{m}\right)$	Level 6	Calculated in Panel G9
(ha)		= 13.85 ML/ha
Panel G11– Calculation of Recycled runoff to crop water supplie	ed ratio (%)	
<b>Recycled runoff to crop water supplied ratio</b> (%)	Level 6	= Recycled runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$
		÷ Crop water supplied <sub>it</sub> $\left(\frac{ML}{ha}\right)$
		= 1.02 ML/ha ÷ 13.85 ML/ha = <b>7.36%</b>
Becycled runoff $\frac{ML}{m}$	Level 7	Calculated in Panel G5
ha)		= 1.02 ML/ha
(ron water supply ( <sup>ML</sup> )	Level 6	Calculated in Panel G9
$(100 \text{ water supply}_{it} (\frac{1}{ha}))$		= 13.85 ML/ha
Panel G12– Calculation of Crop water loss to crop water supplie	ed ratio (%)	
Crop water loss to crop water supply ratio (%)	Level 6	= Crop water loss $_{it}\left(\frac{ML}{ha}\right)$
		÷ Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$
		= 8.03 ML/ha ÷ 13.85 ML/ha = 58.19%
Crop water loss $_{it}\left(\frac{ML}{L}\right)$	Level 6	Calculated in Panel G8
(ha)		= 8.03 ML/ha
Crop water supply <sub>it</sub> $\left(\frac{ML}{L}\right)$	Level 6	Calculated in Panel G9
ha/		= 13.85 ML/ha



Figure 7.1 Levels 6 to 7 of the WESM model (environmental low-level measures)

## Table 7.3 WESM Analysis (year 2012) – Panel F (Level 6 and Level 7: Low-Level Economic Measures)

Panel F1– Calculation of Lint yield		
Variable	WESM Level	Formula
Lint yield <sub>it</sub> $\left(\frac{kg}{ha}\right)$	Level 7	= Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$ × Lint percentage <sub>it</sub> (%) = 4734 kg/ha x 40% = <b>1894 kg/ha</b> (equivalent to 8.4 bales/ha)
Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$	WESM Input	DSSAT Output (simulation) = 4734 kg/ha
Lint percentage <sub>it</sub> (%)	WESM Input	Crop Variety (assumption: SICOT 43BRF) = 40%
Panel F2– Calculation of Seed yield		
Seed yield <sub>it</sub> $\left(\frac{kg}{ha}\right)$	Level 7	= Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$ × Seed percentage <sub>it</sub> (%) = 4734 kg/ha x 60% = <b>2840 kg/ha</b>
Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$	WESM Input	DSSAT Output (simulation) = 4734 kg/ha
Seed percentage <sub>it</sub> (%)	WESM Input	Accounting Data (assumption) = $100\%$ - Lint percentage <sub>it</sub> (%) = $100\%$ - $40\%$ = $60\%$

Panel F3– Calculation of Sales revenue (lint)	Panel F3– Calculation of Sales revenue (lint)			
Sales revenue <sub>it</sub> (lint) $\left(\frac{\$}{ha}\right)$	Level 6	$= \text{Lint yield}_{\text{it}} \left(\frac{\text{kg}}{\text{ha}}\right) \times \text{Lint price}_{\text{it}} \left(\frac{\$}{\text{kg}}\right)$		
		= 1894 kg/ha x \$2.09/kg = <b>\$3959 / ha</b>		
Lint yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$	Level 7	Calculated in Panel F1 = 1894 kg/ha		
Lint price <sub>it</sub> $\left(\frac{\$}{\text{kg}}\right)$	WESM Input	Accounting Data (Assumption) = \$2.09/kg of lint		
Devel E4. Colordation of Solor means (cotton)				
Panel F4– Calculation of Sales revenue (cotton seed)				
Sales revenue <sub>it</sub> (cotton seed) $\left(\frac{\$}{ha}\right)$	Level 6	$= \text{Seed yield}_{\text{it}} \left(\frac{\text{kg}}{\text{ha}}\right) \times \text{Seed price}_{\text{it}} \left(\frac{\$}{\text{kg}}\right)$		
		= 2840  kg/ha x  0.32/  kg = \$909/ha		
Seed yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$	Level 7	Calculated in Panel F2 = 2840 kg/ha		
Seed price <sub>it</sub> $\left(\frac{\$}{\text{kg}}\right)$	WESM Input	Accounting Data (Assumption) = \$0.32/ kg of seed		

Panel F5– Calculation of Traded irrigation volume	2		
Traded irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	= Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$	
		$\times$ Traded irrigation to Irrigation abstracted ratio <sub>it</sub> (%)	
		= 8.66  ML/ha x  0% = 0  ML/ha	
Irrigation withdrawn $_{it}$ $\left(\frac{ML}{m}\right)$	Level 7	Calculated in Panel G6	
(ha)		= 8.66 ML/ha	
Traded irrigation to Irrigation abstracted ratio <sub>it</sub>	WESM	Water management practice (assumption – Current Farming Practice) = $0\%$	
	Input		
Panel F6– Calculation of Licensed irrigation volum	ne		
Licensed irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right) =$	Level 7	= Irrigation withdrawn $_{it}\left(\frac{ML}{ha}\right)$	
		imes Licensed irrigation to Irrigation withdrawn ratio <sub>it</sub> (%)	
		= 8.66 ML/ha x 100% = <b>8.66 ML/ha</b>	
Irrigation withdrawn $\frac{ML}{M}$	Level 7	Calculated in Panel G6	
ha /		= 8.66 ML/ha	
Licensed irrigation to Irrigation abstracted ratio <sub>it</sub>	WESM	Water management practice (assumption for – Current Farming Practice) = 100%	
	Input		
Panel F7– Calculation of Traded irrigation cost			
Traded irrigation $cost_{it}\left(\frac{\$}{ha}\right)$	Level 6	= Traded irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right) \times$ Unit traded irrigation cost <sub>it</sub> $\left(\frac{\$}{ML}\right)$	
		= \$0/ha	

Traded irrigation volume <sub>it</sub> $\left(\frac{ML}{L}\right)$	Level 7	Calculated in Panel F5
(ha)		= 0 ML/ha
Unit traded irrigation cost. $\left(\frac{\$}{-}\right)$	<b>WESM</b>	Accounting Data (assumption)
(ML)	Input	Not applicable since Traded irrigation to Irrigation withdrawn ratio is assumed
		zero
Panel F8– Calculation of Licensed irrigation cost		
Licensed irrigation $cost_{it}\left(\frac{\$}{ha}\right)$	Level 6	$= \text{Licensed irrigation volume}_{\text{it}} \left(\frac{\text{ML}}{\text{ha}}\right) \times \text{Unit traded irrigation cost}_{\text{it}} \left(\frac{\$}{\text{ML}}\right)$
		= 8.66 ML/ha x \$60.30 = <b>\$522</b> /ha
Licensed irrigation volume: $\binom{ML}{-}$	Level 7	Calculated in Panel F6
hier ha		= 8.66 ML/ha
Unit licensed irrigation cost. (\$	<b>WESM</b>	Accounting Data (assumption)
(ML)	<mark>Input</mark>	= \$60.30/ ML
Panel F9– Calculation of Total water cost		
	1	
Total water $cost_{it}\left(\frac{\$}{ha}\right)$	Level 6	= Licensed irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$ + Traded irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$
		= \$522.2 /ha + \$0 /ha = <b>\$522 /ha</b>
Licensed irrigation cost. $\left(\frac{\$}{1}\right)$	Level 6	Calculated in Panel F8
hierised in rigation cost it (ha)		= \$522 /ha

Traded irrigation cost. $\left(\frac{\$}{-}\right)$	Level 6	Calculated in Panel F7		
ha/		= \$0 /ha		
Panel F10 – Calculation of Other variable cost				
Other variable $cost_{it}\left(\frac{\$}{ha}\right)$	Level 6	= Unit other variable $\operatorname{cost}_{it}\left(\frac{\$}{\operatorname{kg of lint}}\right) \times \operatorname{Lint yield}_{it}\left(\frac{\operatorname{kg}}{\operatorname{ha}}\right)$		
		= \$0.12/kg x 1894 kg/ha = <b>\$227/ha</b>		
Unit other variable $\text{cost}_{\text{it}}\left(\frac{\$}{\text{kg of lint}}\right)$	<b>WESM</b>	Accounting Data (assumption)		
	<mark>Input</mark>	= \$0.12/kg of lint		
Lint yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$	Level 7	Calculated in Panel F1		
		= 1894 kg/ha		



Figure 7.1 Levels 6 to 7 of the WESM model (economic low-level measures)

# 7.3 Analysis of water and economic sustainability

Running the WESM model using data provides an opportunity to discuss the analysis of the drivers of profit to water cost ratio, thereby enabling the analysis of water and economic sustainability performance.

The value of a crop business is generated by economic and environmental factors. In Chapter 4, it is theorised that the profit to water cost ratio is a valuable summary measure of the environmental (with regard to water resource use) and economic sustainability of crop businesses. However, in order to understand how to improve water and economic sustainability performance of a crop business, it is essential to analyse drivers of profit to water cost ratio to gain insights into what drives this summary measure, how it will change as a result of a particular decision, and how the change translates into meaningful implications for management decisions.

In this section, following the five levels of analysis that are laid out in Chapter 4, the analysis of profit to water cost ratio of crop production under current management practice for year 2012 is used to demonstrate how the WESM model works.



Figure 7.1 Levels 1 to 3 of the WESM model



Figure 7.1 Levels 3 to 5 of the WESM model



Figure 7.1 Levels 5 to 6 of the WESM model

#### 7.3.1 First-level of analysis

In Table 7.3, Panel A, it is shown that profit to water cost ratio is 2.57. This means for every dollar spent on water, the crop business creates an operating profit of \$2.57. However, by merely looking at the profit to water cost ratio, the crop manager does not know explicitly how efficiently the business uses the water resource to create its economic value and how water is sourced.

In the first-level of analysis, the profit to water cost ratio of 2.57 is produced by a return of \$97 on one ML of water supplied to the cropping system, 63% of which is sourced by abstracted water which costs \$60.30/ML to obtain. In other words, for every one ML of water supplied to the cropping system, 0.63 ML is sourced from water abstracted from the environment. As it costs \$60.30 to obtain one ML of irrigation water, the business has to incur a cost of \$38 for irrigation water (=  $0.63ML \times 60.30/ML$ ) to generate \$97 of operating profit.

Though the first-breakdown of the profit to water cost ratio explains the effect of return on water, water use leverage and weighted average irrigation cost on the summary figure, it is still not clear what drives return on water, how water management practices affect water use leverage, and how irrigation water is sourced. Therefore, it is necessary to further decompose the three drivers of profit to water ratio into their components.

Table 7.3 WESN	l Analysis (year	: 2012) – Panel A	(Level 1 of WESM)
----------------	------------------	-------------------	-------------------

Analysis of Profit to water cost ratio				
Variable	Level	Formula		
Profit to water cost ratio <sub>it</sub>	Level 1	= Return on water <sub>it</sub> $\left(\frac{\$}{ML}\right)$ $\div \left[$ Water use leverage <sub>it</sub> (%) $\times$ Weighted average irrigation cost <sub>it</sub> $\left(\frac{\$}{ML}\right)$		
		$=$ \$97/ ML $\div$ (62.54% x \$60.30) $=$ <b>2.5</b> 7		
Return on water <sub>it</sub> $\left(\frac{\$}{ML}\right)$	Level 2	Calculated in Panel B1 = \$97 / ML		
Water use leverage <sub>it</sub> (%)	Level 2	Calculated in Panel B2 = 62.54%		
Weighted average irrigation cost it $\left(\frac{\$}{ML}\right)$	Level 2	Calculated in Panel B3 = \$60.30		

#### 7.3.2 Second-level of analysis

#### 7.3.2.1 Decomposition of return on water

In Table 7.3, Panel B1, it is demonstrated that a return of \$97 per ML of water resource is explained by economic water use index of \$361 per ML of water resource and profit margin of 27%. This means for every ML of crop water supply, the business generates \$361 of total sales revenue, 27% of which remains after operating costs have been deducted.

The decomposition of return on water into the economic water use index and profit margin reveals the ability of water resource use in crop production to generate sales and the extent to which economic return can be created from each dollar of sales. However, at this level, the crop manager still does not know how the value of the economic water use index (i.e. \$361 per ML of water resource) will change as a result of crop management decisions. By gaining insights into the factors affecting this measure, it helps the crop manager to identify areas for continuous improvement in economic value created for the business with regard to water resource use.

Furthermore, while the profit margin – a ratio of operating profit to total sales revenue – is a useful measure of assessing whether a business stays in business, this measure per se (i.e. 27% in this example) only reveals, in a relative term, the profitability of each dollar of sales the crop business generates. By decomposing the profit margin ratio into its two terms, crop managers can further examine how changes in one or both terms involved affect its change. Accordingly, in the third-level of analysis, economic water use index and profit margin will be further broken into their components.

#### 7.3.2.2 Decomposition of water use leverage

Table 7.3, Panel B2, explains how crop water supply for crop production is sourced. For every 1 ML of water resource used, 30% of it is rainfall, 7.4% of it comes from recycled irrigation water, and the remaining 62.6% is abstracted as irrigation water.

In respect to water resources for crop irrigation, rainfall is considered an uncontrollable water resource, varying across years. In this example, in 2012, the amount of rainfall provided to the business only contributed to 30% of the total crop water needs to be supplied to its cropping system. One way to reduce dependence on irrigation water and reduce the risk of

crop yield loss due to insufficient water availability is improving the management of water runoff from rainfall and furrow irrigation systems. In other words, the more water runoff being recycled by the farm, the less water needs to be abstracted from the environment for crop production.

In this case 68% of crop water is provided by abstracted irrigation water. The higher the percentage of recycled water runoff, the lower the water use leverage. Under current water management practice, it is assumed that the business recycles only 75% of its total water runoff. If the business improves its water management practice by increasing the ratio of recycled water towards 100%, the water use leverage figure will decrease.

#### 7.3.2.3 Decomposition of weighted irrigation water cost

Weighted irrigation water cost represents the average water cost per ML of irrigation water that a crop business has to pay for its two irrigation water sources; licensed irrigation and traded irrigation. The value of this measure is determined by four factors, including licensed irrigation to irrigation abstracted ratio (%), unit licensed irrigation cost (\$/ML), traded irrigation to irrigation abstracted ratio (%), and unit traded irrigation cost (\$/ML).

With an attempt to provide a simple example to illustrate how the developed WESM works, it is assumed that irrigation water is only sourced by licensed irrigation. This means that weighted average irrigation cost and unit licensed irrigation cost are identical.

Given that unit licensed irrigation cost is assumed \$60.30/ML (see Table 7.2), Table 7.3 (Panel B3) reports weighted average irrigation cost value at \$60.30/ML.

In this respect, the effect of irrigation sources and weighted irrigation water cost on the profit to water cost ratio is isolated in this analysis. Therefore, the two key drivers of profit to water cost ratio considered in this example are return on water and water use leverage, the components of which will be further analysed in the lower levels of analysis.

# Table 7.3 WESM Analysis (year 2012) – Panel B (Level 2 of WESM)

Panel B1-Analysis of Return on water				
Variable	Level	Formula		
<b>Return on water</b> <sub>it</sub> $\left(\frac{\$}{ML}\right)$	Level 2	= Profit margin <sub>it</sub> (%) × Economic water use index <sub>it</sub> $\left(\frac{\$}{ML}\right)$ = 26.85% x \$361 /ML = <b>\$97</b> / <b>ML</b>		
Profit margin <sub>it</sub> (%)	Level 3	Calculated in Panel C1 = 26.85%		
Economic water use $index_{it}\left(\frac{\$}{ML}\right)$	Level 3	Calculated in Panel C2 = \$361 /ML		
Panel B2-Analysis of Water use leverage				
Water use leverage (%)	Level 2	= 100% - [Rainfall to crop water supplied ratio (%) + Recycled runoff to crop water supplied ratio (%) ] $= 100% - (30.1% + 7.36%) = 62.54%$		
Rainfall to crop water supplied ratio (%)	Level 6	Calculated in Panel G10 = 30.1%		
Recycled runoff to crop water supplied ratio (%)	Level 6	Calculated in Panel G11 = 7.36%		
Panel B3- Analysis of Weighted average irrigation cost				
Weighted average irrigation $cost_{it} \left(\frac{\$}{ML}\right)$	Level 2	$= \left[ \text{Licensed irrigation to Irrigation withdrawn ratio}_{it} (\%) \right]$		
--	------------	---		
		× Unit licensed irrigation $cost_{it} \left(\frac{\$}{ML}\right)$		
		+ $\left[ \text{Traded irrigation to Irrigation withdrawn ratio}_{it} (\%) \right]$		
		× Unit traded irrigation $\operatorname{cost}_{it}\left(\frac{\$}{ML}\right)$		
		= 100%  x  \$60.30 = \$60.30		
Licensed irrigation to Irrigation withdrawn rat	WESM Input	Water management practice (assumption for – Current Farming Practice)		
(%)		= 100%		
Traded irrigation to Irrigation withdrawn ratio	WESM Input	Water management practice (assumption for – Current Farming Practice)		
(%)		=0%		
Unit licensed irrigation cost. $\begin{pmatrix} \$ \\ - \end{pmatrix}$	WESM Input	Accounting Data (assumption)		
$\frac{\text{Ontencensed in right on cost}_{it}}{\text{ML}}$		= \$60.30/ ML		
Unit traded irrigation cost. $\left(\frac{\$}{-1}\right)$	WESM Input	Accounting Data (assumption)		
(ML)		Not applicable since Traded irrigation to Irrigation withdrawn ratio is assumed		
		zero		

### 7.3.3 Third-level of analysis

### 7.3.3.1 Decomposition of profit margin

In Table 7.3, Panel C1, a profit margin of 27% is explained by the operating profit of \$1,307 per ha and the total sales revenue of \$4,868 per ha.

The profit margin ratio shows that for every dollar of sales, the business earns 27 cents of operating profit. In addition, the decomposition of profit margin reveals that the economic value the crop business generates for each unit of land is \$1,307/ha. As an absolute measure of profitability, operating profit provides a means of translating changes in profitability into value created for the business. However, in order to understand how this profitability figure can be improved as a result of a particular crop management decision, it is essential to understand what drives its two components: total sales revenue and total operating cost. This will be explained in the fourth level of analysis (Section 7.3.4).

### 7.3.3.2 Decomposition of economic water use index

Table 7.3, Panel C2, explains that an economic water use index of \$361 of sales revenue for one ML of crop water supply is produced by a crop water use index of 140 kg of lint per ML of crop water supply, which is scaled up to total yield by the lint percentage of 40% and then translated into a dollar term by incorporating the weighted average cotton price of \$1.02/kg of total yield.

More specifically, there are two steps involved in the decomposition of the economic water use index. First, a crop water use index of 140 kg of lint per ML of crop water supply is converted to 350 kg of total yield per ML of crop water supply by dividing 140 by the lint percentage of 40% (e.g. 140/0.4 = 350). Second, for each ML of crop water supply, total sales revenue generated is a product of 350 kg of total yield and the weighted average cotton price of \$1.02/kg of total yield, which is \$375 of sales revenue (e.g. \$350 x  $1.02 = $357^{99}$ ).

The decomposition of the economic water use index helps to explain the effect of the crop water use index, lint percentage and weighted average cotton price on it. However, at this level, we do not know what factor(s) determine the level of production of the primary crop

<sup>&</sup>lt;sup>99</sup> Slight difference in Economic water use index value occurs between two calculating approaches (e.g., based on its definition versus its decomposition components) due to rounding.

product (i.e. lint yield) produced by each ML of crop water supply. In addition, by only looking at the average figure of cotton price, we do not know the extent to which the value of each of product components affect the average value of the crop products and/or how to determine the break-even points (in lint yield or lint price) for evaluating business risks.

Given that lint percentage is assumed constant (e.g. 40%, see Table 7.2), in the next level of analysis (Section 7.3.4), the two key drivers of economic water use index – crop water use index and weighted average cotton price – are decomposed into their drivers.

### Table 7.3 WESM Analysis (year 2012) – Panel C (Level 3 of WESM)

Panel C1-Analysis of Profit Margin		
Variable	Level	Formula
Profit margin <sub>it</sub> (%)	Level 3	$= \left[ \mathbf{Opearating  profit_{it}} \left( \frac{\$}{ha} \right) \div \mathbf{Total  sales  revenue_{it}} \left( \frac{\$}{ha} \right) \right] \times 100\%$ $= \$1307 / ha \div \$4868 / ha = 26.85\%$
Operating profit (\$)	Level 4	Calculated in Panel D1
Operating prom <sub>it</sub> $(ha)$		= \$1307 /ha
Total calco revenue (\$)	Level 5	Calculated in Panel E4
$\frac{10 \text{ tal sales revenue}_{\text{it}}}{(ha)}$		= \$4868 /ha
Panel C2-Analysis of Economic water use index		
Variable	Level	Formula
Economic water use $index_{it}\left(\frac{\$}{ML}\right)$	Level 3	= Crop water use $index_{it} \left(\frac{kg \text{ of lint}}{ML}\right) \div Lint percentage_{it}(\%)$ × Weighted average cotton $price_{it} \left(\frac{\$}{kg \text{ of total product}}\right)$ = 140 kg/ML ÷ 40% x \$1.03 /kg = <b>\$361</b> / <b>ML</b>
Crop water use $index_{it} \left( \frac{kg \text{ of } lint}{ML} \right)$	Level 4	Calculated in Panel D2 = 140 kg of lint/ML

Lint percentage <sub>it</sub> (%)	WESM	Crop Variety (assumption: SICOT 43BRF)
	Input	=40%
Weighted average cotton price <sub><i>it</i></sub> $\left(\frac{\$}{kg \text{ of total yield}}\right)$	Level 4	Calculated in Panel D3
(kg of total yield)		= \$1.03 /kg

### 7.3.4 Fourth-level of analysis

### 7.3.4. 1 Decomposition of operating profit

Table 7.3, Panel D1, shows that the business's operating profit \$1,307 per ha resulted from \$4,868/ha of total sales revenue after deducting \$3,561/ha of operating cost. As total sales revenue and operating cost are the two drivers of operating profit and profit margin, the break-down of the operating profit figure is useful for looking at potential increases in economic return for the crop business by increasing sales revenue and/or cutting operating cost.

However, by only viewing total sales revenue and operating cost figures, we do not know how total sales revenue and/or operating cost change as a result of management decision(s) and how the change(s) add value to the business. Therefore, in the fifth-level of analysis (Section 7.3.5), information of sales revenue and operating cost is drilled down into more detailed information to allow us to gain more insights into what drives the two quantitative terms that determine profitability of the crop business.

### 7.3.4.2 Decomposition of crop water use index

Table 7.3, Panel D2, explains that a crop water use index of 140 kg of lint per ML of crop water supply is a product of a transpiration index of 2,531 kg of biomass/ML of transpiration, a harvest index of 0.16 kg of lint /kg of biomass, and a water input efficiency of 35%.

More specifically, there are two steps involved in the decomposition of the crop water use index. First, by multiplying 2,531 kg of biomass/ML of transpiration (transpiration index value) and 0.16 kg of lint /kg of biomass (harvest index value), the relationship between the primary crop product produced and the amount of water that contributed to crop production is quantified, which is 405 kg of lint produced per ML of transpiration (2,531 x 0.16 = 405). Second, incorporation of water input efficiency into the ratio of lint to transpiration allows us to quantify the ability of the crop business to generate primary crop product from each unit of water volume supplied to the cropping system. More particularly, a water input efficiency of 35% indicates that for every 1 ML of water resource that is supplied to the cropping system, only 0.37 ML is actually used by plants for producing biomass. This means the amount of lint

product produced from 1 ML of crop water supply is 142 kg of lint (e.g.,  $405 \times 0.35 = 142^{100}$ ).

Of the three components of the crop water use index, the transpiration index and harvest index are driven by nitrogen management and pest and disease management, respectively, while the water input efficiency measure is driven by water management decisions. In order to gain more insights into how well crop managers manage and use water for their crop production, as well as to quantify the impact of water management decisions, particularly irrigation scheduling decisions on water input efficiency, it is assumed that nitrogen is applied sufficiently for crop production and pest and disease are well controlled. This leaves the key component that drives the crop water use index, and hence economic water use index, water input efficiency. In the next section, (Section 7.3.5), the water input efficiency figure is further broken down into its productive and non-productive water components.

### 7.3.4.3 Decomposition of weighted average cotton price

The decomposition of weighted average cotton price into its components in Table 7.3, Panel D3, realises that the average cotton price of \$1.02 per kg of total cotton product is driven by the lint percentage of 40%, the lint price of \$2.09/kg, and cotton seed price of \$0.32/kg.

As presented in Table 7.2, lint price and cotton seed price used for this example are obtained from the sample budget for furrow irrigated cotton in 2014–2015 for the Central and Northern NSW area, reported by NSW Department of Primary Industries (NSW Department of Primary Industries 2015). In practice, lint price is reported in dollar per bale of lint<sup>101</sup>, while cotton seed price is reported in dollar per tonne of cotton seed. Accordingly, \$2.09/kg of lint is equivalent to \$474/bale of lint, and \$0.32/kg of cotton seed is equivalent to \$320/tonne of cotton seed.

As illustrated in Table 7.3, Panel C2, the economic water use index is driven by three components, including crop water use index, weighted average cotton price, and percentage lint. In order to isolate the effect of economic factors (for example, lint price and cotton seed price) and variety factors (for example, lint percentage) from the effect of crop management

<sup>&</sup>lt;sup>100</sup> Slight difference in crop water use index value occurs between two calculating approaches (e.g. based on its definition versus its decomposition components) due to rounding.

 $<sup>^{101}</sup>$  1 bale = 227 kg

decisions (which are captured by the crop water use index) on the economic water use index, in this example, it is assumed that lint price, cotton seed price, and lint percentage remain constant.

### Table 7.3 WESM Analysis (year 2012) – Panel D (Level 4 of WESM)

Panel D1-Analysis of Operating profit		
Variable	Level	Formula
Operating $\operatorname{profit}_{it}\left(\frac{\$}{ha}\right)$	Level 4	= Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$
		- Total operating $cost_{it}\left(\frac{\$}{ha}\right)$
		= \$4868 /ha - \$3561 /ha = <b>\$1307 /ha</b>
Total sales revenue: $\left(\frac{\$}{-}\right)$	Level 5	Calculated in Panel E4
ha/		= \$4868 /ha
Total operating cost $\begin{pmatrix} \$ \\ - \end{pmatrix}$	Level 5	Calculated in Panel E5
(ha)		= \$3561/ ha
Panel D2 - Analysis of Crop water use index		
Crop water use $index_{it} \left( \frac{kg \text{ of } lint}{ML} \right)$	Level 4	$= \text{Transpiration index}_{it} \left( \frac{\text{kg of biomass}}{\text{ML of transpiration}} \right)$
		× Harvest index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{kg of biomass}}\right)$
		× Water input efficiency <sub>it</sub> (%)
		= 2531 x 0.16 x 34.45% = <b>140 kg of lint/ML</b>
Transpiration index. (kg of biomass)	Level 5	Calculated in Panel E2
ML of transpiration/		= 2531 kg/ML

Harvost index. $\left(\frac{\text{kg of lint}}{1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 +$	Level 5	Calculated in Panel E3
harvest muex <sub>it</sub> (kg of biomass)		= 0.16 kg of lint/ kg of biomass
Water input efficiency <sub>it</sub> (%)	Level 5	Calculated in Panel E1
		= 34.45%
Panel D3 - Analysis of Weighted average cotton price		·
Weighted average cotton $\operatorname{price}_{it}\left(\frac{\$}{\operatorname{kg of total yield}}\right)$	Level 4	$= \left[ \text{Lint percentage}_{it} (\%) \times \text{Lint price}_{it} \left( \frac{\$}{\text{kg}} \right) \right]$
		+ [Seed percentage <sub>it</sub> (%)
		× Seed price <sub>it</sub> $\left(\frac{\$}{\text{kg}}\right)$
		= $(40\% \text{ x } \$2.09) + (60\% \text{ x } \$0.32) = \$1.03 / \text{kg}$
Lint percentage <sub>it</sub> (%)	WESM	Crop Variety (assumption: SICOT 43BRF)
	Input	= 40%
Seed percentage <sub>it</sub> (%)	WESM	Accounting Data (assumption)
	Input	= $100\%$ - Lint percentage <sub>it</sub> (%)
		= 100% - 40% = 60%
Lint price $\begin{pmatrix} \$ \\ \end{pmatrix}$	<b>WESM</b>	Accounting Data (Assumption)
Line price it (kg)	Input	= \$2.09/ kg of lint
Seed price. $\left(\frac{\$}{}\right)$	WESM	Accounting Data (Assumption)
kg/	Input	= \$0.32/ kg of seed

### 7.3.5 Fifth-level of analysis

### 7.3.5.1 Decomposition of water input efficiency

The decomposition of water input efficiency presented in Table 7.3, Panel E1, reveals that of 13.85 ML/ha of water resource supplied to the cropping system, 58% of it (equivalent to 8.03 ML/ha) is lost from the cropping system and hence does not contribute to yield production. Given that 7.5% of water loss from runoff is recycled back to the farming system, only 34.5% of water supplied is actually used by the crop for transpiration to generate biomass and ultimately profit for the business.

Furthermore, Table 7.3, Panel G9, reports that of 13.85 ML/ha of crop water supply, 8.66 ML/ha is provided by irrigation water. This illustrates that reducing the amount of water lost will decrease the amount of water that needs to be abstracted from the environment (e.g. irrigation water). Alternatively, the savings can be used to increase crop production elsewhere on the farm assuming available land on site. Either way, minimising water loss provides positive economic and environmental outcomes for the crop business.

Water supplied to the cropping system can be lost via soil evaporation, deep drainage, runoff or change in soil water. In order to quantify the degree to which each of non-productive water components contribute to the total water lost from the cropping system for identifying areas for improving water efficiency, water loss is further decomposed into its components.

Table 7.3, Panel F2, shows that 8.03 ML/ha of water lost includes 5.83 ML/ha of soil evaporation, 1.87 ML/ha of water lost due to deep drainage, 0.35 ML/ha of loss from water runoff and a relatively small decrease in soil water (e.g., 0.02 ML/ha). For a particular soil type, soil evaporation does not vary significantly with regard to water management. Therefore, in order to improve water input efficiency, it is essential to reduce deep drainage and loss from water runoff.

### 7.3.5.2 Decomposition of total operating cost

Table 7.3, Panel E5, reports that the total operating cost of \$3,561/ha consists of \$522/ha of total water cost, \$227/ha of other variable operating costs, and \$2,812/ha of fixed operating costs.

As described in Table 7.3, Panel F5, \$522/ha of total water cost (includes licensed irrigation costs in the example) is a product of irrigation volume of 8.66 ML/ha and the unit cost of licensed irrigation of \$60.30/ML. In addition, \$227/ha of other variable operating costs are obtained by multiplying the lint price of \$0.12/kg by lint yield of 1894 kg/ha.

Given that operating fixed cost, lint price, and unit licensed irrigation costs are assumed constant, total operating cost is driven by the amount of irrigation abstracted and lint produced. Furthermore, since irrigation schedules are designed in DSSAT in a way to ensure sufficient water is supplied to the cropping system, lint yield does not vary significantly with regard to water management. Accordingly, the amount of irrigation abstracted is the key driving factor of operating cost.

### Table 7.3 WESM Analysis (year 2012) – Panel E (Level 5 of WESM)

Panel E1 – Analysis of Water input efficiency		
Variable	WESM Level	Formula
Water input efficiency <sub>it</sub> (%)	Level 5	= 100%
		– [Crop water loss to crop water supplied ratio (%)
		+ Recycled runoff to crop water supplied ratio (%)
		= 100% - (58.19% + 7.36%) = 34.45%
Crop water lost to crop water supplied ratio (%)	Level 6	Calculated in Panel G12
		= 58.19%
Recycled runoff to crop water supplied ratio (%)	Level 6	Calculated in Panel G11
		= 7.36%
Panel E2 – Analysis of Transpiration index	i	·
Transpiration index <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ML of tranpiration}}\right)$	Level 5	$= \text{Biomass yield}_{\text{it}} \left(\frac{\text{kg of biomass}}{\text{ha}}\right)$
		$\div$ Transpiration <sub>it</sub> $\left(\frac{ML \text{ of transpiration}}{ha}\right)$
		= $12201 \text{ kg/ha} \div 4.82 \text{ ML/ha} = 2531 \text{ kg/ML}$
Biomass vield <sub>it</sub> (kg of biomass)	WESM Input	DSSAT Output (simulation)
ha /		= 12,201  kg/ha



Sales revenue. (lint) $\left(\frac{\$}{-}\right)$	Level 6	Calculated in Panel F3
ha)		= \$3959 /ha
Sales revenue. (cotton seed.) $\left(\frac{\$}{-}\right)$	Level 6	Calculated in Panel F4
ha/		= \$909/ha
Panel E5-Analysis of Total operating cost		
Total operating $cost_{it}\left(\frac{\$}{ha}\right)$	Level 5	= Total water $\operatorname{cost}_{it}\left(\frac{\$}{ha}\right)$ + Other variable $\operatorname{cost}_{it}\left(\frac{\$}{ha}\right)$
		+ Fixed operating $cost_{it}\left(\frac{\$}{ha}\right)$
		= \$522 /ha + \$227/ha + \$2812/ha = <b>\$3561</b> / ha
Total water cost. $\left(\frac{\$}{}\right)$	Level 6	Calculated in Panel F9
ha/		= \$522 /ha
Other variable cost $\left(\frac{\$}{-}\right)$	Level 6	Calculated in Panel F10
(ha)		= \$227/ha
Fixed operating cost: $\left(\frac{\$}{-}\right)$	WESM Input	Accounting Data (assumption)
ha/		= \$2812/ha

### 7.3.6 Summary of WESM analysis

The analysis is conducted under the assumptions that economic factors, lint percentage, and irrigation allocation ratios remain unchanged while nitrogen practices and pest and disease are well managed. This allows the crop manager(s) to isolate the effect of water management on the overall water and economic sustainability performance of the crop business. It also enables crop managers to assess whether suboptimal performance of the business's sustainability performance is driven by water management decisions or other factors.

The first-level of analysis shows that the profit to water cost ratio is driven by return on water and water use leverage. In addition, in the second-level of analysis, it is demonstrated that water use leverage is driven by the ratio of recycled runoff. Moreover, the two drivers of return on water, the economic water use index and profit margin, are found to be impacted by crop water use index and total operating cost respectively (the third-level of analysis).

The fourth-level of analysis reveals that the crop water use index is mainly affected by water input efficiency, which is shown to be driven by the amount of deep drainage and loss from water runoff (the fifth and sixth levels of analysis). Furthermore, in the fifth-level of analysis, it is shown that total operating cost is affected by the amount of irrigation abstracted, which is driven by the irrigation infiltrated and ratio of recycled runoff.

The WESM model analysis shows that, in respect to water management, the two key factors that drive the profit to water cost ratio are the amount of water lost through deep drainage and runoff.

### 7.4 Conclusion

Chapter 7 provides an empirical example of the how the developed WESM model works and, based on the two-phased simulation modelling process, how the model generates valid, reliable and useful information that supports crop managers in better sustainability- related decision making and control.

More specifically, this chapter uses the sample year 2012 presented here (under current furrow irrigation practice) to quantify all measures across seven levels of the WESM model and conduct analysis to examine the relationship between high-level and low-level

information. Through the analysis of the summary water and economic sustainability measure – the profit to water cost ratio – over seven levels of analysis for the sample year 2012, water lost from the system due to deep drainage and runoff are identified as the two key factors that drive the profit to water cost ratio. Furthermore, the analysis enables me to generate a first set of performance outputs that can be used for standard and target setting in decision making and control for researchers and farmers.

In the next chapter (Chapter 8), an improved furrow irrigation practice is tested in the same WESM model and the results are compared against that of the current furrow irrigation practice. This identifies both economic and environmental (water) sustainability implications of improving furrow irrigation practices.

### Chapter 8. A step towards sustainability of crop businesses -Improving furrow irrigation management practices

### 8.1 Introduction

This chapter contains two key things. First, following Chapter 7, this chapter examines how the overall environmental and economic sustainability performance of a cotton business can be improved as a result of changing furrow irrigation management practice. To do this, I apply and test theory relating to how irrigation practice can be improved and the kind of impacts this has on environmental sustainability and economic performance.

Second, the output of Scenario 2 could provide the basis for better targets (compared to Scenario 1). By doing this, I further demonstrate the model's usefulness and provide an approach to overcome inherent target setting problems in PMS design.

Section 8.2 presents a comparison of furrow irrigation management practice (i.e. SIRMOD input) and furrow irrigation performance (i.e. SIRMOD output) between the current furrow irrigation practice (Scenario 1) and the improved furrow irrigation practice (Scenario 2).<sup>102</sup> As discussed in Chapter 6, SIRMOD output data are used for crop production simulation modelling (using DSSAT) to generate water and crop production input data for WESM analysis.

Section 8.3 reports the results of univariate tests (using the student *t*-test) on the statistical differences between the Scenario 2 sample (based on the improved furrow irrigation practice) and the Scenario 1 sample (based on the current furrow irrigation practice) across seven levels of WESM, following a bottom-up approach. Furthermore, Section 8.4 contains analyses of environmental and economic differences between the two scenarios. The comparisons between Scenario 1 and Scenario 2 provide insights into how changes in furrow irrigation decisions drive changes in the overall economic and environmental sustainability performance of a crop business, and how the changes can be translated into better

<sup>&</sup>lt;sup>102</sup> While these furrow irrigation parameters were described and reported in Chapter 6 for each of scenarios, in Section 8.2, a comparison between the base line furrow irrigation practice (Scenario 1) and alternative (improved) furrow irrigation practice (Scenario 2) is presented to provide basis for variance analysis between two scenarios in this chapter.

environmental sustainability performance, as well as increased economic value for the crop business.

Finally, Section 8.5 summarises the chapter, including implications for better water management decisions to drive crop businesses to move towards sustainability. I find that moving from the current to the improved furrow irrigation practice improves both environmental and economic performance thereby making the activity more sustainable.

# 8.2 Comparisons of furrow irrigation management practice and furrow irrigation performance between Scenario 1 and Scenario 2

This section provides a comparison between the base line furrow irrigation practice (Scenario 1) and alternative, improved furrow irrigation practice (Scenario 2) to provide basis for variance analysis between two scenarios in later sections in this chapter.

Chapter 7 demonstrated that, given good nitrogen, pest and disease management, and constant economic factors<sup>103</sup>, crop factors<sup>104</sup>, and irrigation allocation ratios<sup>105</sup>; the two key factors that drive the profit to water cost ratio are the loss of water from the system due to deep drainage and water runoff. In addition, in Chapter 4, it is discussed that loss from water runoff can be managed through better water recycling management practices, while loss from deep drainage can be managed improved scheduling and management of irrigation.

As described in Table 8.1, the current furrow irrigation practice (Scenario 1) is characterised by the two irrigation factors; furrow inflow rate of 2.80 l/s and time of cut-off (irrigation time) at 745.00 minutes. As a result of such relatively low furrow inflow and the considerable time irrigating (e.g.173 minutes longer than the advance time<sup>106</sup>), infiltration (irrigation infiltrated) is not uniformly distributed along the furrow (see Figure 8.1). More particularly, there is a significant difference (36 mm) in the amount of irrigation applied between the top (i.e. 91 mm per irrigation) and the bottom (i.e. 55 mm per irrigation) of the furrow with the average of irrigation applied of 79 mm (or 0.79 ML/ha) per irrigation event (see Figure 8.1).

<sup>&</sup>lt;sup>103</sup> Including lint price, cotton seed price, unit irrigation cost, unit other variable cost, operating fixed cost.

<sup>&</sup>lt;sup>104</sup> For example, lint percentage and cotton seed percentage.

<sup>&</sup>lt;sup>105</sup> For example, proportions of licensed irrigation traded irrigation.

<sup>&</sup>lt;sup>106</sup> Advance time is defined as the time required for the irrigation water to reach the end of the furrow (Smith et al. 2005).

Furthermore, it was shown that a considerable deep drainage fraction (ratio of deep drainage to average irrigation applied) of 21% resulted from the current furrow irrigation practice (e.g. 0.17mm/0.79 mm = 21%). This loss of water to deep drainage not only impacts the environment but also represents a significant water and economic loss for the Australian cotton industry.<sup>107</sup>

In order to improve the distribution uniformity of furrow irrigation systems, which will reduce the amount of irrigation infiltrated per event as well as minimise the amount of crop water loss through deep drainage, a more advanced furrow irrigation practice is examined in this chapter. Increasing furrow inflow rates, while reducing time of cut-off below that of current furrow irrigation practices, result in improved irrigation performance<sup>108</sup> (Smith et al. 2005). More specifically, Smith et al. (2005) show that a flow rate of 6 l/s is considered a good irrigation management strategy for improving irrigation performance without inducing erosion. In respect to irrigation time (or cut-off time), it is suggested that stopping irrigation when irrigation water reaches the bottom of the furrow, in other words setting time to cut-off equal to the advance time, leads to significantly improved irrigation performance.

In this thesis, the value of two irrigation factors – the furrow inflow rate and irrigation time – that define the improved irrigation management practice for furrow irrigation on Vertosols were obtained from a real-world example by J. Purcell (Aqua-Tec Consulting<sup>109</sup>, personal communication, 2015). Table 8.1, Panel A, shows that given the same cropping system that is used for Scenario 1 (e.g. same furrow characteristics, soil type and location), the improved irrigation management practice (Scenario 2) is defined by a furrow inflow rate of 6.00 l/s and cut-off time of 310 minutes. While Scenario 2 represents an improved situation compared to Scenario 1 in respect to economic and environmental performance (which will be discussed later in this chapter), it does not necessary reflect optimal water management conditions. In future research, an optimization process can be undertaken in order to determine optimal conditions for maximising the overall water and economic sustainability.

<sup>&</sup>lt;sup>107</sup> Details of net environmental loss and net economic loss of the current irrigation management practice (compared to the improved irrigation management practice) are discussed in Section 8.4.4.

<sup>&</sup>lt;sup>108</sup> Application efficiency and deep drainage losses are used as proxies for irrigation performance (Smith et al. 2005). Accordingly, increased application efficiency and reduced deep drainage losses indicate improvements in irrigation performance.

<sup>&</sup>lt;sup>109</sup> http://www.aquatechconsulting.com.au/jim-purcell.htm

Table 8.1, Panel B, reveals that, by setting a higher flow rate (6 l/s in Scenario 2 as opposed to 2.80 l/s in Scenario 1) while reducing the gap between the cut-off time and the advance time (e.g. from 173 minutes in Scenario 1 to 85 minutes in Scenario 2), the distribution uniformity of irrigation applied is improved to 93%. Furthermore, as illustrated in Figure 8.1, there is a much smaller difference between the infiltration amount per event (less than 10 mm) between the top (e.g. 57 mm per irrigation event) and the bottom (e.g. 48 mm per irrigation event) of the furrow, with an average of infiltration of 55 mm (or 0.55 ML/ha) per irrigation event. Since the amount of water infiltrating along the furrow for Scenario 2 is less variable than that of Scenario 1, it was predicted there would be more uniform total yield produced along the furrow<sup>110</sup> for Scenario 2 than for Scenario 1. Most importantly, water loss from deep drainage was reduced substantially, reducing from 10.1 mm (or 0.101ML/ha) per irrigation event to 0.4 mm (or 0.004 ML/ha) per irrigation event (see Table 8.1, Panel C)

Though moving from the current irrigation management practice to the improved irrigation management practice leads to greater uniformity of irrigation across the furrow, reduced irrigation infiltrated per event, as well decreased loss to deep drainage, water runoff from irrigation increases significantly. Table 8.1, Panel B, shows that there is not much difference in inflow to the cropping system between Scenario 1 (with 125.20 m<sup>3</sup>) and Scenario 2 (with 111.60 m<sup>3</sup>), but the tailwater fraction increases considerably from 10.42% for Scenario 1 to 30.72% for Scenario 2 as a result of higher furrow inflow. Consequently, the amount of water runoff changes from 9 mm (or 0.09 ML/ha) per event for Scenario 1 to 24 mm (or 0.24 ML/ha) per event for Scenario 2 (see Table 8.1, Panel C). However, runoff from irrigated fields must be captured and recycled. Moreover, it is assumed that a better water recycling practice is used in Scenario 2 with 85% recovery compared to 75% in Scenario 1. This also reduces the amount of irrigation water needing to be abstracted from the environment (i.e. by adding water recycled from runoff to the cropping system).

In the next section, simulated results generated by DSSAT for Scenario 1 and Scenario 2 are presented to show the input data for WESM analysis and comparison between the two scenarios.

<sup>&</sup>lt;sup>110</sup> This also results in potential improvement in fibre quality and hence economic value of the crop.

### Table 8.1 Furrow irrigation data: Scenario 1 versus Scenario 2

	Scenario	Scenario 2	Information source
Panel A: SIRMOD Input – Scenario 1 versus Scenario	2	-	1
SIRMOD Input Variables			
Inflow controls	<u> </u>		
Furrow Inflow, l/s	2.80	6.00	
Time of Cut-off, min	745.00	310.00	
Panel B – Simulated results (SIRMOD Output) – Scen	ario 1 versu	s Scenario 2	
Irrigation performance			
Advance time, min	572.10	225.50	
Distribution Uniformity, %	85.45	93.23	
Tailwater Fraction	10.42	30.72	
Volume Balance			
Inflow, m <sup>3</sup>	125.20	111.60	
Infiltration, m <sup>3</sup>	112.40	77.30	
Outflow, m <sup>3</sup>	12.80	34.30	
Panel C –Furrow irrigation input to WESM – Scenari	o 1 versus So	cenario 2	
Irrigation infiltrated per event $\left(\frac{ML \text{ per event}}{ha}\right)$	0.79	0.55	Calculated based on SIRMOD Output
Runoff from irrigation per event $\left(\frac{ML \text{ per event}}{ha}\right)$	0.09	0.24	Calculated based on SIRMOD Output
Ratio of recycled runoff	75%	85%	Assumption
Deep drainage per event $\left(\frac{ML \text{ per event}}{ha}\right)$	0.103	0.004	DSSAT Simulated Output (average over 86 years)



Figure 8.1 Infiltration curve\_ Scenario 2 versus Scenario 1

### 8.3 Statistical test between Scenario 1 and Scenario 2

This section provides results of a statistical test *(t-test)* for the significance of the difference between the means of two samples – Scenario 1 and Scenario 2 as described above.

Each sample includes 90 crop-year observations obtained from DSSAT crop production simulation modelling, from 1924 to 2014. DSSAT results show that for this period there are four years (1928, 1940, 1945 and 2002) which had no crop production due to severe drought. As a result, for each of the two scenarios, the final sample includes 86 crop-years of observations.

Section 8.3.1 reports results of *t-test* for simulated variables of Scenario 1 and Scenario 2 which are used as WESM model input data, followed by results of the *t-test* for low-level environmental variables (Level 7 and Level 6, Section 8.3.2), low-level economic variables (Level 7 and Level 6, Section 8.3.3), middle-level variables (Level 4 and Level 3) (Section 8.3.5), and high-level variables (Level 2 and Level 1) (Section 8.3.6).

The analysis and discussion of the significance of the difference between the means of two samples follows the bottom-up approach to demonstrate how changes in furrow irrigation management decisions at the operational level drive changes in middle-level measures and high-level measures of the WESM model, and eventually the economic and environmental sustainability performance of a crop business.

### 8.3.1 Description of the WESM input variables

In this section, the difference in the means of WESM simulated input variables between Scenario 1 and 2 are presented and discussed (see Table 8.2). More particularly, three sets of simulated data are reported as follows.

First, Table 8.2, Panel A, reveals that by switching from the current furrow irrigation management practice (represented by Scenario 1) to the improved one (Scenario 2), the amount of irrigation infiltrated (infiltration) per irrigation event is reduced by 30% (from 0.79 ML/ha per event to 0.55 ML/ha per event). In contrast, the amount of runoff from irrigation per event increases from 0.09 ML/ha per event to 0.24 ML/ha per event. However, unlike water loss due to deep drainage, a high proportion of water runoff from irrigation can be

recycled to the cropping system, for example through water recycling management, to supplement rainfall and hence reduce the amount of irrigation water needing to be abstracted from the environment. It is assumed that for Scenario 2, better water recycling management allows 85% of runoff from irrigation to be recycled to the cropping system (as opposed to 75% in Scenario 1). In this respect, water loss to irrigation runoff in Scenario 2 is reduced considerably. Details of total water runoff, recycled runoff, and water lost from runoff of the two scenarios are reported in Section 8.3.2.

Second, in respect of economic yield of crop production, there is no statistically significant difference in the means of total yield between Scenario 1 and Scenario 2 (see Table 8.2, Panel B). This suggests that, on average, the crops did not grow any better in Scenario 2 conditions compared to Scenario 1, while it is not unreasonable to have expected the mean yield in Scenario 2 to be higher than the mean yield in Scenario 1. However, for ease of comparison, I chose a common irrigation management which, in preliminary trials, seemed appropriate for both scenarios. The more extensive dataset I now have suggests that Scenario 2 irrigation management might need further refinement to take advantage of the greater uniformity in water application. Since there is not much difference in total yield between Scenario 1 and Scenario 2<sup>111</sup>, it is suggested that changes in economic value are a result of changes in the amount of irrigation abstracted from the environment between the two scenarios. This is explored in Section 8.3.2.

Third, Table 8.2, Panel C, reports a range of crop water components simulated by DSSAT. Since both Scenario 1 and Scenario 2 samples are examined under the same cropping system, same period of time (over the same 86 years), they share the same weather conditions and similar number of irrigation events. Therefore, it is not surprising that there is no statistical difference between parameter values for the two scenarios except for deep drainage and change in soil water status over each season. More specifically, deep drainage decreased by 96% (from 1.01 ML/ha for Scenario 1 to 0.04 ML/ha for Scenario 2) when implementing the improved irrigation management practice. In addition, change in soil water under S2 is -1.19 ML/ha while that for S1 is only -0.06 ML/ha. Accordingly, compared to S1, S2 demonstrates smaller water loss in respect to change in soil water, which means greater use is made of the soil water by the crop. This substantial reduction in deep drainage and change in soil water

<sup>&</sup>lt;sup>111</sup> However, it is still a better management practice than S1 due to a substantial reduction in deep drainage and a greater ratio of recycled runoff, leading to a considerable decrease in total water lost to the cropping system.

helps to explain how the overall economic and environmental sustainability performance of a crop business can change as a result of irrigation management decisions at the operational level, as well as provide economic and environmental benefits, which will be discussed in later subsections of Section 8.3.

Apart from water and crop production input variables, which are obtained from SIRMOD and DSSAT simulation, economic data and crop data are input to WESM to allow quantification of the measures presented in WESM. In order to isolate the effect of water management decisions, it is assumed that economic data and crop data remains constant across the two scenarios. In addition, it is assumed that 100% of irrigation water is sourced from licensed irrigation water in both cases (see Table 7.3 for values of non-simulated WESM input variables).

### Table 8.2 Statistical Test: Univariate Mean Differences in WESM Simulated Input Data between Scenario 1 and Scenario 2

Variable	Information	Mean	Mean	Difference in mean	p-value	
	Source	(Scenario 1)	(Scenario 2)	$\left(\frac{\mathbf{S2}-\mathbf{S1}}{\mathbf{S1}}\right) \times \mathbf{100\%}$		
Panel A: Furrow irrigation simulated data <sup>112</sup>			•			
Irrigation infiltrated per event $\frac{1}{100}$ (ML per event)	SIRMOD	0.79 ML/ha per	0.55 ML/ha per	-30%		
ha /		event	event			
Runoff from irrigation per event $\begin{pmatrix} ML \text{ per event} \end{pmatrix}$	SIRMOD	0.09 ML/ha per	0.24 ML/ha per	166%		
ha )		event	event			
Ratio of recycled runoff (%)	Assumption	75%	85%	13%		
Panel B: Crop production simulated data (reported for a	crop season peri	od)				
Biomass yield <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ha}}\right)$	DSSAT	11,948 kg/ha	12,270 kg/ha	2.7%	0.004	**
Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$	DSSAT	4,507 kg/ha	4,514 kg/ha	0.1%	0.880	
Panel C: Crop water simulated data (reported for a crop s	eason period)	I				
Rainfall it $\left(\frac{ML}{ha}\right)$	DSSAT	4.57 ML/ha	4.57 ML/ha	0%	0.993	
Runoff from rainfall it $\left(\frac{ML}{ha}\right)$	DSSAT	0.52 ML/ha	0.53 ML/ha	0.6%	0.958	
Soil evaporation it $\left(\frac{ML}{ha}\right)$	DSSAT	6.24 ML/ha	6.10 ML/ha	-2.3%	0.254	

<sup>&</sup>lt;sup>112</sup> For each sample, value of each of furrow irrigation simulated variables presented in Panel A is constant across years.

$Transpiration_{it}\left(\frac{ML \text{ of transpiration}}{ha}\right)$	DSSAT	4.61 ML/ha	4.69 ML/ha	1.6%	0.259	
Number of irrigation events $_{it}\left(\frac{events}{crop season}\right)$	DSSAT	9.8	10.2	4%	0.111	
Deep drainage <sub>it</sub> $\left(\frac{ML}{ha}\right)$	DSSAT	1.01 ML/ha	0.04 ML/ha	-95.7%	0.000	***
Change in soil water <sub>it</sub> $\left(\frac{ML}{ha}\right)$	DSSAT	-0.06 ML/ha	-1.19 ML/ha	1,938%	0.000	***

This Table reports univariate tests using the student t-test. Two-tailed test of significance: \*\*\*< 0.001, \*\*< 0.01 and \*< 0.05.

### 8.3.2 Statistical test – Level 6 and Level 7 – Low-level environmental variables

This section reports results of *t-tests* for the low-level (e.g. Level 7 and Level 6) of WESM variables with regards to environmental measurement (see Table 8.3, Panel G).

### 8.3.2.1 Water inflow and outflow components in SIRMOD

Table 8.3, Panel G1 to G3, provide the three water inflow and outflow components given by SIRMOD: irrigation infiltrated, runoff from irrigation, and irrigation applied. First, there is statistically significant difference (*p-value* < 0.001) between the means of irrigation infiltrated <sup>113</sup> for Scenario 1 and Scenario 2. More specifically, there is, on average a substantial reduction (28%) of total irrigation infiltrated over a crop season when shifting from the current irrigation management practice to the improved one. There is also a statistically significant difference (*p-value* < 0.001) in the means of runoff from irrigation between the two scenarios: total runoff from irrigation over a crop season increases by 174% from Scenario 1 to Scenario 2. The significant changes in the amount of total irrigation infiltrated and total runoff from irrigation over a crop season can be explained by the significant decrease in the amount of irrigation infiltrated per event and the significant increase in runoff from irrigation per event, respectively (Section 8.3.1).

Since the irrigation applied to the field is the sum of the infiltration and runoff, it is not surprising that the irrigation applied to Scenario 2 is only 7% less than that applied to Scenario 1, but this difference is statistically different.

### 8.3.2.2 Water runoff, water recycling and irrigation abstracted

Table 8.3, Panel G4 to G7, shows water runoff components of the cropping system and the change in the amount of water abstracted from the environment as a result of recycling management practices.

First, total water runoff from the cropping system is from two sources; runoff from rainfall described in Table 8.2, Panel C, and runoff from irrigation, discussed above. Table 8.3, Panel G4, shows that there is no significant difference in the runoff from rainfall. However, in

<sup>&</sup>lt;sup>113</sup> Note that for irrigation, I am effectively replacing the DSSAT irrigation infiltrated (or infiltration) calculations with SIRMOD's because that provides better realism, particular down the furrow as illustrated in Figure 8.1.

respect to total water runoff over a crop season, the value for Scenario 2 is significantly greater than for Scenario 1 due to the substantial increase in runoff from irrigation of 174%.

Second, as discussed in Section 8.3.1, under the improved water management practice, 85% of water runoff is assumed to be recycled to the cropping system in Scenario 2, while only 75% of water runoff is recycled under the current water management scenario. It is also noted that these ratios of recycled runoff are applied for both runoff from irrigation and runoff from rainfall. Table 8.3, Panel G5, shows that when water recycling management is taken into account, a considerable amount of runoff is recycled to the cropping systems (e.g. 2.55 ML/ha for Scenario 2 compared with 1.07 ML/ha for Scenario 1).

Third, for the entire cropping system, which includes the cropped field and the dam for storing abstracted and recycled water, in an irrigation season, the farmer will have to irrigate the field, which will include both irrigation infiltrated (or infiltration) and runoff from irrigation. However, depending on the efficiency of his operations, and in particular his recycling of runoff, he will able to use a recoverable portion of total water runoff (which is referred to as ratio of recycled runoff) which is returned to the dam. This will reduce the amount of irrigation water that he needs to abstract from the environment. Thus, there is a significant reduction of 27% in irrigation water abstracted from the environment moving from Scenario 1 to Scenario 2 (see Table 8.3, Panel G6).

### 8.3.2.3 Crop water loss and its water components

Table 8.3, Panel G8, shows that there is a statistically significant difference in the amount of crop water lost (*p-value* < 0.001) over a crop season between the Scenario 1 and Scenario 2. Of the four components of crop water loss, including soil evaporation, deep drainage, loss from water runoff<sup>114</sup> and change in soil water, deep drainage and change in soil water are shown to be the key drivers of the reduction in total crop water loss. It is shown that moving from Scenario 1 to Scenario 2 results in a reduction of water lost from the cropping system of 2.15 ML/ha. This implies that the overall water management of S2 has been more effective than for S1. This is important because water lost to deep drainage is truly non-recoverable by the farmer and, in susceptible situations, is the source of irrigation-induced salinization.

<sup>&</sup>lt;sup>114</sup> Loss from water runoff is a portion of total water runoff that is not recycled (calculated in Panel G7).

Detailed analysis and discussion of environmental and economic significance of Scenario 2 will be presented in Section 8.4.

### 8.3.2.4 Crop water supply and its water components

Table 8.3, Panel G9 to Panel G12, compare the means of crop water supply and a range of ratios of crop water components to crop water supply, which provide underpin the discussions of water input efficiency and water use leverage in the later subsections.

About one-third of crop water required (34% for Scenario 1 and 36% for Scenario 2), is provided by rainfall. It is evident that there is a significant difference in the means of recycled runoff to crop water supply ratio (*p-value* < 0.001) between Scenario 1 and Scenario 2 (see Panel G11). More specifically, when moving from Scenario 1 to Scenario 2, there is a significant increase in the means of recycled runoff to crop water supply ratio, from 8% for Scenario 1 to 20% for Scenario 2.

As a result of this, a significant reduction in the proportion of abstracted water in total crop water supply is expected when moving from Scenario 1 to Scenario 2. This relates to the water use leverage that will be discussed in Section 8.3.6.

Finally, Table 8.3, Panel G12, shows the crop water loss to crop water supply ratio. As shown in equation 6.10, crop water supply is equal to crop water loss, transpiration and recycled runoff. Mathematically, the ratios of crop water loss and recycled runoff are the complement of the ratio of transpiration to crop water supply (which is defined as water input efficiency). However, as crop water loss is reported to take up around half of crop water supply in both scenarios (57% for S1 and 43% for S2) (see Table 8.3, Panel G12), the ratio of crop water loss to crop water supply is considered as a key driver of water input efficiency. The lower the ratio of crop water loss to crop water supply, the more efficient the water resource is in producing crop production.

Table 8.3, Panel G12, shows a significant difference in the means of crop water loss to crop water supply ratio (*p*-value < 0.001) between Scenario 1 and Scenario 2.

## Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 – Panel G (Level 6 and Level 7: Low-Level Environmental Measures)

Variable	WESM level	Mean	Mean	Difference in means	<i>p</i> -value	
		(Scenario 1)	(Scenario 2)	$\left(\frac{\mathbf{S2}-\mathbf{S1}}{\mathbf{S1}}\right) \times 100\%$		
				$(S1)^{100}$		
Panel G1: Calculation of irrigation infiltrated						
Irrigation infiltrated it $\left(\frac{ML}{ha}\right)$	Level 7	7.76 ML/ha	5.59 ML/ha	-28%	0.000	***
= Irrigation infiltrated per event it $\left(\frac{ML \text{ per event}}{ha}\right)$						
$\times$ Number of irrigation events <sub>it</sub>						
Irrigation infiltrated per event it $\left(\frac{ML \text{ per event}}{ha}\right)$	WESM Input	0.79 ML/ha	0.55 ML/ha	-30%		
Number of irrigation events	WESM Input	9.8	10.2	4%	0.111	
Panel G2: Calculation of runoff from irrigation						
Runoff from irrigation it $\left(\frac{ML}{ha}\right) =$	Level 7	0.90 ML/ha	2.47 ML/ha	174%	0.000	***
= Runoff from irrigation per event it $\left(\frac{ML \text{ per event}}{ha}\right)$						
× Number of irrigation events						
Runoff from irrigation per event it $\left(\frac{ML \text{ per event}}{ha}\right)$	WESM Input	0.09 ML/ha	0.24 ML/ha	166%		
Number of irrigation events	WESM Input	9.8	10.2	4%	0.111	
					1	

WESM level	Mean	Mean	Difference in means	<i>p</i> -value	
	(Scenario 1)	(Scenario 2)	$\left(\frac{\mathbf{S2}-\mathbf{S1}}{\mathbf{S1}}\right) \times \mathbf{100\%}$		
Level 7	8.66 ML/ha	8.06 ML/ha	-7%	0.000	**
Level 7	7.76 ML/ha	5.59 ML/ha	-28%	0.000	***
Level 7	0.90 ML/ha	2.47 ML/ha	174%	0.000	***
Level 7	1.42 ML/ha	3.00 Ml/ha	110%	0.000	***
WESM Input	0.52 ML/ha	0.53 ML/ha	0.6%	0.958	
Level 7	0.90 ML/ha	2.47 ML/ha	174%	0.000	***
	WESM level         Level 7         Level 7         Level 7         WESM Input         Level 7	WESNTIEVEIMean (Scenario 1)Level 78.66 ML/haLevel 77.76 ML/haLevel 70.90 ML/haUevel 70.90 ML/haUevel 70.52 ML/haUevel 70.90 ML/ha	WESN levelMean (Scenario 1)Mean (Scenario 2)Level 78.66 ML/ha8.06 ML/haLevel 77.76 ML/ha5.59 ML/haLevel 70.90 ML/ha2.47 ML/haLevel 71.42 ML/ha3.00 Ml/haWESM Input0.52 ML/ha0.53 ML/haLevel 70.90 ML/ha2.47 ML/ha	WESM level       Mean (Scenario 1)       Mean (Scenario 2)       Difference in means $\left(\frac{S2 - S1}{S1}\right) \times 100\%$ Level 7       8.66 ML/ha       8.06 ML/ha       -7%         Level 7       7.76 ML/ha       5.59 ML/ha       -28%         Level 7       0.90 ML/ha       2.47 ML/ha       174%         Level 7       1.42 ML/ha       3.00 Ml/ha       110%         WESM Input       0.52 ML/ha       0.53 ML/ha       0.6%         Level 7       0.90 ML/ha       2.47 ML/ha       174%	WESM level         Mean (Scenario 1)         Mean (Scenario 2)         Difference in means $(\frac{S2 - S1}{S1}) \times 100\%$ $p$ -value           Level 7         8.66 ML/ha         8.06 ML/ha         -7%         0.000           Level 7         7.76 ML/ha         5.59 ML/ha         -28%         0.000           Level 7         0.90 ML/ha         2.47 ML/ha         174%         0.000           Level 7         1.42 ML/ha         3.00 Ml/ha         110%         0.000           WESM Input         0.52 ML/ha         0.53 ML/ha         0.6%         0.958           Level 7         0.90 ML/ha         2.47 ML/ha         0.6%         0.958

Variable	WESM level	Mean	Mean	Difference in means	<i>p</i> -value	
		(Scenario 1)	(Scenario 2)	$\left(\frac{\mathbf{S2}-\mathbf{S1}}{\mathbf{S1}}\right) \times \mathbf{100\%}$		
Panel G5: Calculation of recycled runoff		•			•	
Recycled runoff it $\left(\frac{ML}{ha}\right)$	Level 7	1.07 ML/ha	2.55 ML/ha	138%	0.000	***
= Total water runoff $_{it}\left(\frac{ML}{ha}\right)$ × Ratio of recycled runoff						
Total water runoff $_{it}\left(\frac{ML}{ha}\right)$	Level 7	1.43 ML/ha	3.00 Ml/ha	110%	0.000	***
Ratio of recycled runoff <sub>it</sub> (%)	WESM Input	75%	85%	13%		
Panel G6: Calculation of irrigation abstracted						
Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$	Level 7	7.59 ML/ha	5.51 ML/ha	-27%	0.000	***
= Irrigation applied it $\left(\frac{ML}{ha}\right)$ - Recycled runoff it $\left(\frac{ML}{ha}\right)$						
Irrigation applied it $\left(\frac{ML}{ha}\right)$	Level 7	8.66 ML/ha	8.06 ML/ha	-7%	0.000	**
Recycled runoff it $\left(\frac{ML}{ha}\right)$	Level 7	1.07 ML/ha	2.55 ML/ha	138%	0.000	***
Panel G7: Calculation of loss from water runoff	<u> </u>	l	<u> </u>		<u> </u>	<u> </u>

Variable	WESM level	Mean	Mean	Difference in means	<i>p</i> -value	
		(Scenario 1)	(Scenario 2)	$\left(\frac{\mathbf{S2}-\mathbf{S1}}{\mathbf{S1}}\right) \times \mathbf{100\%}$		
Loss from water runoff it $\left(\frac{ML}{ha}\right)$	Level 7	0.36 ML/ha	0.45 ML/ha	26%	0.000	***
= Total water runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$ - Recycled runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$						
Total water runoff it $\left(\frac{ML}{ha}\right)$	Level 7	1.43 ML/ha	3.00 ML/ha	110%	0.000	***
Recycled runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	1.07 ML/ha	2.55 ML/ha	138%	0.000	***
Panel G8: Calculation of crop water loss				1		
$Crop water loss_{it} \left(\frac{ML}{ha}\right) =$	Level 6	7.55 ML/ha	5.40 ML/ha	-28%	0.000	***
Soil evaporation it $\left(\frac{ML}{ha}\right)$ + Deep drainage <sub>it</sub> $\left(\frac{ML}{ha}\right)$ +						
Loss from water runoff it $\left(\frac{ML}{ha}\right)$ +						
Change in soil water $_{it}\left(\frac{ML}{ha}\right)$						
Soil evaporation it $\left(\frac{ML}{ha}\right)$	WESM Input	6.24 ML/ha	6.10 ML/ha	-2.3%	0.254	
Deep drainage <sub>it</sub> $\left(\frac{ML}{ha}\right)$	WESM Input	1.01 ML/ha	0.04 ML/ha	-95.7%	0.000	***
Loss from water runoff it $\left(\frac{ML}{ha}\right)$	Level 7	0.36 ML/ha	0.45 ML/ha	26%	0.000	***
Change in soil water $_{it}\left(\frac{ML}{ha}\right)$	WESM Input	-0.06 ML/ha	-1.19 ML/ha	1938%	0.000	***

Panel G9: Calculation of crop water supply										
Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 6	13.23 ML/ha	12.63 ML/ha	-5%	0.000	***				
= Rainfall <sub>it</sub> $\left(\frac{ML}{ha}\right)$ + Irrigation abstracted <sub>it</sub> $\left(\frac{ML}{ha}\right)$										
+ Recycled runoff										
Rainfall it $\left(\frac{ML}{ha}\right)$	WESM Input	4.57 ML/ha	4.57 ML/ha	0%	0.993					
Irrigation abstracted it $\left(\frac{ML}{ha}\right)$	Level 7	7.59 ML/ha	5.51 ML/ha	-27%	0.000	***				
Recycled runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	1.07 ML/ha	2.55 ML/ha	138%	0.000	***				
Panel G10: Calculation of rainfall to crop water supply ratio (%)		•								
Rainfall to crop water supply ratio (%)	Level 6	34%	36%	5%	0.000	***				
= Rainfall it $\left(\frac{ML}{ha}\right)$ ÷ Crop water supply it $\left(\frac{ML}{ha}\right)$										
Rainfall it $\left(\frac{ML}{ha}\right)$	WESM Input	4.57 ML/ha	4.57 ML/ha	0%	0.993					
Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 6	13.23	12.63 ML/ha	-5%	0.000	***				
		ML/ha								
Panel G11: Calculation of recycled runoff to crop water supply ratio	(%)									
Recycled runoff to crop water supply ratio (%)	Level 6	8%	20%	151%	0.000	***				
= Recycled runoff $_{it}\left(\frac{ML}{ha}\right)$										
÷ Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$										
Recycled runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	1.07 ML/ha	2.55 ML/ha	138%	0.000	***				
--	---------	------------	-------------	------	-------	-----	--	--	--	--
Crop water supply <sub>it</sub> $\left(\frac{ML}{T}\right)$	Level 6	13.23	12.63 ML/ha	-5%	0.000	***				
(ha)		ML/ha								
Panel G12: Calculation of crop water loss to crop water supply ratio (%)										
Crop water loss to crop water supply ratio (%)	Level 6	57%	43%	-23%	0.000	***				
$= \text{Crop water loss}_{\text{it}}\left(\frac{\text{ML}}{\text{ha}}\right)$										
÷ Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$										
Crop water $loss_{it}\left(\frac{ML}{ha}\right)$	Level 6	7.55 ML/ha	5.40 ML/ha	-28%	0.000	***				
Crop water supply <sub>it</sub> $\left(\frac{ML}{L}\right)$	Level 6	13.23	12.63 ML/ha	-5%	0.000	***				
ha/		ML/ha								

#### 8.3.3 Statistical test – Level 6 and Level 7 – Low-level economic variables

This section provides results of *t-test* for the low-level (e.g. Level 7 and Level 6) of WESM variables with regards to measurement of economic variables, including operating cost and sales revenue components. This is summarised in Table 8.3, Panel F.

#### 8.3.3.1 Measurement of sales revenue

For cotton crops, lint and cotton seed are the two products that can create value for a crop business. Of these two products, lint is considered a higher value product with a selling price of \$2.09/kg, compared with \$0.32/kg for cotton seed (see Table 7.3). Assuming that lint selling price and cotton seed selling price remains unchanged across years and between the two samples, sales revenues generated from lint product and from cotton seed product are driven by lint yield and cotton seed yield respectively.

Furthermore, both lint yield and cotton seed yield are driven by the total yield produced per hectare of land, assuming that lint percentage and seed percentage are constant for a particular variety grown, for example 40% and 60% respectively for SICOT 43BRF (see Table 7.3). Since there is no significant difference in the means of total yield between the two scenarios (as discussed in Section 8.3.1), there is no significant difference in the means of lint yield and cotton seed yield, or the sales revenue of lint and cotton seed, between the two scenarios (See Table 8.3, Panel F1 to F4).

#### 8.3.3.2 Measurement of total water cost

In the Chapters 3 and 4, I discussed how irrigation abstracted can be sourced from licensed irrigation and traded irrigation, and therefore, total irrigation cost comprises licensed irrigation cost and traded irrigation cost. Since unit licensed irrigation cost and unit traded irrigation cost are typically not the same, how irrigation is allocated (which is reflected by the ratios of licensed irrigation and traded irrigation to total irrigation abstracted) will determine the irrigation cost.

To make it simpler, in the empirical chapters of this thesis (Chapter 7 and Chapter 8), it is assumed that 100% of irrigation is sourced from licensed irrigation, equal to the amount of irrigation abstracted calculated in Section 8.3.2 (see Table 8.3, Panel F5 and F6). Furthermore, zero rainfall cost and zero recycling costs are assumed. As a result, the

irrigation cost represents total water cost<sup>115</sup> incurred by a crop business for crop production, reported in Table 8.3, Panel F9.

There is a significant difference in the means of total water cost (*p-value* < 0.001) between the two scenarios, with a substantial reduction of total water cost (27%) when moving from Scenario 1 to Scenario 2.

# 8.3.3.3 Measurement of other variable costs

Table 8.3, Panel F10, shows that other variable cost is a product of unit other variable cost and lint yield. Since there is no significant difference in the means of lint yield, there is no significant difference in other variable costs when moving from Scenario 1 to Scenario 2.

Sections 8.3.2 and 8.3.3 lay out the cause and effect of changes in the means of environmental and economic measures at the low level (e.g. Level 7 and 6) of WESM. This allows further analysis of changes in the means of middled-level and high-level variables of WESM in the next sections (Sections 8.3.4 to 8.3.6).

<sup>&</sup>lt;sup>115</sup> Water cost will change as a result of changes in irrigation abstracted which is dependent on water management decisions.

 Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 – Panel F (Level 6 and Level 7: Low-Level Economic Measures)

Variable	WESM	Mean	Mean	Mean difference	p-value	
	level	(Scenario 1)	(Scenario 2)	$\left(\frac{S2-S1}{S1}\right) \times 100\%$		
Panel F1: Calculation of lint yield						
Lint yield, $\left(\frac{\text{kg}}{\text{m}}\right)$	Level 7	1,803 kg/ha	1,806 kg/ha	0.1%	0.880	
ha/		(= 8.0	(= 8.0  bales/ha)			
$= \text{Total yield}_{\text{it}}\left(\frac{\text{kg of total product}}{\text{ha}}\right)$		bales/ha)				
× Lint percentage <sub>it</sub> (%)						
Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$	WESM Input	4,507 kg/ha	4,514 kg/ha	0.1%	0.880	
Lint percentage <sub>it</sub> (%)	WESM	40%	40%	0%		
	Input					
Panel F2: Calculation of seed yield					· ·	
Seed yield <sub>it</sub> $\left(\frac{kg}{ha}\right)$	Level 7	2,704 kg/ha	2,708 kg/ha	0.1%	0.880	
$= \text{Total yield}_{\text{it}}\left(\frac{\text{kg of total product}}{\text{ha}}\right)$						
× Seed percentage <sub>it</sub> (%)						
Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$	WESM Input	4,507 kg/ha	4,514 kg/ha	0.1%	0.880	

Variable	WESM	Mean	Mean	Mean difference	p-value
	level	(Scenario 1)	(Scenario 2)	$\left(\frac{S2-S1}{S1} ight)  imes 100\%$	
Seed percentage <sub>it</sub> (%)	WESM	60%	60%	0%	
	Input				
Panel F3: Calculation of sales revenue (lint)					
Sales revenue <sub>it</sub> (lint) $\left(\frac{\$}{ha}\right)$	Level 6	\$3,765 /ha	\$3,770 /ha	0.1%	0.880
= Lint yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right) \times \text{Lint price}_{\text{it}} \left(\frac{\$}{\text{kg}}\right)$					
Lint yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$	Level 7	1,803 kg/ha	1,806 kg/ha	0.1%	0.880
Lint price $\begin{pmatrix} \$ \\ - \end{pmatrix}$	WESM	\$2.09/ kg of	\$2.09/ kg of	0%	
$\operatorname{Lint price_{it}}\left(\overline{\mathrm{kg}}\right)$	Input	lint	lint		
Panel F4: Calculation of sales revenue (cotton seed)					
Sales revenue <sub>it</sub> (cotton seed) $\left(\frac{\$}{ha}\right)$	Level 6	\$865/ha	\$867/ha	0.1%	0.880
= Seed yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right) \times \text{Seed price}_{\text{it}} \left(\frac{\$}{\text{kg}}\right)$					
Seed yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$	Level 7	2,704 kg/ha	2,708 kg/ha	0.1%	0.880

eed price <sub>it</sub> $\left(\frac{\$}{1-2}\right)$	WESM	\$0.32/kg of	\$0.32/kg of	0%						
seeu price <sub>it</sub> (kg)	<mark>Input</mark>	seed	seed							
Panel F5: Calculation of traded irrigation volume										
Traded irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	0 ML/ha	0 ML/ha							
= Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$										
$\times$ Traded irrigation to Irrigation abstracted ratio <sub>it</sub> (%)										
Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$	Level 7	7.59 ML/ha	5.51 ML/ha	-27%	0.000	***				
Traded irrigation to Irrigation abstracted ratio <sub>it</sub> (%)	WESM	0%	0%							
	Input									
Panel F6: Calculation of licensed irrigation volume										
Licensed irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	7.59 ML/ha	5.51 ML/ha	-27%	0.000	***				
= Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$										
$\times$ Licensed irrigation to Irrigation abstracted ratio <sub>it</sub> (%)										
Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$	Level 7	7.59 ML/ha	5.51 ML/ha	-27%	0.000	***				
Licensed irrigation to Irrigation abstracted ratio <sub>it</sub> (%)	WESM	100%	100%	0%						
	Input									

Panel F7: Calculation of traded irrigation cost									
Traded irrigation $cost_{it} \left(\frac{\$}{ha}\right)$	Level 6	\$0 /ha	\$0 /ha						
= Traded irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$									
× Unit traded irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ML}}\right)$									
Traded irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	0 ML/ha	0 ML/ha						
Unit traded irrigation $\operatorname{cost}_{it}\left(\frac{\$}{M}\right)$	WESM	N/A	N/A						
(ML)	Input								
Panel F8: Calculation of licensed irrigation cost									
Licensed irrigation $cost_{it} \left(\frac{\$}{ha}\right)$	Level 6	\$458/ha	\$333/ha	-27 <b>%</b>	0.000	***			
= Licensed irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$									
× Unit traded irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ML}}\right)$									
Licensed irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	7.59 ML/ha	5.51 ML/ha	-27%	0.000	***			
Unit licensed irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ML}}\right)$	WESM Input	\$60.30/ML	\$60.30/ML	0%					
Panel F9: Calculation of total water cost		· · · · · ·			L	1			

Total water $cost_{it} \left(\frac{\$}{ha}\right)$	Level 6	\$458/ha	\$333/ha	-27%	0.000	***
= Licensed irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$						
+ Traded irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$						
Licensed irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$	Level 6	\$458/ha	\$333/ha	-27%	0.000	***
Traded irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$	Level 6	\$0/ha	\$0/ha			
Panel F10: Calculation of other variable cost						
Other variable $cost_{it} \left(\frac{\$}{ha}\right)$	Level 6	\$216/ha	\$217/ha	0.1%	0.880	
= Unit other variable $cost_{it} \left(\frac{\$}{kg \text{ of lint}}\right)$						
$\times$ Lint yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$						
Unit other variable $cost_{it} \left(\frac{\$}{kg \text{ of lint}}\right)$	WESM Input	\$0.12/kg of lint	\$0.12/kg of lint	0%		
Lint yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$	Level 7	1,803 kg/ha	1,806 kg/ha	0.1%	0.880	

#### 8.3.4 Statistical test – Level 5 of WESM

This section discusses the results of *t-test* for five WESM measures presented in Level 5 (see Table 8.3, Panel E).

# 8.3.4.1 Water input efficiency (%)

In Chapter 4, water input efficiency is defined as the degree to which crop water supply is used for carbon assimilation to generate lint product, measured as a ratio of transpiration to crop water supply. It is also evident in Section 8.3.2 that water input efficiency is mainly driven by crop water loss to crop water supply ratio.

Table 8.3, Panel E1, shows that there is a statistically significant (*p-value* < 0.001) difference in the means of water input efficiency between Scenario 1 and Scenario 2. While there is an increase by 6% in water input efficiency when switching from Scenario 1 water management conditions to Scenario 2 water management conditions, on average, the water input efficiency of 37% for Scenario 2 is still relatively low (e.g. compared to the theoretical value of 100%) as nearly half of the water (43%) that is applied to the cropping system is non-productive and lost from the system<sup>116</sup>.

# 8.3.4.2 Transpiration index

Table 8.3, Panel E2, shows that there is no statistical difference in the means of transpiration index between Scenario 1 and Scenario 2. On average, the numerator of the index (biomass yield) slightly increases by 2.7% while its denominator (transpiration) only increases by 1.6%. As a result, the mean of transpiration index only slightly increases by 1.1% when moving from Scenario 1 to Scenario 2.

The results reported for changes in transpiration index are as expected since the simulation designed in DSSAT to enable crops to grow under non-limiting nitrogen conditions (a total of 300 kg of nitrogen is applied over a crop season) in both scenarios. As discussed in Chapter 3, nitrogen management is considered as a driving factor of the transpiration index. Given the same nitrogen applications, it is expected there should be no significant difference in transpiration index between the two scenarios.

<sup>&</sup>lt;sup>116</sup> As discussed in Section 8.3.2, 20% of water supply is from recycled runoff.

# 8.3.4.3 Harvest index

Similar to the transpiration index, on average the harvest index only slightly decreases by 2.5% from 0.151 under Scenario 1 to 0.147 in Scenario 2. This is because the DSSAT software is designed in a way that crops grow under well controlled pest and disease management, which is considered as the key driver of harvest index. Therefore, as expected, there should be no significant difference in the means of the harvest index between the Scenario 1 and Scenario 2 samples

#### 8.3.4.4 Total sales revenue

Since it is shown in the previous section (Section 8.3.3) that there is no significant difference in the means of both sales revenue generated from lint and cotton seed, it is expected that there should be no significant difference in the means of the sum of them – total sales revenue.

#### 8.3.4.5 Total operating cost

Table 8.3, Panel E5, reveals that there is significant difference in the means of total operating cost (p-value < 0.001) between Scenario 1 and Scenario 2. By definition, fixed operating costs do not change while it is evident that there is no significant difference in the means of other variable costs (see Section 8.3.3) between the two scenarios. It is shown that a reduction of \$125/ha in total operating cost is a result of the decrease of \$125/ha in total water cost.

Though water cost only contributes a small portion to total operating cost, and a significant reduction in water cost (e.g. 27%) merely results in a 4% of reduction in total operating cost, when looking at the business level and the industry level, a reduction of \$125/ha in total operating cost provides significant economic implications for crop businesses. Detailed analysis and discussion of economic significance of the Scenario 2 sample will be presented in Section 8.4.

The analysis of five measures in Level 5 of WESM reveals that by switching from Scenario 1 to Scenario 2 conditions, there is statistical and materially significant difference in the means of water input efficiency and total operating cost. This indicates that these two measures and their associated drivers (deep drainage and total water cost respectively) can be used to explain changes in higher level measures of WESM.

Variable	WESM Level	Mean (Scenario 1)	Mean (Scenario 2)	$\frac{\text{Difference in means}}{\left(\frac{\text{S2}-\text{S1}}{\text{S1}}\right) \times 100\%}$	p-value	
Panel E1: Analysis of water input efficiency						
Water input efficiency <sub>it</sub> (%) = 100% – Crop water loss to crop water supply ratio% – Recycled runoff to crop water supply ratio (%)	Level 5	35%	37%	6%	0.000	***
Crop water loss to crop water supply ratio (%)	Level 6	57%	43%	-23%	0.000	***
Recycled runoff to crop water supply ratio (%)	Level 6	8%	20%	151%	0.000	***
Panel E2: Analysis of transpiration index		l			L	
Transpiration index <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ML of tranpiration}}\right)$	Level 5	2,600 kg of biomass/ML	2,629 kg of biomass/ML	1.1%	0.200	
$= \text{Biomass yield}_{\text{it}} \left(\frac{\text{kg of biomass}}{\text{ha}}\right)$ $\div \text{Transpiration}_{\text{it}} \left(\frac{\text{ML of transpiration}}{\text{ha}}\right)$						
$\frac{(11)}{(11)}$ Biomass yield <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ha}}\right)$	WESM Input	11,948 kg/ha	12,270 kg/ha	2.7%	0.004	*
Transpiration <sub>it</sub> $\left(\frac{ML \text{ of transpiration}}{ha}\right)$	WESM Input	4.61 ML/ha	4.69 ML/ha	1.6%	0.259	
Panel E3: Analysis of harvest index				·		

# Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 – Panel E (Level 5 of WESM)

Variable	WESM	Mean	Mean	Difference in means	p-value	
	Level	(Scenario 1)	(Scenario 2)	$\left(\frac{\mathbf{S2}-\mathbf{S1}}{\mathbf{S1}}\right) \times \mathbf{100\%}$		
Harvest index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{kg of biomass}}\right)$	Level 5	0.151	0.147	-2.5%	0.000	***
= Lint yield <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{ha}}\right)$						
$\div$ Biomass yield <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ha}}\right)$						
Lint yield <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{ha}}\right)$	Level 7	1,803 kg/ha	1,806 kg/ha	0.1%	0.880	
Biomass yield <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ha}}\right)$	WESM Input	11,948 kg/ha	12,270 kg/ha	2.7%	0.004	*
Panel E4: Analysis of total sales revenue						
Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$	Level 5	\$4,630 /ha	\$4,637 /ha	0.1%	0.880	
= Sales revenue <sub>it</sub> (lint) $\left(\frac{\$}{ha}\right)$						
+ Sales revenue <sub>it</sub> (cotton seed) $\left(\frac{\$}{ha}\right)$						
Sales revenue <sub>it</sub> (lint) $\left(\frac{\$}{ha}\right)$	Level 6	\$3,765 /ha	\$3,770 /ha	0.1%	0.880	
Sales revenue <sub>it</sub> (cotton seed) $\left(\frac{\$}{ha}\right)$	Level 6	\$865/ha	\$867/ha	0.1%	0.880	
Panel E5: Analysis of total operating cost		1		1		

Variable	WESM	Mean	Mean	Difference in means	p-value	
	Level	(Scenario 1)	(Scenario 2)	$\left(\frac{\mathbf{S2}-\mathbf{S1}}{\mathbf{S1}}\right) \times \mathbf{100\%}$		
Total operating $cost_{it}\left(\frac{\$}{ha}\right)$	Level 5	\$3,486/ha	\$3,361/ha	-4%	0.000	***
= Total water $cost_{it}\left(\frac{\$}{ha}\right)$ + Other variable $cost_{it}\left(\frac{\$}{ha}\right)$						
+ Fixed operating $cost_{it} \left(\frac{\$}{ha}\right)$						
Total water $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$	Level 6	\$458/ha	\$333/ha	-27%	0.000	***
Other variable $\operatorname{cost}_{it}\left(\frac{\$}{ha}\right)$	Level 6	\$216/ha	\$217/ha	0.1%	0.880	
Fixed operating $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$	WESM Input	\$2,812/ha	\$2,812/ha	0%		

#### 8.3.5 Statistical test – Level 3 and Level 4 of WESM

This section discusses the results of the *t-test* for the two middle-levels of WESM, Level 3 and Level 4, which are reported in Table 8.3, Panel D and Panel C respectively.

#### 8.3.5.1 Level 4 of WESM

In the fourth level of WESM, three measures are presented, including operating profit, crop water use index, and weighted average cotton price. First, Table 8.3, Panel D1, shows that there is a significant difference (*p-value* < 0.01) in the means of operating profit between Scenario 1 and Scenario 2. The table also shows that, on average, while there is not much difference (e.g. \$7/ha) in the value of total sales revenue between the two scenarios, total operating cost reduces by 4% or \$125/ha when moving from Scenario 1 to Scenario 2. As a result, operating profit, which is total sales revenue after subtracting total operating cost, for Scenario 2 increases considerably by 12% or \$132/ha compared to that of Scenario 1.

Similarly, Table 8.3, Panel D2, also shows that there is significant (*p-value* < 0.01) difference in the means of crop water use index – a measure of how productive a crop is in producing lint product from one unit of volume (e.g. ML) of water supplied to the cropping system – between Scenario 1 and Scenario 2. Inspections of the data show that a 5% increase in crop water use index, from 137 kg of lint/ML to 144 kg of lint/ML, is a result of a slight increase (e.g. 1.1%) in transpiration index, a relatively low decrease in harvest index (e.g. 2.5%), and a considerable increase (e.g. 6%) in water input efficiency. In contrast, there is no difference in weighted average cotton price between Scenario 1 and Scenario 2 (see Table 8.3, Panel D3). This is because it is assumed that the economic factors (lint price and seed price), as well as variety factors (lint percentage and seed percentage), remain constant across years and between the two scenarios.

The comparisons presented above indicate that of the three measures of Level 4 of WESM, on average, both operating profit – which represents profitability of a crop business – and crop water use index – which reflects productivity of the crop business in respect to water resource use – improve when switching from Scenario 1 conditions to Scenario 2 conditions. This provides empirical evidence for further discussions of causes of changes of Level 3 measures in Section 8.3.5.2.

Variable	WESM level	Mean	Mean	Mean difference	p-value	
		(Scenario 1)	(Scenario 2)	$\left(\frac{S2-S1}{S1}\right) \times 100\%$		
Panel D1: Analysis of operating profit						
Operating profit <sub>it</sub> $\left(\frac{\$}{ha}\right)$	Level 4	\$1,144/ha	\$1,276/ha	12%	0.000	**
= Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$						
- Total operating $cost_{it}\left(\frac{\$}{ha}\right)$						
Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$	Level 5	\$4,630 /ha	\$4,637 /ha	0.1%	0.880	
Total operating $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$	Level 5	\$3,486/ha	\$3,361/ha	-4%	0.000	***
Panel D2: Analysis of crop water use index						
Crop water use index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{ML}}\right)$	Level 4	137 kg of	144 kg of	5%	0.000	**
$= \text{Transpiration index}_{it} \left( \frac{\text{kg of biomass}}{\text{ML of transpiration}} \right)$		lint/ML	lint/ML			
× Harvest index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{kg of biomass}}\right)$						
× Water input efficiency <sub>it</sub> (%)						

# Table 8.1 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 – Panel D (Level 4 of WESM)

Variable	WESM level	Mean	Mean	Mean difference	p-value	
		(Scenario 1)	(Scenario 2)	$\left(\frac{S2-S1}{S1}\right) \times 100\%$		
Transpiration index. ( <u>kg of biomass</u> )	Level 5	2,600 kg of	2,629 kg of	1.1%	0.200	
ML of transpiration/		biomass/ML	biomass/ML			
Harvest index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{kg of biomass}}\right)$	Level 5	0.151	0.147	-2.5%	0.000	***
Water input efficiency <sub>it</sub> (%)	Level 5	35%	37%	6%	0.000	***
Panel D3: Analysis of weighted average cotton price Weighted average cotton $\text{price}_{it}\left(\frac{\$}{\text{kg of total yield}}\right)$	Level 4	\$ 1.03/kg	\$ 1.03/kg	0%	0.085	
$= \begin{bmatrix} \text{Lint percentage}_{it} (\%) \\ (\$) \end{bmatrix}$						
× Lint price <sub>it</sub> $\left(\frac{1}{\text{kg}}\right)$						
+ Seed percentage <sub>it</sub> (%)						
× Seed price <sub>it</sub> $\left(\frac{\$}{\text{kg}}\right)$						
Lint percentage <sub>it</sub> (%)	WESM Input	40%	40%	0%		
Seed percentage <sub>it</sub> (%)	WESM Input	60%	60%	0%		

Variable	WESM level	Mean	Mean	Mean difference	p-value	
		(Scenario 1)	(Scenario 2)	$\left(\frac{\mathbf{S2}-\mathbf{S1}}{\mathbf{S1}}\right)\times\mathbf{100\%}$		
Lint price, $\left(\frac{\$}{-}\right)$	WESM Input	\$2.09/ kg of	\$2.09/ kg of	0%		
$\operatorname{Lint price}_{\mathrm{it}}\left(\overline{\mathrm{kg}}\right)$		lint	lint			
Seed price: $\left(\frac{\$}{-}\right)$	WESM Input	\$0.32/kg of	\$0.32/kg of	0%		
kg/		seed	seed			

# 8.3.5.2. Level 3 of WESM

Level 3 of WESM consists of two drivers of return on water (which is presented in Level 2 of WESM) – profit margin and economic water use index – both of which show a significant difference in their means between Scenario 1 and Scenario 2 (see Table 8.3, Panel C). This is expected since operating profit, a driver of profit margin, and crop water use index, a driver of economic water use index, are significantly different when moving from Scenario 1 to Scenario 2.

More particularly, Table 8.3, Panel C1, shows that though total sales revenue is very much the same between the two scenarios, operating profit, as reported in Section 8.3.5.2, shows a 12% increase. As a result, profit margin increases by 12%, from 24% for Scenario 1 to 27% for Scenario 2.

Economic water use index – a ratio of sales revenue to crop water supply – is shown to increase by 5% or \$17/ML of crop water supply (see Table 8.3, Panel C2). Since its other two drivers – lint percentage and weighted average cotton price – are assumed to remain constant within and between the two scenarios, crop water use index, with an increase of 5%, is the only driving factor to cause the change in economic water use index.

Variable	WESM level	Mean (Scenario 1)	Mean (Scenario 2)	Mean difference (S2 – S1)	p-value	
	10,01	(80011110-1)	(80011010)	$\left(\frac{-1}{51}\right) \times 100\%$		
Panel C: Analysis of profit margin						
Profit margin <sub>it</sub> (%)	Level 3	24%	27%	12%	0.000	***
$= \left[ \text{Opearating profit}_{\text{it}} \left( \frac{\$}{\text{ha}} \right) \right]$						
÷ Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right) \times 100\%$						
Operating $\operatorname{profit}_{it}\left(\frac{\$}{ha}\right)$	Level 4	\$1,144/ha	\$1,276/ha	12%	0.000	**
Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$	Level 5	\$4,630 /ha	\$4,637 /ha	0.1%	0.880	
Panel C2: Analysis of economic water use index						
Economic water use $index_{it}\left(\frac{\$}{ML}\right)$	Level 3	\$352 /ML	\$369 /ML	5%	0.000	**
= Crop water use index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{ML}}\right)$						
$\div$ Lint percentage <sub>it</sub> (%)						
× Weighted average cotton price <sub><i>it</i></sub> $\left(\frac{\$}{\text{kg of total product}}\right)$						
Crop water use index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{kg of lint}}\right)$	Level 4	137 kg of	144 kg of	5%	0.000	**
ML /		lint/ML	lint/ML			

# Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 – Panel C (Level 3 of WESM)

Variable	WESM level	Mean (Scenario 1)	Mean (Scenario 2)	$\frac{\text{Mean difference}}{\left(\frac{\text{S2}-\text{S1}}{2}\right) \times 100\%}$	p-value	
				$(s_1)^{\times 100\%}$		
Lint percentage <sub>it</sub> (%)	WESM	40%	40%	0%		
	Input					
Weighted average cotton price, $\left( \frac{\$}{} \right)$	Level 4	\$ 1.03/kg	\$ 1.03/kg	0%	0.085	
(kg of total yield)						

#### 8.3.6 Statistical test – Level 1 and Level 2 of WESM

This section reports results of the *t-test* for the two high-levels of WESM – Level 2 and Level 1 – which are reported in Table 8.3, Panel B and Panel A, respectively.

#### 8.3.6.1. Level 2 of WESM

In the second level of WESM, the three drivers of the overall economic and environmental sustainability measure – profit to water cost ratio – are discussed.

First, it is shown that there is a significant (*p-value* < 0.001) difference in the means of return on water – a measure of economic return on water resource use – between the two scenarios (see Table 8.3, Panel B1). The 17% increase in return on water, which is equivalent to an additional gain of \$14/ML (= 101 – 87), is a result of a 12% increase in profit margin and 5% increase in economic water use index when switching from Scenario 1 to Scenario 2.

Second, Table 8.3, Panel B2, shows that there is a significant difference in the means of water use leverage between Scenario 1 and Scenario 2. Since rainfall to crop water supply, on average, is not significantly different between the two scenarios, it is the significant change in recycled runoff to crop water supply ratio driving the change in water use leverage. More particularly, water use leverage decreases by 24%, from 58% for Scenario 1 to 44% for Scenario 2, as a result of a substantial increase in recycled runoff to crop water supply ratio. The results reported in Table 8.3, Panel B2, emphasise the impact of runoff recycling decisions on both environmental and economic performance of the crop business. The higher recycled runoff to crop water supply ratio, the lower water use leverage, indicating that lower abstractions/withdrawals from the environment are required and lower irrigation water costs will be incurred (as a result of the reduction in the amount of abstracted water).

Variable	WESM	Mean	Mean	Mean difference	p-value	
	level	(Scenario 1)	(Scenario 2)	$\left(\frac{S2-S1}{S2-S1}\right) \times 100\%$		
				<b>S1</b>		
Panel B1: Analysis of return on water						
Return on water <sub>it</sub> $\left(\frac{\$}{ML}\right)$	Level 2	\$87/ML	\$101/ML	17%	0.000	***
= Profit margin <sub>it</sub> (%)						
× Economic water use $index_{it}\left(\frac{\$}{ML}\right)$						
Profit margin <sub>it</sub> (%)	Level 3	24%	27%	12%	0.000	***
Economic water use $index_{it}\left(\frac{\$}{ML}\right)$	Level 3	\$352 /ML	\$369 /ML	5%	0.000	**
Panel B2: Analysis of water use leverage						
Water use leverage (%)	Level 2	58%	44%	-24%	0.000	***
= 100%						
<ul> <li>– [Rainfall to crop water supply ratio (%)</li> </ul>						
+ Recycled runoff to crop water supply ratio (%) ]						
Rainfall to crop water supply ratio (%)	Level 6	34%	36%	5%	0.000	
Recycled runoff to crop water supply ratio (%)	Level 6	8%	20%	151%	0.000	***
Panel B3: Analysis of weighted average irrigation cost		1	1			

# Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 – Panel B (Level 2 of WESM)

Variable	WESM	Mean	Mean	Mean difference	p-value	
	level	(Scenario 1)	(Scenario 2)	$\left(\frac{\mathrm{S2}-\mathrm{S1}}{\mathrm{S1}}\right) \times 100\%$		
Weighted average irrigation $cost_{it} \left(\frac{\$}{ML}\right)$	Level 2	\$60.30 /ML	\$60.30 /ML	0%		
$= \left[ \text{Licensed irrigation to Irrigation abstracted ratio}_{\text{it}} (\%) \right]$						
× Unit licensed irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ML}}\right)$						
+ $\left[ \text{Traded irrigation to Irrigation abstracted ratio}_{it} (\%) \right]$						
× Unit traded irrigation $cost_{it} \left(\frac{\$}{ML}\right)$						
Licensed irrigation to Irrigation abstracted ratio <sub>it</sub>	WESM	100%	100%	0%		
(%)	Input					
Traded irrigation to Irrigation abstracted ratio <sub>it</sub>	WESM	0%	0%			
(%)	Input					
Unit licensed irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ML}}\right)$	<mark>WESM</mark> Input	\$60.30 /ML	\$60.30 /ML	0%		
Unit traded irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ML}}\right)$	WESM Input	N/A	N/A			

# 8.3.6.2 Level 1 of WESM

In the top level of the WESM model, there is a significant (*p-value* < 0.001) difference in the means of profit to water cost ratio between Scenario 1 and Scenario 2 (see Table 8.3, Panel A). This can be explained by a significant increase in the numerator (return on water) accompanied by a significant decrease in the denominator (water use leverage) when moving from Scenario 1 to Scenario 2. More specifically, it is empirically evident that a substantial increase of 56% in profit to water cost ratio is a result of a considerable increase of 17% in return on water and a significant reduction of 24% in water use leverage.

The 17% change in return on water is driven by a 12% increase in profit margin and a 5% increase in economic water use index; this is driven by a reduction of 27% in water cost and increase in operating profit, and by an increase of 6% in water input efficiency, respectively. In addition, it is reported that the considerable decrease in total water cost is driven by a reduction of 27% in the amount of irrigation abstracted from the environment, while the improvement of water input efficiency is mainly driven by a substantial reduction of 96% in water loss from deep drainage. At the operational level, it is empirically evident that such a significant reduction (27%) in the amount of irrigation abstracted and the substantial reduction (96%) in deep drainage is a result of runoff recycling decisions and irrigation scheduling decisions, respectively.

On the other hand, the decrease of 24% in water use leverage is driven by a significant increase of 151% in recycled runoff to crop water supply ratio; this is driven by a substantial increase of 138% in the amount of recycle runoff. Similar to the change in return on water, it is empirically evident that such a considerable decrease (24%) in water use leverage is a result of runoff recycling decisions at the operational level.

Variable	WESM	Mean	Mean	Mean difference	p-value	
	level	(Scenario 1)	(Scenario 2)	$\left(\frac{\mathbf{S2}-\mathbf{S1}}{\mathbf{S1}}\right)\times\mathbf{100\%}$		
Profit to water cost ratio <sub>it</sub>	Level 1	2.62	4.10	56%	0.000	***
= Return on water <sub>it</sub> $\left(\frac{\$}{ML}\right)$						
÷ [Water use leverage <sub>it</sub> (%)						
× Weighted average irrigation cost it $\left(\frac{\$}{ML}\right)$						
Return on water <sub>it</sub> $\left(\frac{\$}{ML}\right)$	Level 2	\$87/ML	\$101/ML	17%	0.000	***
Water use leverage <sub>it</sub> (%)	Level 2	58%	44%	-24%	0.000	***
Weighted average irrigation cost it $\left(\frac{\$}{ML}\right)$	Level 2	\$60.30 /ML	\$60.30 /ML	0%	0.395	

#### Table 8.3 Statistical Test: Univariate Mean Differences between Scenario 1 and Scenario 2 – Panel A (Level 1 of WESM)

# 8.4 Environmental and economic implications of moving from Scenario 1 to Scenario 2

In Section 8.3, results of the statistical test *(t-test)* for the significant difference in the means of all measures presented in WESM (from Level 7 to Level 1) between Scenario 1 and Scenario 2 are reported and discussed.

The key findings, based on 86 years for the period of 1924–2014 (excluding 1928, 1940, 1945 and 2002 since crops could not be grown due to drought conditions), show that there is a statistically significant difference in the means of a range of key economic and environmental sustainability measures between the two scenarios. They include:

- Irrigation abstracted, crop water loss and total water cost (Level 6 and Level 7 of WESM)
- ii. Water input efficiency and total operating cost (Level 5 of WESM)
- iii. Crop water use index and operating profit (Level 4 of WESM)
- iv. Profit margin and economic water use index (Level 3 of WESM)
- v. Water use leverage and return on water (Level 2 of WESM) and
- vi. Profit to water cost ratio (Level 1 of WESM).

It is empirically evident<sup>117</sup> that statistically, the choice of improved management practices in furrow irrigation systems results in better economic and environmental performance than that of current practices in furrow irrigation systems.

The comparison between the means of the two samples across seven levels of WESM are represented in Figure 8.2, which shows the top-down approach to how the overall organisational decisions, with regards to economic and water sustainability, made at the organisational level, link to the key management decisions made at an operational level.

<sup>&</sup>lt;sup>117</sup> By 'empirically evident', I mean I apply simulation method to obtain data for testing theory relating to how crop water and economic sustainability performance can be improved by changing water management practices.



Figure 8.2 Levels 1 to 2 of the WESM model



Figure 8.2 Levels 3 to 5 of the WESM model



Figure 8.2 Levels 5 to 6 of the WESM model



Figure 8.2 Levels 6 to 7 of the WESM model (economic low-level measures)



Figure 8.2 Levels 6 to 7 of the WESM model (environmental low-level measures)

# 8.4.1 Difference in environmental resource use and environmental sustainability performance between Scenario 1 and Scenario 2

As discussed in the previous sections, environmental and economic performance of the crop business will change significantly as a result of changes in irrigation scheduling decisions (e.g. increasing furrow inflow rate from 2.8 ML/ha to 6.0 ML/ha while reducing irrigation time from 745 minutes to 310 minutes) and runoff recycling management decisions (e.g. increasing the ratio of recycled runoff from 75% to 85%). This section focuses on discussions of changes in use of environmental resources, more specifically water resources, and water sustainability performance between Scenario 1 and Scenario 2 as a result of these two water management decisions.

#### 8.4.1.1 Difference in recycled water runoff and loss from water runoff

Table 8.4, Panel A1, reports that runoff from irrigation increases considerably from 0.90 ML/ha in Scenario 1 to 2.47 ML/ha in Scenario 2. This is because furrow inflow rate increases rapidly from 2.8 l/s to 6.0 l/s (which can be observed via SIRMOD animation). Consequently, total water runoff increases substantially by 1.58 ML/ha.

However, water runoff can be recycled and stored in a dam belonging to the farm. Assuming that 75% of water runoff is recycled under Scenario 1 while 85% of water runoff can be recycled under Scenario 2 due to a better runoff recycling management practice, a considerable increase in the amount of runoff is recycled to the dam (e.g. 1.48 ML/ha) with relatively low increases in water loss from runoff (e.g. 0.10 ML/ha) observed when moving from Scenario 1 to Scenario 2. This means runoff recycling management implemented in Scenario 2 not only enables more water to be added to the dam to supplement irrigation that is required for a crop, but also reduces the gap in loss from water runoff between the two scenarios.

#### 8.4.1.2 Difference in irrigation applied and irrigation abstracted

It is shown in Table 8.4, Panel A2, that there is not much difference (e.g. only a slight decrease of 0.60 ML/ha) in the amount of irrigation needs to be applied to the furrows to supplement seasonal rainfall when moving from the current furrow irrigation practices to the improved furrow irrigation practices. In this sense, Scenario 2 does not provide any better irrigation performance (e.g. the amount of withdrawals) than that of Scenario 1.

However, an increase of 1.48 ML/ha in recycled runoff results in an additional decrease of 1.48 ML/ha in irrigation water that needs to be abstracted from the environment. In total, a significant decrease in withdrawals (e.g. 2.08 ML/ha) is reported when moving from Scenario 1 to Scenario 2. This not only implies less water impost of cotton cultivation on the environment, but also a significant reduction in water cost (for example \$125/ha, assuming irrigation cost of \$60.30/ML).

#### 8.4.1.3 Difference in crop water loss components

Table 8.4, Panel A3, shows that a reduction of 2.15 ML/ha is mainly driven by a decrease of deep drainage (e.g. 0.97 ML/ha) and a decrease in change in soil water (e.g., 1.13 ML/ha). A significant decrease in deep drainage indicates that Scenario 2 does provide better irrigation performance with regard to losses to deep drainage than that of Scenario 1. In addition, a substantial decrease in change in soil water under S2 indicates that better use of soil water is made by the crop.

On the one hand, changes in irrigation scheduling from Scenario 1 to Scenario 2 leads to significant increase in water runoff (but it can be managed via better recycling management practices). On the other hand, the improved irrigation management practices result in a substantial reduction in deep drainage and change in soil water. Water loss to deep drainage, which is unlike water loss to runoff, cannot be managed after it occurs, and therefore, provides significant economic and environmental consequences. For example, the amount of crop water saved from reductions in loss to deep drainage could be potentially used to grow more cotton given that farms are growing under capacity due to limitations on available water<sup>118</sup> and add more value to the crop business. In addition, deep drainage causes potential harm to ecosystems and the environment as a whole (Smith 2005).

Furthermore, the change in crop water loss reported in the section will translate to important economic and environmental outcomes for individual farms and the cotton industry as a whole. This will be discussed in detail in Chapter 9.

<sup>&</sup>lt;sup>118</sup> However, this is a huge debate. Environmentalists argue that the water should be returned as environmental flows. The industry wants to grow more cotton. The government wants more taxes.

#### 8.4.1.4 Difference in crop water supply components

Crop water supply can be viewed from two different perspectives, depending on the purpose for which the crop manager/farmer wants to know the information. Typically, he/she looks at two different questions: how dependent her/his crop is on the water abstracted from the environment, and how efficient the crop is at converting water resources into biomass. For the former question, it is useful to decompose crop water supply into rainfall, irrigation abstracted and recycled runoff components. For the latter question, it is useful to view crop water supply as a sum of a non-productive water component (e.g. crop water loss), recycled runoff, and a productive water component (e.g. transpiration).

#### Water use leverage

In order to gain insights into what drives water use leverage, crop water supply needs to be considered from the crop water inflow point of view. Among rainfall, irrigation abstracted and recycled runoff, irrigation abstracted is the only component that reflects the dependency of crop water use on withdrawals from the environment. Moreover, the degree to which crop water used is sourced by water abstracted from the environment (which is defined as water use leverage) depends on the proportion of rainfall and recycled water runoff in total crop water supply. The higher the ratios of rainfall and/or recycled runoff to crop water supply, the lower the proportion of crop water requirements that need to be abstracted from the environment.

Table 8.4, Panel A4, shows that given the same seasonal rainfall (cumulative rainfall from the planting date to the harvesting) and a slight decrease in crop water supply (0.6 ML/ha, see Section 8.3.2), the improvement of water use leverage from 58% under Scenario 1 conditions to 44% under Scenario 2 conditions is mainly driven by a significant increase in recycled runoff to crop water supply ratio (from 8% to 20%). This implies that water sustainability performance of a crop, represented by the water use leverage measure, can be improved as a result of better water recycling management practices.

#### Water input efficiency

Table 8.4, Panel A5, reports ratios of crop water supply components, from the crop water outflow point of perspective, to the total crop water supply, to gain a better understating of what drives water input efficiency. It is empirically evident that the improvement of water input efficiency, from 35% under Scenario 1 conditions to 37% under Scenario 2 conditions,

is driven by a decrease in crop water loss to crop water supply ratio (from 57% to 43%) and an increase in recycled runoff to crop water supply ratio (from 8% to 20%).

 Table 8.4 Economic and environmental differences between Scenario 1 and Scenario 2 – Panel A (Difference in environmental resource use)

Variable	WESM level	Mean	Mean	Difference in means				
		(Scenario 1)	(Scenario 2)	( <b>S2 – S1</b> )				
Panel A1: Difference in recycled runoff and loss from water runoff								
Runoff from irrigation it $\left(\frac{ML}{ha}\right)$	Level 7	0.90	2.47	1.57				
Runoff from rainfall it $\left(\frac{ML}{ha}\right)$	WESM Input	0.52	0.53	0.01				
Total water runoff it $\left(\frac{ML}{ha}\right)$	Level 7	1.42	3.00	1.58				
Recycled runoff it $\left(\frac{ML}{ha}\right)$	Level 7	1.07	2.55	1.48				
Loss from water runoff it $\left(\frac{ML}{ha}\right)$	Level 7	0.35	0.45	0.10				
Panel A2: Difference in irrigation abstracted								
			·					
Irrigation applied it $\left(\frac{ML}{ha}\right)$	Level 7	8.66	8.06	-0.60				
Recycled runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	1.07	2.55	1.48				
Irrigation abstracted it $\left(\frac{ML}{ha}\right)$	Level 7	7.59	5.51	-2.08				
Variable	WESM level	Mean	Mean	Difference in means				
--	------------	--------------	--------------	---------------------	--	--		
		(Scenario 1)	(Scenario 2)	( <b>S2 – S1</b> )				
Panel A3: Difference in crop water loss								
Soil evaporation it $\left(\frac{ML}{ha}\right)$	WESM Input	6.24	6.10	-0.15				
Loss from water runoff it $\left(\frac{ML}{ha}\right)$	Level 7	0.36	0.45	0.09				
Deep drainage <sub>it</sub> $\left(\frac{ML}{ha}\right)$	WESM Input	1.01	0.04	-0.97				
Change in soil water <sub>it</sub> $\left(\frac{ML}{ha}\right)$	WESM Input	-0.06	-1.19	-1.13				
Crop water $loss_{it}\left(\frac{ML}{ha}\right)$	Level 6	7.55	5.40	-2.15				
Panel A4: Difference in crop water supply components (from crop water inflow point of view) and water use leverage								
Crop water supply $\left(\frac{ML}{ha}\right)$	Level 6	13.23	12.63	-0.60				
Rainfall to crop water supply ratio (%)	Level 6	34%	36%	5%				
Recycled runoff to crop water supply ratio (%)	Level 6	8%	20%	151%				

Variable	WESM level	Mean	Mean	Difference in means		
		(Scenario 1)	(Scenario 2)	( <b>S2 – S1</b> )		
Water use leverage <sub>it</sub> (%)	Level 2	58%	44%	-14%		
Panel A5: Difference in crop water supply (from crop water outflow point of view) and water input efficiency						
Crop water loss to crop water supply ratio (%)	Level 6	57%	43%	-14%		
Recycled runoff to crop water supply ratio (%)	Level 6	8%	20%	12%		
Water input efficiency <sub>it</sub> (%)	Level 5	35%	37%	2%		

# 8.4.2 Difference in water cost and economic sustainability performance between Scenario 1 and Scenario 2 samples

Table 8.4, Panel B, summarises changes in total sales revenue, total operating costs, and operating profit, as a result of water management decisions under Scenario 2 conditions, outlined in Sections 8.3.2, 8.3.3 and 8.3.4, in order to provide better understanding of economic implications of moving from Scenario 1 to Scenario 2.

It is shown that the change of total water cost (e.g. \$125/ha) is the key driver of change in profitability (e.g. \$132/ha). Furthermore, a reduction of water cost from \$458/ha to \$333/ha is explained by a reduction of irrigation water abstracted (e.g. 2.08 ML/ha) – a result of better runoff recycling management practices.

The change in operating profit, a measure of the economic sustainability performance of a crop business, will translate into economic value created for the crop business by moving from the current water management practice to the more advanced water management practice, and provide economic implications for individual farms and the cotton industry as a whole. This will be discussed in details in Section 8.4.4.

Variable	WESM Level	Mean	Mean	Difference in the means	
		(Scenario 1)	(Scenario 2)	(S2 - S1)	
Difference in economic yield and total sales revenue					
Lint yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$	Level 7	1,803	1,806	2.6	
Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$	WESM Input	4,507	4,514	7	
Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$	Level 5	\$4,630	\$4,637	\$7	
Difference in irrigation cost and total operating cost					
Total water $\text{cost}_{it}\left(\frac{\$}{\text{ha}}\right)$	Level 6	\$458	\$333	-\$125	
Other variable $\operatorname{cost}_{it}\left(\frac{\$}{ha}\right)$	Level 6	\$216	\$217	\$1	
Total operating $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$	Level 5	\$3,486	\$3,362	-\$126	
Difference in operating profit and profit margin					
Operating $\operatorname{profit}_{it}\left(\frac{\$}{ha}\right)$	Level 4	\$1,144	\$1,276	\$132/ha	
Profit margin <sub>it</sub> (%)	Level 3	24%	27%	3%	

Table 8.4 Economic and environmental differences between Scenario 1 and Scenario 2 – Panel B (Difference in economic performance)

#### 8.4.3 Economic and environmental sustainability performance

This section reports changes in four key measures that imply both economic and environmental performance – crop water use index, economic water use index, return on water and profit to water cost ratio – when moving from Scenario 1 to Scenario 2 (see Table 8.4, Panel C).

First it is demonstrated that there are small increases in the crop water use index and economic water use index. This is because the irrigation rules chosen, on average, do not favour Scenario 2 more than Scenario 1, leading to a slight change in economic yield between Scenario 1 and Scenario 2. Given that total crop water supply for Scenario 2 is not much different from that of Scenario 1 and the economic factors (e.g. lint price and cotton seed price) are assumed constant within and between the two examined samples, the crop water use index and economic water use index do not shown significant improvements. We might expect that future crops, using a refined/optimal irrigation management, should result in higher economic yield, and therefore, higher a crop water use index and economic water use index.

On the other hand, it is reported that return on water increases by 17% from \$87/ML to \$101/ML. Since return on water measures the economic return on water resource use (in volume), the value of return on water for Scenario 2 is used to quantify the economic gain from water saving when moving from Scenario 1 to Scenario 2. This will be discussed in detail in Section 8.4.4.

Lastly, the profit to water cost ratio is shown to improve significantly by 56% or 1.48 (e.g. from 2.62 for Scenario 1 to 4.10 for Scenario 2). Since the numerator of this ratio – operating profit – is used as a proxy for economic sustainability, and its denominator – water cost or irrigation cost – is used as an indicator of the water impost of cotton cultivation on the environment, this result implies improvements to both economic and environmental sustainability. We have maintained economic sustainability while improving environmental sustainability, which is a move towards sustainability.

Table 8.4 Economic and environmental differences between Scenar	io 1 and Scenario 2 – Panel C (higher level measures)
---	---

Variable	WESM level	Mean	Mean	Mean difference
		(Scenario 1)	(Scenario 2)	( <b>S2 – S1</b> )
Crop water use index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{ML}}\right)$	Level 4	137 kg	144 kg	7 kg
Economic water use index <sub>it</sub> $\left(\frac{\$}{ML}\right)$	Level 3	\$352	\$369	\$17
Return on water <sub>it</sub> $\left(\frac{\$}{ML}\right)$	Level 2	\$87	\$101	\$14
Profit to water cost ratio <sub>it</sub> $\left(\frac{\text{\$ operating profit}}{\text{\$ water cost}}\right)$	Level 1	2.62	4.10	1.48

## 8.5 Refining simulation experimental design

There are a number of ways in which simulation experimental design can be refined in future research. These are outlined below.

#### a. Improvement of irrigation applied

For the particular improved scenario examined in this thesis, as explained in Section 8.2, there is not much improvement/reduction in the amount of irrigation water applied to the cropping system compared to the current scenario.<sup>119</sup> As a result, on average, total crop water supply only reduces by 5% or 0.6 ML/ha. It is expected that future crops, using a refined/optimal furrow irrigation management practice, would result in more significant decreases in irrigation required and hence crop water supply for the crop business.

### b. Improvement of crop water loss

Since under the improved water management practice, deep drainage decreases significantly to nearly zero, while loss to water runoff is reduced by 85% and only makes up 6% of total crop water loss (e.g. 0.45ML/ha of loss from water runoff in 6.59 ML/ha of total water lost, see Table 8.3, Panel G8), the only water loss component that needs to be reduced considerably in order to reduce total crop water loss is soil evaporation. However, under furrow irrigation systems, soil surface is still dramatically wet and water lost from soil evaporation is still a major cause of crop water loss.

Since the large residual soil evaporation cannot be reduced whenever the irrigation process results in a wet soil surface, the sprinkler irrigation method may not improve water input efficiency. Other irrigation methods need to be considered to resolve this issue. With current technology, subsurface irrigation may be the best irrigation management strategy to improve water input efficiency.

#### c. Improvement of lint yield

One way to improve crop production and hence total sales revenue for future crops is to refine irrigation scheduling under Scenario 2 conditions to increase potential yield for a crop. Associated with the improvement of yield production will be a greater transpiration and hence possibly greater crop water requirement. This then leads to some increase in total

<sup>&</sup>lt;sup>119</sup> However, it is worth recalling that deep drainage reduces substantially and a greater amount of runoff is recycled under Scenario 2.

variable costs, including water cost and other variable cost. However, it is expected that the value of the increased lint exceeds the cost of the additional water and additional other variable costs. This can be tested in future research. Given that Scenario 2 examined in this thesis is considered as the improved, not optimal, management practice, the empirical results and implications provided in this thesis offer a step towards sustainability. In future research, further refinements in irrigation management can be examined in an optimisation process to identify the best management practice to maximise economic and water sustainability performance for a crop business.

#### d. Weighted average irrigation cost

There is no difference in the means of weighted average irrigation cost – the third driver of the profit to water cost ratio (see Table 8.3, Panel B3). This is because it is assumed that irrigation allocations are the same (e.g. 100% of irrigation abstracted is sourced by licensed irrigation) and licensed irrigation cost per ML remains unchanged (e.g. \$60.30) across years and between the two scenarios. By doing that, the effects of irrigation scheduling decisions and runoff recycling decisions on profit to water cost ratio are isolated from irrigation allocation decisions. Future research can relax the assumption of irrigation allocation decisions that are described above to gain more insights into the impact of irrigation allocation allocations on the overall economic and environmental performance of a crop business.

Future research can consider other scenarios of irrigation allocations, for example, irrigation is sourced from only traded irrigation or a mix of licensed irrigation and traded irrigation. This will enable gaining insights into the magnitude effect of various irrigation allocation decisions on the water cost incurred by a crop business and the economic implications of such decisions

#### e. Improvement of profit to water cost ratio

It can be argued that an increase of 56% in profit to water cost ratio is a result of **runoff recycling decisions** and **irrigation scheduling decisions** made at the operational level. Given that Scenario 2 still reflects suboptimal irrigation management decisions, further improvement in profit to water cost ratio is expected for future crops by using optimal conditions.

#### f. Incorporation of energy cost

These figures reported above are quantified without taking into consideration costs saved from reductions in energy use (for example, diesel fuel used for pumping irrigation water from rivers/bores/dams to the field). Furthermore, it is expected that irrigation performance under Scenario 2 conditions can be further improved through a process of optimization, leading to a further potential social, environmental and economic gain for a crop business and the industry as a whole.

## 8.6 Conclusion

In this chapter, in order to demonstrate the utility of the designed WESM, I apply and test theory around how irrigation practice can be improved (Scenario 2) and provide empirical evidence as to how and to what extent changing management practice at the operational level of a crop business has impacts on its overall water and economic sustainability performance.

The novel two-phased simulation modelling approach enables a comparison between two scenarios. Furthermore, the hierarchical WESM model allows me to analyse and identify what activities at the operational level are driving the change in the overall water and economic performance of a crop business when moving from Scenario 1 to Scenario 2.

It is empirically evident that there is a statistically, economically and environmentally significant difference when switching from the current water management practice to the improved water management practice described in this chapter. More specifically, the organisational level measure of profit to water cost ratio increases by 56%, operating profit increases by 30%, and 28% of water volume can be saved. In Chapter 9, detailed discussions on how individual cotton farms, as well as the industry as a whole, can improve their economic and environmental performance by moving from the current management practice to the improved management practice.

Finally, the output of Scenario 2 could provide the basis for setting better targets (compared to using Scenario 1). However, future crops, using a refined/optimal furrow irrigation management practice, should result in more significant improvement in the overall water and economic sustainability measure as well as other key sustainability measures contained in the

designed WESM model. To do this, better targets can be set to enable crop managers to move towards sustainability.

# **Chapter 9. Conclusions and Implications**

## 9.1 Research Summary

The central research question addressed in this thesis was how can Environmental Performance Measurement Systems (EPMS) be *designed* and *used* in an agricultural setting to support managers in water and economic sustainability-related decision making and control.

Water plays a crucial role in crop agriculture in enabling crop businesses to maintain profitability in the long-term. On the other hand, as a substantial user of water, agriculture has a significant impact on the sustainability of freshwater at both global and local levels. Therefore, improvement of water management in agriculture is vital for global economic and environmental sustainability. Despite this, extant research provides little guidance on how EPMS can be designed and used to support improved environmental (water) and economic sustainability in the crop production context.

Cotton production in Australia is selected as the empirical setting in this thesis for four main reasons: (i) cotton is a high value crop and is of economic significance; (ii) cotton production provides fibre, a primary cotton product to meet one of the most essential needs of human beings<sup>120</sup>; (iii) cotton production (both yield and quality) is sensitive to water availability; and (iv) cotton is perceived to be a high user of water, therefore, improvement of water sustainability of cotton production implies economic and environmental significance at the regional and global levels.

In the context of cotton businesses, extant research has not identified valid crop water performance measures which are informative about both water and economic sustainability performance. As outlined in Chapter 3, the literature surrounding efficiency of water used by plants is littered with either inconsistent or invalid terminology. Further, given such issues, standards and targets developed based on existing measures are unlikely to be useful for managers to support sustainability- related decision making and control at the organisational level.

<sup>&</sup>lt;sup>120</sup> Another product from cotton production is cotton seed which can be used as cooking oil.

From a management accounting (MA) research perspective, there has been very little MA research conducted in an agricultural setting. In addition, extant EPMS research presents two underlying theoretical problems, which are also reflected at a more general level in broader performance measurement systems (PMS) research. The first relates to the validity issue of EPMS design. The second relates to the target setting issue of EPMS design.

#### 9.1.1 Validity issue of EPMS (contribution 1)

In respect to the validity characteristics of EPMS, there is little work on how to design valid environmental performance measures to provide managers more precise information enabling better environmentally sustainable decision making and control. I argue that there are two key reasons for this. First, from the MA research to date, there is still little work on integration between PMS and natural science when researchers define the concept of environmental sustainability and/or develop environmental sustainability measures (Bebbington & Thomson 2013; Lamberton 2005). This means MA researchers do not always have a good scientific understanding of the underlying environmental issues of the phenomena they are studying, and hence they do not know what should be measured in regard to environmental sustainability performance (e.g. Figge & Hahn 2013). Second, while environmental management typically occurs at an operational level in organisations – the level at which environmental sustainability issues arise and need to be managed – there has been little recognition of the connection between this and the organisational level where EPMS typically reside.<sup>121</sup> This means there is still little understanding about how the environmental issues at the operational level can inform decision making and control at the organisational level (Kolk & Mauser 2002).

To address this, I apply an interdisciplinary approach – an integration of water and crop science and accounting – to develop a theoretical discussion of water sustainability at the global and regional levels and apply it to an examination of crop water sustainability in a cotton production setting. Based on this, I develop a number of key water sustainability measures that capture crop water sustainability at an operational level, the level at which water sustainability issues related to cotton production occur and need to be managed.

<sup>&</sup>lt;sup>121</sup> While EPMS studies at the higher organizational levels provide aggregate information about environmental performance of an organization, for example resource use efficiency (Figge & Hahn 2013), such aggregate, high level information itself does not capture the underlying phenomenon being examined. In contrast, the examination of EPMS at the lower level, an operational level, will provide disaggregate and detailed information in relation to environmental sustainability and a better understanding of the actual drivers of the overall environmental performance of the organization.

Furthermore, I develop an organisational level summary measure (profit to water cost ratio), which incorporates information about both economic sustainability and environmental sustainability and can be used to assess the overall sustainability performance of a crop business.

Then, I take a decomposition ratio analysis approach to develop and articulate theoretical and logical links between measures at an operational level through to the organisational level and tie them together in a clear and structured way. By doing this, a theoretical connection between the organisational level summary measure and the operational level measures is established. This forms a new theoretical construction of EPMS – which I label the Water and Economic Sustainability Performance Measurement (WESM) model – comprising measures spanning across all levels of the organisation, providing guidance at an operational level, as well as informing and being informed by the organisational level.

The new designed WESM overcomes the two key problems with EPMS validity addressed in extant research. The first key problem is the missing link between science and MA research as identified by Bebbington & Thomson (2013) and Lamberton (2005), who call for more work addressing this issue. My study is one of the first to establish the theoretical links between crop and water science with PMS. The second key problem is the lack of connection between operational (low) and organisational (high) level measures (Kolk & Mauser 2002). In their review of existing environmental management research, Kolk and Mauser (2002) note that environmental management models are focused at a broad and conceptual level, and, despite the evidence of a close link between organisational activities at the operational level and an organisation's environmental performance, there is still lack of EPMS that can be operationalized at an organisation's operational level<sup>122</sup> (Kolk & Mauser 2002). My study is one of the first to establish the theoretical connection between organisational (high) level sustainability measures and operational (low) level sustainability measures in a hierarchical, logical and structured way. By addressing these two key problems in the context of crop production, I also contribute to the PMS literature by providing a new theoretical construct that overcomes the validity issue of PMS addressed by (Flamholtz, Das & Tsui 1985; Ittner & Larcker 2003; Nørreklit, Nørreklit & Israelsen 2006; Otley 1999).

<sup>&</sup>lt;sup>122</sup> Kolk and Mauser (2002) also raised a concern that despite several calls from environmental management researchers for more attention in this field in order to improve the insights into organizations' environmental performance measurement and management, this area of research has still been understudied – which continues to be the case.

This study is the first to develop a scientifically and hierarchically valid EPMS model for decision making and control for water and economic sustainability in broad acre agriculture. By providing the theoretical development of a new, valid, science-based model, I make a theoretical contribution of overcoming the validity issue of EPMS, one of the key issues of EPMS design that has been raised in the literature.

Furthermore, the research makes a methodological contribution to sustainability- related MA research by demonstrating a new approach – the interdisciplinary approach, an integration of environmental science and accounting – to studying EPMS to overcome the validity issue of EPMS design. This study is the first to incorporate water and crop production science into accounting research to develop theoretical explanations as to what water sustainability means at the global and regional levels. From this, a theoretical development of water and economic sustainability for crop businesses is provided as a foundation for the design of WESM. Other MA researchers who are interested in studying issues related to environmental sustainability may find the interdisciplinary approach pioneered in this study a useful guide.

#### 9.1.2 Target setting issue of EPMS (contribution 2)

The second issue of EPMS design addressed in this thesis is the target setting issue. Extant MA research has discussed the critical role of target setting in managing and directing organisational performance, and the problem of constructing valid targets (Ahmed & Rafiq 1998; Dekker, Groot & Schoute 2012; Elnathan, Lin & Young 1996; Figge & Hahn 2013; Ittner & Larcker 2001; Josée & Louis 2004). While there is a lot of discussion around target setting in the literature, there is little research around how to construct targets.

In the development of EPMS, issues related to appropriate target values for performance measures become an even greater challenge as decision makers have to deal with a more complex environment where economic and environmental performance aspects are simultaneously taken into consideration (Virtanen, Tuomaala & Pentti 2013). From the research to date, there is still lack of an all-encompassing and comprehensive approach to steer managerial decision making towards sustainability (Figge & Hahn 2013). Extant research provides little evidence how to develop environmental performance standards that can be used as *targets* to support managers in better sustainability-related decision making and control.

I find that simulation models, coupled with valid performance measures, enable the construction of standards, which may be used as the basis for targets. Specifically, I use the CROPGRO-Cotton embedded in the DSSAT system, a well-recognised and widely-used cotton production model (Thorp et al. 2014), to simulate a cotton cropping system to generate crop production and water data, which provides input data to the WESM model. Furthermore, I select furrow irrigation practice, the most common form of irrigation used in the cotton industry, as an irrigation method for the model, to examine the impact of irrigation management practice on sustainability performance of a crop business. In order to improve the quality of simulated information, I designed a new two-phased simulation modelling process to simulate furrow irrigation practice more accurately.

In addition, I implemented the designed simulation modelling process to simulate the current furrow irrigation practice. By doing this, I demonstrated that the theoretical WESM model works and was able to quantify a number of valid crop water and economic sustainability performance measures. In principle, these measures could be used for standard and target setting in decision making and control for researchers and farmers.

In order to demonstrate the utility of the designed WESM, I apply and test theory around how irrigation practice can be improved and what kind of impacts this has on environmental sustainability and economic performance. The novel two-phased simulation modelling approach enables a comparison between base line (current) furrow irrigation practice and alternative (improved) irrigation practice. I find statistically significant improvements in water use and economic outcomes. More specifically, the organisational level measure of profit to water cost ratio increases by 56%, operating profit increases by 30%, and 28% of water volume can be saved. The output of Scenario 2 could provide the basis for setting better targets (compared to using Scenario 1). Moreover, these results provide significant implications to the cotton industry (and potentially agriculture more broadly) with the potential to save hundreds of gigalitres of water and increase profitability by tens of millions of dollars per crop season for cotton farming in Australia.

By applying simulation modelling to quantify a range of valid crop water and economic sustainability performance measures presented in the designed WESM, I show how to overcome the challenge identified by Figge and Hahn (2013) and Virtanen, Tuomaala & Pentti (2013) of how targets can be set more rigorously in the context of EPMS. Furthermore,

this is one of the first studies to specifically focus on how to design and construct performance targets, addressing the critisim by (Bol et al. 2010; Dekker, Groot & Schoute 2012; Ittner & Larcker 2001), that this is an underresearched topic in the PMS literature more broadly.

In addition, this study is the first to develop a novel two-phased simulation modelling process to examine the impact of water management practice on water and economic sustainability performance of a crop business. By doing this, the thesis also explains how the quality of information can be improved and how the designed WESM can be used to support managers in better water and economic sustainability-related decision making and control. Further to this, I demonstrate the usefulness of simulation modelling as a research method which has not had a great deal of application in research beyond a few costing studies (Balachandran, Balakrishnan & Sivaramakrishnan 1997; Balakrishnan & Sivaramakrishnan 2001, 2002; Christensen & Demski 1995; Dhavale 2005; Hansen & Banker 2002; Kaplan & Thompson 1971; Labro & Vanhoucke 2007; Noreen & Burgstahler 1997). In this respect, this thesis makes a method contribution to sustainability- related management accounting research by demonstrating a new way to model the effect of various decision choices on the economic and environmental sustainability performance of a business, even when those decision choices have not been use in practice.

This research also provides a theoretical contribution to the accounting literature through the development and application of theory from science which has the potential to contribute to overcoming validity and target setting problems in PMS design; this further increases the explanatory power of our theory. To the author's knowledge, this is the first study to provide empirical evidence on a statistically, economically and environmentally significant difference between the modelled scenarios. This demonstrates that the designed WESM can be used to help decision-makers make better water and economic decisions. My approach reflects the sentiment expressed by Davis, Eisenhardt & Bingham (2007) and Harrison et al. (2007), that simulation is a useful method for theory development and theory testing when the examined phenomena involve longitudinal, nonlinear and/or complex processes, or when empirical data are challenging to obtain elsewhere – which is the study examined in this thesis<sup>123</sup>.

<sup>&</sup>lt;sup>123</sup> Since my simulation approach enables my work to make contribution to the subset literature of EPMS, it also makes contribution to the broader sustainability accounting literature.

Finally, this is the first study which provides evidence that a comprehensive WESM can be empirically derived, and hence may be utilised by growers in decision making and control. Practically, the results also have implications for the cotton industry as to how changing water management practices can improve economic and environmental sustainability performance at the industry level.

#### 9.1.3 Agriculture- related management accounting research (contribution 3)

While agriculture is economically and socially significant in meeting human needs for food and clothing, it is surprising that there has been very little management accounting research conducted within an agricultural setting and almost none on its role in environmental sustainability. This study addresses Jack's (2005, p. 60) contention that "*agricultural academics tend not to be interested in accounting and accounting researchers tend to stay clear of agriculture and related industries*". In addition, this study reflects the concern raised by Argilés and Slof (2001) that despite the important role of agriculture, accounting researchers have given it little attention, and hence there is little empirical research in the agricultural setting.

This thesis is the first study, to the author's knowledge in the management accounting literature, which incorporates water and crop science into accounting to develop EPMS model for the agricultural sector, specifically the cotton industry. The study has provided an insight into how management accounting has a key role to play in understanding how the sector may move towards water and economic sustainability. In doing so, a new avenue of investigation is opened up for studying EPMS in the agriculture sector more broadly, as discussed in Section 9.3.

#### 9.1.4 Novel EPMS model for agriculture (contribution 4)

The contribution to the agriculture literature relates to the novel theoretical framework of EPMS – the WESM model – developed and provided to crop agriculture research.

While the water use efficiency concept has been used in crop science research to describe how efficiently water can be converted into crop output, the literature surrounding efficiency of water used by plants is influenced by inconsistent terminology (Blum 2009; Cossani, Slafer & Savin 2012; Hochman, Holzworth & Hunt 2009; Igbadun et al. 2006; Liu-Kang & Hsiao 2004; Moore, Robertson & Routley 2011; Passioura 2006; Pereira, Cordery & Iacovides 2012; Tennakoon & Milroy 2003; van Halsema & Vincent 2012). More specifically, I find three issues related to crop water efficiency definitions in the literature: (i) different terminologies are used to express the concept of "water use efficiency" – the relationship between crop yield and crop water use – including WUE, CWP and CWUI; (ii) different water terms are used to define the denominator (e.g. water component) and/or the numerator (e.g. crop output) of water efficiency measures; (iii) the same WUE measure is used to express two different concepts of water efficiency. Accordingly, the concept of water use efficiency remains ambiguous and inconsistent.

Furthermore, since improving water use efficiency with regards to irrigation water has become an important issue in agricultural water management research and practice, new indicators have been developed in an attempt to capture the efficient use of irrigation (Cammarano et al. 2012; Pereira, Cordery & Iacovides 2012). While an irrigation measure (e.g. the WP<sub>Irrig</sub> indicator in Pereira et al.'s 2012 study) is proposed as a measure of how efficient irrigation water is used to generate economic yield, this measure, expressed as the ratio between total economic yield obtained by the crop business and irrigation amount, suffers a validity issue since the economic value of irrigation use is over-represented.<sup>124</sup> In addition, by presenting the comparisons of lint yield (kg/ha) response to irrigation amount (mm) among three different cotton locations with different rainfall, Cammarano et al. (2012) provide an invalid comparison and hence an invalid evaluation of economic outcomes of cotton production with respect to irrigation application across regions.<sup>125</sup>

In order to overcome the ambiguity, inconsistency and, most importantly, the validity issue of water use efficiency measurement in the extant literature (Cammarano et al. 2012; Igbadun et al. 2006; Pereira, Cordery & Iacovides 2012), I develop, based on water and crop science, a set of key water sustainability measures that more precisely capture water sustainability-related activities and management practices occurring at the crop production level, the level at which water sustainability issues arise and need to be managed. By doing this, I not only overcome the validity issue of how to measure water use efficiency, but also establish a theoretical connection between water coming into a cropping system and crop production coming out the cropping system.

<sup>&</sup>lt;sup>124</sup> Section 3.3.1.2 provides detailed discussion on this issue.

<sup>&</sup>lt;sup>125</sup> Similar to Pereira et al.'s 2012 study, in Cammarano et al.'s study, the contribution of rainfall to the economic outcomes of the crop is not separated from that of irrigation.

In addition, I incorporate an accounting perspective in the discussion of crop water and economic sustainability to develop a summary measure of water and economic sustainability performance of a crop business (profit to water cost ratio) and its building block (the WESM model). This links the summary measure at the organisational level to its drivers across all levels of the organisation and in particular to the operational level. To the author's knowledge, this study is the first study to provide a theoretical development of a set of valid water sustainability measures at the operational level, and the designed WESM is the first EPMS to provide a comprehensive conceptual framework for crop water and economic sustainability performance. This can be used to support better sustainability- related decision making and control for crop agriculture.

Moreover, the new WESM model uses a novel two-phased crop simulation modelling process to enable the development of more valid and precise standards for crop businesses' economic and environmental performance measures. In this respect, I provide a new tool to allow crop businesses to evaluate their actual performance against a target and identify areas for improvement in water and economic sustainability performance. The model also has utility at the regional level where the value of the summary sustainability measure of individual crops will signal how profitable a crop is in respect to water resource use compared to other crops grown in the same region. This will support policy makers in relevant decisions related to water allocation under water constraint.

Finally, by designing the novel two-phased crop simulation modelling process, I provide the crop simulation modelling literature a new approach to improve the quality of simulated information and establish an end-to-end connection between irrigation strategies and their economic and environmental outcomes. Both issues have not been examined in the literature to date (Jones et al. 2003; Keating et al. 2003; Smith, Raine & Minkevich 2005; Wang et al. 2002)

## 9.2 Implications of the research

This section discusses the implications of the research findings. There are two key practical implications from this research. First, the novel WESM model developed is a new tool which practitioners can use for on-farm decision making and control for water and economic

sustainability management. Second, it is empirically evident that moving from current to improved practice has industry-level environmental and economic significance.

#### 9.2.1 Use of new tool

There are two key components of the novel WESM model developed. First, the WESM model provides a theoretical framework which practitioners may use in implementing improved EPMS. In addition, the developed crop water use index can enable crop managers and farmers to identify areas for improved sustainability performance.<sup>126</sup> The sustainability measures developed in this thesis can be translated into guidance for crop managers to enable them to improve their economic and environmental performance, including identifying areas for improved sustainability performance, and making appropriate/optimal operational decisions to achieve better economic and environmental performance.

Second, this thesis provides cotton growers and other industry's stakeholders (including crop agronomists, crop researchers, industry partners) a simulation tool to enable them:

- Identify strategies to improve the environmental and economic performance under furrow irrigation systems, such as by forecasting and using 'what if' scenario analysis. For example, test and experiment with changes in furrow irrigation as demonstrated in Chapter 8.
- b. Test and experiment changes in irrigation practice, such as sub-surface irrigation.
- c. Estimate and select targets to work towards environmental sustainability.
- d. Utilise diagnostic control through variance analysis, where simulated performance (budget) is compared with actual output.
- e. Construct improved information to support capital investment decisions regarding irrigation systems.

In summary, the developed WESM incorporating the two-phased simulation modelling process provides crop managers a tool for risk management and the identification of new

<sup>&</sup>lt;sup>126</sup> For example, when a farmer looks at the crop water use index, if the crop water use index in respect to water input efficiency turns out to be poor, irrigation management needs to be examined. On the other hand, if the harvest index is poor (i.e. very low compared to the theoretical value), it suggests that things such as pest control and disease control need to be better managed. Alternatively, if the transpiration index is poor, this means the farmer needs to improve nitrogen management practices. The detailed discussion of crop water use index with the constructed decision tree is provided in Section 3.3.3.

possibilities and potentials in farming, in particular improvements in water and economic sustainability.

## 9.2.2 Economic and environmental significance

In this section I demonstrate that moving from current to improved practice has industry level environmental and economic significance. To do this I provide a naive estimate of the potential effect if Scenario 2 was adopted by applying the results from Chapter 8 (calculated on a per ha basis) to a typical 400 ha farm, and also a conservative estimate of the industry size. For the purpose of this analysis, I use the average irrigated area for the Australian cotton industry from 2010–2011 crop season to the 2013–2014 season, based on data from the Australian Bureau of Statistics (2015) f 365,365 hectares (see Appendix E, Table E1) for further information of industry size from different sources). The environmental and economic gains from moving from Scenario 1 to Scenario 2 at both business level and industry level are summarised in Appendix E, Table E2.

#### Potential environmental gain:

At the farm level, it is evident in this thesis that by simply increasing furrow inflow from 2.80 L/s to 6.0 L/s and reducing irrigation time (time of cut-off) from 745 minutes to 310 minutes, individual cotton farms with a size of 400 irrigated hectares can save up to 860 ML of water, produce an additional 550 bales of lint, and gain an additional operating profit of nearly \$139,600 per crop season (given a lint price of \$474/bale, seed price of \$0.32/kg, and licensed water cost of \$60.3/ML).

At the industry level, it is reported that, in respect to environmentally significance, Scenario 1 results in a net water loss of 2.15 ML per ha, 860 ML per business, or nearly 786 GL in total at the industry level. This means by simply moving from the current water management practice (Scenario 1) to the improved water management practice (Scenario 2), the cotton industry as a whole can save up to 786 GL per crop season (i.e. every 6 months).

#### Potential crop production gain:

The amount of crop water saved could be potentially used to grow more cotton<sup>127</sup>. Given that the crop water use index for Scenario 2 is 144 kg of lint per ML, an additional of 310 kg of

<sup>&</sup>lt;sup>127</sup> However, this is a huge debate. Environmentalists argue that the water should be returned as environmental flows. The industry wants to grow more cotton. The government wants more taxes.

lint is produced per ha as a result of 2.15 ML saved per ha (i.e. 2.15 ML/ha x 144 kg/ML = 310 kg/ha). This implies that when scaled up to business size and industry size, an additional 124,000 kg of lint (equivalent to 546 bales of lint) per business or an addition of 113,263,150 kg of lint across the industry (equivalent to 498,957 bales of lint), can be produced.

Furthermore, as a result of a slight increase in lint yield (e.g. 2.6 kg/ha), an additional 4.6 bales of lint can be produced at the business level, equivalent to 4,185 bales of lint at the industry level. In total, by simply moving from the current water management practice to the improved water management practice, a potential gain of 125,040 kg of lint (equivalent to 550 bales of lint) at the business level or 114,213,099 kg (equivalent to 503,141 bales of lint) at the industry level can be achieved.

#### Potential economic gain:

In terms of operating profit, by simply improving water management practices (e.g. from S1 to S2), a crop business can gain an additional profit of \$217/ha from additional water saved of 2.15 ML//ha, giving a return on water for S2 of \$101/ha (e.g. 2.15 ML/ha x \$101/ML = \$217/ha). This translates to additional profit of \$86,800 at the business level or \$79.3 million at the industry level can be achieved.

Moreover, as a result of the reduction in water costs when switching from Scenario 1 to Scenario 2, operating profit is reported to increase by \$132/ha – equivalent to \$52,800 (at the business level) and \$48.2 million (at the industry level). In total, a potential gain of \$94,000 operating profit at the business level or \$85,860,775 operating profit at the industry level can be achieved.

In summary, by simply moving from current to improved water management practices an average cotton business, with 400 irrigated hectares, can save up to 860ML of water, produce an additional 550 bales of lint, and gain an additional operating profit of nearly \$140,000 per crop season (given a lint price of \$474/bale, seed price of \$0.32/kg, and water licensed cost of \$60.3/ML). For a crop business, these figures will be increased by the increase in size of irrigated land. At the industry level, these figures are aggregated to an additional 786 GL of

water saved, an additional 503,141 bales of lint produced, and an additional operating profit of about \$127.5 million gained per crop season.

## 9.3 Limitations, delineations and future research

Discussions of limitations and delineations of the research, and how the findings of the research are able to be extended in future research, are presented in this section.

#### 9.3.1 Limitations and future research

One of the core contributions of this research is to provide empirical evidence on statistical, economic and environmental differences between the current management practice and the improved management practice with regard to water resource use in the cotton industry. However, these results have a set of limitations associated with it, which suggest avenues for future research.

First, it is assumed that economic factors such as lint price, cotton seed price, and unit licensed irrigation cost remain constant across 86 years of study and are based on the data of a sample budget for furrow irrigated cotton in 2014–2015 for the Central and Northern NSW area (NSW Government - Department of Primary Industries 2015). Since the analysis of economic and environmental implications of moving from Scenario 1 to Scenario 2 use the average crop water and production data over 86 years for each scenario, this assumption would impact the results of the economic significance from the analysis. Future research can explore the generalisability and sensitivity of the results to variations in economic factors.

Second, the irrigation scheduling set in DSSAT for both Scenario 1 and Scenario 2 do not favour Scenario 2 more than Scenario 1 since it is shown that, on average, the crops did not grow any better in Scenario 2. Furthermore, the way the irrigation scheduling is set is based on the assumption that in both cases crops grow under non-limiting irrigation supply. In practice, irrigation supply could be a challenge, particularly in low rainfall years; for example, in Scenario 1 27% more of irrigation needs to be abstracted compared to Scenario 2. Future research could refine irrigation scheduling to allow better economic yield produced under Scenario 2 conditions and place greater emphasis on irrigation-related risk management by moving to Scenario 2.

Third, it is noted that on average lint yield produced under both scenarios is around 8 bales/ha, which is lower than what can be obtained in practice (e.g. 10 bales/ha). Since the analysis conducted in the research is comparative analysis, this does not detract from the significance of the findings from the study. Since both SIRMOD and DSSAT software is calibrated and validated, and widely used internationally for water and crop production research (Gillies & Smith 2005; Hoogenboom et al. 2015; Jones et al. 2003; Smith, Raine & Minkevich 2005; Walker 2003), it is evident that these two software packages are reliable for undertaking simulation modelling studies. However, ongoing research is necessary to further calibrate these models in this respect.

Fourth, regarding irrigation allocation, it is assumed that the amount of irrigation abstracted in both scenarios is merely sourced by licensed irrigation. In practice, crop businesses usually need to source irrigation from both licensed and traded irrigation. In this respect, the impact of irrigation allocation on weighted irrigation cost particularly, and on the overall economic and environmental performance of a crop business generally, is not accounted for in this research. Future research could examine various alternatives of irrigation allocation to gain more insights into the impact of irrigation allocation of crop businesses' sustainability performance.

Fifth, when accounting for water cost, it is assumed that rain is a "free" water resource and recycling costs are equal to zero. In future research, costs related to capture and storage of rain and recycling costs need to be taken into consideration and be included in the denominator of profit to water cost ratio – total water cost.

Finally, while it is expected that fibre quality varies along the furrow in Scenario 1, and between Scenario 1 and Scenario 2, as a result of differences in irrigation distribution within and between the two scenarios, the lint price is assumed to be the same in the comparative analysis in Chapter 8. Future research could consider changes in lint price due to varied fibre quality.

While these limitations of the research are acknowledged, they do not devalue the significance of the findings from the study. In addition, this lays a basis to extend the research in future.

#### 9.3.2 Delineations and future research

There are a number of choices I have made which delineate the scope of the research. First, I deal with water only, not energy or other sustainability issues. Second, I deal with agriculture, not other industry settings such as manufacturing. Third, cotton is examined, not other broad acre crops or other forms of agriculture. Fourth, the geographic location of the simulation model (i.e. Myall Vale), soil type (i.e. clay vertosol) and other choices (i.e. nitrogen application, plant density) were made in developing this model. Finally, I only consider furrow irrigation, and consider only two furrow irrigation strategies (two scenarios). While future research may explore these issues, in this section I will identify a number of potential avenues to expand the scope of the research.

First, while I use water as an environmental resource to be examined in this thesis, the research approach (i.e. interdisciplinary approach), the simulation method, and the hierarchical WESM model developed in this thesis can be expanded to incorporate other key resources; for example, energy and soil. In addition, further development of the WESM model designed in this thesis can be undertaken to enable soil sustainability and energy sustainability aspects to be included into the WESM, as well as linkages between soil, energy and water sustainability to be made, to provide an integrated and comprehensive decision tool that can be used to support crop businesses in making better sustainability- related decisions

Second, while the empirical setting considered in this thesis is agriculture, and the new EPMS – the WESM model – is designed specifically for crop agriculture, the application of simulation modelling for target setting and supporting decision making can be used as a research method in other settings where science related to production activity can be drawn upon. This includes studying energy sustainability- related PMS design and target construction in the manufacturing setting to expand the work of Virtanen, Tuomaala & Pentti (2013).

Third, while cotton is selected as the empirical crop in this thesis, the findings have implications for other cropping systems. The scientifically and hierarchically valid WESM model, designed for cotton, could be expanded and the two-phased simulation modelling process developed in this thesis could be applied to other crops, for example wheat, soybean, and rice, in order to support other crop industries and crop agriculture as a whole to improve sustainability- related decision making and control.

Fourth, while only one Australian cotton area, Myall Vale, and one soil type, clay vertosol, is considered in this thesis, this study could be expanded to replicate the designed WESM model, coupled with the two-phased simulation modelling process, to examine other cotton regions and their associated soil type across Australia, in order to examine the effect of cotton locations and weather conditions on water and economic sustainability of crop businesses. Other forms of crop management, including nitrogen application, could also be examined to identify the optimal crop management conditions with respect to water use and crop yield.

Finally, in relation to furrow irrigation strategies and target setting, there are a number of ways the empirical work of this thesis can be extended. For example, refining the optimal furrow irrigation management under Scenario 2 conditions can be undertaken through a process of optimisation using simulation modelling (e.g. via SIRMOD and DSSAT) in order to improve/optimise furrow irrigation performance and irrigation scheduling so that crop yield can be further improved/maximised while irrigation applied can be further reduced/minimised. As a result, it is expected that higher profit to water cost ratio and further economic and environmental gains at both business and industry levels can be achieved, compared to the current management practice described in the Scenario 1.

However, even with a more refined/optimal furrow irrigation management, under furrow irrigation systems, the amount of water loss to soil evaporation is still substantial. The problem is that the large residual soil evaporation cannot be reduced while the irrigation process results in a wet soil surface. This means water input efficiency is still far from the target value of 1.0 when furrow irrigation systems are in use. In this respect, future research can consider other irrigation systems in order to reduce/minimise water loss from soil evaporation (while still maintaining the low level of deep drainage and further reducing runoff from irrigation described in Scenario 2) so that water input efficiency can be improved. This will provide both economic and environmental implications for crop businesses and the industry as a whole, since less water will be lost to the environment and more water can be saved to grow more cotton. With current technology, sprinkler irrigation systems and sub-surface drip irrigation systems can be considered as a replacement of furrow irrigation systems in the process of improving the sustainability performance with regard to water sustainability. However, it is expected that sprinkler irrigation systems may not improve water input efficiency since while there would be zero irrigation runoff to recycle

and there would be equally small deep drainage (compared to Scenario 2), the soil surface will still be wet, and possibly more continuously. In contrast, it is hoped that sub-surface drip irrigation systems can provide water with no surface wetting from irrigation, and hence can improve water input efficiency and water sustainability for crop businesses and the whole industry. While it is recommended to explore the potential of sub-surface drip irrigation systems in improving water sustainability performance, capital investments as well as energy cost associated with system operation need to be taken into consideration. Though a sub-surface drip irrigation system could result in some increase in energy cost (and system depreciation), it is hoped that the value of the increase in economic yield and decrease in water cost exceeds the additional costs incurred when shifting from furrow irrigation systems to sub-surface drip irrigation system.

Another way to extend the target setting research, which also relates to the two points discussed above, is to develop staged sustainability targets which include reference to technology used in crop cultivation. Setting sustainability performance targets play an important role in providing guidance and goals to assist a manager in identifying what the business should aim for in the long-term, and directing organisational change to move towards sustainability. This research has laid out Stage 1 and Stage 2 sustainability targets which result from the application of the current furrow irrigation practice and the improved furrow irrigation practice, respectively. While this explicates how a crop business can move from the status quo to a more sustainable performance, it is shown that the improved management practice examined in this research is still sub-optimal and hence the question of how the crop business can move towards sustainability still remains open. Future research can undertake Stage 3 and Stage 4 sustainability targets which are based on the application of an optimal furrow irrigation practice and a sub-surface drip irrigation system, respectively.

## 9.4 Conclusion

This piece of research has been undertaken to answer the question: how can Environmental Performance Measurement Systems (EPMS) be *designed* and *used* in an agricultural setting to support managers in water and economic sustainability-related decision making and control.

This research is initially stimulated by the concept of sustainable development introduced by the Brundtland Commission as "*development that meets the needs of the present without compromising the ability of future generations to meet their own needs*" (The Brundtland Commission 1987, p. 1). It is evidenced that some core activities at the organisation's operational level are roots of a number of environmental problems at the global and regional levels. Therefore, in order to enable future generations to meet their own needs, at present, organisations in many industries, such as agriculture, manufacturing or petroleum industries, need to operate in a more efficient and more sustainable manner so that the use of natural resources can be minimised and discharges to the natural environment can be mitigated.

The agricultural sector, a significant user of water, has a significant impact on the sustainability of freshwater at both global and local levels. On the other hand, agriculture is economically and socially significant in meeting human needs for food and clothing. Therefore, improvement of environmental (water) and economic sustainability of crop businesses is of crucial importance for society to become more sustainable. Management accounting researchers can contribute to crop agriculture and to broader society by applying PMS to sustainability issues to construct more valid and comprehensive models to support crop businesses in water and economic sustainability- related decision making and control.

Within these broad issues, this piece of research has taken initial and tentative steps to address more overtly than previous research the design of EPMS in an agricultural setting in two ways.

First, having undertaken an interdisciplinary approach – an integration of water and crop science and accounting, and a decomposition ration analysis approach, an accounting technique to enable the articulation of the theoretical and logical links between measures at the operational level through to the organisational level – I construct a theoretical EPMS model (WESM model) that has the potential to overcome validity issue in EPMS design. By doing this, the WESM model enables capture of water sustainability-related activities and management practices occurring at the operational level. This will provide the organisational level with more precise information to support economic and environmental sustainability decision making and control.

Second, the study examines the usefulness of the WESM model by designing a two-phase crop production simulation modelling approach, considering its application to the furrow irrigation practice (Scenario 1) and the improved furrow irrigation practice (Scenario 2). In doing so, I demonstrate that the theoretical model works and explains how targets can be set more rigrously in the context of EPMS. In this respect, the study lays the foundation for opening up the idea of consutructing targets based on simulation modelling and providing an approach to setting tagets to steer managerial decision making towards sustainability.

The study also provides economically and environmentally significant implications for the Australian cotton industry – the empirical setting considered in the thesis (and agriculture more broadly). By simply moving from the current to the improved water management practices, the industry can potentially save hundreds of gigalitres of water and increase profitability by tens of millions of dollars per crop season.

By developing a new theoretical model that potentially overcomes the validity and target setting issues of EPMS to support crop managers in water and economic sustainability-related decision making and control, the study lays the groundwork for future research in EPMS design and use in increasing depth and scope.

# References

- Abbott, A.M., Higgerson, G.J., Long, R.L., Lucas, S.R., Naylor, G.R.S., Tischler, C.R. & Purmalis, M.M. 2010, 'An Instrument for Determining the Average Fiber Linear Density (Fineness) of Cotton Lint Samples', *Textile Research Journal*, vol. 80, no. 9, pp. 822–33.
- Ahmed, P.K. & Rafiq, M. 1998, 'Integrated benchmarking: a holistic examination of select techniques for benchmarking analysis', *Benchmarking for Quality Management & Technology*, vol. 5 no. 3, pp. 225-42.
- Al-Tuwaijri, S.A., Christensen, T.E. & Hughes Ii, K.E. 2004, 'The relations among environmental disclosure, environmental performance, and economic performance: a simultaneous equations approach', *Accounting, Organizations and Society*, vol. 29, no. 5–6, pp. 447-71.
- Alexandratos, N. & Bruinsma, J. 2012, *World agriculture towards 2030/2050: the 2012 revision*, Food and Agriculture Organization of the United Nations (FAO).
- Anderson, S.W., Dekker, H.C. & Sedatole, K.L. 2010, 'An Empirical Examination of Goals and Performance-to-Goal Following the Introduction of an Incentive Bonus Plan with Participative Goal Setting', *Management Science*, vol. 56, no. 1, pp. 90-109.
- Aranda, C., Arellano, J. & Davila, A. 2014, 'Ratcheting and the Role of Relative Target Setting', *Accounting Review*, vol. 89, no. 4, pp. 1197-226.
- Argilés, J.M. & Slof, E.J. 2001, 'New opportunities for farm accounting', *European* Accounting Review, vol. 10, no. 2, pp. 361-83.
- Arnold, M.C. & Artz, M. 2015, 'Target difficulty, target flexibility, and firm performance: Evidence from business units' targets', *Accounting, Organizations and Society*, vol. 40, pp. 61-77.
- Artz, M., Homburg, C. & Rajab, T. 2012, 'Performance-measurement system design and functional strategic decision influence: The role of performance-measure properties', *Accounting, Organizations and Society*, vol. 37, no. 7, pp. 445-60.
- Atkinson, A.A., Balakrishnan, R., Booth, P., Cote, J.M., Groot, T., Malmi, T., Roberts, H., Enrico, Uliana & Wu, A. 1997, 'New Directions in Management Accounting Research', *Journal of Management Accounting Research*, vol. 9, pp. 79-108.
- Australasian Legal Information Institute 2014, *Water Sharing Plan For The NSW Border Rivers Unregulated And Alluvial Water Sources 2012*, Australasian Legal Information Institute, NSW, Australia, viewed 15th January 2014, <<u>http://www.austlii.edu.au/au/legis/nsw/consol\_reg/wspftnbruaaws2012800/s44.html</u> >.
- Australian Bureau of Statistics 2015, 7121.0 Agricultural Commodities, Australia, 2014-15, viewed 15 December, <<u>http://www.abs.gov.au/ausstats/abs@.nsf/mf/7121.0></u>.
- Avissar, R., Avissar, P., Mahrer, Y. & Bravdo, B.A. 1985, 'A model to simulate response of plant stomata to environmental conditions', *Agricultural and Forest Meteorology*, vol. 34, no. 1, pp. 21-9.
- Azapagic, A. & Perdan, S. 2000, 'Indicators of Sustainable Development for Industry: A General Framework', *Process Safety and Environmental Protection*, vol. 78, no. 4, pp. 243-61.
- Baillie, C. 2009, *Energy and Carbon Accounting Case Study (Keytah)*, National Centre for Engineering in Agriculture Publication 1003254/1, USQ, Toowoomba.
- Baillie, C. & Chen, G. 2008, 'Reducing energy input costs and associated greenhouse gas emissions in cotton', *14th Australian Cotton Conference*, Gold Coast.

- Balachandran, B.V., Balakrishnan, R. & Sivaramakrishnan, K. 1997, 'On the Efficiency of Cost-Based Decision Rules for Capacity Planning. (cover story)', *Accounting Review*, vol. 72, no. 4, p. 599.
- Balakrishnan, R. & Sivaramakrishnan, K. 2001, 'Sequential Solutions to Capacity-Planning and Pricing Decisions', *Contemporary Accounting Research*, vol. 18, no. 1, pp. 1-26.
- Balakrishnan, R. & Sivaramakrishnan, K. 2002, 'A Critical Overview of the Use of Full-Cost Data for Planning and Pricing', *Journal of Management Accounting Research*, vol. 14, p. 3.
- Bange, M. & Constable, G. 2006, 'Crop physiology producing a better fibre', paper presented to the *Product, Production, Profit - Progressing our Natural Advantage, Proceedings 13th Australian Cotton Growers Conference* Gold Coast, Australia, 8-10 August.
- Bange, M.P., Long, R.L., Constable, G.A. & Gordon, S.G. 2010, 'Minimizing Immature Fiber and Neps in Upland Cotton', *Agronomy journal*., vol. 102, no. 2, pp. 781-9.
- Barney, J. 1991, 'Firm Resources and Sustained Competitive Advantage', *Journal of Management*, vol. 17, no. 1, p. 99.
- Barth, J.A.C., Filimonau, V., Bayer, P., Struckmeier, W. & Grathwohl, P. 2010, 'Global Water Balance', in T.E. Graedel & E. van der Voet (eds), *Linkages of Sustainability*, The MIT Press, Cambridge, England, pp. 221-32.
- Bartolomeo, M., Bennett, M., Bouma, J.J., Heydkamp, P., James, P. & Wolters, T. 2000, 'Environmental management accounting in Europe: current practice and future potential', *European Accounting Review*, vol. 9, no. 1, pp. 31-52.
- Bebbington, J. & Larrinaga, C. 2014, 'Accounting and sustainable development: An exploration', *Accounting, Organizations and Society*, vol. 39, no. 6, pp. 395-413.
- Bebbington, J. & Thomson, I. 2013, 'Sustainable development, management and accounting: Boundary crossing', *Management Accounting Research*, vol. 24, no. 4, pp. 277-83.
- Bell, S. & Quiggin, J. 2008, 'The limits of markets: the politics of water management in rural Australia', *Environmental Politics*, vol. 17, no. 5, pp. 712-29.
- Bhargava, M., Dubelaar, C. & Ramaswami, S. 1994, 'Reconciling diverse measures of performance: A conceptual framework and test of a methodology', *Journal of Business Research*, vol. 31, no. 2–3, pp. 235-46.
- Bhargava, M., Dubelaar, C. & Scott, T. 1998, 'Predicting bankruptcy in the retail sector: an examination of the validity of key measures of performance', *Journal of Retailing and Consumer Services*, vol. 5, no. 2, pp. 105-17.
- Blum, A. 2009, 'Effective use of water (EUW) and not water-use efficiency (WUE) is the target of crop yield improvement under drought stress', *Field Crops Research*, vol. 112, no. 2–3, pp. 119-23.
- Boiral, O. & Sala, J.-M., 'Environmental management: Should industry adopt ISO 14001?', *Business Horizons*, vol. 41, no. 1, pp. 57-64.
- Bol, J.C., Keune, T.M., Matsumura, E.M. & Jae Yong, S. 2010, 'Supervisor Discretion in Target Setting: An Empirical Investigation', *Accounting Review*, vol. 85, no. 6, pp. 1861-86.
- Bone, J., Head, M., Barraclough, D., Archer, M., Scheib, C., Flight, D. & Voulvoulis, N. 2010, 'Soil quality assessment under emerging regulatory requirements', *Environment International*, vol. 36, no. 6, pp. 609-22.
- Bourne, M., Mills, J., Wilcox, M., Neely, A. & Platts, K. 2000, 'Designing, implementing and updating performance measurement systems', *International Journal of Operations & Production Management*, vol. 20, no. 7, pp. 754-71.
- Boyce Chartered Accountants 2012, *Australian Cotton Comparative Analysis 2012 Crop*, Cotton Research and Development Corporation.

- Braunack, M.V., Bange, M.P. & Johnston, D.B. 2012, 'Can planting date and cultivar selection improve resource use efficiency of cotton systems?', *Field Crops Research*, vol. 137, pp. 1-11.
- Brix-Asala, C., Hahn, R. & Seuring, S. 2016, 'Reverse logistics and informal valorisation at the Base of the Pyramid: A case study on sustainability synergies and trade-offs', *European Management Journal*, vol. 34, no. 4, pp. 414-23.
- Brown, D.A. & Sundin, H. 2016, '(Forthcoming) Greening the black box: Integrating the environment and management control systems', *Accounting, Auditing and Accountability Journal*.
- Bucks, D., Nakayama, F. & Warrick, A. 1982, 'Principles, practices, and potentialities of trickle (drip) irrigation', *Advances in irrigation*.
- Burritt & Schaltegger 2001, 'Eco-efficiency in corporate budgeting', *Environmental Management and Health*, vol. 12, no. 2, pp. 158-74.
- Burritt, R.L., Hahn, T. & Schaltegger, S. 2002, 'Towards a Comprehensive Framework for Environmental Management Accounting — Links Between Business Actors and Environmental Management Accounting Tools', *Australian Accounting Review*, vol. 12, no. 27, pp. 39-50.
- Burritt, R.L. & Saka, C. 2006, 'Environmental management accounting applications and ecoefficiency: case studies from Japan', *Journal of Cleaner Production*, vol. 14, no. 14, pp. 1262-75.
- Burritt, R.L. & Schaltegger, S. 2010, 'Sustainability accounting and reporting: Fad or trend?', *Accounting, Auditing and Accountability Journal*, vol. 23, no. 7, pp. 829-46.
- Cai, X., Yang, Y.-C.E., Ringler, C., Zhao, J. & You, L. 2011, 'Agricultural water productivity assessment for the Yellow River Basin', *Agricultural Water Management*, vol. 98, no. 8, pp. 1297-306.
- Cammarano, D., Payero, J., Basso, B., Wilkens, P. & Grace, P. 2012, 'Agronomic and economic evaluation of irrigation strategies on cotton lint yield in Australia', *Crop & Pasture Science*, vol. 63, pp. 647–55.
- Cassman, K.G., Dobermann, A., Walters, D.T. & Yang, H.S. 2003, 'Meeting cereal demand while protecting natural resources and improving environmental quality', *Annual Review of Environment and Resources*, vol. 28, pp. 315–58.
- Chalmers, K., Godfrey, J.M. & Lynch, B. 2012, 'Regulatory theory insights into the past, present and future of general purpose water accounting standard setting', *Accounting, Auditing & Accountability Journal*, vol. 25, no. 6, pp. 1001-24.
- Chapagain, A.K., Hoekstra, A.Y., Savenije, H.H.G. & Gautam, R. 2006, 'The water footprint of cotton consumption: An assessment of the impact of worldwide consumption of cotton products on the water resources in the cotton producing countries', *Ecological Economics*, vol. 60, no. 1, pp. 186-203.
- Cheeroo-Nayamuth, F.C., Robertson, M.J., Wegener, M.K. & Nayamuth, A.R.H. 2000, 'Using a simulation model to assess potential and attainable sugar cane yield in Mauritius', *Field Crops Research*, vol. 66, no. 3, pp. 225-43.
- Chen, G. & Baillie, C. 2009, 'Development of a framework and tool to asses on-farm energy uses of cotton production', *Energy Conversion and Management*, vol. 50, no. 5, pp. 1256-63.
- Chenoweth, J., Hadjikakou, M. & Zoumides, C. 2013, 'Review article: Quantifying the human impact on water resources: a critical review of the water footprint concept', *Hydrology & Earth System Sciences Discussions*, vol. 10, no. 7, pp. 9389-433.
- Christensen, J. & Demski, J.S. 1995, 'The classical foundations of 'modern' costing', *Management Accounting Research*, vol. 6, no. 1, pp. 13-32.

Clarkson, P.M., Li, Y., Richardson, G.D. & Vasvari, F.P. 2008, 'Revisiting the relation between environmental performance and environmental disclosure: An empirical analysis', *Accounting, Organizations and Society*, vol. 33, no. 4–5, pp. 303-27.

- Conforti, P. 2011, *Looking ahead in world food and agriculture: perspectives to 2050*, Food and Agriculture Organization of the United Nations (FAO).
- Cossani, C.M., Slafer, G.A. & Savin, R. 2012, 'Nitrogen and water use efficiencies of wheat and barley under a Mediterranean environment in Catalonia', *Field Crops Research*, vol. 128, no. 0, pp. 109-18.
- Cotton Australia 2008, Annual Report 2007-2008, Cotton Australia, Sydney.
- Cotton Australia 2015, *Australian cotton production statistics* viewed 15 December, <<u>http://cottonaustralia.com.au/cotton-library/statistics</u>.
- Cotton Seed Distributors 2013, *Turn Out Percentage Factors Involved*, viewed 15 December 2015,

<<u>http://www.csd.net.au/system/fofs/attachments/000/000/218/original/FOF260413.pd</u> <u>f?1398921936></u>.

- Davis, J.P., Eisenhardt, K.M. & Bingham, C.B. 2007, 'Developing theory through simulation methods', *Academy of Management Review*, vol. 32, no. 2, pp. 480-99.
- de Oliveira, O.J., Serra, J.R. & Salgado, M.H., 'Does ISO 14001 work in Brazil?', *Journal of Cleaner Production*, vol. In Press, Corrected Proof.
- De Wever, H. 2010, 'Interactions of the Water Cycle with Energy, Material Resources, Greenhouse Gas Production, and Land Use', in T.E. Graedel & E. van der Voet (eds), *Linkages of Sustainability*, The MIT Press, Cambridge, England, pp. 243-65.
- Dekker, H.C., Groot, T. & Schoute, M. 2012, 'Determining Performance Targets', *Behavioral Research in Accounting*, vol. 24, no. 2, pp. 21-46.
- Department of Environment and Primary Industries 2014, *Trading Water Licences for Groundwater and Unregulated Surface Water in Southern Victoria*, Department of Environment and Primary Industries, Victoria, Australia, viewed 15 January 2014, <<u>http://www.dpi.vic.gov.au/agriculture/farming-management/soil-</u> water/water/solutions/technical-resources/trading-water-licences>.
- Department of Natural Resources and Mines 2014, *Licensing of water bore drillers*, Queensland Goevernment, Queensland, Australia, viewed 15th January 2014, <<u>http://www.dnrm.qld.gov.au/water/access/bores/licensing-bore-drillers?SQ\_DESIGN\_NAME=print-guide></u>.
- Department of Science- Information-Technology Innovation and the Arts 2014, *SILO Climate Data*, Department of Science, Information Technology, Innovation and the Arts, Queensland, <<u>http://www.longpaddock.qld.gov.au/silo/</u>>.
- Department of the Environment 2014, *Water entitlement purchasing in the Murray-Darling Basin*, Australian Government, viewed 28th January 2014, <<u>http://www.environment.gov.au/topics/water/rural-water/restoring-balance-murray-darling-basin></u>.
- Dhavale, D.G. 2005, 'Cost Considerations in Optimal Capacity Acquisition: An Option Pricing Approach', *Journal of Management Accounting Research*, vol. 17, pp. 75-93.
- Dias, M.A.d.P., Vianna, J.N.d.S. & Felby, C. 2016, 'Sustainability in the prospective scenarios methods: A case study of scenarios for biodiesel industry in Brazil, for 2030', *Futures*, vol. 82, pp. 1-14.
- Dubelaar, C., Bhargava, M. & Ferrarin, D. 2002, 'Measuring retail productivity: what really matters?', *Journal of Business Research*, vol. 55, no. 5, pp. 417-26.
- Elkington, J. 1997, Cannibals with Forks: The Tripple Bottom Line of 21st Century Business, Capstone, Oxford.

- Elnathan, D., Lin, T.W. & Young, S.M. 1996, 'Benchmarking and Management Accounting: A Framework for Research', *Journal of Management Accounting Research*, vol. 8, pp. 37-54.
- Epstein, M.J. 1996, 'You've got a great environmental strategy—Now what?', *Business Horizons*, vol. 39, no. 5, pp. 53-9.
- Epstein, M.J. & Roy, M.-J. 2001, 'Sustainability in Action: Identifying and Measuring the Key Performance Drivers', *Long Range Planning*, vol. 34, no. 5, pp. 585-604.
- FAO 2013, *FAO Statistical Yearbook 2013: World food and agriculture*, Food and Agriculture Organization of the United Nations (FAO).
- Farrell, A. & Hart, M. 1998, 'What does sustainability really mean? The search for useful indicators', *Environment*, vol. 40, no. 9, p. 4.
- Ferreira, A. & Otley, D. 2009, 'The design and use of performance management systems: An extended framework for analysis', *Management Accounting Research*, vol. 20, no. 4, pp. 263-82.
- Figge, F. & Hahn, T. 2013, 'Value drivers of corporate eco-efficiency: Management accounting information for the efficient use of environmental resources', *Management Accounting Research*, vol. 24, no. 4, pp. 387-400.
- Fisher, J. 1995, 'Contingency-based research on management control systems: categorization by level of complexity', *Journal of accounting literature*, vol. 14, p. 24.
- Fisher, J.G., Peffer, S.A., Sprinkle, G.B. & Williamson, M.G. 2015, 'Performance Target Levels and Effort: Reciprocity across Single- and Repeated-Interaction Settings', *Journal of Management Accounting Research*, vol. 27, no. 2, pp. 145-64.
- Flamholtz, E.G., Das, T.K. & Tsui, A.S. 1985, 'Toward an integrative framework of organizational control', *Accounting, Organizations and Society*, vol. 10, no. 1, pp. 35-50.
- Floridi, M., Pagni, S., Falorni, S. & Luzzati, T. 2011, 'An exercise in composite indicators construction: Assessing the sustainability of Italian regions', *Ecological Economics*, vol. 70, no. 8, pp. 1440-7.
- Forbes, J.C. & Watson, R.D. 1992, *Plants in Agriculture*, Cambridge University Press, Great Britian.
- Franco-Santos, M., Kennerley, M., Micheli, P., Martinez, V., Mason, S., Marr, B., Gray, G. & Neely, A. 2007, 'Towards a definition of a business performance measurement system', *International Journal of Operations & Production Management*, vol. 27, no. 8, pp. 784 - 801.
- French, R.J. & Schultz, J.E. 1984, 'Water use efficiency of wheat in a Mediterranean type environment. I. The relation between yield, water use and climate.', *Aust. J. Agric. Res.*, vol. 35, pp. 743–64.
- Gaydon, D.S., Meinke, H. & Rodriguez, D. 2012, 'The best farm-level irrigation strategy changes seasonally with fluctuating water availability', *Agricultural Water Management*, vol. 103, no. 0, pp. 33-42.
- Gillies, M.H. 2008, 'Managing the Effect of Infiltration Variability on the Performance of Surface Irrigation', Doctor of Philosophy thesis, University of Southern Queensland, Australia.
- Gillies, M.H. & Smith, R.J. 2005, 'Infiltration parameters from surface irrigation advance and run-off data', *Irrigation Science*, vol. 24, no. 1, pp. 25-35.
- Gleick, P. 1998, 'The world's water', Issues in Science & Technology, vol. 14, no. 4, p. 80.
- Gleick, P.H. 2003, 'WATER USE', *Annual Review of Environment and Resources*, vol. 28, pp. 275-314.

- Graedel, T.E. & van der Voet, E. 2010, 'Linkages of Sustainability: An Introduction', in T.E. Graedel & E. van der Voet (eds), *Linkages of Sustainability*, The MIT Press, Cambridge, England, pp. 1-8.
- Grassini, P., Hall, A.J. & Mercau, J.L. 2009, 'Benchmarking sunflower water productivity in semiarid environments', *Field Crops Research*, vol. 110, no. 3, pp. 251-62.
- Grassini, P., Yang, H., Irmak, S., Thorburn, J., Burr, C. & Cassman, K.G. 2011, 'High-yield irrigated maize in the Western US Corn Belt: II. Irrigation management and crop water productivity', *Field Crops Research*, vol. 120, no. 1, pp. 133-41.
- Gray, R. 1992, 'Accounting and environmentalism: An exploration of the challenge of gently accounting for accountability, transparency and sustainability', *Accounting, Organizations and Society*, vol. 17, no. 5, pp. 399-425.
- Gray, R. 2010, 'Is accounting for sustainability actually accounting for sustainability...and how would we know? An exploration of narratives of organisations and the planet', *Accounting, Organizations and Society*, vol. 35, no. 1, pp. 47-62.
- Green, S.G. & Welsh, M.A. 1988, 'Cybernetics and Dependence: Reframing the Control Concept', *Academy of Management Review*, vol. 13, no. 2, pp. 287-301.
- Greene, D.L. 2010, 'Measuring Energy Sustainability', in T.E. Graedel & E. van der Voet (eds), *Linkages of Sustainability*, The MIT Press, Cambridge, England, pp. 355 73.
- Hamdy, A., Ragab, R. & Scarascia-Mugnozza, E. 2003, 'Coping with water scarcity: water saving and increasing water productivity', *Irrigation and Drainage*, vol. 52, pp. 3–20.
- Hansen, S.C. & Banker, R.D. 2002, 'The Adequacy of Full-Cost-Based Pricing Heuristics', Journal of Management Accounting Research, vol. 14, p. 33.
- Harrison, J.R., Zhiang, L.I.N., Carroll, G.R. & Carley, K.M. 2007, 'SIMULATION MODELING IN ORGANIZATIONAL AND MANAGEMENT RESEARCH', *Academy of Management Review*, vol. 32, no. 4, pp. 1229-45.
- Hart, S.L. 1995, 'A natural resource based view of the firm', *Academy of Management Review*, vol. 20, no. 4, pp. 986-1014.
- Haslam, C., Butlin, J., Andersson, T., Malamatenios, J. & Lehman, G. 2014, 'Accounting for carbon and reframing disclosure: A business model approach', *Accounting Forum*, vol. 38, no. 3, pp. 200-11.
- Hearn, A.B. 1994, 'OZCOT: A simulation model for cotton crop management', *Agricultural Systems*, vol. 44, no. 3, pp. 257-99.
- Hearn, A.B., Ives, P.M., Room, P.M., Thomson, N.J. & Wilson, L.T. 1981, 'Computer-based cotton pest management in Australia', *Field Crops Research*, vol. 4, pp. 321-32.
- Henri, J.-F. & Journeault, M. 2008, 'Environmental performance indicators: An empirical study of Canadian manufacturing firms', *Journal of Environmental Management*, vol. 87, no. 1, pp. 165-76.
- Henri, J.-F. & Journeault, M. 2010, 'Eco-control: The influence of management control systems on environmental and economic performance', *Accounting, Organizations and Society*, vol. 35, no. 1, pp. 63-80.
- Henzell, T. 2007, *Australian Agriculture : Its History and Challenges*, Collingwood, Vic. : CSIRO Publishing, 2007.
- Herbohn, K. 2005, 'A full cost environmental accounting experiment', *Accounting, Organizations and Society*, vol. 30, no. 6, pp. 519-36.
- Hochman, Z., Gobbett, D., Holzworth, D., McClelland, T., van Rees, H., Marinoni, O., Garcia, J.N. & Horan, H. 2013, 'Reprint of "Quantifying yield gaps in rainfed cropping systems: A case study of wheat in Australia", *Field Crops Research*, no. 0.
- Hochman, Z., Holzworth, D. & Hunt, J.R. 2009, 'Potential to improve on-farm wheat yield and WUE in Australia', *Crop and Pasture Science*, vol. 60, no. 8, pp. 708-16.

- Hofstede, G. 1978, 'The Poverty of Management Control Philosophy', *Academy of Management Review*, vol. 3, no. 3, pp. 450-61.
- Hoogenboom, G., Jones, J.W., Wilkens, P.W., Porter, C.H., Boote, K.J., Hunt, L.A., Singh, U., Lizaso, J.I., White, J.W., Uryasev, O., Ogoshi, R., Koo, J., Shelia, V. & Tsuji, G.Y. 2015, 'Decision Support System for Agrotechnology Transfer (DSSAT) Version 4.6 (<u>www.DSSAT.net</u>)', DSSAT Foundation, Prosser, Washington.
- Hopwood, A.G. 2009, 'Accounting and the environment', *Accounting, Organizations and Society*, vol. 34, no. 3–4, pp. 433-9.
- Ibaraki, M. 2010, 'Issues of Unsustainability Related to Water', in T.E. Graedel & E. van der Voet (eds), *Linkages of Sustainability*, The MIT Press, Cambridge, England, pp. 267-90.
- Igbadun, H.E., Mahoo, H.F., Tarimo, A.K.P.R. & Salim, B.A. 2006, 'Crop water productivity of an irrigated maize crop in Mkoji sub-catchment of the Great Ruaha River Basin, Tanzania', *Agricultural Water Management*, vol. 85, no. 1–2, pp. 141-50.
- Indjejikian, R.J., Matějka, M. & Schloetzer, J.D. 2014, 'Target Ratcheting and Incentives: Theory, Evidence, and New Opportunities', *Accounting Review*, vol. 89, no. 4, pp. 1259-67.
- Intergovernmental Panel on Climate Change 2011, Intergovernmental Panel on Climate Change Report.
- Intergovernmental Panel on Climate Change 2013, *Climate Change 2013: The Physical Science Basis*.
- Ittner, C.D. & Larcker, D.F. 2001, 'Assessing empirical research in managerial accounting: a value-based management perspective', *Journal of Accounting and Economics*, vol. 32, no. 1–3, pp. 349-410.
- Ittner, C.D. & Larcker, D.F. 2003, 'Coming Up Short on Nonfinancial Performance Measurement', *Harvard Business Review*, vol. 81, no. 11, pp. 88-95.
- Jabbour, C.J.C. 2010, 'Non-linear pathways of corporate environmental management: a survey of ISO 14001-certified companies in Brazil', *Journal of Cleaner Production*, vol. 18, no. 12, pp. 1222-5.
- Jack, L. 2005, 'Stocks of knowledge, simplification and unintended consequences: the persistence of post-war accounting practices in UK agriculture', *Management Accounting Research*, vol. 16, no. 1, pp. 59-79.
- Jackson, T.M., Hanjra, M.A., Khan, S. & Hafeez, M.M. 2011, 'Building a climate resilient farm: A risk based approach for understanding water, energy and emissions in irrigated agriculture', *Agricultural Systems*, vol. 104, no. 9, pp. 729-45.
- Jackson, T.M., Khan, S. & Hafeez, M. 2010, 'A comparative analysis of water application and energy consumption at the irrigated field level', *Agricultural Water Management*, vol. 97, no. 10, pp. 1477-85.
- Jones, J. & Corral de Zubielqui, G. 2016, 'Doing well by doing good: A study of universityindustry interactions, innovationess and firm performance in sustainability-oriented Australian SMEs', *Technological Forecasting and Social Change*.
- Jones, J.W., Hoogenboom, G., Porter, C.H., Boote, K.J., Batchelor, W.D., Hunt, L.A., Wilkens, P.W., Singh, U., Gijsman, A.J. & Ritchie, J.T. 2003, 'The DSSAT cropping system model', *European Journal of Agronomy*, vol. 18, no. 3–4, pp. 235-65.
- Josée, S.-P. & Louis, R. 2004, 'Short-term effects of benchmarking on the manufacturing practices and performance of SMEs', *International Journal of Productivity and Performance Management*, vol. 53, no. 8, pp. 681-99.
- Joshi, S., Krishnan, R. & Lave, L. 2001, 'Estimating the Hidden Costs of Environmental Regulation', *Accounting Review*, vol. 76, no. 2, p. 171.
- Jun, K.S., Chung, E.-S., Sung, J.-Y. & Lee, K.S. 2011, 'Development of spatial water resources vulnerability index considering climate change impacts', *Science of The Total Environment*, vol. 409, no. 24, pp. 5228-42.
- Kanae, S. 2010, 'Measuring and Modelling the Sustainability of Global Water Resources', in T.E. Graedel & E. van der Voet (eds), *Linkages of Sustainability*, The MIT Press, Cambridge, England, pp. 291-306.
- Kang, Y., Ryu, M.-H. & Kim, S. 2010, 'Exploring sustainability management for telecommunications services: A case study of two Korean companies', *Journal of World Business*, vol. 45, no. 4, pp. 415-21.
- Kaplan, R.S. & Norton, D.P. 1996, 'Using the balanced scorecard as a strategic management system', *Harvard Business Review*, vol. 74, no. 1, pp. 75-85.
- Kaplan, R.S. & Thompson, G.L. 1971, 'Overhead Allocation via Mathematical Programming Models', *Accounting Review*, vol. 46, no. 2, pp. 352-64.
- Kato, T., Kimura, R. & Kamichika, M. 2004, 'Estimation of evapotranspiration, transpiration ratio and water-use efficiency from a sparse canopy using a compartment model', *Agricultural Water Management*, vol. 65, no. 3, pp. 173-91.
- Keating, B.A., Carberry, P.S., Hammer, G.L., Probert, M.E., Robertson, M.J., Holzworth, D., Huth, N.I., Hargreaves, J.N.G., Meinke, H., Hochman, Z., McLean, G., Verburg, K., Snow, V., Dimes, J.P., Silburn, M., Wang, E., Brown, S., Bristow, K.L., Asseng, S., Chapman, S., McCown, R.L., Freebairn, D.M. & Smith, C.J. 2003, 'An overview of APSIM, a model designed for farming systems simulation', *European Journal of Agronomy*, vol. 18, no. 3–4, pp. 267-88.
- Khabbaz, B.G. 2010, 'Life Cycle Energy Use and Greenhouse Gas Emissions of Australian Cotton: Impact of Farming Systems', Master of Engineering thesis, University of Southern Queensland.
- Khan, S. & Hanjra, M. 2008, 'Sustainable land and water management policies and practices: a pathway to environmental sustainability in large irrigation systems', *Land Degradation and Development*, vol. 19, pp. 469–87.
- Khan, S., Tariq, R., Yuanlai, C. & Blackwell, J. 2006, 'Can irrigation be sustainable?', *Agricultural Water Management*, vol. 80, no. 1–3, pp. 87-99.
- Knepper, T.P. & Ternes, T.A. 2010, 'Water Quality as a Component of a Sustainable Water Supply', in T.E. Graedel & E. van der Voet (eds), *Linkages of Sustainability*, The MIT Press, Cambridge, England, pp. 233-41.
- Kolk, A. & Mauser, A. 2002, 'The evolution of environmental management: from stage models to performance evaluation', *Business Strategy & the Environment (John Wiley & Sons, Inc)*, vol. 11, no. 1, pp. 14-31.
- Labro, E. & Vanhoucke, M. 2007, 'A Simulation Analysis of Interactions among Errors in Costing Systems', *Accounting Review*, vol. 82, no. 4, pp. 939-62.
- Lamberton, G. 2005, 'Sustainability accounting—a brief history and conceptual framework', *Accounting Forum*, vol. 29, no. 1, pp. 7-26.
- Langousis, A. & Kaleris, V. 2012, 'Theoretical framework to estimate spatially averaged rainfalls conditional on river discharges and point rainfall measurements from a single location: an application to Western Greece', *Hydrology & Earth System Sciences Discussions*, vol. 9, no. 11, pp. 12463-522.
- Lindner, K., Knepper, T.P., Ahmed, M.T., Barth, J.A.C., Crutzen, P.J., Dayrit, F.M., De Wever, H., Ibaraki, M., Kanae, S., Schmidt, M. & Ternes, T.A. 2010, 'Stocks, Flows and Prospects of Water', in T.E. Graedel & E. van der Voet (eds), *Linkages of Sustainability*, The MIT Press, Cambridge, England, pp. 309-19.

- Lisson, S.N., Inman-Bamber, N.G., Robertson, M.J. & Keating, B.A. 2005, 'The historical and future contribution of crop physiology and modelling research to sugarcane production systems', *Field Crops Research*, vol. 92, no. 2–3, pp. 321-35.
- Liu-Kang, X. & Hsiao, T.C. 2004, 'Predicted versus measured photosynthetic water-use efficiency of crop stands under dynamically changing field environments\*This paper is an expansion with much additional material of a preliminary report appearing in 2000 in Acta Horticulturae 537', *Journal of Experimental Botany*, vol. 55, no. 407, pp. 2395-411.
- Long, R.L., Bange, M.P., Gordon, S.G., van der Sluijs, M.H.J., Naylor, G.R.S. & Constable, G.A. 2010, 'Fiber Quality and Textile Performance of Some Australian Cotton Genotypes', *Crop Science*, vol. 50, no. 4, pp. 1509-18.
- Malmi, T. & Brown, D. 2008, 'Management control systems as a package—Opportunities, challenges and research directions', *Management Accounting Research*, vol. 19, no. 4, pp. 287-300.
- Massoud, M.A., Fayad, R., El-Fadel, M. & Kamleh, R. 2010, 'Drivers, barriers and incentives to implementing environmental management systems in the food industry: A case of Lebanon', *Journal of Cleaner Production*, vol. 18, no. 3, pp. 200-9.
- Mateos, L., González-Dugo, M.P., Testi, L. & Villalobos, F.J., 'Monitoring evapotranspiration of irrigated crops using crop coefficients derived from time series of satellite images. I. Method validation', *Agricultural Water Management*, no. 0.
- Mehta, V.K., Haden, V.R., Joyce, B.A., Purkey, D.R. & Jackson, L.E. 2013, 'Irrigation demand and supply, given projections of climate and land-use change, in Yolo County, California', *Agricultural Water Management*, vol. 117, no. 0, pp. 70-82.
- Miles, M.P., Munilla, L.S. & McClurg, T. 1999, 'The impact of ISO 14000 environmental management standards on small and medium sized enterprises', *Journal of Quality Management*, vol. 4, no. 1, pp. 111-22.
- Milne, M.J. 1996, 'On sustainability; the environment and management accounting', *Management Accounting Research*, vol. 7, no. 1, pp. 135-61.
- Milroy, S.P., Bange, M.P. & Hearn, A.B. 2004, 'Row configuration in rainfed cotton systems: modification of the OZCOT simulation model', *Agricultural Systems*, vol. 82, no. 1, pp. 1-16.
- Moore, A.D., Robertson, M.J. & Routley, R. 2011, 'Evaluation of the water use efficiency of alternative farm practices at a range of spatial and temporal scales: A conceptual framework and a modelling approach', *Agricultural Systems*, vol. 104, no. 2, pp. 162-74.
- Moore, D.R.J. 2013, 'Sustainability, institutionalization and the duality of structure: Contradiction and unintended consequences in the political context of an Australian water business', *Management Accounting Research*, vol. 24, no. 4, pp. 366-86.
- Moreno-Pérez, M.F. & Roldán-Cañas, J. 2012, 'Assessment of irrigation water management in the Genil-Cabra (Córdoba, Spain) irrigation district using irrigation indicators', *Agricultural Water Management*.
- Naramngam, S. & Tong, S.T.Y. 2013, 'Environmental and economic implications of various conservative agricultural practices in the Upper Little Miami River basin', *Agricultural Water Management*, vol. 119, no. 0, pp. 65-79.

National Farmers' Federation 2012, NFF Farm Facts: 2012, ACT, Australia.

- Neely, A., Gregory, M. & Platts, K. 1995, 'Performance measurement system design: A literature review and research agenda', *International Journal of Operations & Production Management*, vol. 15, no. 4, pp. 80-116.
- Neely, A., Mills, J., Platts, K., Richards, H., Gregory, M., Bourne, M. & Kennerley, M. 2000, 'Performance measurement system design: developing and testing a process-based

approach', *International Journal of Operations & Production Management*, vol. 20, no. 10, pp. 1119-45.

- Neely, A., Richards, H., Mills, J., Platts, K. & Bourne, M. 1997, 'Designing performance measures: a structured approach', *International Journal of Operations & Production Management*, vol. 17, no. 11, pp. 1131-52.
- Nissim, D. & Penman, S.H. 2001, 'Ratio Analysis and Equity Valuation: From Research to Practice', *Review of Accounting Studies*, vol. 6, no. 1, pp. 109-54.
- Noreen, E.W. & Burgstahler, D. 1997, 'Full-Cost Pricing and the Illusion of Satisficing', Journal of Management Accounting Research, vol. 9, pp. 239-63.
- Nørreklit, L., Nørreklit, H. & Israelsen, P. 2006, 'The validity of management control topoi: Towards constructivist pragmatism', *Management Accounting Research*, vol. 17, no. 1, pp. 42-71.
- NSW Department of Primary Industries 2015, *Cotton Gross Margins: Furrow Irrigated Cotton - Central and Northern NSW 2014-2015*, NSW Department of Primary Industries.
- Ogden, S.G. 1997, 'Accounting for organizational performance: The construction of the customer in the privatized water industry', *Accounting, Organizations and Society*, vol. 22, no. 6, pp. 529-56.
- Otley, D. 1999, 'Performance management: a framework for management control systems research', *Management Accounting Research*, vol. 10, no. 4, pp. 363-82.
- Oweis, T.Y., Farahani, H.J. & Hachum, A.Y. 2011, 'Evapotranspiration and water use of full and deficit irrigated cotton in the Mediterranean environment in northern Syria', *Agricultural Water Management*, vol. 98, no. 8, pp. 1239-48.
- Passioura, J. 1996, 'Simulation Models: Science, Snake Oil, Education, or Engineering?', *Agronomy Journal*, vol. 88, no. 5, pp. 690-4.
- Passioura, J. 2006, 'Increasing crop productivity when water is scarce—from breeding to field management', *Agricultural Water Management*, vol. 80, no. 1–3, pp. 176-96.
- Patten, D.M. 2002, 'The relation between environmental performance and environmental disclosure: a research note', *Accounting, Organizations and Society*, vol. 27, no. 8, pp. 763-73.
- Patzek, T. 2004, 'Thermodynamics of the Corn-Ethanol Biofuel Cycle', *Critical Reviews in Plant Sciences*, vol. 23, no. 6, pp. 519-67.
- Patzek, T.W. & Pimentel, D. 2005, 'Thermodynamics of Energy Production from Biomass', *Critical Reviews in Plant Sciences*, vol. 24, no. 5/6, pp. 327-64.
- Pearce, D. 1976, 'THE LIMITS OF COST-BENEFIT ANALYSIS AS A GUIDE TO ENVIRONMENTAL POLICY', *KYKLOS*, vol. 29, no. 1, pp. 97-112.
- Penman, S.H. 2003, *Financial Statement Analysis and Security Valuation*, 2nd edn, McGraw-Hill, New York, USA.
- Pereira, A.R., Green, S. & Villa Nova, N.A. 2006, 'Penman–Monteith reference evapotranspiration adapted to estimate irrigated tree transpiration', *Agricultural Water Management*, vol. 83, no. 1–2, pp. 153-61.
- Pereira, L.S., Cordery, I. & Iacovides, I. 2012, 'Improved indicators of water use performance and productivity for sustainable water conservation and saving', *Agricultural Water Management*, vol. 108, no. 0, pp. 39-51.
- Pink, B. 2012, 2012 Year Book Australia: A comprehensive source of information about Australia, vol. 92, Canberra, ABS Catalogue No. 1301.0.
- Pondeville, S., Swaen, V. & De Rongé, Y. 2013, 'Environmental management control systems: The role of contextual and strategic factors', *Management Accounting Research*, vol. 24, no. 4, pp. 317-32.

Purcell, J. & Currey, A. 2003, *Gaining Acceptance of Water Use Efficiency Framework, Terms and Definitions*, National Program for Sustainable Irrigation.

- Rainey, D.L. 2010, Sustainable Business Development: Inventing the Future Through Strategy, Innovation, and Leadership, Cambridge University Press.
- Ramankutty, N. 2010, 'Agriculture and Forests: Recent Trends, Future Prospects', in T.E. Graedel & E. van der Voet (eds), *Linkages of Sustainability*, The MIT Press, Cambridge, England, pp. 11-31.
- Raskin, P., Gleick, P., Kirshen, P., Pontius, G. & Strzepek, K. 1997, *Water Futures:* Assessment of Long-range Patterns and Prospects, Stockholm Environment Institute, Stockholm, Sweden.
- Reidsma, P., Feng, S., van Loon, M., Luo, X., Kang, C., Lubbers, M., Kanellopoulos, A., Wolf, J., van Ittersum, M.K. & Qu, F. 2012, 'Integrated assessment of agricultural land use policies on nutrient pollution and sustainable development in Taihu Basin, China', *Environmental Science & amp; Policy*, vol. 18, no. 0, pp. 66-76.
- Rifkin, J. 2005, 'Is Sustainable Development Compatible with Human Welfare?', in Easton 2011 (ed.), *Taking Sides: Clashing Views on Environemntal Issues*, 14th edn, Mc Graw Hill, New York, pp. 22-3.
- Rijsberman, F.R. 2006, 'Water scarcity: Fact or fiction?', *Agricultural Water Management*, vol. 80, no. 1–3, pp. 5-22.
- Roer, A.-G., Korsaeth, A., Henriksen, T.M., Michelsen, O. & Strømman, A.H. 2012, 'The influence of system boundaries on life cycle assessment of grain production in central southeast Norway', *Agricultural Systems*, vol. 111, no. 0, pp. 75-84.
- Roth, G. 2010, *Economic, Environmental and Social Sustainability Indicators of the Australian Cotton Industry*, Cotton Research and Development Corporation.
- Roth, G., Harris, G., Gillies, M., Montgomery, J. & Wigginton, D. 2013, 'Water-use efficiency and productivity trends in Australian irrigated cotton: a review', *Crop and Pasture Science*, vol. 64, no. 12, pp. 1033-48.
- Salvador, R., Martínez-Cob, A., Cavero, J. & Playán, E. 2011, 'Seasonal on-farm irrigation performance in the Ebro basin (Spain): Crops and irrigation systems', *Agricultural Water Management*, vol. 98, no. 4, pp. 577-87.
- Salzmann, O., Ionescu-somers, A. & Steger, U. 2005, 'The Business Case for Corporate Sustainability:: Literature Review and Research Options', *European Management Journal*, vol. 23, no. 1, pp. 27-36.
- Samson, A. & Cotton CRC 2009, *Qualitative and Quantitative Analysis of the BMP Trial Evaluation Project*, Cotton Industry.
- Schäfer, A.I. & Beder, S. 2006, 'Relevance of the precautionary principle in water recycling', *Desalination*, vol. 187, no. 1–3, pp. 241-52.
- Schaltegger, S. & Burritt, R.L. 2010, 'Sustainability accounting for companies: Catchphrase or decision support for business leaders?', *Journal of World Business*, vol. 45, no. 4, pp. 375-84.
- Schaltegger, S. & Synnestvedt, T. 2002, 'The link between 'green' and economic success: environmental management as the crucial trigger between environmental and economic performance', *Journal of Environmental Management*, vol. 65, no. 4, pp. 339-46.
- Scherer, F.M. 2001, 'Firm Behavior', in J.S. Neil & B.B. Paul (eds), *International Encyclopedia of the Social & Behavioral Sciences*, Pergamon, Oxford, pp. 5676-81.
- Sinclair, T.R., Tanner, C.B. & Bennett, J.M. 1984, 'Water-Use Efficiency in Crop Production', *BioScience*, vol. 34, no. 1, pp. 36-40.

- Smith, R.J., Raine, S.R. & Minkevich, J. 2005, 'Irrigation application efficiency and deep drainage potential under surface irrigated cotton', *Agricultural Water Management*, vol. 71, no. 2, pp. 117-30.
- Soliman, M.T. 2008, 'The Use of DuPont Analysis by Market Participants', *Accounting Review*, vol. 83, no. 3, pp. 823-53.
- Sommer, R., Piggin, C., Haddad, A., Hajdibo, A., Hayek, P. & Khalil, Y. 2012, 'Simulating the effects of zero tillage and crop residue retention on water relations and yield of wheat under rainfed semiarid Mediterranean conditions', *Field Crops Research*, vol. 132, no. 0, pp. 40-52.
- Stewart, J.M., Oosterhuis, D.M., Heitholt, J.J. & Mauney, J.R. (eds) 2010, *Physiology of Cotton*, Springer Science+Business Media.
- Tennakoon, S.B. & Milroy, S.P. 2003, 'Crop water use and water use efficiency on irrigated cotton farms in Australia', *Agricultural Water Management*, vol. 61, no. 3, pp. 179-94.
- The Brundtland Commission 1987, Our Common Future, The Report of the World Comission on Environment and Development, Oxford University Press, Oxford.
- The Cotton Industry Development and Delivery Team 2012, *Australian Cotton Production Manual 2012*, Cotton Catchement Communities (CRC).
- The Economics of Ecosystems and Biodiversity 2010, *The Economics of Ecosystems and Biodiversity: Mainstreaming the Economics of Nature: A Synthesis of the Approach, Conclusions and Recommendations of TEEB.*
- The Economics of Ecosystems and Biodiversity 2013, *The Economics of Ecosystems and Biodiversity for Water and Wetlands*, Institute for European Environmental Policy (IEEP) & Ramsar Secretariat.
- The Economics of Ecosystems and Biodiversity 2014, *Making Nature's Values Visible*, viewed 4th February 2014, <<u>http://www.teebweb.org/></u>.
- Thorp, K.R., S. Ale, M. P. Bange, E. M. Barnes, G. Hoogenboom, R. J. Lascano, A. C. McCarthy, S. Nair, J. O. Paz, N. Rajan, K. R. Reddy, G. W. Wall & White, J.W. 2014, 'Development and Application of Process-based Simulation Models for Cotton Production: A Review of Past, Present, and Future Directions ', *Journal of Cotton Science* vol. 18, no. 1, pp. 10-47.
- Tingey-Holyoak, J. 2014, 'Sustainable water storage by agricultural businesses: Strategic responses to institutional pressures', *Journal of Business Research*, no. 0.
- UNCTAD 2014, *Key Statistics and Trends in Inernational Trade 2014*, United Nations Conference on Trade and Development (UNCTAD).
- US Department of Homeland Security 2013, *Food and Agriculture Sector*, US Department of Homeland Security, viewed 17th December 2013, <a href="https://www.dhs.gov/food-and-agriculture-sector">https://www.dhs.gov/food-and-agriculture-sector</a>.
- van Halsema, G.E. & Vincent, L. 2012, 'Efficiency and productivity terms for water management: A matter of contextual relativism versus general absolutism', *Agricultural Water Management*, vol. 108, pp. 9-15.
- van Ittersum, M.K., Cassman, K.G., Grassini, P., Wolf, J., Tittonell, P. & Hochman, Z. 2012, 'Yield gap analysis with local to global relevance—A review', *Field Crops Research*, no. 0.
- Veleva, V. 1994, 'Development of Environemntal Sustainability Indicators for Use in Envirollmental Impact Assessment', MSc Thesis thesis, University of Manchester, UK, Manchester.
- Verhalen, L.M., Greenhagen, B.E. & Thacker, R.W. 2003, 'Lint Yield, Lint Percentage, and Fiber Quality Response in Bollgard, Roundup Ready, and Bollgard/Roundup Ready Cotton', *The Journal of Cotton Science*, vol. 7, no. 2, pp. 23-38.

- Virtanen, T., Tuomaala, M. & Pentti, E. 2013, 'Energy efficiency complexities: A technical and managerial investigation', *Management Accounting Research*, vol. 24, no. 4, pp. 401-16.
- Wagner, M. & Schaltegger, S. 2004, 'The Effect of Corporate Environmental Strategy Choice and Environmental Performance on Competitiveness and Economic Performance:: An Empirical Study of EU Manufacturing', *European Management Journal*, vol. 22, no. 5, pp. 557-72.
- Walker, W.R. 2003, "Surface irrigation simulation, evaluation and design." User Guide and *Technical Documentation*, Utah State University, Logan, Utah.
- Wang, E., Robertson, M.J., Hammer, G.L., Carberry, P.S., Holzworth, D., Meinke, H., Chapman, S.C., Hargreaves, J.N.G., Huth, N.I. & McLean, G. 2002, 'Development of a generic crop model template in the cropping system model APSIM', *European Journal of Agronomy*, vol. 18, no. 1–2, pp. 121-40.
- Weber, M. 2008, 'The business case for corporate social responsibility: A company-level measurement approach for CSR', *European Management Journal*, vol. 26, no. 4, pp. 247-61.
- Wei, Q., Roger, B. & Gary, M. 2011, 'Environmental management accounting in local government; A case of waste management', *Accounting, Auditing & Accountability Journal*, vol. 24, no. 1, p. 93.
- Wells, A.T. & Hearn, A.B. 1992, 'OZCOT: A cotton crop simulation model for management', *Mathematics and Computers in Simulation*, vol. 33, no. 5–6, pp. 433-8.
- Williams, J. & Martin, P. (eds) 2011, *Defending the Social Licence of Farming: Issues, Challenges and New Directions for Agriculture*, CSIRO Publishing.
- Wisner, P.S., Epstein, M.J. & Bagozzi 2006.
- WWAP 2009, The United Nations World Water Development Report 3: Water in a Changing World, UNESCO, Paris.
- WWAP 2012, The United Nations World Water Development Report 4: Managing Water under Uncertainty and Risk, UNESCO, Paris.
- WWAP 2014, *The United Nations World Water Development Report 2014: Water and Energy*, UNESCO, Paris.
- Yeates, S.J., Constable, G.A. & McCumstie, T. 2010, 'Irrigated cotton in the tropical dry season. I: Yield, its components and crop development', *Field Crops Research*, vol. 116, no. 3, pp. 278-89.
- Zhang, L., Walker, G.R. & Dawes, W.R. 2002, 'Water Balance Modelling: Concepts and Applications', in T.R. McVicar, L. Rui, J. Walker, R.W. Fitzpatrick & L. Changming (eds), *Regional water and soil assessment for managing sustainable agriculture in China and Australia*, Australian Centre for International Agricultural Research Canberra, pp. 31-47.
- Zhao, H., Li, X., Wang, X. & Tian, D. 2010, 'Grain size distribution of road-deposited sediment and its contribution to heavy metal pollution in urban runoff in Beijing, China', *Journal of Hazardous Materials*, vol. 183, no. 1-3, pp. 203-10.

# Appendices

# Appendix A: Supplementary information for Chapter 4

**Appendix A1: An example of Cotton Gross Margins** (Source: NSW Department of Primary Industries (2015))

NSW Depa Prima	ary Inc	dustries	ottoni		Cotto	n Gro	ss Mai	rgins
Furrow Irr	igate	d Cotto	n - (Ro	oundu	p Ready	Flex® Bo	llgard ll®	)
Central & Nort	hern N	SW ~ 2014-	2015				-	-
GROSS MARGIN	BUDGE	т				1	Sample	Your
							Budget	Budge
INCOME:							\$/ha	\$/ha
11.00 bale	es/ha at	Lint' \$	474	/bale (a	it gin)		\$5,214	
		Seed \$	80	/bale (a	it gin)		\$880	
	()	ess discount) \$	-	/bale (e	stimate)		\$0	
		\$	554					
		A. 1	TOTAL IN	COME \$	/ha:	[	\$6,094	
VARIABLE COST	S:						Sample	Your
See Calendar of Oper	ations for	details					Budget	budget
		Cultivation					\$129	
		Planting					\$115	
		Irrigation (9.79ML	.)				\$573	
		Fertiliser & applic	ation				\$3/4	
		herbicide & appli	cation				\$02 \$125	
		l icence fees	iicauon				\$370	
		Defoliation					\$103	
Picking & cartage to gin							\$334	
		Ginning & levies	-				\$756	
	1	Refuge crop, pige	eon pea, 5	5% of Bt (	cotton area.		\$31	
		Crop insurance					\$110	
		Other (e.g. consu	ltants)				\$74	
			В.	TOTAL \	ARIABLE (	COSTS \$/ha:	\$3,177	
				C. GRO	SS MARGI	I (A-B) \$/ha:	\$2,917	
			D. GROSS	S MARG	IN (C/ML ap	plied) \$/ML:	\$298	
EFFECT OF YIEL	D AND F	PRICE ON GR	OSS M/	ARGIN	PER HEC	TARE:		
Approxi	imate Brea	akeven Yield:	5.73	bales/h	а	(based on t	he lint, seed, disco	unt prices abo
Approxi	mate Brea	keven Price: \$	289	/bale		(based o	on 11 bales/ha. Ov	erhead costs a
SENSITIVITY THE	BLE				perman	ent labour are NOT	considered in the	breakeven pr
		PR	ICE (\$/ba	ale incl.	of seed &	discounts)		
<u> </u>		454	504		554	604	654	
	1.50	511	4 534		1,201	1,636	2,011	
/he	9.00	1,071	1,521		1,9/1	2,421	2,871	
a les	11.00	1.817	2,367		2,917	3 467	4.017	
t pa	12.00	2,190	2,790	<u> </u>	3,390	3,990	4 590	
E.	13.00	2,563	3,213		3,863	4,513	5,163	
	14.50	3,123	3,848		4,573	5,298	6,023	

### Central & Northern NSW ~ 2014-2015

# Furrow Irrigated Cotton (Roundup Ready Flex® Boligard II®)

			Ma	chinery				In	puts		TOTAL
		Calendar of Operations			Total	Rate		Band	Cost	Cost	Cost
					\$/ha	/ha	Unit	Width	\$/unit	\$/ha	\$/ha
Jul	Farming:	Discing	180kW PTO		11.43						11.43
Jul	Farming:	Hill up	180kW PTO		12.72						12.72
Jul	Fertiliser:	MAP	with above			120	kg	100%	0.72	86.64	86.64
Aug	Fertiliser:	Urea	135kW PTO	140kg N	4.27	304	kg	100%	0.50	152.41	156.67
Aug	Farming:	Rubber tyre roller	135kW PTO		11.07						11.07
Sep	Irrigation:	Pre Plant				1.60	ML	100%	60.30	96.48	96.48
Oct	Planting:	Precision planter	180kW PTO		11.13						11.13
Oct	Planting:	Seed <sup>9</sup> : Roundup Ready Flex Bollgard II	with above			13	kg	100%	8.00	104.00	104.00
Oct	Herbicide:	Fluometuron + prometryn (440g/L + 400g/L)	Self Propelled		3.67	2.00	kg	100%	17.56	35.12	38.79
Nov	Crop Insura	nce:		Premium	depends or	variou	s fact	ors			110.00
Nov	Irrigation:	In Crop				1.30	ML	100%	60.30	78.39	78.39
Nov	Fertiliser:	Urea (water run)	with above	120kg N		260	kg	100%	0.50	130.35	130.35
Nov	Herbicide:	Roundup Ready® Plantshield® (690g/kg Glyphosate)	Self Propelled		3.67	1.20	kg	100%	6.85	8.22	11.89
Dec	Insecticide:	Fipronil (200g/L), target: mirids	Self Propelled		3.67	0.10	L	100%	285.17	28.52	32.18
Dec	Irrigation:	In Crop				1.00	ML	100%	60.30	60.30	60.30
Dec	Farming:	Cultivation: Inter row	180kW PTO		11.85						11.85
Dec	Other:	Chipping or Spot Spray <sup>4</sup>			5.00						5.00
Dec	Irrigation:	In Crop				1.00	ML	100%	60.30	60.30	60.30
Jan	Irrigation:	In Crop				1.00	ML	100%	60.30	60.30	60.30
Jan	Insecticide:	Abamectin (18g/L), target: mites	Self Propelled		3.67	0.60	L	100%	19.11	11.47	15.13
Jan	Herbicide:	Roundup Ready® Plantshield® (690g/kg Glyphosate)	Self Propelled		3.67	1.20	kg	100%	6.85	8.22	11.89
Jan	Irrigation:	In Crop				1.00	ML	100%	60.30	60.30	60.30
Jan	Insecticide:	Dimethoate (400g/L), target: GVB, aphids	Aerial Spraying		14.00	0.40	L	100%	9.66	3.86	17.86
Jan	Irrigation:	In Crop				1.00	ML	100%	60.30	60.30	60.30
Feb	Insecticide:	Diafenthiuron (500g/L), target: SLW, aphids	Aerial Spraying		14.00	0.70	L	100%	65.32	45.72	59.72
Feb	Irrigation:	In Crop				0.80	ML	100%	60.30	48.24	48.24
Feb	Irrigation:	In Crop				0.80	ML	100%	60.30	48.24	48.24
Mar	Licence:	Bollgard II stacked RRF Licence Fee <sup>5</sup>									370.00
Mar	Defoliation:	Thidiazuron + Diuron (120g + 60g/L)	Aerial Spraying		14.00	0.25	L	100%	177.14	44.28	58.28
Mar	Defoliation:	Ethephon (720g/L)	with above			1.50	L	100%	6.42	9.63	9.63
Mar	Defoliation:	Crop oil	with above			2.00	L	100%	4.31	8.62	8.62
Mar	Defoliation:	Ethephon (720g/L)	Aerial Spraying		14.00	2.00	L	100%	6.42	12.85	26.85
Apr	Picking:	Own plant: round baler	Picker: JD7760		67.24	per ha					67.24
Apr	Picking:	plus fuel				28.00	L	100%	1.04	29.03	29.03
Apr	Picking:	plus wrap		40.00 /	round bale					9.41 / lint bale	103.53
Мау	Cartage:	Lift	costs related to	7.00 /	round bale	12 rour	nd bai	es/truck		1.65 / lint bale	18.12
May	Cartage:	Freight	yleid	45.00 /	round bale	50 km 1	from g	gin		10.59 / lint bale	116.47
May	Ginning:			65.00	/ lint bale					65.00 / lint bale	715.00
Jun	Levies:	Research Levy & Cotton Australia Levy		3.75	/ lint bale					3.75 / lint bale	41.25
Jun	Farming:	Mulcher with root cutter	135kW PTO		14.90						14.90
Jun	Other:	Consultant	Contractor		65.00						65.00
Jun	Other:	Soil moisture monitoring	Contractor		4.00						4.00
Jun	Farming:	Desiting & grading channels	Contractor		50.00						50.00
Jun	Herbicide:	Steniising channels	135kW PTO		19.09						19.09
Jun	Farming:	Pupae busting	180kW PTO		17.39	_					17.39
Jun	Heruge:	Reruge Crop: Pigeon peas 5%, see page 4				0.29	ML				31.39
		B. TOTAL VARIABLE COSTS \$/ha:				0.70	10.7				3,177
		Total Irrigation Water Use ML/ha:				9.79	ML'				
											Page 2

FOOTNOTES: 2014-15 Central & Northern NSW, Furrow Irrigated Cotton Gross Margin Budget \$474/bale is the five-year average cotton lint price 2010-2015. 2. The cotton seed price is given indicatively as a per bale value. \$80/bale for seed (prior to ginning costs being subtracted) is the equivalent of \$320/t, assuming an average of 250kg of cotton seed per bale of lint. 3. Seed costs per kg will vary with the time of ordering and seed treatments chosen. 4. Chipping or spot spray can be used to control any surviving weeds. 5. The technology licence fee for Bollgard II<sup>®</sup> stacked with Roundup Ready<sup>®</sup> Flex for 2014-15 is \$370 per green hectare (GST exclusive). This example uses Monsanto Cotton Choices<sup>™</sup> Option 1, which provides a discount on the technology licence fees for 2014-15. However, you should choose the option which best suits your operation. See http://www.cottonchoices.com.au. 6. Assumptions made for module cartage are: property distance from the gin 50km, road train carting 12 round bales per trip, 4.25 lint bales per round bale. 7. Irrigation requirements typically range from 8-13ML/ha. The largest impact on water use is in-crop rainfall. GENERAL INFORMATION For a complete guide to cotton management, see the Australian Cotton Production Manual 2014. A gross margin represents the difference between gross income and the variable costs of producing a crop. Gross margin budgets do not take into account risk, overhead costs (including permanent labour) and do not calculate farm profit. This budget should be used as a GUIDE ONLY. It is designed to give an indication of operations and costs required to grow a cotton crop. A grower should alter this budget to take account of individual field management plans, movements in crop and input prices and changes in seasonal conditions. In all instances, operations should be tailored to the requirements of individual paddocks. BALE: The industry term 'per bale' is in reference to a ginned lint bale of 227kg. New picking technology picks the cotton and packs it into round or rectangular bales on-farm. Bt: A licence fee is paid to Monsanto for cotton seed that uses Bollgard technology. CHEMICALS: Always read chemical labels and follow directions, as it is your legal responsibility to do so. Use of a particular brand name or active ingredient does NOT imply a recommendation by NSW DPI. DEFOLIANT: Good conditions are required to get the best performance. The choice of defoliant and rate used depends on the moisture status of the plant and seasonal conditions. FERTILISER REQUIREMENTS: All fertiliser strategies include comprehensive soil testing prior to sowing. HERBICIDES: The cornerstone of weed management and managing herbicide resistance risks is controlling survivors and preventing new weed seeds from entering the seed bank. To reduce the likelihood of herbicide resistance, rotate herbicide groups and weed management techniques. Aim to plant into clean fields. See the Herbicide Resistance Management Strategy (found in the CottonInfo Cotton Pest Management Guide) and the Monsanto Roundup Ready Flex Cotton Weed Management Guide. INSECTICIDES: Insecticides suggested in this budget and spray timing are examples only and strategies will vary with individual circumstances. Individual paddocks need careful monitoring to determine pest and beneficial insect populations. Use recommended thresholds for all pests. Avoid using broad spectrum sprays and continuously using chemicals from the same group. Follow the Insecticide Resistance Management Strategy (found in the CottonInfo Cotton Pest Management Guide) to protect the value of insecticide technologies for the future. Conserving and utilising beneficial insects is a key aspect of long-term effective pest management. IRRIGATION: The cost of water varies considerably depending on the source, location and number of times water is pumped. A cost of \$60.30/ML is used assuming pumping from the regulated Namoi River (fees \$32.24/ML) with water 'lifted' twice using diesel pumps (pumping cost \$28.06/ML). This pumping cost takes into consideration fuel and maintenance. Pumping costs can be in excess of \$120/ML for bores. LABOUR: Labour is assumed to be an overhead cost and not included in this budget. An estimate of labour costs is \$250-\$300/ha.

LEVIES: The Research Levy (\$2.25/bale) is a compulsory levy that is invoiced by the ginning organisation following ginning. Funds collected through this levy are used to fund vital industry research by the Cotton Research and Development Corporation (CRDC). The Cotton Australia Levy (\$1.50/bale) is a voluntary levy, which funds the peak industry body Cotton Australia that provides a valuable policy and advocacy role, farmer support and promotes the Australian cotton industry.

MACHINERY: The cost of each farming pass reflects variable costs only (fuel, repairs and maintenance), labour and depreciation are considered overhead costs, so are not included in this budget.

ROTATION: Whilst cotton can be grown in various rotations, this budget assumes a two-year rotation of cotton-wheatlong fallow.

YIELD: A yield of 11 bales/ha is achievable considering the long fallow, 'best practice' operations and a five-year average yield for the variety S74BRF in Cotton Seed Distributors (CSD) commercial trial results.

Page 3

# Appendix A2: WESM Analysis

# Table A1 WESM model - Panel D (Level 4 of WESM)

Panel D1: Analysis of Crop water use index		
Variable	WESM	Formula
	Level	
Crop water use $index_{it} \left( \frac{kg \text{ of } lint}{ML} \right)$	Level 4	$= \text{Lint yield}_{\text{it}} \left(\frac{\text{kg of lint}}{\text{ha}}\right) \div \text{Crop water supply}_{\text{it}} \left(\frac{\text{ML}}{\text{ha}}\right)$
		$= Transp \qquad index_{it} \left(\frac{kg \text{ of biomass}}{ML \text{ of tranpiration}}\right)$
		× Harvest index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{kg of biomass}}\right)$
		× Water input efficiency <sub>it</sub> (%)
Transpiration index <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ML of transpiration}}\right)$	Level 5	$= \text{Biomass yield}_{\text{it}} \left(\frac{\text{kg of biomass}}{\text{ha}}\right) \div \text{Transpiration}_{\text{it}} \left(\frac{\text{ML of transpiration}}{\text{ha}}\right)$
Harvest index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{kg of biomass}}\right)$	Level 5	$= \text{Lint yield}_{\text{it}} \left(\frac{\text{kg of lint}}{\text{ha}}\right) \div \text{Biomass yield}_{\text{it}} \left(\frac{\text{kg of biomass}}{\text{ha}}\right)$

Water input efficiency <sub>it</sub> (%)	Level 5	$= \left[ \text{Transpiration}_{\text{it}} \left( \frac{\text{ML of transpiration}}{\text{ha}} \right) \div \text{Crop water supply}_{\text{it}} \left( \frac{\text{ML}}{\text{ha}} \right) \right]$
		$\times 100\%$
Panel D2: Analysis of Weighted average cotton pri-	ce	
Variable	WESM	Formula
	Level	
Weighted average cotton $price_{it}$ $\left(\frac{\$}{1-1-1-1-1}\right)$	Level 4	= Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$ ÷ Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{ha}\right)$
\kg of total yield/		$= \left[ \text{Lint percentage}_{it} (\%) \times \text{Lint price}_{it} \left( \frac{\$}{kg} \right) \right]$
		+ $\left[\text{Seed percentage}_{it}(\%) \times \text{Seed price}_{it}\left(\frac{\$}{kg}\right)\right]$
Lint percentage <sub>it</sub> (%)	WESM Input	$= \left[ \text{Lint yield}_{\text{it}} \left( \frac{\text{kg of lint}}{\text{ha}} \right) \div \text{Total yield}_{\text{it}} \left( \frac{\text{kg of total product}}{\text{ha}} \right) \right] \times 100\%$
	WEOM	
Seed percentage <sub>it</sub> (%)	WESM Input	$= \left[ \text{Seed yield}_{\text{it}} \left( \frac{\text{kg of seed}}{\text{ha}} \right) \div \text{ Total yield}_{\text{it}} \left( \frac{\text{kg of total product}}{\text{ha}} \right) \right] \times 100\%$
		$= 100\% - \text{Lint percentage}_{it}$ (%)
		Crop Data
Lint price: $\left(\frac{\$}{-}\right)$	WESM	Accounting Data
kg/	<mark>Input</mark>	

Seed price (\$)	WESM	Accounting Data
Seed price <sub>it</sub> $\left( \frac{kg}{kg} \right)$	<mark>Input</mark>	

# Table A1 WESM model - Panel E (Level 5 of WESM)

Panel E1: Water input efficiency		
Variable	WESM	Formula
	Level	
Water input efficiency <sub>it</sub> (%)	Level 5	$= \left[ \text{Plant transpiration}_{\text{it}} \left( \frac{\text{ML of transpiration}}{\text{ha}} \right) \div \text{Crop water supply}_{\text{it}} \left( \frac{\text{ML}}{\text{ha}} \right) \right]$
		× 100%
		= 100% – [Crop water loss to crop water supplied ratio (%)
		+ Recycled runoff to crop water supplied ratio (%)
Crop water loss to crop water supply ratio <sub>it</sub>	Level 6	= Crop water $loss_{it}\left(\frac{ML}{ha}\right) \div Crop water supply_{it}\left(\frac{ML}{ha}\right)$
Recycled runoff to crop water supply ratio <sub>it</sub>	Level 6	$= \text{Recyled runoff}_{\text{it}}\left(\frac{\text{ML}}{\text{ha}}\right) \div \text{Crop water supply}_{\text{it}}\left(\frac{\text{ML}}{\text{ha}}\right)$
Panel E2: Transpiration index		
Variable	WESM	Formula
	Level	

Traspiration indexkg of biomassML of transpiration	Level 5	$= \text{Biomass yield}_{\text{it}} \left( \frac{\text{kg of biomass}}{\text{ha}} \right)$
		$\div Transpiration_{it}\left(\frac{ML of transpiration}{ha}\right)$
$Biomass yield_{it} \left(\frac{kg \text{ of biomass}}{ha}\right)$	WESM Input	DSSAT Output (simulation)
$Transpiration_{it}\left(\frac{ML \text{ of transpiration}}{ha}\right)$	WESM Input	DSSAT Output (simulation)
Panel E3: Harvest index		
Variable	WESM	Formula
	Level	
Harvest index <sub>i</sub> $\left(\frac{\text{kg of lint}}{\text{kg of biomass}}\right)$	Level 5	$= \text{Lint yield}_{\text{it}} \left(\frac{\text{kg of lint}}{\text{ha}}\right) \div \text{Biomass yield}_{\text{it}} \left(\frac{\text{kg of biomass}}{\text{ha}}\right)$
Lint yield <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{ha}}\right)$	Level 7	= Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right) \times \text{Lint percentage}_{\text{it}}$ (%)
Biomass yield <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ha}}\right)$	WESM Input	DSSAT Output (simulation)
Panel E4: Analysis of Operating profit		
Operating profit <sub>it</sub> $\left(\frac{\$}{ha}\right)$	Level 4	$= \text{ Total sales re} \qquad \qquad _{it} \left(\frac{\$}{ha}\right) - \text{ Total operating } \text{cost}_{it} \left(\frac{\$}{ha}\right)$

Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$	Level 5	= Sales revenue <sub>it</sub> (lint) $\left(\frac{\$}{ha}\right)$ + Sales revenue <sub>it</sub> (cotton seed) $\left(\frac{\$}{ha}\right)$
Total operating $\text{cost}_{it}\left(\frac{\$}{ha}\right)$	Level 5	= Total water $\operatorname{cost}_{it}\left(\frac{\$}{ha}\right)$ + Other variable $\operatorname{cost}_{it}\left(\frac{\$}{ha}\right)$
		+ Fixed operating $cost_{it} \left(\frac{\$}{ha}\right)$
Panel E5: Analysis of Total sales revenue		
Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$	Level 5	= Sales revenue <sub>it</sub> (lint) $\left(\frac{\$}{ha}\right)$ + Sales revenue <sub>it</sub> (cotton seed) $\left(\frac{\$}{ha}\right)$
Sales revenue <sub>it</sub> (lint) $\left(\frac{\$}{ha}\right)$	Level 6	= Lint yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right) \times \text{Lint price}_{\text{it}} \left(\frac{\$}{\text{kg}}\right)$
Sales revenue <sub>it</sub> (cotton seed) $\left(\frac{\$}{ha}\right)$	Level 6	= Seed yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$ × Seed price <sub>it</sub> $\left(\frac{\$}{\text{kg}}\right)$
Panel E6: Analysis of Total operating cost		·
Total operating $cost_{it}\left(\frac{\$}{ha}\right)$	Level 5	= Total water $cost_{it}\left(\frac{\$}{ha}\right)$ + Other variable $cost_{it}\left(\frac{\$}{ha}\right)$
		+ Fixed operat $\cos t_{it} \left(\frac{\$}{ha}\right)$
Total water $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$	Level 6	= Licensed irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$ + Traded irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$

Other variable $ cost_{it} \left(\frac{\$}{ha}\right) $	Level 6	= Unit other variable $\text{cost}_{it}\left(\frac{\$}{\text{kg of lint}}\right) \times \text{Lint yield}_{it}\left(\frac{\text{kg}}{\text{ha}}\right)$
Fixed operating $cost_{it} \left(\frac{\$}{ha}\right)$	<mark>WESM</mark> Input	Accounting Data

# Table A1 WESM model - Panel G - Level 6

Panel G1: Calculation of Crop water supply		
Variable	WESM Level	Formula
Crop water supply <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 6	$= \text{Rainfall}_{\text{it}} \left(\frac{\text{ML}}{\text{ha}}\right) + \text{Irrigation abstracted}_{\text{it}} \left(\frac{\text{ML}}{\text{ha}}\right)$
		+ Recycled runoff $_{it}\left(\frac{ML}{ha}\right)$
Rainfall it $\left(\frac{ML}{ha}\right)$	WESM Input	DSSAT Output (simulation)
Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$	Level 7	= Irrigation applied it $\left(\frac{ML}{ha}\right)$ - Recycled runoff it $\left(\frac{ML}{ha}\right)$
Recycled runoff $_{it}\left(\frac{ML}{ha}\right)$	Level 7	= Total water runoff $_{it}\left(\frac{ML}{ha}\right)$ × Ratio of recycled runoff
Panel F2: Calculation of Crop water loss		
Variable	WESM Level	Formula

$Crop water loss_{it} \left(\frac{ML}{ha}\right)$	Level 6	= Soil evaporation it $\left(\frac{ML}{ha}\right)$ + Deep drainage <sub>it</sub> $\left(\frac{ML}{ha}\right)$
		+ Loss from water runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$
		+ Change in soil water it $\left(\frac{ML}{ha}\right)$
Soil evaporation it $\left(\frac{ML}{ha}\right)$	WESM Input	DSSAT Output (simulation)
Deep drainage <sub>it</sub> $\left(\frac{ML}{ha}\right)$	WESM Input	DSSAT Output (simulation)
Change in soil water <sub>it</sub> $\left(\frac{ML}{ha}\right)$	WESM Input	DSSAT Output (simulation)
Loss from water runoff it $\left(\frac{ML}{ha}\right)$	Level 7	= Total water runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$ - Recycled runoff <sub>it</sub> $\left(\frac{ML}{ha}\right)$
Panel F3: Calculation of Sales revenue (lint)		
Variable	WESM Level	Formula
Sales revenue <sub>it</sub> (lint) $\left(\frac{\$}{ha}\right)$	Level 6	$= \text{Lint yield}_{it} \left(\frac{kg}{ha}\right) \times \text{Lint price}_{it} \left(\frac{\$}{kg}\right)$
Lint yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$	Level 7	= Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right) \times \text{Lint percentage}_{\text{it}} (\%)$
Lint price <sub>it</sub> $\left(\frac{\$}{\text{kg}}\right)$	WESM Input	Accounting Data
Panel F4: Calculation of Sales revenue (cotton	seed)	

Variable	WESM Level	Formula
Sales revenue <sub>it</sub> (cotton seed) $\left(\frac{\$}{ha}\right)$	Level 6	$= \text{Seed yield}_{it} \left(\frac{kg}{ha}\right) \times \text{Seed price}_{it} \left(\frac{\$}{kg}\right)$
Seed yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$	Level 7	$= \text{Total yield}_{\text{it}}\left(\frac{\text{kg of total product}}{\text{ha}}\right) - \text{Lint yield}_{\text{it}}\left(\frac{\text{kg}}{\text{ha}}\right)$
Seed price <sub>it</sub> $\left(\frac{\$}{\text{kg}}\right)$	WESM Input	Accounting Data
Panel F5: Calculation of Total water cost		
Variable	WESM Level	Formula
Total water $cost_{it}\left(\frac{\$}{ha}\right)$	Level 6	= Licensed irrigation $cost_{it}\left(\frac{\$}{ha}\right)$ + Traded irrigation $cost_{it}\left(\frac{\$}{ha}\right)$
Licensed irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$	Level 6	Calculated in Panel G
Traded irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$	Level 6	Calculated in Panel G
Panel F6: Calculation of Licensed irrigation co	st	
Variable	WESM Level	Formula
Licensed irrigation $cost_{it} \left(\frac{\$}{ha}\right)$	Level 6	= Licensed irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$
		× Unit l irrigation $cost_{it} \left(\frac{\$}{ML}\right)$

P	1	1
Licensed irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	= Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$
		$\times$ Licensed irrigation to Irrigation abstracted ratio <sub>it</sub> (%)
Unit licensed irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ML}}\right)$	WESM Input	Accounting Data
Panel F7: Calculation of Traded irrigation cost	-	
Variable	WESM Level	Formula
Traded irrigation $cost_{it} \left(\frac{\$}{ha}\right)$	Level 6	$= \text{Traded irrigation volume}_{\text{it}} \left(\frac{\text{ML}}{\text{ha}}\right)$
		× Unit traded irrigation $cost_{it}\left(\frac{\$}{ML}\right)$
Traded irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	= Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$
		$\times$ Traded irrigation to Irrigation abstracted ratio <sub>it</sub> (%)
Unit traded irrigation $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ML}}\right)$	WESM Input	Accounting Data
Panel F8: Calculation of Other variable cost		
Variable	WESM Level	Formula

Other variable $cost_{it}\left(\frac{\$}{ha}\right)$	Level 6	$= \text{Unit other variable cost}_{it} \left(\frac{\$}{\text{kg of lint}}\right) \times \text{Lint y} \qquad \qquad _{it} \left(\frac{\text{kg}}{\text{ha}}\right)$
Unit other variable $\text{cost}_{\text{it}}\left(\frac{\$}{\text{kg of lint}}\right)$	WESM Input	Accounting Data
Lint yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$	Level 7	= Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right) \times \text{Lint percentage}_{it}$ (%)

# Table A1 WESM model - Panel G - Level 7

Panel G1: Calculation of Seed yield				
Variable	WESM Level	Formula		
Seed yield <sub>it</sub> $\left(\frac{kg}{ha}\right)$	Level 7	$= \text{Total yield}_{it} \left( \frac{\text{kg of total product}}{\text{ha}} \right) - \text{Lint yield}_{it} \left( \frac{\text{kg}}{\text{ha}} \right)$		
Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$	WESM Input	DSSAT Output (simulation)		
Lint yield <sub>it</sub> $\left(\frac{\text{kg}}{\text{ha}}\right)$	Level 7	Calculated in G2		
Panel G2: Calculation of Lint yield				
Variable	WESM Level	Formula		

Lint yield <sub>it</sub> $\left(\frac{kg}{ha}\right)$	Level 7	$= \text{Total yield}_{it}\left(\frac{\text{kg of total product}}{\text{ha}}\right) \times \text{Lint pecentage}_{it} (\%)$
Total yield <sub>it</sub> $\left(\frac{\text{kg of total product}}{\text{ha}}\right)$	WESM Input	DSSAT Output (simulation)
Lint percentage <sub>it</sub> (%)	WESM Input	Crop Data
Panel G3: Calculation of Licensed irrigation ve	olume	
Variable	WESM Level	Formula
Licensed irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	= Irrigation abstrated $_{it}\left(\frac{ML}{ha}\right)$
		$\times$ Licensed irrigation to Irrigation abstracted ratio _it (%)
Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$	Level 7	Calculated in Panel F
Licensed irrigation to Irrigation abstracted <sub>it</sub>	WESM Input	Irrigation allocation data
ratio <sub>it</sub>		
Panel G4: Calculation of Traded irrigation vol	ume	
Variable	WESM Level	Formula

Traded irrigation volume <sub>it</sub> $\left(\frac{ML}{ha}\right)$	Level 7	= Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$
		imes Traded irrigation to Irrigation abstracted ratio <sub>it</sub> (%)
Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$	Level 7	Calculated in Panel F
Traded irrigation to Irrigation abstracted <sub>it</sub>	WESM Input	Irrigation allocation data
ratio <sub>it</sub>		
Panel G5: Calculation of Loss from water runc	off	
Variable	WESM Level	Formula
Loss from water runoff it $\left(\frac{ML}{ha}\right)$	Level 7	= Total water runoff $_{it}\left(\frac{ML}{ha}\right)$ - Recycled runoff $_{it}\left(\frac{ML}{ha}\right)$
Total water runoff $_{it}\left(\frac{ML}{ha}\right)$	Level 7	Calculated in Panel F
Recycled runoff it $\left(\frac{ML}{ha}\right)$	Level 7	Calculated in Panel F
Panel G6: Calculation of Irrigation abstracted		
Variable	WESM Level	Formula
Irrigation abstracted $_{it}\left(\frac{ML}{ha}\right)$	Level 7	= Irrigatio applied it $\left(\frac{ML}{ha}\right)$ - Recycled runoff it $\left(\frac{ML}{ha}\right)$
Irrigation infiltrated it $\left(\frac{ML}{ha}\right)$	Level 7	Calculated in G10

Recycled runoff $_{it}\left(\frac{ML}{ha}\right)$	Level 7	Calculated in G7
Panel G7: Calculation of Recycled runoff		
Variable	WESM Level	Formula
Recycled runoff $_{it}\left(\frac{ML}{ha}\right)$	Level 7	= Total water runoff <sub>it</sub> $\left(\frac{ML}{ha}\right) \times Ratio of recycled runoff$
Total water runoff $_{it}\left(\frac{ML}{ha}\right)$	Level 7	Calculated in G8
Ratio of recycled runoff	WESM Input	Farming practice
Panel G8: Calculation of Total water runoff		
Variable	WESM Level	Formula
Total water runoff $_{it}\left(\frac{ML}{ha}\right)$	Level 7	= Runoff from rainfall <sub>it</sub> $\left(\frac{ML}{ha}\right)$ + Runoff from irrigation <sub>it</sub> $\left(\frac{ML}{ha}\right)$
Runoff from rainfall it $\left(\frac{ML}{ha}\right)$	WESM Input	DSSAT Output (simulation)
Runoff from irrigation it $\left(\frac{ML}{ha}\right)$	WESM Input	SIRMOD Output (simulation)
Panel G9: Calculation of Runoff from irrigatio	n	
Variable	WESM Level	Formula
Runoff from irrigation it $\left(\frac{ML}{ha}\right)$	Level 7	$= \text{Runoff from irrigation per event}_{it} \left(\frac{\text{ML per event}}{\text{ha}}\right)$
		× Number of irrigat events it

$\left(\frac{\text{ML per event}}{\text{ha}}\right)$		
Number of irrigation events	WESM Input	DSSAT Output (simulation)
Panel G10: Calculation of Irrigation infiltrated		
Variable	WESM Level	Formula
Irrigation infiltrated it $\left(\frac{ML}{ha}\right)$	Level 7	$= \text{Irrigation infiltrated per event}_{\text{it}} \left(\frac{\text{ML per event}}{\text{ha}}\right)$
		$\times$ Number of irrigation events <sub>it</sub>
Irrigation infiltrated per event <sub>it</sub>	WESM Input	SIRMOD Output (simulation)
$\left(\frac{ML \text{ per event}}{ha}\right)$		
Number of irrigation events	WESM Input	DSSAT Output (simulation)



### Table A2 WESM Input Variables (DSSAT output parameters)



## Table A3 WESM Input Variables (Non-DSSAT output parameters)

# Table A4 Summary of High-Level WESM measures

Variables	Level	Formula	Definition
Profit to water cost ratio: $\left(\frac{\text{$ operating profit}}{\text{$ operating profit}}\right)$	Level 1	= Operating profit: $\left(\frac{\$}{-}\right)$	The amount of operating
water cost /		(ha)	profit generated by each
		$\div$ Total water cost <sub>it</sub> $\left(\frac{\$}{}\right)$	dollar spent on water
		ha/	resource.
Beturn on water, $\left(\frac{\$}{1}\right)$	Level 2	$-$ Operating profit. $\left(\frac{\$}{-}\right)$	The degree to which
(ML)		ha/	operating profit is generated
		÷ Crop water supply <sub>it</sub> $\left(\frac{ML}{L}\right)$	by one unit of volume of all
		ha /	sources of water supplied to
			the crop business.
Water use leverage <sub>it</sub> (%)	Level 2	= Irrigation abstracted it $\left(\frac{ML}{ha}\right)$ ÷	The degree to which Crop
		(ML) × 1000(	water supply to a cropping
		Crop water supply <sub>it</sub> $\left(\frac{1}{ha}\right)$ × 100%	system is sourced by water
			withdrawn from the
			environment (i.e., irrigation
			water).
Weighted average irrigation cost $\left(\frac{\$}{-}\right)$	Level 2	= Total water cost. $\left(\frac{\$}{-}\right)$	The water rate (i.e., \$ of
(ML)		ha)	water cost per ML of water)
		$\div$ Irrigation abstracted it $\left(\frac{ML}{M}\right)$	that a crop business has to
		ha/	pay on average to its two



Total sales revenue <sub>it</sub> $\left(\frac{\$}{ha}\right)$	Level 5	$= \text{Sales revenue}_{it}(\text{lint })\left(\frac{\$}{\text{ha}}\right)$	Consisting of sales revenues generated from the primary
		+ Sales revenue <sub>it</sub> (cotton seed) $\left(\frac{\$}{ha}\right)$	product (i.e., lint product) and sales revenue generated
			from the by-product (i.e., cotton seed product).
Total operating $\text{cost}_{\text{it}}\left(\frac{\$}{\text{ha}}\right)$	Level 5	= Total water $\operatorname{cost}_{it}\left(\frac{\$}{ha}\right)$	Consisting of cost components that are incurred
		+ Other variable $\operatorname{cost}_{it}\left(\frac{\$}{ha}\right)$	as a result of operating activities undertaken for crop
		+ Fixed operating $\operatorname{cost}_{it}\left(\frac{\$}{ha}\right)$	production.
Transpiration index <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ML of tranpiration}}\right)$	Level 5	= Biomass yield <sub>it</sub> $\left(\frac{\text{kg of biomass}}{\text{ha}}\right)$ $\div$ Transpiration <sub>it</sub> $\left(\frac{\text{ML of transpiration}}{\text{ha}}\right)$	The amount of total dry mater (biomass) generated by each unit of volume of water actually used by a crop.
Harvest index <sub>it</sub> $\left(\frac{\text{kg of lint}}{\text{kg of biomass}}\right)$	Level 5	$= \text{Lint yield}_{it} \left(\frac{\text{kg of lint}}{\text{ha}}\right)$ $\div \text{Biomass yield}_{it} \left(\frac{\text{kg of biomass}}{\text{ha}}\right)$	The extent to which biomass is converted into primary crop product.

Water input efficiency <sub>it</sub> (%)	Level 5	$= \begin{bmatrix} \text{Transpiration}_{\text{it}} \left( \frac{\text{ML of transpiration}}{\text{ha}} \right) \\ (\text{ML}) \end{bmatrix}$	The degree to which Crop water supply is actually used
		$\left[ \div \text{Crop water supply}_{it}\left(\frac{1}{ha}\right) \right]$	for carbon assimilation to
		× 100%	generate lint product.

#### Table A5 Summary of Low-Level WESM measures





Runoff from irrigation it $\left(\frac{ML}{ha}\right)$	Level 7	= Runoff from irrigation per event it $\left(\frac{ML \text{ per event}}{ha}\right)$
		× Number of irrigation events
Irrigation infiltrated it $\left(\frac{ML}{ha}\right)$	Level 7	= Irrigation infiltrated per event it $\left(\frac{ML \text{ per event}}{ha}\right)$
		$\times$ Number of irrigation events <sub>it</sub>

# Appendix B: Supplementary information for Chapter 5

#### **Appendix B1: Description of cotton production operations**

An Australian cotton farming system typically consists of six main cotton production processes, including field preparation (fallow), planting, in-crop, irrigation, harvest and post-harvest (Chen & Baillie 2009; Khabbaz 2010).

In addition, each cotton process comprises of a range of operations (activities) which can be further classified into different types of practices. For example, a field preparation process (a pre-planting process to create a suitable seedbed for germination) may consist of three different operations; tillage, fertilising and spraying. Tillage can involve discing, soil regrading and deep ripping practices (Chen & Baillie 2009; Khabbaz 2010).

Decisions on which operations are required and/or which type of practice needs to be carried out in a particular cotton production process is contingent on a wide range of factors, such as climatic conditions (i.e., rainfall amount and frequency, ambient temperature), soil types, irrigation availability, choices of different tillage methods, irrigation systems and cotton varieties (Chen & Baillie 2009; Khabbaz 2010). For example, as opposed to a tradition tillage method, a reduced tillage approach does not involve discing and soil regrading in an attempt to reducing adverse effects on soil conditions (such as soil compacting) and energy use. Furthermore, the irrigation process may or may not be included in a cotton farming system, depending on management decisions as to whether irrigated cotton or dry land cotton is grown (which is based on rainfall conditions and irrigation availability). While dry land cotton reduces the risk of irrigation shortage and saves water supply costs, this kind of approach may reduce fibre quality (such as fibre length and strength) and cotton production yield. In addition, most cotton operations and practices are machine-intensive activities, and involve the use of energy (mainly diesel fuel and electricity) for tractors and/or pumps (Chen & Baillie 2009; Khabbaz 2010).

#### 1. Fallow

A fallow operation, which is conducted before planting, involves tillage and other soil preparation activities, such as fertilizing, pesticide and herbicide spraying. The purpose of a fallow operation is to create a suitable seedbed with warm soil for germination. This requires the use of heavy machinery (choices of tractors are depending on load and soil conditions) and a significant level of diesel fuel consumption (Chen & Baillie 2009; Khabbaz 2010).

### Tillage

A tillage operation involves mechanical modification of soil structure to prepare a favorable seed bed. In recent years the trend in tillage operations has been towards a zero till practice. The advantage of reduced tillage operations is not only fuel cost savings but also a retention of soil organic matter and soil moisture, and reduced soil erosion application (The Cotton Industry Development and Delivery Team 2012). A comparison of farm operations by Baillie (2009) shows about 12% fuel savings with reduced tillage and a further 13% in the case of zero till practice.

### 2. Planting

A planting operation involves decisions on sowing density (i.e., number of cotton seeds per m) and row configurations. There are a range of row configurations that are used by cotton growers across the Australian cotton industry, such as single skip, double skip, and super single (The Cotton Industry Development and Delivery Team 2012).

In general, wider row spacing configurations result in less water use for growing a crop (per hectare) and more water use efficiency, and hence reducing the variable cost of water inputs. However, it may also limit yield potential. While a trade-off between water costs and cotton yield is a key consideration of cotton growers when making a row configuration choice, there are a number of other factors also need to be taken into account, such as soil type and environment (i.e., rainfall availability, temperature, weather). Wider row spacing configurations can be applied to help cotton growers manage water risk in dry land cotton production, but as mentioned above, this may lead to a reduction in cotton yield and fibre quality (The Cotton Industry Development and Delivery Team 2012).

Row configurations and sowing density also drive energy use in irrigation activities, such as pumping and sowing. By using dry land cotton production, cotton growers can achieve energy savings as irrigation is not required. However, there is a trade-off between these energy costs and cotton yield and fibre quality that needs to be taken into consideration (Chen & Baillie 2009; Khabbaz 2010).

### 3. In-crop

Once the plant has become established, various operations to maintain the crop are undertaken. Fertilizers and pesticides are applied with a tractor via boom spraying, and interrow cultivation is used to help weed control. Some chemicals such as herbicides and insecticides may also be applied via aircraft. Controlled traffic farming, which uses a global positioning system (GPS) to establish fixed tractor passages and precision drilling, can save 50% to 70% of fuel costs over the season.

## Fertilization

Fertigation, an application of liquid fertilizer through the irrigation system, which can be applied using the aerial spraying method. This practice can be used to replace traditional fertilizer applied via tillage to reduce energy consumption while increasing efficiency of fertilization (Chen & Baillie 2009; Khabbaz 2010).

## **Pesticide Treatment**

The use of biotechnology and the introduction of genetically modified crops such as Bollgard II, Roundup Ready and Liberty Link, has led to a significant reduction in insecticide and herbicide use, resulting not only in reduced costs but also in reduced energy usage in the application (The Cotton Industry Development and Delivery Team 2012). However, these genetically modified crops require adherence to strict insecticide resistance management strategies, including pupae busting after harvesting. Pupae busting is required as the cotton season normally ends in March or April (the autumn season), when shorter days and cooler temperatures can enable Helicoverpa pupae<sup>128</sup> to develop. As a result, the moths that emerge at the start of the next cotton season in September (the spring season) may be resistant to the limited pesticide applications at the end of the previous season. Cultivation to control the pupae under cotton stubble is therefore of critical importance to ensure that any moths emerging from pupation will become trapped and die. This practice, however, requires the use of machinery and diesel fuels.

## 4. Irrigation

Determining the timing and amount of irrigation can be a challenge due to unpredictable weather and difference in soil structure for each field (Roth 2010). Each soil type has a

<sup>&</sup>lt;sup>128</sup> A pupa is an insect in the stage of development between a larva and an adult insect such as a moth.

different soil moisture capacity that is available to the plant (like a sponge holding water). The goal of irrigation is to make enough moisture available for the plant's daily use so that the plant does not become stressed. Stressed plants can result in yield reductions of up to 1.9 bales per hectare (The Cotton Industry Development and Delivery Team 2012).

Compensating for estimated rainfall with water applied through irrigation can be a difficult balancing act. Too much water causes deep drainage (where water below the root line is wasted), or even water logging (where excess water on the field causes low soil oxygen levels). This is especially the case when water is over applied through surface irrigation to ensure the water reaches the end of the furrow or if it rains shortly after an irrigation cycle (Chen & Baillie 2009).

#### 5. Harvesting

A harvest process usually involves two main operations, cotton picking and module building, and requires two types of heavy machinery, cotton pickers and module builders. The use of such large machines means that energy consumption for a harvest process can make up 20-25% of on-farm direct energy use.

#### **Cotton picking**

While cotton grown in Africa is still mainly harvested by hand, cotton in Australia and other countries such as the USA and Brazil is harvested with large machines. Though it takes less time to harvest cotton using machine pickers as opposed to human pickers, this contemporary farming practice requires a considerable amount of energy and releases GHG emissions in the natural environment. In addition, the machine may take not only the mature fibre bolls but also unopened bolls, plant sticks, and other foreign matter (such as soil, leaves, and twigs),

#### Module building

After a cotton picker takes the seed cotton off the bolls via a spindle, the harvested cotton is transported to module builders at the edge of the field. Module builders press the cotton into square long modules, which are covered with a tarpaulin until transported to the ginning plant. Modern cotton pickers have built in module builders that can help reduce labour, machine and energy requirements.

#### 6. Post-harvest

The process of pulling out the plant stubbles and burning them is no longer custom (The Cotton Industry Development and Delivery Team 2012). Instead, practices such as root cutting (residual cotton stalks are cut from the roots), mulching the stalk (shredding of cotton stalks) and pupae busting (pupae of insects in soil are destroyed by tillage) are promoted. After root cutting and mulching practices, the residues are incorporated into the soil, improving soil fertility and nitrogen content. As mentioned above, where Bollgard II cotton is grown, pupae destruction through tillage operation must be implemented to avoid over-wintering of the pest, which could lead to increasing resistance.
### Appendix B2: The validity issue of the OZCOT model

As this research aims to examine cotton production simulation in an agricultural setting, the APSIM-OZCOT model would seem to be the first choice for the thesis given its roots in Australian research. As discussed in the Section 5.3, good crop production models allows generation of valid (reliable) estimate of crop parameters that can be used to support better decision making and control. Therefore, it is crucial importance to examine whether the selected crop production simulation model can produce scientifically defensible estimate of interested crop variables.

In this Appendix the APSIM-OZCOT model was tested by simulating cotton crops grown on black vertosol soil at Myall Vale, NSW for a 100-year run of weather data obtained from SILO<sup>129</sup>. Furthermore, furrow irrigation was used as the irrigation method for simulating water and economic performance of irrigated crops. The cotton crops were also simulated under the assumptions that there were no nutrient or pest limitations.

The models for crop water process and water economic sustainability performance presented in Chapters 3 and 4 place emphasis on transpiration, the productive flux of water out of the farming system. Usually, this is confounded with soil evaporation to estimate evapotranspiration. While this latter, an aggregated parameter, can be relatively easily be measured or estimated from the sort of crop water balance that astute farmers should be maintaining, it conceals the real value of transpiration.

In crop simulation modelling, transpiration can be easily calculated as a function of environmental evaporative demand, energy interception by the crop and stomatal closure due to water stress. The theory has been well developed and, while it is rare to have experimental data measuring transpiration directly, it is an appropriate test of model validity to see that its results are consistent with the theory.

The common calculating sequence follows three steps:

(i) Calculation of environmental demand, with potential evapotranspiration as the output parameter.

<sup>&</sup>lt;sup>129</sup> SILO is an online database of more than 100 years of continuous daily weather records. <u>https://www.longpaddock.qld.gov.au/silo/</u>

- (ii) Partitioning potential evapotranspiration into potential transpiration and potential soil evaporation on the basis of radiation interception by the crop canopy. The partition is generally represented by the relationship between normalized transpiration - the ratio between potential transpiration and potential evapotranspiration- and leaf area index - a measure of the leaf area displayed by the crop.
- (iii) Adjusting potential transpiration for the effects of water stress, to estimate actual transpiration. This is generally represented as the relationship between normalized transpiration - the ratio between actual transpiration and potential evapotranspiration and percentage of available soil water remaining.

The initial model that was used was the APSIM version of OZCOT, which was a porting of the OZCOT code (Hearn 1994) into the APSIM simulation environment.

- (i) Potential evapotranspiration was based on the Priestly-Taylor method rather than the more modern Penman-Monteith equation. However, this is unlikely to introduce significant errors.
- (ii) The radiation partitioning process was modelled using unpublished data from Mateos (Hearn 1994)
- (iii) The effect of soil moisture deficit on transpiration is not described in Hearn (1994), but the source code of the OZCOT moiety was available and the following relationships are based on the parameter values used in that code.

Ritchie (1973) proposed a single model for this effect, where stomatal function and hence the ratio of actual transpiration and potential evapotranspiration would remain maximal at or near 1.0 until a threshold value of plant available water, in his case, 80%, had been withdrawn from the root zone. As soil water declined past this point, this ratio reduced to zero, irrespective of atmospheric conditions. In contrast, Denmead and Shaw (1962) proposed that the threshold would depend on atmospheric conditions represented by potential evapotranspiration and the threshold value for plant available water would decline as potential evapotranspiration declined. This more nuanced model has been supported by Sadras and Milroy (1996).



# Figure B1 Ratio of potential transpiration to potential evapotranspiration as a function of both soil moisture index and evaporative demand (potential evapotranspiration)

Figure B.1 shows that the OZCOT parameter soil moisture index (smi) is functionally similar to plant available water. The ratio of potential transpiration to potential evapotranspiration was plotted against smi for all values of "plant status=alive", that is, a crop was actually growing.

The analytical description above shows three steps. The first is accommodated by normalising the simulated transpiration data against daily potential evaporative demand. The next two stages are separated by filtering the data into, initially, useful ranges of LAI. The data in Fig. B1 are for the range of LAI 1.0-1.5, which should give a maximal ratio of potential transpiration to potential evapotranspiration (horizontal maximum in Fig. B1) of 0.4-0.54. The range of potential evapotranspiration for this subset of data was 0.5-12mm/day and this was filtered to 2.5-10mm/day to allow comparison with Fig. B2, which implies a tail of data descending towards the origin for plant available water, represented here as smi in the range of 0.2-0.5, depending on potential evapotranspiration.

The filtered data are depicted in figure B2. The characteristics described above are present a maximal zone with values of ratio of potential transpiration to potential evapotranspiration ranging from 0.4 to 0.54 and a plant available water greater than approximately 0.2-0.4.

However, there are many simulated data in the region for smi greater than 0.2 and ratio of potential transpiration to potential evapotranspiration less than 0.4 for which there is no theoretical or coding reason. These represent days of grossly underestimated transpiration and, given the primary interest in transpiration as a pivotal parameter in our WESM, this result undermines the reliability and validity of APSIM-OZCOT as the crop model of choice.



Figure B2 Simulated ratio of potential transpiration to potential evapotranspiration as a function of smi when leaf area index was restricted to 1.0-1.5 and potential evapotranspiration to 2.5-10 mm/day

## Appendix B3: The validity of the DSSAT model

Simulation of the growth and water use of cotton crops grown on black vertosols at Myall Vale, NSW, using the same 100-year run of weather was conducted using the CROPGRO-Cotton model in the DSSAT system. The crop management rules are also set under furrow irrigation approach and with the assumptions of no nutrient or pest limitations.

Similar filtering of the data was performed, with one difference. DSSAT does not use smi to describe soil water status, instead using fraction of plant available water remaining. This is used on the x-axis in figure B3.



## Figure B3 Ratio of potential transpiration to potential evapotranspiration calculated from DSSAT simulation for leaf area index of 1-1.5 and potential evapotranspiration of 2.5-10 mm/day

The relationship shows no data points in the region forbidden by theory. The simulation is, to that extent, more valid than the APSIM-OZCOT model. The simulation conditions maintained the crop in a well-, possibly excessively-watered condition, as shown by plant available water values > 100%. The irrigation frequency also meant that dry conditions were not experienced, so that portion of the transpiration response cannot be shown.

## **Appendix C: Supplementary information for Chapter 6**

## Appendix C1: Overview of subsurface (drip) irrigation systems

In subsurface (drip) irrigation, water is carried by pipes. Figure C1 provides an example of subsurface (drip) irrigation systems in a cotton farm.

These pipes are typically light weight flexible plastic tubing, so that water is released directly to the soil in the root zone. The release is via emitters attached to, or built into, the pipe and spaced to provide one emitter for each or a few plants in the care of row crops. The emitters are engineered to have reasonably uniform water release rates from a low (1-2 bar) water pressure in the lateral line along each row. The purpose is to provide an intentionally heterogeneous wetting pattern with the water applied only to the zone of active roots. Ideally, irrigation scheduling should allow the average application rate to match the average transpiration rate, with little or no excess to go to deep drainage. In the case of sub-surface drip irrigation, the lateral is buried beneath the row of crop plants at 30+ centimetres, typically, to avoid surface wetting and hence eliminate soil evaporation and runoff from irrigation (Bucks, Nakayama & Warrick 1982).



Figure C1 An example of subsurface (drip) irrigation systems in a cotton farm (Source: Khabbaz 2010)

Of the three irrigation systems, the subsurface drip irrigation system consumes the most energy as it involves pressurised irrigation and requires increased head pressure. In contrast, surface irrigation system requires less energy as it works by simply applying water from a head ditch, which is filled with water from a bore, river or catchment with a pump, into the furrows where water spreads due to gravity and the slope of the field. Research on the impact of farming systems on energy use and associated GHG emissions reports that energy usage for irrigation using a furrow (or border check) irrigation system is about two-thirds of that used in lateral move (or centre pivot) irrigation and less than half of the energy used in subsurface drip irrigation (Khabbaz 2010).

#### Appendix C2: Overview of lateral-move and centre pivot irrigation systems

Lateral-move and centre pivot irrigation systems are self-propelled irrigation systems. A centre pivot irrigation system rotates around a fixed central point and moves at a constant speed up and down a paddock (Khabbaz 2010). Figure C2 provides an example of a lateral-move system on a cotton farm. While about 80% of Australian cotton farms use surface irrigation systems, there is growing use of lateral-move and centre pivot irrigation systems due to their improved water use efficiency compared to surface irrigation systems (Roth et al. 2013).

Compared to modern irrigation systems like subsurface irrigation systems and lateral move irrigation systems, the traditional furrow irrigation systems are considered the least efficient systems with respect to water use (Jackson, Khan & Hafeez 2010). However, while deep drainage is reported nearly zero under lateral-move and centre pivot irrigation system (Roth et al. 2013), the soil surface is still wet, and hence water loss to soil evaporation is unavoidable. Accordingly, lateral-move and centre pivot irrigation systems are less water use efficient but more energy use efficient than subsurface (drip) irrigation systems. In contrast, lateral-move and centre pivot irrigation systems.

Farm managers need to consider the trade-offs between energy use efficiency and water use efficiency between irrigation systems when making decisions on which type of irrigation systems are applied in their cotton farms. Other factors that also need to be taken into account include climatic factors, soil structure, and local irrigation policy.



Figure C2 An example of lateral move irrigation systems in a cotton farm (Source: Khabbaz 2010)

Soil Lavar	Water Content <sup>130</sup>			Soil components		Other soil properties					
Layer (cm)	LL15	DUL	SAT	Clay %	Silt %	Stone %	pН	Root	Saturated	Bulk	Organic
(CIII)	$(cm^3/cm^3)$	$(cm^3/cm^3)$	$(cm^3/cm^3)$				1:5	Factor	hydraulic	density	content
									conductivities	$(g/cm^3)$	(%)
									(Ks, cm/hr)		
40	0.29	0.43	0.47	61.6	27.6	10.8	8.0	1	94.00	1.22	0.71
80	0.38	0.46	0.48	65.6	23.5	10.9	8.5	0.301	6.00	1.50	0.46
125	0.38	0.44	0.46	66.1	23.7	10.2	8.7	0.129	6.00	1.46	0.45
145	0.37	0.46	0.49	65.8	25.8	8.4	8.9	0.067	6.00	1.44	0.40
180	0.37	0.46	0.49	63.4	27.7	8.8	8.9	0.039	6.00	1.44	0.24

# Table C1 Soil properties of the Myall Valve soil (clay vertosol)

<b>Furrow Points</b>	Furrow Distance (m)	Infiltration(mm)_Scenario 1	Infiltration(mm)_Scenario 2
1	0	91	57
2	10	91	58
3	20	91	58
4	30	91	58
5	40	91	58
6	50	91	58
7	60	91	58
8	70	91	58
9	80	90	58
10	90	90	58
11	100	90	58
12	110	90	58
13	120	90	58
14	130	89	58
15	140	89	58
16	150	89	58
17	160	89	57
18	170	88	57
19	180	88	57
20	190	88	57
21	200	88	57
22	210	88	57
23	220	87	57
24	230	87	57
25	240	8/	57
26	250	86	57
27	260	86	56
28	270	85	57
29	280	85	57
30	290	85	56
15	300	84	50
32	310	84 	50
24	220	00	50
25	240	00	50
	240 2E0	<u>ຽງ</u> ຄຸ	50 Ec
27	350	۵۷ وی	50
20	270	02 Q1	
20	200	01 Q1	55
39	200	۵۱ ۵۱	
40 //1	330	80 20	55
42	410	79	54

Table C2 Simulated infiltration – Scenario 1 versus Scenario 2

<b>Furrow Points</b>	Furrow Distance (m)	Infiltration(mm)_Scenario 1	Infiltration(mm)_Scenario 2
43	420	79	54
44	430	78	54
45	440	78	54
46	450	77	54
47	460	76	54
48	470	76	54
49	480	75	53
50	490	74	53
51	500	74	53
52	510	73	53
53	520	72	53
54	530	72	53
55	540	71	52
56	550	71	52
57	560	70	52
58	570	69	52
59	580	68	51
60	590	67	51
61	600	67	51
62	610	65	51
63	620	64	50
64	630	63	50
65	640	62	50
66	650	62	50
67	660	61	49
68	670	60	49
69	680	58	49
70	690	57	48
71	700	56	48
72	710	55	48

## **Appendix D: Supplementary information for Chapter 7**

**Appendix D1: Calculations of fixed operating cost per ha and unit other variable cost** In this appendix I explain how fixed operating cost per ha and unit variable cost (\$/bale) is calculated based on economic data obtained from a cotton gross margin budget for furrow irrigated cotton in 2014 -2015(NSW Department of Primary Industries 2015).

An accounting technique (Cost-Volume-Profit (CVP) modelling) is coupled with a statistical technique (regression), is applied to undertake the analysis.

First, the CVP model is expressed as follow: Gross margin (\$/ha) = Total sales revenue (\$/ha) - Total operating cost (\$/ha) [D.1]

Or

 $Gross margin (\$/ha) = Selling price^{131} (\$/bale) x Lint yield (bales/ha) - Fixed operating cost (\$/ha) + Unit variable cost (\$/bale) x Lint yield (bales/ha) [D.2]$ 

Or

Gross margin (\$/ha) = Unit contribution margin (\$/bale) x Lint yield (bales/ha) - Fixed operating cost (\$/ha) [D.3]

I assume that contribution margin (the difference between selling price and unit variable cost) and fixed operating cost are constant in the short-term - therefore equation (D.3) shows that gross margin is a function of lint yield.

A data set of gross margin and associated lint yield is collected to establish the gross margin function. I assume a selling price of \$554.00/bale. This set of data is shown in Table D1.

<sup>&</sup>lt;sup>131</sup> The selling price provided by NSW Department of Primary Industries (2015) is already included cotton seed.

#### Table D1 The effect of lint yield on gross margin

Lint yield (bales/ha)		Gross Margin (\$/ha)
	7.50	736.00
	9.00	1445.00
	10.00	1918.00
	11.00	2391.00
	12.00	2864.00
	13.00	3337.00
	14.50	4047.00

(Source: NSW Department of Primary Industries (2015)

Based on the data set in Table D1, the gross margin function is determined using the regression technique (see Figure D1) as follow:

Gross margin 
$$(\$/ha) = 473(\$/bale) \times Lint yield (bales/ha) - 2812 (\$/ha)$$
 [D.4]



Figure D1 Cotton gross margin function

The equation (D.4) shows that fixed operating cost is \$2812 per ha and unit contribution margin is \$473 per bale.

In addition, given the selling price of \$554 per bale, unit variable cost is equal to \$81 per bale (= \$554 -\$473).

Furthermore, given licensed irrigation cost of \$60.30 per ML(including licence fees of \$32.24/ML and pumping cost of \$28.06/ML),<sup>132</sup> or \$53.67 per bale<sup>133</sup>(=\$60.30 x 0.89), unit variable cost, excluding irrigation cost is equal to \$27.33 per bale (= \$81 - \$53.67). This figure is equivalent to \$0.12 per kg of lint.

<sup>&</sup>lt;sup>132</sup> (NSW Department of Primary Industries (2015).

<sup>&</sup>lt;sup>133</sup> As shown in the example of cotton gross margin (NSW Department of Primary Industries, 2015), 9.79ML of irrigation required to produce 11 bales or 0.89 ML of irrigation per bale.

# Appendix E:Supplementary information for Chapter 9

## Table E1 Cotton irrigated area

Information	Cotton	2010-11	2011-12	2012-13	2013-14	Average
Source	area					(2010-
						2014)
Cotton	Total ha	599,630	566,000	425,786	414,000	501,354
Australia						
ABS	Total ha	588,294	596,497	437,818	390,031	503,160
ABS	Irrigated	<mark>359,280</mark>	<mark>397,221</mark>	<mark>365,268</mark>	<mark>339,692</mark>	<mark>365,365</mark>
	<mark>ha</mark>					
ABS	Dryland ha	229,015	199,276	72,549	50,340	137,795

Source: (Australian Bureau of Statistics 2015; Cotton Australia (2015)

## Table E2 Environmental and economic gain by moving from Scenario 1 to Scenario 2

Size of analysis		Per ha	<b>Business Level</b>	Industry Level	
Cotton irrigated area	(ha)		1	400	365,365
Potential	Water volume saving	= Decrease in crop water lost (from S1 to S2) =	2.15 ML	860 ML	785,535 ML
environmental gain		2.15 ML/ha			(785.5 GL)
Potential crop	Gain in lint production (from	= Crop water saved (ML/ha) (from S1 to S2) x	310 kg	124,000 kg	113,263,150 kg
production gain	additional available water	Crop water use index (kg/ML) (S2)	(1.36 bale)	(546 bale)	(498,957 bale)
	resource)	= 2.15 ML/ha x 144 kg/ML			
		= 310 kg/ha			
	Gain in lint production (based	= Increase in lint yield (from S1 to S2)	2.6 kg	1,040 kg	949,949 kg
	on improved practice)	= 2.6 kg/ha	(0.01 bale)	(4.6 bale)	(4,185 bale)
	Total gain in lint production		312.6 kg	125,040 kg	114,213,099 kg
				(550 bale)	(503,141 bale)
Potential economic	Gain in operating profit = Crop water saved (ML/ha) (from S1 to S2) x		\$217	\$86,800	\$79,284,205
gain	(from additional available	Return on water (\$/ML) (S2)			
	water resource)	= 2.15 ML/ha x \$101/ha			
		= \$217/ha			
	Gain in operating profit (\$)	= Increase in operating profit (from S1 to S2)	\$132	\$52,800	\$48,228,180
	(based on improved practice)	= \$132/ha			
	Total gain in operating profit	(\$)	\$349	\$139,600	\$127,512,385